X

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q
OUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934.

For the transition period from ______ to _____

For the quarterly period ended March 31, 2023

Commission file number 001-38825

LIVEVOX HOLDINGS, INC.

(Exact Name of Registrant as Specified in Its Charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization) 82-3447941 (I.R.S. Employer Identification No.)

655 Montgomery Street, Suite 1000, San Francisco, CA 94111 (Address of principal executive offices) (Zip Code) (415) 671-6000

(Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading symbol(s)	Name of each exchange on which registered
Class A common stock, par value \$0.0001 per share	LVOX	The Nasdaq Stock Market LLC
Redeemable Warrants, each whole Warrant exercisable to purchase one share of Class A common stock at an exercise price of \$11.50	LVOXW	The Nasdaq Stock Market LLC
Units, each consisting of one share of Class A common stock and one- half of one redeemable Warrant	LVOXU	The Nasdaq Stock Market LLC

nan of one redecinable warrant			
, .	· / 1	ired to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the pr subject to such filing requirements for the past 90 days. Yes \boxtimes No \square	eceding 12 months (or for such shorter
		every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-red to submit such files). Yes \boxtimes No \square	Γ (§232.405 of this chapter) during the
		accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth rging growth company" in Rule 12b-2 of the Exchange Act.	company. See the definitions of "large
Large accelerated filer		Accelerated filer	
Non-Accelerated filer	\boxtimes	Smaller reporting company	\boxtimes
Emerging growth company	\boxtimes		

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Indicate by check mark whether the Registrant is a shell company (as defined in Rule 12b-2 of the Securities Exchange Act of 1934). Yes 🗆 No 🗵

As of May 4, 2023, the registrant had 92,938,649 shares of Class A common stock, par value \$0.0001 per share, issued and outstanding (100,482,399 shares of common stock, less 7,543,750 shares held in escrow).

Table of Contents

TABLE OF CONTENTS

	Page
Part I - Financial Information	<u>1</u>
Item 1. Consolidated Financial Statements	<u>1</u>
Consolidated Balance Sheets	<u>1</u>
Consolidated Statements of Operations and Comprehensive Loss	<u>3</u>
Consolidated Statements of Stockholders' Equity	<u>4</u>
Consolidated Statements of Cash Flows	4 5 7
Notes to the Consolidated Financial Statements	<u>7</u>
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>38</u>
Item 3. Quantitative and Qualitative Disclosure About Market Risk	<u>51</u>
Item 4. Controls and Procedures	<u>51</u>
Part II - Other Information	<u>53</u>
Item 1, Legal Proceedings	
Item 1A. Risk Factors	<u>53</u>
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	<u>53</u>
Item 3. Defaults Upon Senior Securities	53 53 53 53 53 53 53 53
Item 4. Mine Safety Disclosures	<u>53</u>
Item 5. Other Information	<u>53</u>
Item 6. Exhibits	<u>53</u>
<u>Signatures</u>	<u>55</u>
i	
•	

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

This quarterly report on Form 10-Q (this "Quarterly Report") includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). All statements, other than statements of present or historical fact included in this Quarterly Report, regarding the future financial performance of LiveVox Holdings, Inc. ("LiveVox" or the "Company"), as well as LiveVox's strategy, future operations, future operating results, financial position, expectations regarding revenue, losses, and costs, prospects, plans and objectives of management are forward-looking statements. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. The words "anticipate," "believe," "continue," "could," "estimate," "expect," "intends," "may," "might," "plan," "possible," "protential," "project," "should," "will," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements in this Quarterly Report are subject to risks and uncertainties that may include, for example:

- the business, operations and financial performance of the Company, including market conditions and global and economic factors beyond the Company's control, such as a tight labor market, inflationary pressures, rising interest rates, volatility in foreign exchange rates, supply chain constraints, recessionary fears, and impacts from the invasion of Ukraine by Russia;
- the impact of COVID-19 and related significant market volatility in the Company's business, our industry and the global economy;
- the high level of competition in the cloud contact center industry and the intense competition and competitive pressures from other companies in the industry in which the Company operates;
- · the effect of legal, tax and regulatory changes;
- the Company's ability to maintain its listing on The Nasdaq Stock Market LLC ("Nasdaq"), including its ability to comply with the requirement that the bid price for the Class A common stock be above \$1.00 for a period of 30 consecutive trading days;
- · the Company's ability to raise financing or complete acquisitions in the future;
- · the Company's success in retaining or recruiting, or changes required in, its officers, key employees or directors;
- the future financial performance of the Company;
- the Company's reliance on third-party telecommunications and internet service providers to provide its products and for other aspects of its business;
- · the outcome of any legal proceedings that may be instituted against the Company;
- reliance on information systems and the ability to properly maintain the confidentiality and integrity of data;
- · the occurrence of cyber incidents or a deficiency in cybersecurity protocols; and
- the ability to obtain third-party software licenses for use in or with the Company's products.

The foregoing review of important factors should not be construed as exhaustive and should be read in conjunction with the other risk factors included in our filings with the SEC, including our Annual Report on Form 10-K filed with the SEC on March 2, 2023. Forward-looking statements reflect current views about LiveVox's plans, strategies and prospects, which are based on information available as of the date of this Quarterly Report. Except to the extent required by applicable law, LiveVox undertakes no obligation (and expressly disclaims any such obligation) to update or revise the forward-looking statements whether as a result of new information, future events or otherwise.

PART I—FINANCIAL INFORMATION

ITEM 1. CONSOLIDATED FINANCIAL STATEMENTS

LIVEVOX HOLDINGS, INC.

Consolidated Balance Sheets (In thousands, except per share data)

	As of			
	March 31, 2023		Do	ecember 31, 2022
	(1	Unaudited)		
ASSETS				
Current assets:				
Cash and cash equivalents	\$	19,266	\$	20,742
Marketable securities—available for sale debt securities, current (amortized cost of \$44,702 and \$49,593 as of March 31, 2023 and December 31, 2022, respectively)		43,718		48,182
Accounts receivable, net of allowance of credit losses of \$1,868 and \$1,459 as of March 31, 2023 and December 31, 2022, respectively		21,000		21,447
Deferred sales commissions, current		3,375		3,171
Prepaid expenses and other current assets		5,966		5,211
Total current assets		93,325		98,753
Property and equipment, net		2,405		2,618
Goodwill		47,481		47,481
Intangible assets, net		15,854		16,655
Operating lease right-of-use assets		4,069		4,920
Deposits and other		320		371
Deferred sales commissions, net of current		7,728		7,356
Deferred tax asset, net		23		1
Total assets	\$	171,205	\$	178,155
LIABILITIES & STOCKHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$	5,892	S	5,987
Accrued expenses		11,296	Ť	12,399
Deferred revenue, current		1,456		1,318
Term loan, current		1,122		982
Operating lease liabilities, current		1,500		1,655
Finance lease liabilities, current		5		11
Total current liabilities	_	21,271	_	22,352
Line of credit		320		
Deferred revenue, net of current		395		338
Term loan, net of current		53.041		53,585
Operating lease liabilities, net of current		3,416		3,649
Warrant liability		566		633
Other long-term liabilities		363		363
Total liabilities		79,372		80,920
	_	17,512	_	30,720
Commitments and contingencies (Note 9 and 21)				
Stockholders' equity:				
Preferred stock, \$0.0001 par value per share; 25,000 shares authorized and none issued and outstanding as of March 3 2023 and December 31, 2022.	1,	_		_

Table of Contents

 $Common\ stock, \$0.0001\ par\ value\ per\ share; 500,000\ shares\ authorized\ and\ 92,937\ shares\ issued\ and\ outstanding\ as\ of\ March\ 31,\ 2023;\ 500,000\ shares\ authorized\ and\ 92,729\ shares\ issued\ and\ outstanding\ as\ of\ December\ 31,\ 2022.$ 267,493 264,919 Additional paid-in capital Accumulated other comprehensive loss (1,703)(2,196)Accumulated deficit (173,966) (165,497) Total stockholders' equity 91,833 97,235 Total liabilities & stockholders' equity 171,205 178,155

Consolidated Statements of Operations and Comprehensive Loss (Unaudited) (In thousands, except per share data)

For the three months ended March 2023 2022 Revenue 36,866 \$ 32,093 Cost of revenue 13,262 13,632 Gross profit 23,604 18,461 Operating expenses Sales and marketing expense 13,480 14,652 General and administrative expense 9,171 7,468 Research and development expense 7,983 8,490 Total operating expenses 30,634 30,610 Loss from operations (7,030) (12,149)Interest expense, net 1,096 750 Change in the fair value of warrant liability (67)(392)Other income, net (70)(64)Total other expense, net 959 294 (7,989)Pre-tax loss (12,443)Provision for income taxes 480 544 Net loss (8,469) \$ (12,987)Comprehensive loss \$ (8,469) \$ (12,987)Net loss Other comprehensive income (loss), net of tax Foreign currency translation adjustment 66 (49)Net unrealized gain (loss) on marketable securities 427 (888)Total other comprehensive income (loss), net of tax 493 (937) Comprehensive loss (7,976) (13,924) Net loss per share Net loss per share—basic and diluted (0.09) \$ (0.14)\$ 92,842 Weighted average shares outstanding—basic and diluted 91,478

Consolidated Statements of Stockholders' Equity (Unaudited) (In thousands)

	Commo	n Stoc	k	Additional Paid-in	Accumulated Other Comprehensive	Accumulated	
	Shares		Amount	Capital	Loss	Deficit	Total
Balance at December 31, 2021	90,697	\$	9	\$ 253,468	\$ (477)	\$ (128,022)	\$ 124,978
Foreign currency translation adjustment	_		_	_	(49)	_	(49)
Net unrealized loss on marketable securities	_		_	_	(888)	_	(888)
Stock-based compensation	_		_	2,479	_	_	2,479
Net loss	_		_	_	_	(12,987)	(12,987)
Balance at March 31, 2022	90,697	\$	9	\$ 255,947	\$ (1,414)	\$ (141,009)	\$ 113,533

	Commo	n Sto	ock		Additional Paid-in	Accumulated Other Comprehensive		Accumulated		
	Shares		Amount	Capital		Loss		Deficit		Total
Balance at December 31, 2022	92,729	\$	9	\$	264,919	\$ (2,196)	\$	(165,497)	\$	97,235
Gross issuance of shares upon vesting of stock-based awards	316		_		_	_		_		_
Shares withheld to cover withholding taxes for stock-based awards	(108)		_		(294)	_		_		(294)
Net transfer from LiveVox TopCo	_		_		219	_		_		219
Foreign currency translation adjustment	_		_		_	66		_		66
Net unrealized gain on marketable securities	_		_		_	427		_		427
Stock-based compensation	_		_		2,649	_		_		2,649
Net loss								(8,469)		(8,469)
Balance at March 31, 2023	92,937	\$	9	\$	267,493	\$ (1,703)	\$	(173,966)	\$	91,833

Consolidated Statements of Cash Flows (Unaudited) (Dollars in thousands)

	For the three months 31.	ended March
	2023	2022
Operating activities:		
Net loss	\$ (8,469) \$	(12,987)
Adjustments to reconcile net loss to net cash used in operating activities:		
Depreciation and amortization	243	274
Amortization of identified intangible assets	800	1,073
Amortization of deferred loan origination costs	36	27
Amortization of deferred sales commissions	845	740
Non-cash lease expense	349	453
Stock-based compensation expense	2,649	2,479
Credit loss expense	413	33
Loss on disposition or impairment of asset	509	_
Deferred income tax benefit	(22)	(78)
Net realized loss on sale of marketable securities	50	10
Amortization of premium paid on marketable securities	46	131
Change in the fair value of the warrant liability	(67)	(392)
Changes in assets and liabilities		
Accounts receivable	34	655
Other assets	(702)	(19)
Deferred sales commissions	(1,420)	(695)
Accounts payable	(95)	(2,134)
Accrued expenses	(662)	(3,716)
Deferred revenue	195	(16)
Operating lease liabilities	(389)	(464)
Net cash used in operating activities	(5,657)	(14,626)
Investing activities:	(=,==,)	(= 1,===)
Purchases of property and equipment	(9)	(536)
Purchases of marketable securities	(8,600)	(1,477)
Proceeds from sale of marketable securities	7,604	1,515
Proceeds from maturities and principal paydowns of marketable securities	5,791	242
Net cash provided by (used in) investing activities	4,786	(256)
Financing activities:	4,780	(230)
Repayments on loan payable	(140)	(140)
Proceeds from drawdown on line of credit	320	(140)
Debt issuance costs	(299)	_
Repayments on finance lease obligations	(7)	(6)
Payments of employees' withholding taxes on net share settlement of share-based awards	(294)	(0)
	· /	_
Principal payments under the structured payable arrangement	(441) 219	
Net transfer from LiveVox TopCo		(1.4.6)
Net cash used in financing activities	(642)	(146)
Effect of foreign currency translation	37	(97)
Net decrease in cash, cash equivalents and restricted cash	(1,476)	(15,125)
Cash, cash equivalents, and restricted cash beginning of period	20,742	47,317
Cash, cash equivalents, and restricted cash end of period	\$ 19,266 \$	32,192

	For t	he three mor	ded March
		2023	2022
Supplemental disclosure of cash flow information:			
Interest paid	\$	1,300	\$ 801
Income taxes paid		47	56
Supplemental schedule of noncash investing activities:			
Net unrealized loss (gain) on marketable securities	\$	(427)	\$ 888

Reconciliation of cash, cash equivalents and restricted cash to the consolidated balance sheets (dollars in thousands):

	As of M	arch	31,
	2023		2022
Cash and cash equivalents	\$ 19,266	\$	32,092
Restricted cash, current	_		100
Total cash, cash equivalents and restricted cash	\$ 19,266	\$	32,192

Notes to the Consolidated Financial Statements (Unaudited)

1. Organization

LiveVox Holdings, Inc. (formerly known as Crescent Acquisition Corp("Crescent")), and its subsidiaries (collectively, the "Company," "LiveVox," "we," "us" or "our") is engaged in the business of developing and marketing a cloud-hosted Contact Center as a Service ("CCaaS") customer engagement platform that leverages microservice technology to rapidly innovate and scale digital engagement functionality that also incorporates the capabilities of fully integrated omnichannel customer connectivity, multichannel enabled Customer Relationship Management and Workforce Optimization applications. LiveVox's customers are located primarily in the United States. LiveVox's services are used to initiate and manage customer contact campaigns primarily for companies in the accounts receivable management, tele-sales and customer care industries.

On June 18, 2021 (the "Closing Date" or "Closing"), Crescent, a Delaware corporation, consummated the business combination pursuant to an Agreement and Plan of Merger, dated January 13, 2021 (the "Merger Agreement"), by and among Crescent, Function Acquisition I Corp, a Delaware corporation and direct, wholly owned subsidiary of Crescent ("First Merger Sub"), Function Acquisition II LLC, a Delaware limited liability company and a direct, wholly owned subsidiary of Crescent ("Second Merger Sub"), LiveVox Holdings, Inc., a Delaware corporation ("Old LiveVox"), and GGC Services Holdco, Inc., a Delaware corporation, solely in its capacity as the representative, agent and attorney-in-fact (in such capacity, the "Stockholder Representative") of LiveVox TopCo, LLC ("LiveVox TopCo"), a Delaware limited liability company and the sole stockholder of Old LiveVox as of immediately prior to Closing (the "LiveVox Stockholder"). Pursuant to the Merger Agreement, a business combination between Crescent and Old LiveVox was effected through (a) the merger of First Merger Sub with and into Old LiveVox, with Old LiveVox continuing as the surviving corporation (the "First Merger") and (b) immediately following the First Merger and as part of the same overall transaction as the First Merger, the merger of Old LiveVox with and into Second Merger Sub, with Second Merger Sub continuing as the surviving entity (the "Second Merger", and collectively with the other transactions described in the Merger Agreement, the "Merger"). On the Closing Date, Crescent changed its name to "LiveVox Holdings, Inc." and Second Merger Sub changed its name to "LiveVox Intermediate LLC"

On June 22, 2021, the Company's ticker symbols on The Nasdaq Stock Market LLC ("Nasdaq") for its Class A common stock, warrants to purchase Class A common stock and public units were changed to "LVOX", "LVOXW" and "LVOXU", respectively.

LiveVox, Inc. was a direct, wholly owned subsidiary of Old LiveVox prior to the Merger and is a wholly owned subsidiary of the Company after the Merger. LiveVox, Inc. was first incorporated in Delaware in 1998 under the name "Tools for Health" and in 2005 changed its name to "LiveVox, Inc." On March 21, 2014, LiveVox, Inc. and its subsidiaries were acquired by Old LiveVox. The principal United States operations of the Company are located in San Francisco, California. The Company has five main operating subsidiaries: LiveVox, Inc., which is wholly owned and incorporated in Delaware, LiveVox Colombia SAS which is wholly owned with an office located in Medellin, Colombia, LiveVox Solutions Private Ltd with an office located in Bangalore, India, Speech IQ, LLC which is wholly owned and organized in Ohio, and Engage Holdings, LLC (d/b/a BusinessPhone.com) ("BusinessPhone.com") which is wholly owned and organized in Ohio. Additionally, the Company has a wholly owned subsidiary, LiveVox International, Inc., that is incorporated in Delaware. LiveVox, Inc. and LiveVox International, Inc. own 99.99% and 0.01%, respectively, of LiveVox Solutions Private Ltd.

2. Summary of Significant Accounting Policies

a) Basis of Presentation and Principles of Consolidation

The accompanying unaudited consolidated financial statements have been prepared in accordance with generally accepted accounting principles in the United States ("U.S. GAAP") and applicable rules and regulations of the Securities and Exchange Commission ("SEC") regarding interim financial reporting. Certain information and note disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted pursuant to such rules and regulations or if they substantially duplicate the disclosures contained in the Company's annual audited consolidated financial statements. Therefore, these unaudited interim consolidated financial statements should be read in conjunction with the Company's audited consolidated financial statements and related notes as of and for the year ended December 31, 2022 included in the Company's Annual Report on Form 10-K ("Annual Report") filed with the SEC on March 2, 2023. The information as of December 31, 2022 included in the consolidated balance sheets was derived from those audited consolidated financial statements.

In the opinion of management, the unaudited consolidated financial statements reflect all adjustments, which are normal and recurring in nature, necessary for fair financial statement presentation. All intercompany transactions and balances have been

Notes to the Consolidated Financial Statements (Unaudited)

eliminated in consolidation. Results of operations for the three months ended March 31, 2023 and 2022 are not necessarily indicative of the results to be expected for the full annual periods.

b) Use of Estimates

The preparation of the consolidated financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from those estimates, and such differences could be material to the Company's consolidated financial position and results of operations, requiring adjustment to these balances in future periods. Significant items subject to such estimates and assumptions include, but are not limited to, the determination of the useful lives of long-lived assets, period of benefit of deferred sales commissions, allowances for credit losses, fair value of marketable securities, fair value of goodwill and long-lived assets, fair value of incentive awards, fair value of warrants, establishing standalone selling price, valuation of deferred tax assets, income tax uncertainties and other contingencies, including the Company's ability to exercise its right to repurchase incentive options from terminated employees.

c) Segment Information

The Company has determined that its Chief Executive Officer ("CEO") is its chief operating decision maker. The Company's CEO reviews financial information presented on a consolidated basis for purposes of assessing performance and making decisions on how to allocate resources. Accordingly, the Company has determined that it operates in a single reportable segment.

d) Foreign Currency Translation

The financial position and results of operations of the Company's international subsidiaries are measured using the local currency as the functional currency. Revenue and expenses have been translated into U.S. dollars at average exchange rates prevailing during the periods. Assets and liabilities have been translated at the rates of exchange on the balance sheet date. The resulting translation gain and loss adjustments are recorded directly as a separate component of stockholders' equity (accumulated other comprehensive loss), unless there is a sale or complete liquidation of the underlying foreign investments, or the adjustment is inconsequential.

e) Fair Value of Financial Instruments

Fair value is defined as the price that would be received from the sale of an asset or the transfer of a liability in an orderly transaction between market participants at the measurement date. The Company utilizes a fair value hierarchy to classify fair value amounts of the Company's assets and liabilities recognized or disclosed in the Company's consolidated financial statements based on the lowest level of input that is significant to the fair value measurement. The levels of the hierarchy are described below:

- Level 1—Observable inputs that reflect quoted prices (unadjusted) for identical assets or liabilities in active markets.
- Level 2—Includes other inputs that are directly or indirectly observable in the marketplace.
- Level 3—Unobservable inputs that are supported by little or no market activity.

In determining fair value, the Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs to the extent possible. Observable or market inputs reflect market data obtained from independent sources, while unobservable inputs reflect the Company's assumptions based on the best information available. The Company recognizes transfers into and out of the levels as of the end of each reporting period. Refer to Note 19 for additional information regarding the fair value measurements.

f) Liquidity and Capital Resources

LiveVox's consolidated financial statements have been prepared assuming the Company will continue as a going concern for the 12-month period from the date of issuance of the consolidated financial statements, which contemplates the realization of assets and the settlement of liabilities and commitments in the normal course of business. The Company's main sources of liquidity include:

Notes to the Consolidated Financial Statements (Unaudited)

- Available-for-sale ("AFS") debt securities, which are all classified as short-term securities to fund current operations and may be liquidated at the Company's
 discretion if the need arises. The Company held AFS debt securities of \$43.7 million and \$48.2 million as of March 31, 2023 and December 31, 2022, respectively.
 See Note 4 for more information;
- The term loan and revolving credit facility that the Company entered into with PNC Bank on November 7, 2016 (as amended, the "Credit Facility"), which has been amended several times. On March 31, 2023, the Company entered into the Eighth Amendment to the Credit Facility (the "Eighth Amendment"). As of the Eighth Amendment effective date, the term loan commitment is \$54.9 million, the revolver commitment is \$5.0 million and the letter of credit sublimit is \$1.5 million. See Note 9 and 10 for more information.

The Company's primary use of cash is for operating and administrative activities including employee-related expenses, and general, operating and overhead expenses. Future capital requirements will depend on many factors, including the Company's customer growth rate, customer retention, timing and extent of development efforts, the expansion of sales and marketing activities, the introduction of new and enhanced services offerings, the continuing market acceptance of the Company's services, effective integration of acquisition activities, if any, and maintaining the Company's bank credit facility. Additionally, the duration and extent of the impact from the current macroeconomic and geopolitical conditions continues to depend on future developments that cannot be accurately predicted at this time, such as tight labor market, inflationary pressures, rising interest rates, volatility in foreign exchange rates, supply chain constraints, recessionary fears and the specific impact of these and other factors on LiveVox's business, employees, customers and partners. While those factors have caused operational difficulties, and may continue to create challenges for the Company's performance, they have not, thus far, had a substantial net impact on the Company's liquidity position.

The Company believes it has sufficient financial resources for at least the next 12 months from the date these consolidated financial statements are issued.

g) Debt Discount and Issuance Costs

The Company's debt issuance costs and debt discount are recorded as a direct reduction of the carrying amount of the debt liability and are amortized to interest expense over the contractual term of the term loan.

h) Cash, Cash Equivalents and Restricted Cash

Cash and cash equivalents are stated at fair value. The Company considers all highly liquid investments with an original maturity of 90 days or less to be cash equivalents. The Company limits its credit risk associated with the cash and cash equivalents by placing investments with banks it believes are highly creditworthy. The Company has exposure to credit risk to the extent cash balances exceed amounts covered by Federal deposit insurance. At March 31, 2023 and December 31, 2022, the Company had no cash equivalents or restricted cash. Cash consists of bank deposits.

i) Marketable Securities

The Company invests in various marketable securities. As of March 31, 2023 and December 31, 2022, the Company designated all of these marketable securities as debt securities and classified them as available-for-sale ("AFS"). No debt securities were classified as held-to-maturity ("HTM") or trading. Debt securities are classified as current or non-current, based on maturities and the Company's expectations of sales and redemptions in the next 12 months. The Company determines the appropriate classification of marketable securities at the time of purchase and re-evaluates such designation as of each balance sheet date.

Debt securities classified as AFS are reported at fair value with unrealized gains and losses, net of income taxes, as a separate component of stockholders' equity (accumulated other comprehensive loss) in the consolidated balance sheets until the securities are sold or the unrealized losses are related to credit losses. Gains and losses on sales of AFS debt securities are recorded on the trade date in other income (expense), net, in the consolidated statements of operations and comprehensive loss. The cost of AFS debt securities sold or the amount reclassified out of accumulated other comprehensive loss into earnings is determined using the specific identification method.

At each reporting date, the Company evaluates the amortized cost of AFS debt securities compared to their fair value to determine whether an AFS debt security is impaired. The Company first assesses whether it intends to sell the security or whether it is more likely than not that the Company will be required to sell the security before the recovery of its entire amortized cost basis. If either of these criteria is met, the security's amortized cost basis is written down to fair value through other income (expense), net, in the consolidated statements of operations and comprehensive loss. If neither of these criteria is met, the Company evaluates whether the decline in fair value below amortized cost basis has resulted from credit losses or other factors. In making

Notes to the Consolidated Financial Statements (Unaudited)

this assessment, the Company considers factors such as the extent to which fair value is less than amortized cost basis, any changes to the rating of the security by a rating agency, and any adverse conditions specifically related to the security. Credit related unrealized losses are recognized as an allowance for credit losses in the consolidated balance sheets with a corresponding charge in other income (expense), net, in the consolidated statements of operations and comprehensive loss. Non-credit related unrealized losses are recorded in other comprehensive income (loss), as applicable, net of applicable taxes.

On January 1, 2023, the Company adopted Accounting Standards Update ("ASU") No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments and all subsequent updates (collectively, the "ASC 326") and applied to its AFS debt securities. Please refer to Note 2(aa) for more information about ASC 326. See Note 4 for additional information relating to the Company's marketable securities.

In connection with the adoption of new standards, the Company elected to exclude accrued interest from both the fair value and the amortized cost basis of AFS debt securities and present it within prepaid expenses and other current assets in the Company's consolidated balance sheets. The Company elected not to measure an allowance for credit losses for accrued interest receivable. As such, accrued interest receivable is written off in a timely manner when deemed uncollectible, by reversing previously recognized interest income.

j) Accounts Receivable

Trade accounts receivable are stated net of any write-offs and the allowance for credit losses, at the amount the Company expects to collect. The Company performs ongoing credit evaluations of its customers and generally does not require collateral unless a customer has previously defaulted. The Company maintains an allowance for credit losses, which represents the best estimate of lifetime expected credit losses against the existing accounts receivable, inclusive of unbilled receivables, based on certain factors including the age of the receivable balance, historical write-off experience, past collection experience with the customer, credit quality of the customer, current economic conditions, and reasonable and supportable forecasts of future economic conditions. If the financial condition of the Company's customers were to deteriorate, adversely affecting their ability to make payments, additional allowances would be required. Accounts receivable are charged off against the allowance for credit losses after all means of collection have been exhausted and the potential for recovery is considered remote. Recoveries of accounts receivable previously written off are recorded as income when received.

The allowance for credit losses activity for the three months ended March 31, 2023 and 2022 are as follows (dollars in thousands):

	Three Mor	nths Er ch 31,	ıded	
	 2023	2022		
Balance, beginning of period	\$ 1,459	\$	1,282	
Credit loss expense	413		33	
Accounts receivable write-offs	(4)		(18)	
Accounts receivable recoveries	_		88	
Balance, end of period	\$ 1,868	\$	1,385	

On January 1, 2023, the Company applied ASC 326 to its trade accounts receivable. The Company determined that the allowance for credit losses as of December 31, 2022 recorded under the accounting standards in effect during that period is sufficient and that no additional allowance for credit losses on its trade accounts receivable is required upon adoption of ASC 326. Please refer to Note 2(aa) for more information about ASC 326.

k) Property and Equipment

Property and equipment are stated at cost less accumulated depreciation. Expenditures for major renewals and betterments that extend the useful lives of property and equipment are capitalized. Expenditures for maintenance and repairs, including planned major maintenance activities, are charged to expense as incurred. When assets are retired or disposed, the asset's original cost and related accumulated depreciation are eliminated from the accounts and any gain or loss is reflected in the consolidated statements of operations and comprehensive loss. Amortization expense on capitalized software is included in depreciation expense. Depreciation of leasehold improvements is recorded over the shorter of the estimated useful life of the leasehold improvement or lease terms that are reasonably assured.

Notes to the Consolidated Financial Statements (Unaudited)

Depreciation of property and equipment is provided using the straight-line method based on the following estimated useful lives:

	Years
Computer equipment	3 - 5
Computer software	3
Furniture and fixtures	5 - 10
Leasehold improvements	5
Website development	2

l) Identified Intangible Assets

On March 21, 2014, LiveVox, Inc. and subsidiaries were acquired by LiveVox Holdings, Inc. On October 16, 2019, the Company acquired the rights to certain assets of Teckst Inc. On December 16, 2019, the Company acquired the rights to Speech IQ, LLC. On February 5, 2021, the Company completed its asset acquisition of BusinessPhone. The acquisitions resulted in identified marketing-based, technology-based, customer-based, trademark-based, and workforce-based intangible assets. The fair value of the identified assets was determined as of the date of the acquisition by management with the assistance of an independent valuation firm. The identified intangible assets are being amortized using the straight-line method based on the following estimated useful lives:

	Years
Marketing-based	7
Technology-based	4 - 10
Customer-based	7 - 16
Trademark-based	4
Workforce-based	10

m) Goodwill

Goodwill represents the excess of the purchase price of acquired business over the fair value of the underlying net tangible and intangible assets. The Company performed its annual impairment review of goodwill on October 1 of each year, and when a triggering event occurs between annual impairment tests.

In testing for goodwill impairment, the Company has the option to first assess qualitative factors to determine if it is more likely than not that the fair value of the Company's single reporting unit is less than its carrying amount, including goodwill, or bypass the qualitative assessment and proceed directly to the quantitative impairment test in accordance with Accounting Standards Codification ("ASC") 350-20-35, as amended by ASU No. 2017-04, Intangibles-Goodwill and Other (Topic 350): Simplifying the Test for Goodwill Impairment, to determine if the fair value of the reporting unit exceeds its carrying amount. If the fair value is determined to be less than the carrying value, an impairment charge is recorded for the amount by which the reporting unit's carrying amount exceeds its fair value, limited to the total amount of goodwill allocated to that reporting unit.

During the three months ended March 31, 2023 and 2022, no triggering events have occurred that would require an impairment review of goodwill outside of the required annual impairment review, and therefore, no impairment charges were recorded during those periods. Refer to Note 6 for more information.

n) Impairment of Long-Lived Assets

Long-lived assets to be held and used are reviewed for impairment whenever events or changes in circumstances indicate that the related carrying amount may not be recoverable. When required, impairment losses on assets to be held and used are recognized based on the fair value of the asset and long-lived assets to be disposed of are reported at the lower of the carrying amount or fair value. No impairment losses were recognized during the three months ended March 31, 2023 and 2022.

o) Amounts Due to Related Parties

In the ordinary course of business, the Company has and expects to continue to have transactions with its stockholders and affiliates. Refer to Note 11 for more information.

Notes to the Consolidated Financial Statements (Unaudited)

p) Concentration of Risk

Concentration of Credit and Customer Risk

Financial instruments that potentially subject the Company to significant concentrations of credit risk consist primarily of cash and cash equivalents, marketable securities and accounts receivable. Risks associated with cash and cash equivalents and marketable securities are mitigated using what the Company considers creditworthy institutions. The Company performs ongoing credit evaluations of its customers' financial condition. Substantially all of the Company's assets are in the United States.

As of March 31, 2023 and December 31, 2022, no single issuer represented more than 10% of the Company's marketable securities.

The Company's customers are primarily in the receivables management, tele-sales and customer care industries. During the three months ended March 31, 2023 and 2022, substantially all the Company's revenue was generated in the United States. For the three months ended March 31, 2023 and 2022, no single customer represented more than 10% of the Company's revenue. As of March 31, 2023 and December 31, 2022, no single customer represented more than 10% of the Company's accounts receivable.

Concentration of Supplier Risk

The Company relies on third parties for telecommunication, bandwidth, and co-location services that are included in cost of revenue.

As of March 31, 2023, two vendors accounted for approximately34% of the Company's accounts payable. No other single vendor exceeded 10% of the Company's accounts payable at March 31, 2023. At December 31, 2022, one vendor accounted for approximately 38% of the Company's accounts payable. No other single vendor exceeded 10% of the Company's accounts payable at December 31, 2022. The Company believes there could be a material impact on future operating results should a relationship with an existing significant supplier cease.

q) Revenue Recognition

The Company recognizes revenue in accordance with ASC 606, Revenue from Contracts with Customers.

The Company derives substantially all of its revenue by providing cloud-based contact center products under a usage-based model, with prices calculated on a per-call, per-seat, or, more typically, a per-minute basis and contracted minimum usage in accordance with the terms of the underlying agreements. Other immaterial ancillary revenue is derived from call recording, local caller identification packages, performance/speech analytics, text messaging services and professional services billed monthly on primarily usage-based fees and, to a lesser extent, fixed fees. Revenue is recognized when control of these services is transferred to the Company's customers, in an amount that reflects the consideration it expects to be entitled to in exchange for those services excluding amounts collected on behalf of third parties such as sales taxes, which are collected on behalf of and remitted to governmental authorities based on local tax law.

The Company determines revenue recognition through the following steps:

- a. Identification of the contract, or contracts, with a customer;
- b. Identification of the performance obligations in the contract;
- c. Determination of the transaction price;
- d. Allocation of the transaction price to the performance obligations in the contract; and
- e. Recognition of revenue when, or as, the performance obligations are satisfied.

The Company enters into contracts that can include various combinations of services, each of which are distinct and accounted for as separate performance obligations. The Company's cloud-based contact center solutions typically include a promise to provide continuous access to its hosted technology platform solutions through its data centers. Arrangements with customers do not provide the customer with the right to take possession of the Company's software platform at any time. LiveVox's

Notes to the Consolidated Financial Statements (Unaudited)

performance obligations are satisfied over time as the customer simultaneously receives and consumes the benefits and the Company performs its services. The Company's contract terms typically range from one to three years with payment terms of net 10-60 days. As the services provided by the Company are generally billed monthly there is not a significant financing component in the Company's arrangements.

The Company's arrangements typically include monthly minimum usage commitments and specify the rate at which the customer must pay for actual usage above the monthly minimum. Additional usage in excess of contractual minimum commitments is deemed to be specific to the month that the usage occurs, since the minimum usage commitments reset at the beginning of each month. The Company has determined these arrangements meet the variable consideration allocation exception and therefore, it recognizes contractual monthly commitments and any overages as revenue in the month they are earned.

The Company has service-level agreements with customers warranting defined levels of uptime reliability and performance. Customers may receive credits or refunds if the Company fails to meet such levels. If the services do not meet certain criteria, fees are subject to adjustment or refund representing a form of variable consideration. The Company records reductions to revenue for these estimated customer credits at the time the related revenue is recognized. These customer credits are estimated based on current and historical customer trends, and communications with its customers. Such customer credits have not been significant to date.

For contracts with multiple performance obligations, the Company allocates the contract price to each performance obligation based on its relative standalone selling price ("SSP"). The Company generally determines SSP based on the prices charged to customers. In instances where SSP is not directly observable, such as when the Company does not sell the service separately, the SSP is determined using information that generally includes market conditions or other observable inputs.

Professional services for configuration, system integration, optimization or education are billed on a fixed-price or time and material basis and are performed by the Company directly or, alternatively, customers may also choose to perform these services themselves or engage their own third-party service providers. Professional services revenue, which represents approximately 3% of revenue, is recognized over time as the services are rendered.

Deferred revenue represents billings or payments received in advance of revenue recognition and is recognized upon transfer of control. Balances consist primarily of annual or multi-year minimum usage agreements not yet provided as of the balance sheet date. Deferred revenue that will be recognized during the succeeding twelve-month period is recorded as deferred revenue, current in the consolidated balance sheets, with the remainder recorded as deferred revenue, net of current in the Company's consolidated balance sheets.

r) Costs to Obtain Customer Contracts (Deferred Sales Commissions)

Sales commissions are paid for initial contracts and expansions of existing customer contracts. Sales commissions and related expenses are considered incremental and recoverable costs of acquiring customer contracts. These costs are capitalized and amortized on a straight-line basis over the anticipated period of benefit, which the Company has estimated to be five years. The Company determined the period of benefit by taking into consideration the length of the Company's customer contracts, the customer attrition rate, the life of the technology provided and other factors. Amortization expense is recorded in sales and marketing expense within the Company's consolidated statements of operations and comprehensive loss. Amortization expense for the three months ended March 31, 2023 and 2022 was approximately \$0.8 million and \$0.7 million, respectively. No impairment losses were recognized during the three months ended March 31, 2023 and 2022.

s) Advertising

The Company expenses non-direct response advertising costs as they are incurred. There were no advertising costs capitalized during the three months ended March 31, 2023 and 2022. For the three months ended March 31, 2023 and 2022, advertising expense was approximately \$0.6 million and \$0.6 million, respectively. Advertising expense is included under sales and marketing expense in the accompanying consolidated statements of operations and comprehensive loss.

t) Research and Development Costs

Notes to the Consolidated Financial Statements (Unaudited)

Research and development costs not related to the development of internal use software are charged to operations as incurred. Research and development expenses primarily include payroll and employee benefits, consulting services, travel, and software and support costs.

u) Software Development Costs

The Company capitalizes costs of materials, consultants, payroll, and payroll-related costs of employees incurred in developing internal-use software after certain capitalization criteria are met and includes these costs in the computer software. Refer to Note 5 for additional information. Software development costs are expensed as incurred until preliminary development efforts are successfully completed, management has authorized and committed project funding, it is probable that the project will be completed, and the software will be used as intended. To date, all software development costs have been charged to research and development expense in the accompanying consolidated statements of operations and comprehensive loss. There were no capitalized software development costs related to internal-use software during the three months ended March 31, 2023 and 2022.

v) Income Taxes

Deferred Taxes

The Company accounts for income taxes using the asset and liability approach. Deferred tax assets and liabilities are recognized for the future tax consequences arising from the temporary differences between the tax basis of an asset or liability and its reported amount in the consolidated financial statements, as well as from net operating loss and tax credit carryforwards. Deferred tax amounts are determined by using the tax rates expected to be in effect when the taxes will be paid or refunds received, as provided for under currently enacted tax law. A valuation allowance is provided for deferred tax assets that, based on available evidence, are not expected to be realized.

Enactment of the Tax Cuts and Jobs Act in 2017 subjects a U.S. shareholder to current tax on global intangible low-taxed income ("GILTI") earned by certain foreign subsidiaries. Under U.S. GAAP, an entity can make an accounting policy election to either recognize deferred taxes for temporary differences expected to reverse as GILTI in future years or provide for the tax expense related to GILTI resulting from those items in the year of the GILTI inclusion (i.e., as a period expense). The Company has elected to recognize the tax on GILTI as a period expense in the period of inclusion. As such, no deferred taxes are recorded on the Company's temporary differences that might reverse as GILTI in future years.

Uncertain Tax Positions

The Company recognizes the effect of income tax positions only if those positions are more likely than not to be sustained in a court of last resort. Recognized income tax positions are measured at the largest amount that is greater than 50% likely of being realized. Changes in recognition or measurement are reflected in the period in which the change in judgment occurs. The Company does not believe its consolidated financial statements include any uncertain tax positions. It is the Company's policy to recognize interest and penalties accrued on any unrecognized tax benefit as a component of income tax expense.

w) Stock-Based Compensation

The Company measures compensation expense for stock awards granted to employees and nonemployees in accordance with ASC 718 Compensation—Stock Compensation. Stock-based compensation is measured at fair value on grant date. The Company classified all stock awards as equity awards at the grant date, and reassesses the liability versus equity treatment on a quarterly basis for any changes that have occurred during the period that may result in a reclassification. Equity-classified awards are recognized as stock-based compensation expense over an employee's requisite service period or a nonemployee's vesting period on the basis of the grant date fair value. The Company elects to account for forfeitures as they occur, rather than making estimates of future forfeitures.

Management Incentive Units

During 2019, LiveVox TopCo established a Management Incentive Unit program whereby the LiveVox TopCo board of directors has the power and discretion to approve the issuance of Class B Units that represent management incentive units ("MIUs") to any manager, director, employee, officer or consultant of the Company or its subsidiaries. Vesting begins on the date of issuance, and the MIUs vest ratably over five years with 20% of the MIUs vesting on each anniversary of a specified vesting commencement

Notes to the Consolidated Financial Statements (Unaudited)

date, subject to the grantee's continued employment with the Company on the applicable vesting date. Vesting of the MIUs will accelerate upon consummation of a "sale of the company", which is defined by the LiveVox TopCo limited liability company agreement.

The Company measures stock-based compensation expense for MIUs based on the grant date fair value of the award estimated by using a Monte Carlo simulation. Monte Carlo simulation is a widely accepted approach for financial instruments with path dependencies. The Company records stock-based compensation expense for the issued and outstanding MIUs based on the service condition on a straight-line basis over the requisite service period of five years, reduced for actual forfeited MIUs.

Please see Note 15 for further detail about stock-based compensation expenses related to MIUs under the Management Incentive Unit program.

2021 Equity Incentive Plan

On June 16, 2021, the stockholders of the Company approved the 2021 Equity Incentive Plan (the "2021 Plan"), which became effective upon the closing of the Merger on June 18, 2021. As of March 31, 2023, the number of shares reserved for issuance is 19,695,679. The Company grants Restricted Stock Units ("RSUs") and Performance-based Restricted Stock Units ("PSUs") awards to employees, executives, directors, and eligible consultants of the Company.

RSUs are subject to service conditions only. The Company estimates the grant date fair value of RSUs using the closing price of the Company's Class A common stock on Nasdaq on the measurement date. Stock-based compensation expense for RSUs issued to employees is recognized on a straight-line basis over the vesting period for the entire award, reduced for actual forfeited RSUs. Stock-based compensation expense for RSUs issued to nonemployees is recognized as the goods are received or services are performed. The requisite service period typically ranges from one to six years based on the grantee's role in the Company. The amount of cumulative compensation cost recognized at any date must at least equal the portion of the grant-date value of the award that is vested at that date.

PSUs, which are granted to certain key employees, vest either based on the achievement of predetermined market conditions or based on both service and market conditions. The Company estimates the grant date fair value of PSUs using the Monte Carlo simulation. The Company recognizes stock-based compensation expense for PSUs on a tranche-by-tranche basis (i.e., the accelerated attribution method) over an employee's requisite service period, which is the longer of the time-vesting period or the derived service period inferred from the valuation model. Stock-based compensation expense of equity-classified PSUs is recognized provided that the good is delivered or the service is rendered, regardless of when, if ever, the market conditions are satisfied.

Payment of the underlying shares in connection with the vesting of employee RSUs and PSUs generally triggers a tax obligation for the employee, which is required to be remitted to the relevant tax authorities. The Company withholds otherwise deliverable RSU or PSU shares having a fair value at the vest date equal to the maximum statutory withholding tax amount and remits the remaining RSU or PSU shares to the employee recipients. Any cash received and paid to meet an employees' statutory withholding tax requirement is reflected as a financing activity within the consolidated statements of cash flows. During the three months ended March 31, 2023, the Company withheld 107,733 shares to cover employee recipients' withholding tax obligations.

Nonemployee directors acting in their role as members of a board of directors are treated as employees for purpose of ASC 718 if (a) those directors were elected by the Company's shareholders and (b) the awards granted to nonemployee directors are for their services as directors but not for other services. While a nonemployee director may be considered an employee under ASC 718, he or she is not considered an employee under the IRS statutory withholding requirements. As a result, no shares are withholding taxes for an award issued to a nonemployee director. Independent consultants are nonemployees under the IRS statutory withholding requirements. As a result, no shares are withheld to cover withholding taxes for an award issued to an independent consultant.

Please see Note 15 for further detail about stock-based compensation expenses related to RSUs and PSUs under the 2021 Plan.

x) Acquisitions

The Company evaluates acquisitions of assets and other similar transactions to assess whether or not the transaction should be accounted for as a business combination or asset acquisition by first applying a screen test to determine if substantially all of the

Notes to the Consolidated Financial Statements (Unaudited)

fair value of the gross assets acquired is concentrated in a single identifiable asset or group of similar identifiable assets. If the screen is met, the transaction is accounted for as an asset acquisition. If the screen is not met, further determination is required as to whether or not the Company has acquired inputs and processes that have the ability to create outputs which would meet the definition of a business. Significant judgment is required in the application of the screen test to determine whether an acquisition is a business combination or an acquisition of assets.

y) Public and Forward Purchase Warrants

Prior to the Merger, Crescent issued7,000,000 private placement warrants ("Private Warrants") and 12,499,995 public warrants ("Public Warrants") at the close of Crescent's initial public offering ("IPO") on March 7, 2019. As an incentive for LiveVox to enter into the Merger Agreement, pursuant to the Sponsor Support Agreement dated January 13, 2021, Crescent's sponsor agreed to the cancellation of all of the Private Warrants prior to the Closing Date. In addition, 833,333 Forward Purchase Warrants ("Forward Purchase Warrants") were issued pursuant to the Forward Purchase Agreement dated January 13, 2021 between Crescent and Old LiveVox. The 12,499,995 Public Warrants and the 833,333 Forward Purchase Warrants (collectively, the "Warrants") remain outstanding after the Merger. Each whole Warrant entitles the holder to purchase one share of the Company's Class A common stock at a price of \$1.50 per share, subject to adjustments. The Warrants are exercisable at any time prior to June 18, 2026. See Note 12 for further information on stock warrants.

Upon consummation of the Merger, the Company concluded that (a) the Public Warrants meet the derivative scope exception for contracts in the Company's own stock and are recorded in stockholders' equity and (b) the Forward Purchase Warrants do not meet the derivative scope exception and are accounted for as derivative liabilities. Specifically, the Forward Purchase Warrants contain provisions that cause the settlement amounts to be dependent upon the characteristics of the holder of the Warrant which is not an input into the pricing of a fixed-for-fixed option on equity shares. Therefore, the Forward Purchase Warrants are not considered indexed to the Company's stock and should be classified as a liability. Since the Forward Purchase Warrants meet the definition of a derivative, the Company recorded the Forward Purchase Warrants as liabilities on the consolidated balance sheets at fair value upon the Merger, with an offsetting entry to additional paid-in capital. The gain or loss resulting from decrease or increase in the fair value of the Forward Purchase Warrants in the subsequent periods are recognized in the consolidated statements of operations and comprehensive loss. The fair value of the Forward Purchase Warrants was measured using the Black-Scholes option-pricing model at each measurement date. See Note 19 for further information on fair value

z) Restructuring Charges

Restructuring charges associated with management-approved restructuring plans may include employee severance and termination benefits, lease and non-lease contract termination costs, impairment of long-lived assets, and other related costs associated with restructuring activities.

The Company accounts for employee severance and termination benefits that represent a one-time benefit in accordance with ASC 420 Exit or Disposal Cost Obligations. The Company accrues employee severance and termination benefits associated with an one-time benefit arrangement when employees are notified of their termination benefits.

The Company records employee severance and termination benefits in accordance with ASC 712, Compensation - Nonretirement and Postemployment Benefits, if it pays the benefits as part of an on-going benefit arrangement, which includes benefits provided as part of its established severance policies, a consistent past practice or in accordance with statutory requirements. The Company accrues employee severance and termination benefits associated with an on-going benefit arrangement when the payment is probable and the amount is reasonably estimable.

Non-lease contract termination costs and other costs associated with restructuring activities are recorded in accordance with ASC 420.

Please see Note 22 for a full description of the Company's restructuring actions.

aa) Recently Adopted Accounting Pronouncements

As an emerging growth company ("EGC"), the Jumpstart Our Business Startups Act ("JOBS Act") allows the Company to delay adoption of new or revised accounting pronouncements applicable to public companies until such pronouncements are applicable to private companies. The Company has elected to use this extended transition period under the JOBS Act until such time the Company is no longer considered to be an EGC. The Company adopted the following new accounting pronouncements during the three months ended March 31, 2023:

Notes to the Consolidated Financial Statements (Unaudited)

ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326) and Codification Improvement Amendments

In June 2016, the Financial Accounting Standards Board ("FASB") issued ASU No. 2016-13, which changes the impairment model for most financial assets, which includes the Company's accounts receivable and other financial instruments. The new standard replaces the existing incurred loss impairment model with a current expected credit loss impairment model. It also eliminates the concept of other-than-temporary impairment and requires credit losses related to AFS debt securities to be recorded through an allowance for credit losses rather than as a reduction in the amortized cost basis of the securities. In November 2018, the FASB issued ASU No. 2018-19, Codification Improvements to Topic 326, Financial Instruments—Credit Losses, which clarifies that receivables arising from operating leases are not within the scope of Topic 326, Financial Instruments—Credit Losses. Instead, impairment of receivables arising from operating leases should be accounted for in accordance with Topic 842Leases. In April 2019, the FASB issued ASU No. 2019-04, Codification Improvements to Topic 326, Financial Instruments—Credit Losses, Topic 815, Derivatives and Hedging, and Topic 825, Financial Instruments, which clarifies treatment of certain credit losses. In May 2019, the FASB issued ASU No. 2019-05 Financial Instruments—Credit Losses (Topic 326): Targeted Transition Relief, which permits an entity, upon adoption of ASU No. 2016-13, to irrevocably elect the fair value option (on an instrument-byinstrument basis) for eligible financial assets measured at amortized cost basis. In November 2019, FASB issued ASU No. 2019-10, Financial Instruments—Credit Losses (Topic 326), Derivatives and Hedging (Topic 815), and Leases (Topic 842), which changes the effective dates for Topic 326 to give implementation relief to certain types of entities. In November 2019, the FASB issued ASU No. 2019-11, Codification Improvements to Topic 326, Financial Instruments—Credit Losses, which includes various narrow-scope improvements and clarifications. In March 2020, the FASB issued ASU No. 2020-03, Codification Improvements to Financial Instruments, which clarifies and improves certain financial instruments guidance. In March 2022, the FASB issued ASU No. 2022-02, Financial Instruments—Credit Losses (Topic 326): Troubled Debt Restructurings and Vintage Disclosures, which eliminates the recognition and measurement guidance for a troubled debt restructuring for creditors that have adopted ASU No. 2016-13 and also requires public business entities to present current-period gross write-offs (on a current year-to-date basis for interim-period disclosures) by year of origination in their vintage disclosures. ASU 2016-13 and codification improvement amendments to ASC 2016-13 are effective for public business entities for fiscal years beginning after December 15, 2019, and interim periods within those fiscal years. For all other entities, for annual reporting periods beginning after December 15, 2022 and interim periods within those fiscal years. The Company adopted these new standards effective January 1, 2023 on a modified retrospective basis, and the adoption did not result in a material cumulative-effect adjustment in its consolidated financial statements. Please refer to Note 2(i) and Note 2(j) for additional information relating to the Company's application of new standards to its marketable securities and accounts receivable, respectively.

ASU No. 2022-04, Liabilities—Supplier Finance Programs (Subtopic 405-50)

In September 2022, the FASB issued ASU No. 2022-04, Liabilities—Supplier Finance Programs (Subtopic 405-50): Disclosure of Supplier Finance Program Obligations, which requires entities that use supplier finance programs in connection with the purchase of goods and services to disclose the key terms of the programs and information about their obligations outstanding at the end of the reporting period, including a rollforward of those obligations. The guidance is effective for all entities for fiscal years beginning after December 15, 2022, including interim periods in those fiscal years, except for the rollforward requirement, which is effective for fiscal years beginning after December 15, 2023. The Company adopted this standard effective January 1, 2023 utilizing a retrospective method of transition, except for the rollforward requirement which the Company applied prospectively. The adoption did not have a material impact on the Company's consolidated financial statements and related disclosures. See Note 7 for further information on the Company's supplier finance program.

bb) Recently Issued Accounting Pronouncements

ASU No. 2020-06, Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity (Subtopic 815-40)

In August 2020, the FASB issued ASU No. 2020-06, Debt—Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging—Contracts in Entity's Own Equity, (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity, which simplifies the accounting for convertible instruments and contracts on an entity's own equity, including removing certain conditions for equity classification, and amending certain guidance on the computation of EPS for contracts on an entity's own equity. The guidance is effective for public business entities for fiscal years beginning after December 15, 2021, and interim periods within those fiscal years. For all other entities, the guidance is effective for annual reporting periods beginning after December 15, 2023, and interim periods within fiscal years. Early adoption is permitted, but no earlier than fiscal years beginning after December 15, 2020, and interim periods within those fiscal years. Entities can elect to

Notes to the Consolidated Financial Statements (Unaudited)

adopt the guidance through either a modified retrospective method of transition or a fully retrospective method of transition. The Company is currently evaluating the impact this pronouncement will have on its consolidated financial statements and plans to adopt this standard effective January 1, 2024.

3. Revenue

Contract Balance

The following table provides information about accounts receivable, net, and contract liabilities from contracts with customers. The Company did not have any contract assets as of March 31, 2023 and December 31, 2022 (dollars in thousands):

	Mar	ch 31, 2023	December 31, 2022		
Accounts receivable, net	\$	21,000	\$	21,447	
Contract liabilities, current (deferred revenue)		1,456		1,318	
Contract liabilities, non-current (deferred revenue)		395		338	

Changes in the contract liabilities balances are as follows (dollars in thousands):

	March 31, 2023			ber 31, 2022	- \$	Change
Contract liabilities (deferred revenue)	\$	1,851	\$	1,656	\$	195

The increase in deferred revenue was due to billings in advance of performance obligations being satisfied, net of revenue recognized for services rendered during the period. Revenue of \$0.8 million and \$0.5 million was recognized during the three months ended March 31, 2023 and 2022 which was included in the deferred revenue balance at the beginning of the period.

Remaining Performance Obligations

Remaining performance obligations represent the contracted minimum usage commitments and do not include an estimate of additional usage in excess of contractual minimum commitments. The Company's contract terms typically range from one to three years. Revenue as of March 31, 2023 that has not yet been recognized was approximately \$167.9 million, of which \$90.1 million and \$77.8 million is expected to be recognized as revenue withinone year and beyond one year, respectively. As of March 31, 2023, the Company expects to recognize revenue on the remaining performance obligations over the next 51 months.

4. Marketable Securities

As of March 31, 2023 and December 31, 2022, the Company designated all marketable securities as debt securities and classified them as AFS. There were no transfers of debt securities among AFS, HTM and trading categories during the three months ended March 31, 2023 and 2022.

The following table presents the amortized cost, gross unrealized gains and losses, and fair value of the Company's AFS debt securities at March 31, 2023 aggregated by major security type (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	Amortized Cost		Gross Unrealized Gain		Gross Unrealized Loss		F	air Value
U.S. corporate securities	\$	34,169	\$	18	\$	(785)	\$	33,402
U.S. government securities		2,442		21		_		2,463
Asset-backed securities		7,343		16		(226)		7,133
Other debt securities		748		_		(28)		720
Total available for sale securities	-	44,702		55		(1,039)		43,718
Total debt securities	\$	44,702	\$	55	\$	(1,039)	\$	43,718

The following table presents the amortized cost, gross unrealized gains and losses, and fair value of the Company's AFS debt securities at December 31, 2022 aggregated by major security type (dollars in thousands):

	Amortized Cost		Gross Unrealized Gain		Gross Unrealized Loss		F	air Value
U.S. corporate securities	\$	40,186	\$	4	\$	(1,112)	\$	39,078
U.S. government securities		1,479		_		(2)		1,477
Asset-backed securities		7,181		8		(277)		6,912
Other debt securities		747		_		(32)		715
Total available for sale securities		49,593		12		(1,423)		48,182
Total debt securities	\$	49,593	\$	12	\$	(1,423)	\$	48,182

Refer to Note 19 for additional information regarding the fair value measurements of the Company's marketable securities.

Maturity Analysis

The following table presents the amortized cost and fair value of the Company's AFS debt securities by contractual maturities at March 31, 2023 (dollars in thousands):

As of March 31, 2023	Amo	ortized Cost	Fair Value		
Due in one year or less	\$	18,898	\$	18,399	
Due after one year through five years		25,804		25,319	
Total available for sale securities		44,702		43,718	
Total debt securities	\$	44,702	\$	43,718	

Sales of Marketable Securities

Proceeds from sale of AFS debt securities and the associated gains and losses realized in earnings during the three months ended March 31, 2023 and 2022 are listed below (dollars in thousands):

	Three Months Ended March 31,					
		2023	2022			
Proceeds from sale of available for sale debt securities	\$	7,604	\$	1,515		
Gross realized gain	\$	_	\$	_		
Gross realized loss		(50)		(10)		
Net realized loss on sale of available for sale debt securities	\$	(50)	\$	(10)		

Notes to the Consolidated Financial Statements (Unaudited)

Allowance for Credit Losses

At March 31, 2023, the Company reviewed 67 individual AFS debt securities in unrealized loss positions and determined that it does not intend to sell these securities and it is not more likely than not that it will be required to sell these securities before recovery of their amortized cost bases. The Company concluded that the unrealized losses identified as of March 31, 2023 are due to short term interest rate fluctuations, and not credit losses. Further, the Company noted that the present value of future cash flows discounted using the effective interest rate at the date the security was acquired was equal to or greater than the book value of the security, which further supports the conclusion that there is no credit loss. As such, the Company determined no credit loss existed and did not record an allowance for credit losses for its AFS debt securities at March 31, 2023. The Company will continue to monitor its AFS debt securities on a quarterly basis to assess whether there have been any additional indicators of credit losses.

The following table presents the fair value and unrealized losses of the Company's AFS debt securities that are in unrealized loss positions and for which an allowance for credit losses has not been recorded at March 31, 2023 (dollars in thousands):

	I	n Unrealized Less Thai		1	In Unrealized Loss Position For 12 Months Or Longer			
	Gross Unrealized Fair Value Loss				F	air Value	Gross	Unrealized Loss
U.S. corporate securities	\$	22,612	\$	(582)	\$	10,790	\$	(203)
Asset-backed securities		1,396		(33)		5,737		(193)
Other debt securities		720		(28)		_		_
Total available for sale securities		24,728		(643)		16,527		(396)
Total debt securities	\$	24,728	\$	(643)	\$	16,527	\$	(396)

At December 31, 2022, the Company reviewed 83 individual AFS debt securities in unrealized loss positions and determined that it did not intend to sell these securities and it is not more likely than not that it would be required to sell these securities before recovery of their amortized cost bases. The Company concluded that the unrealized losses identified as of December 31, 2022 were due to short term interest rate fluctuations, and did not result from credit losses. Further, the Company noted that the present value of future cash flows discounted using the effective interest rate at the date the security was acquired was equal to or greater than the book value of the security, which further supports the conclusion that there was no credit loss. As such, the unrealized loss remains appropriately recorded in other comprehensive income (loss) as of December 31, 2022 and is not adjusted at adoption of ASC 326.

The following table presents the fair value and unrealized losses of the Company's AFS debt securities that are in an unrealized loss position at December 31, 2022 (dollars in thousands):

	In Unrealized Loss Position For Less Than 12 Months					In Unrealized Loss Position For 12 Months Or Longer			
	Gross Unrealized Fair Value Loss				F	air Value	Gross	Unrealized Loss	
U.S. corporate securities	\$	23,625	\$	(464)	\$	15,453	\$	(648)	
U.S. government securities		995		(2)		482		_	
Asset-backed securities		1,034		(13)		5,878		(264)	
Other debt securities		_		_		715		(32)	
Total available for sale securities		25,654		(479)		22,528		(944)	
Total debt securities	\$	25,654	\$	(479)	\$	22,528	\$	(944)	

Accrued Interest

Notes to the Consolidated Financial Statements (Unaudited)

Accrued interest receivable on AFS debt securities at March 31, 2023 and December 31, 2022 was \$0.3 million and \$0.2 million, respectively, and was excluded from both the fair value and the amortized cost basis of AFS debt securities and was recorded within prepaid expenses and other current assets in the Company's consolidated balance sheets. There was no interest reversed during the three months ended March 31, 2023 and 2022.

5. Property and Equipment

Property and equipment consisted of the following at March 31, 2023 and December 31, 2022 (dollars in thousands):

	Marc	h 31, 2023	Decen	nber 31, 2022
Computer software	\$	426	\$	426
Computer equipment		3,452		3,408
Furniture and fixtures		1,741		1,736
Leasehold improvements		1,530		1,525
Total		7,149		7,095
Less: accumulated depreciation and amortization		(4,744)		(4,477)
Property and equipment, net	\$	2,405	\$	2,618

Depreciation and amortization expense for property and equipment for the three months ended March 31, 2023 and 2022 totaled \$0.2 million and \$0.3 million, respectively. Amortization of computer software charged to operations for the three months ended March 31, 2023 and 2022 was immaterial for both periods, and is included in depreciation expense.

6. Goodwill and Identified Intangible Assets

Goodwill

Goodwill was recorded as a result of the acquisition of the Company in 2014 by funds affiliated with Golden Gate Capital and the acquisitions made by the Company in 2019 of Teckst Inc. and SpeechIQ LLC.

Subsequent to the annual impairment test completed during the fourth quarter of 2022, the Company believes there have been no triggering events that would require an impairment review of goodwill outside of the required annual impairment review. For the three months ended March 31, 2023 and 2022, there was no impairment to the carrying value of the Company's goodwill.

There were no changes in the carrying amount of goodwill during the three months ended March 31, 2023 or the year ended December 31, 2022.

Identified Intangible Assets

Intangible assets were acquired in connection with the acquisition of the Company in March 2014 by Golden Gate Capital, and the Company's acquisition of Teckst Inc., SpeechIQ LLC and BusinessPhone in October 2019, December 2019, and February 2021, respectively.

Amortization expense related to the Company's identified intangible assets was \$0.8 million and \$1.1 million for the three months ended March 31, 2023 and 2022, respectively. On the face of the consolidated statements of operations and comprehensive loss the amortization of technology-based intangible assets is included within cost of revenue, the amortization of marketing-based and customer-based intangible assets are included within sales and marketing expense, and the amortization of the acquired workforce is included within cost of revenue and research and development expense.

Identified intangible assets consisted of the following at March 31, 2023 (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	Cost		Accumulated Amortization		Carrying Amount	Weighted Average Remaining Life (In Years)
Marketing-based	\$ 1,400	\$	(1,347)	\$	53	0.71
Technology-based	18,300		(17,337)		963	1.02
Customer-based	27,700		(13,100)		14,600	7.13
Workforce-based	380		(142)		238	6.28
	\$ 47,780	\$	(31,926)	\$	15,854	

Identified intangible assets consisted of the following at December 31, 2022 (dollars in thousands):

	Cost	Accumulated Amortization		Carrying Amount	Weighted Average Remaining Life (In Years)
Marketing-based	\$ 1,400	\$ (1,328)	\$	72	0.96
Technology-based	18,300	(17,082)		1,218	1.25
Customer-based	27,700	(12,581)		15,119	7.38
Workforce-based	380	(134)		246	6.48
	\$ 47,780	\$ (31,125)	\$	16,655	

Future amortization of identified intangible assets at March 31, 2023 is shown below (dollars in thousands):

As of March 31, 2023	Amount
Remaining 2023	\$ 2,381
2024	2,321
2025	2,106
2026	2,088
2027	2,020
2028 and beyond	4,938
Total future identified intangible asset amortization	\$ 15,854

7. Accrued Expenses

Accrued expenses consisted of the following at March 31, 2023 and December 31, 2022 (dollars in thousands):

	Mar	ch 31, 2023	Decen	nber 31, 2022
Accrued bonuses	\$	1,716	\$	4,078
Accrued paid time off		2,457		2,743
Accrued commissions		2,406		1,726
Principal and interest payable under the structured payable arrangement		_		444
Other accrued expenses		4,717		3,408
Total accrued expenses	\$	11,296	\$	12,399

In July 2022, the Company entered into a financing arrangement with a third-party intermediary to establish a structured payable arrangement related to the Company's commercial insurance policy on directors and officers. The structured payable arrangement was paid through three quarterly installment payments on August 22, 2022, November 22, 2022 and February 22,

Notes to the Consolidated Financial Statements (Unaudited)

2023. The following table summarizes the changes in the principal and interest payable under the structured payable arrangement during the three months ended March 31, 2023 (dollars in thousands):

	Structure arrang	
Liability as of December 31, 2022	\$	444
Cash payments		(444)
Liability as of March 31, 2023	\$	_

8. Leases

The Company accounts for operating leases and finance leases in accordance with ASC 842, Leases.

The Company has leases for offices, data centers and other computer and networking equipment that expire at various dates through 2027. The Company's leases have remaining terms of one to four years, and some of the leases include a Company option to extend the leases. As the Company's leases do not provide an implicit rate, the Company uses an incremental borrowing rate based on the information available at the commencement date in determining the present value of lease payments. The Company has elected the practical expedient on not separating lease components from non-lease components for right-of-use assets.

As of March 31, 2023 and December 31, 2022, the balance of the Company's finance lease was immaterial.

The operating lease cost was as follows (dollars in thousands):

	Three Months Ended March 31,			
	 2023		2022	
Operating lease cost	\$ 930	\$	520	

On January 30, 2023, the Company announced the closure of certain underutilized physical offices within the United States as more of Company's employees shift to a hybrid or remote work environment. During the three months ended March 31, 2023, the Company recognized an impairment charge of \$0.5 million to reflect the write-down of the carrying amount excess over the fair value of the right-of-use asset for the offices closed, which is included in the operating lease cost discussed above. No impairment charge was recorded during the three months ended March 31, 2022.

Supplemental cash flow information related to operating leases was as follows (dollars in thousands):

	Thr	Three Months Ended March 31,			
	2	023		2022	
Cash paid for amounts included in the measurement of lease liabilities:					
Operating cash used in operating leases	\$	485	\$	565	
Right-of-use assets obtained in exchange for lease obligations:					
Operating leases	\$	_	\$	_	

Supplemental balance sheet information related to operating leases was as follows (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	M	March 31, 2023				ember 31, 2022
Operating lease right-of-use assets	\$	4,069	\$	4,920		
Operating lease liabilities:						
Operating lease liabilities—current	\$	1,500	\$	1,655		
Operating lease liabilities—less current portion		3,416		3,649		
Total operating lease liabilities	\$	4,916	\$	5,304		

Weighted average remaining terms were as follows:

	March 31, 2023	December 31, 2022
Weighted average remaining lease term		
Operating leases	3.63 years	3.77 years

Weighted average discount rates were as follows:

	March 31, 2023	December 31, 2022
Weighted average discount rate		
Operating leases	9.0 %	8.9 %

Maturities of lease liabilities were as follows (dollars in thousands):

As of March 31, 2023	Operating leases
Remaining 2023	\$ 1,436
2024	1,217
2025	1,055
2026	1,009
2027	512
Total lease payments	 5,229
Less: imputed interest	(313)
Total	\$ 4,916

As of March 31, 2023, the Company had no operating leases which had been executed but not yet commenced.

9. Borrowings Under Term Loan and Line of Credit

At March 31, 2023 and December 31, 2022, term loan borrowings were as follows (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	March 31, 2023		Decen	nber 31, 2022
Total term loan obligations	\$	54,163	\$	54,567
Less: current portion of term loan		(1,122)		(982)
Long-term term loan obligations	\$	53,041	\$	53,585

The Company entered into a term loan and revolving credit facility with PNC Bank on November 7, 2016 (as amended, the "Credit Facility"), which has been amended several times. The term loan is due December 31, 2025. On March 31, 2023, the Company entered into the Eighth Amendment to the Credit Facility (the "Eighth Amendment,"). The Eighth Amendment, among other changes, (a) resets the existing annual recurring revenue financial covenant beginning with the testing period ending March 31, 2023, and (b) amends the interest rate provisions, including to replace the existing LIBOR benchmark interest rate with an adjusted SOFR benchmark interest rate (as defined in the agreement governing the Credit Facility). As amended, borrowings under the Credit Facility will accrue interest at a per annum rate, at the Company's option, based on either (a) a base rate (as defined in the agreement governing the Credit Facility) plus a margin of 4.0% or (b) an adjusted term SOFR (based on one- or three-month interest periods) plus a margin of 5.0%. The Company elected a LIBOR rate at March 31, 2023 and the interest rate for the term loan portion of the Credit Facility was 9.6%.

As of the Eighth Amendment effective date, the term loan commitment is \$5.9 million, the revolver commitment is \$5.0 million and the letter of credit sublimit is \$1.5 million. Term loan repayments made by the Company totaled \$0.1 million and \$0.1 million during the three months ended March 31, 2023 and 2022, respectively.

The Company incurred original issue discount of \$0.3 million related to the Eighth Amendment, which, along with the remaining balance of the previously deferred original issue discount and loan fees in the amount of \$0.3 million related to the original Credit Facility and all previous amendments, is amortized and recorded to interest expense over the remaining term of the amended Credit Facility using the effective interest method. Third party loan fees associated with the Eighth Amendment were \$0.1 million and expensed upon close of the loan. Total unamortized loan costs associated with the term loan totaled \$0.6 million and \$0.3 million at March 31, 2023 and December 31, 2022, respectively and are recorded within term loan, net of current.

The Company was in compliance with all debt covenants at March 31, 2023 and December 31, 2022 and was in compliance with all debt covenants as of the date of issuance of these consolidated financial statements. There was no unused borrowing capacity under the term loan portion of the Credit Facility at March 31, 2023 and December 31, 2022. On March 31, 2023, the Company drew down approximately \$0.3 million under the revolving portion of the Credit Facility to pay for the original issue discount related to the Eighth Amendment. The interest rate for the revolving portion of the Credit Facility on March 31, 2023 was 12.0%. There were no amounts outstanding under the revolving portion of the Credit Facility as of December 31, 2022.

Aggregate principal maturities of the term loan as of March 31, 2023 was as follows (dollars in thousands):

As of March 31, 2023	Amount to Mature
Remaining 2023	\$ 841
2024	1,753
2025	52,158
Total	\$ 54,752

The net carrying amount of the liability component of the term loan was as follows (dollars in thousands):

	Mai	March 31, 2023		cember 31, 2022
Principal	\$	54,752	\$	54,893
Less: Unamortized issuance costs		(589)		(326)
Net carrying amount	\$	54,163	\$	54,567

Notes to the Consolidated Financial Statements (Unaudited)

10. Letters of Credit

On November 8, 2016, the Company established an irrevocable standby letter of credit using a sub-facility under the Credit Facility, to serve as a security deposit for the Company's San Francisco office. The letter of credit automatically extends for one-year periods from the original expiration date, September 10, 2017, unless written notice is presented to the beneficiary at least 60 days prior to the current expiration date. The irrevocable standby letter of credit has been amended several times, and the latest amendment on July 20, 2022 decreased the total amount to \$0.2 million. All other terms and conditions remained unchanged.

On February 11, 2022, the Company established another irrevocable standby letter of credit in the amount of \$0.3 million using a sub-facility under the Credit Facility, to serve as a guarantee in connection with the buildout of the office that LiveVox Colombia SAS leases in Medellin, Colombia. The standby letter of credit expired on February 11, 2023, which is one year from the issuance date.

11. Related Party Transactions

The Company pays monthly board of director fees plus reimbursement of expenses incurred on behalf of the Company to members of the Company's board of directors. During the three months ended March 31, 2023, board of director fees totaled \$0.3 million and expense reimbursements were immaterial. The Company also granted RSU awards to directors on August 18, 2021 and August 5, 2022 under the 2021 Plan. During the three months ended March 31, 2023, stock-based compensation expense relating to the RSU awards to the board of directors totaled \$0.5 million. During the three months ended March 31, 2022, board of director fees totaled \$0.2 million, expense reimbursements were immaterial, and stock-based compensation expense relating to the RSU awards to the board of directors totaled \$0.1 million. The unpaid balance of board of director fees due to related parties was \$0.1 million and immaterial as of March 31, 2023 and December 31, 2022.

There were no related party accounts receivable as of March 31, 2023 and December 31, 2022.

The related party transactions with affiliate companies were immaterial during the three months ended March 31, 2023 and 2022.

12. Stock Warrants

Public and Forward Purchase Warrants

Immediately following the Merger, LiveVox assumed 833,333 Forward Purchase Warrants and 12,499,995 Public Warrants that had been previously issued by Crescent. Each whole Warrant entitles the holder to purchase one share of the Company's Class A common stock at a price of \$\mathbb{1}\$ 1.50 per share, subject to adjustments.

The Company may redeem the outstanding Public Warrants, in whole and not in part, upon a minimum of 30 days' prior written notice of redemption ("Redemption Period"). For purposes of the redemption, "Redemption Price" shall mean the last reported sales price of the Company's common stock for any twenty trading days within the thirty trading-day period ending on the third trading day prior to the date on which notice of the redemption is given. The Company may redeem the outstanding Public Warrants for cash at a price of \$0.01 per Warrant if the Reference Value equals or exceeds \$18.00 per share. The warrant holders have the right to exercise their outstanding Warrants prior to the scheduled redemption date during the Redemption Period at \$11.50 per share. If the Company calls the Public Warrants for redemption, the Company will have the option to require all holders that wish to exercise the Public Warrants to do so on a "cashless basis", as described in the warrant agreement.

The Forward Purchase Warrants and the shares of Class A common stock issuable upon the exercise of the Forward Purchase Warrants are transferable, assignable and salable, subject to certain limited exceptions. Additionally, the Forward Purchase Warrants are exercisable for cash or on a cashless basis, at the holder's option, and are non-redeemable so long as they are held by the initial purchasers or their permitted transferees. If the Forward Purchase Warrants are held by someone other than the initial purchasers or their permitted transferees then such Warrants will be redeemable by the Company and exercisable by the warrant holders on the same basis as the Public Warrants.

As of March 31, 2023, there were 13,333,328 Warrants outstanding, and no Warrants have been exercised.

13. Stockholders' Equity

Notes to the Consolidated Financial Statements (Unaudited)

Common Stock

On June 22, 2021, the Company's Class A common stock, publicly traded warrants and publicly traded units began trading on Nasdaq under the ticker symbols "LVOX", "LVOXW" and "LVOXU," respectively. Pursuant to the Company's certificate of incorporation, the Company is authorized to issue 500,000,000 shares of Class A common stock with a par value of \$0.0001 per share. As of March 31, 2023, the Company had 92,937,386 shares of Class A common stock issued and outstanding (100,481,136 shares of common stock, less 7,543,750 of which are held in an escrow account to be released only if the price of Class A common stock trading on Nasdaq exceeds certain thresholds during the seven-year period beginning June 18, 2021 (the "Escrowed Shares")). As of December 31, 2022, 500,000,000 shares of Class A common stock were authorized, and 92,729,127 shares were issued and outstanding (100,272,877 shares of common stock, less 7,543,750 Escrowed Shares).

The accumulated other comprehensive loss and accumulated deficit are included in stockholders' equity. At March 31, 2023 and December 31, 2022, the accumulated other comprehensive loss totaled \$1.7 million and \$2.2 million, respectively. The Company's accumulated deficit totaled \$174.0 million and \$165.5 million at March 31, 2023 and December 31, 2022, respectively.

Preferred Stock

Pursuant to the Company's certificate of incorporation, the Company is authorized to issue 25,000,000 shares of preferred stock having a par value of \$0.0001 per share. As of March 31, 2023, no shares of LiveVox preferred stock were issued and outstanding. As of December 31, 2022,25,000,000 shares of preferred stock were authorized, and no shares of preferred stock were issued and outstanding.

Net Transfer from LiveVox TopCo

On March 6, 2023, the Company received \$0.2 million from LiveVox TopCo to pay for the bonuses to the Company's Chief Financial Officer in connection with the consummation of the Merger, and recorded it as additional paid-in-capital in the consolidated balance sheets.

14. Analysis of the Changes in Accumulated Other Comprehensive Income (Loss)

Accumulated other comprehensive income (loss) in the Company's consolidated balance sheets includes foreign currency translation items associated with the Company's foreign operations, and unrealized gain (loss) on the Company's AFS debt securities. Following is an analysis of the changes in accumulated other comprehensive loss, net of applicable taxes, at March 31, 2023 and 2022 (dollars in thousands):

			Mai	rch 31, 2022			
	Foreign currency translation adjustment		(loss) o	realized gain on marketable ecurities	Total accumulated other comprehensive loss		
Balance, beginning of period	\$	(300)	\$	(177)	\$	(477)	
Other comprehensive loss		(49)		(888)		(937)	
Balance, end of period	\$	(349)	\$	(1,065)	\$	(1,414)	
Balance, end of period	<u> </u>	(5.5)					
Balance, end of period	<u>Ψ</u>	(5.5)	÷				
Balance, end of period	<u> </u>		Net un	rch 31, 2023 realized gain	T		
Balance, end of period		gn currency	Net un (loss) o			cumulated other	
Balance, beginning of period		gn currency	Net un (loss) o	realized gain on marketable			
	translatio	gn currency on adjustment	Net un (loss) o	realized gain on marketable ecurities	comp	rehensive loss	

Notes to the Consolidated Financial Statements (Unaudited)

Components of other comprehensive income (loss) and related taxes for the three months ended March 31, 2023 and 2022 are as follows (dollars in thousands):

Three Months Ended March 31, 2023 2022 Before tax Tax effect Net of tax Before tax Tax effect Net of tax Foreign currency translation adjustment \$ \$ 62 \$ (4) \$ 66 \$ (47)\$ 2 (49)403 Net unrealized gain (loss) on marketable securities (24)427 (851)37 (888)Total other comprehensive income (loss) 39 465 493 \$ (898)(28)\$ (937)

The amount of net realized gain (loss) on sale of marketable securities that has been previously included as net unrealized gain (loss) in other comprehensive income (loss) and then reclassified out of other comprehensive income (loss) into earnings during the three months ended March 31, 2023 and 2022 is immaterial.

15. Stock-Based Compensation

Management Incentive Units

As of March 31, 2023, all MIUs are classified as equity. MIU activities for the three months ended March 31, 2023 are summarized as follows (in thousands, except for per share data):

	Number of Shares	Weighted-average Grant Date Fair Value (per share)	Weighted-average Remaining Contractual Term (1)
Outstanding at December 31, 2022	1,896	\$ 0.79	
Granted	_	_	
Vested	(632)	0.79	
Forfeited	(215)	0.79	
Outstanding at March 31, 2023	1,049	\$ 0.79	1.25 years

⁽¹⁾ The weighted-average remaining contractual term is calculated as the sum of the weighted amount of time between the reporting period end and the vest date divided by the sum of the shares that are outstanding by the end of the reporting period.

2021 Equity Incentive Plan

Restricted Stock Units

As of March 31, 2023, all RSUs granted to employees and nonemployees are classified as equity.

 $Employee\ RSU\ activities\ for\ the\ three\ months\ ended\ March\ 31,\ 2023\ are\ summarized\ as\ follows\ (in\ thousands,\ except\ for\ per\ share\ data):$

Notes to the Consolidated Financial Statements (Unaudited)

	Equity-classified RSUs - employee (1)							
	Number of Shares	Weighted-average Grant Date Fair Value (per share)	Weighted-average Remaining Contractual Term (2)					
Outstanding at December 31, 2022	8,110	\$ 4.00						
Granted	98	2.88						
Vested	(316)	5.84						
Forfeited	(877)	4.44						
Outstanding at March 31, 2023	7,015	\$ 3.84	1.36 years					

- (1) Represents awards granted to employees, executives and directors of the Company.
- (2) The weighted-average remaining contractual term is calculated as the sum of the weighted amount of time between the reporting period end and the vest date divided by the sum of the shares that are outstanding by the end of the reporting period.

The aggregate fair value of employee RSUs outstanding as of March 31, 2023, based on the fair value at the reporting period end, was \$1.7 million. The aggregate fair value of employee RSUs vested during the three months ended March 31, 2023, based on the fair value on the vest date, was \$0.9 million.

Nonemployee RSU activities for the three months ended March 31, 2023 are summarized as follows (in thousands, except for per share data):

	Equity-classified RSUs - nonemployee (1)							
	Number of Shares	Grant	ghted-average Date Fair Value (per share)	Weighted-average Remaining Contractual Term (2)				
Outstanding at December 31, 2022	19	\$	2.91					
Granted	_		_					
Vested	_		5.97					
Forfeited	_		_					
Outstanding at March 31, 2023	19	\$	2.84	1.30 years				

- (1) Represents awards granted to eligible consultants of the Company.
- (2) The weighted-average remaining contractual term is calculated as the sum of the weighted amount of time between the reporting period end and the vest date divided by the sum of the shares that are outstanding by the end of the reporting period.

The aggregate fair value of nonemployee RSUs outstanding as of March 31, 2023, based on the fair value at the reporting period end, was \$1.1 million. The aggregate fair value of nonemployee RSUs vested during the three months ended March 31, 2023, based on the fair value on the vest date, was immaterial.

Performance-Based Restricted Stock Units

As of March 31, 2023, all PSUs granted to employees are classified as equity. PSU activities for the three months ended March 31, 2023 are summarized as follows (in thousands, except for per share data):

Notes to the Consolidated Financial Statements (Unaudited)

	Equity-classified PSUs - employee (1)						
	Number of Shares	Grant l	hted-average Date Fair Value oer share)	Weighted-average Remaining Contractual Term (2)			
Outstanding at December 31, 2022	1,707	\$	6.39				
Granted	_		_				
Vested	_		_				
Forfeited	(170)		6.28				
Outstanding at March 31, 2023	1,537	\$	6.40	9.69 years			

- (1) Represents awards granted to employees and executives of the Company.
- (2) The weighted-average remaining contractual term is calculated as the sum of the weighted amount of time between the reporting period end and the vest date divided by the sum of the shares that are outstanding by the end of the reporting period.

The aggregate fair value of PSUs outstanding as of March 31, 2023, based on the fair value at the reporting period end, was \$4.7 million. None of the PSUs vested during the three months ended March 31, 2023.

Stock-Based Compensation Expense

The following tables present the Company's stock-based compensation expense by financial statement line item and by award type for the three months ended March 31, 2023 and 2022 (dollars in thousands):

	Tl	Three Months Ended March 31,				
		2023		2022		
Cost of revenue	\$	108	\$	312		
Sales and marketing expense		319		607		
General and administrative expense		1,377		660		
Research and development expense		845		900		
Total stock-based compensation	\$	2,649	\$	2,479		

	Ti	Three Months Ended March 31,			
		2023	2022		
Equity-classified awards:					
MIUs	\$	103	\$	139	
RSUs - employee		2,451		2,067	
RSUs - nonemployee		5		9	
PSUs - employee		90		264	
Total equity-classified awards		2,649		2,479	
Total stock-based compensation	\$	2,649	\$	2,479	

There were no income tax benefits recognized for the three months ended March 31, 2023, related to tax deductions from RSU awards vesting during that period. Due to the Company's net operating loss, the related tax deductions result in deferred tax assets that are fully offset with a valuation allowance. None of the RSUs vested during the three months ended March 31, 2022, and, therefore, there were no income tax benefits recognized during that period.

As of March 31, 2023, unrecognized stock-based compensation expense related to nonvested awards by award type and their expected weighted-average recognition periods are summarized in the following table (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	based	ognized Stock- Compensation Expense	Weighted-average Recognition Period (1)
Equity-classified awards:			
MIUs	\$	726	1.75 years
RSUs - employee		23,248	2.67 years
RSUs - nonemployee		47	2.92 years
PSUs - employee		8,304	9.69 years
Total equity-classified awards		32,325	
Total unrecognized stock-based compensation	\$	32,325	

(1) The weighted-average recognition period is calculated as the sum of the weighted remaining period to recognize expense for nonvested awards divided by the sum of the shares that are expected to vest for all awards that have not vested or expired by the end of the reporting period. For awards for which the straight-line method is used for expense recognition, the remaining recognition period is the amount of time between the end of the reporting period and the end of time between the end of the reporting period is the amount of time between the end of the reporting period and the end of each separately vesting portion of the award.

16. Geographic Information

Disaggregation of Revenue

The following table disaggregates the Company's revenue by geographic area for the three months ended March 31, 2023 and 2022 (dollars in thousands):

	Three Months Ended March 31,				
	 2023		2022		
United States	\$ 34,397	\$	29,550		
Americas (excluding United States)	911		1,052		
Asia	1,535		1,458		
Europe	23		33		
Total revenue	\$ 36,866	\$	32,093		

In addition, 99.5% of the Company's revenue is denominated in U.S. dollars and 0.5% is denominated in foreign currencies.

Property and Equipment

The following table summarizes total property and equipment, net in the respective locations at March 31, 2023 and December 31, 2022 (dollars in thousands):

	Marc	h 31, 2023	Decen	nber 31, 2022
United States	\$	1,147	\$	1,291
Americas (excluding United States)		294		309
Asia		964		1,018
Property and equipment, net	\$	2,405	\$	2,618

Notes to the Consolidated Financial Statements (Unaudited)

The geographical location of the Company's customers affects the nature, amount, timing and uncertainty of revenue and cash flows due to the potential for unfavorable and uncertain regulatory, political, economic and tax conditions. These uncertainties can impact the amount of revenue recognized through price adjustments and uncertainty of cash flows that may arise due to local regulations.

17. Income Taxes

The effective tax rate used for interim periods is the estimated annual effective consolidated tax rate, based on the current estimate of full year results, except that taxes related to specific discrete events, if any, are recorded in the interim period in which they occur. The annual effective tax rate is based upon several significant estimates and judgments, including the estimated annual pre-tax income of the Company in each tax jurisdiction in which it operates, and the development of tax planning strategies during the year. A valuation allowance is provided when it is more likely than not that some portion or all of a deferred tax asset will not be realized. In addition, the Company's tax expense can be impacted by changes in tax rates or laws and other factors that cannot be predicted with certainty. As such, there can be significant volatility in interim tax provisions.

The provision for income taxes was \$0.5 million and \$0.5 million for the three months ended March 31, 2023 and 2022, respectively. The provision for income taxes for the three months ended March 31, 2023 and 2022 consisted primarily of foreign and state income taxes.

The effective tax rates were (6.06)% and (4.37)% for the three months ended March 31, 2023 and 2022, respectively. The change in the effective tax rate for the three months ended March 31, 2023, relative to 2022, was primarily attributable to the change in valuation allowance, an increase in foreign taxes and certain nondeductible expenses.

The Company files income tax returns in the U.S. federal jurisdiction, various state jurisdictions, India and Colombia. The tax returns are subject to statutes of limitations that vary by jurisdiction. At March 31, 2023, the Company remains subject to U.S. and certain state income tax examinations for tax years 2019 through 2022, and in certain other states for tax years 2018 through 2022. However, due to the Company's net operating loss carryforwards in various jurisdictions, tax authorities have the ability to adjust carryforwards related to closed years.

18. Retirement Benefit Plan

The Company amended its existing 401(k) plan (the "Plan") effective on July 1, 2018. The amended Plan covers eligible employees immediately upon employment with the Company. Participants may contribute up to a maximum percentage of their annual compensation to the Plan as determined by the Company limited to the maximum annual amount set by the Internal Revenue Service. The Plan provides for traditional tax-deferred and Roth 401(k) contribution options. Prior to the Plan amendment, the Company did not provide a matching contribution. The Company began matching fifty percent of the employee contribution up to a maximum of \$200 per pay period, limited to \$4,800 annually, upon adoption of the Plan. One hundred percent of the employer match vests immediately. Effective January 1, 2022, the Company removed the maximum \$200 per pay period matching contribution limitation, and began matching fifty percent of the employee contribution up to a maximum of \$4,800 per plan year. The Company made matching contributions totaling \$0.3 million and \$0.3 million during the three months ended March 31, 2023 and 2022, respectively.

19. Fair Value Measurement

The following table sets forth the fair value of the Company's assets and liabilities at March 31, 2023 (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	Level 1	Level 2		Level 3		Totals
Cash and cash equivalents	\$ 19,266	\$		\$		\$ 19,266
Marketable securities—available for sale debt securities	_		43,718		_	43,718
Total assets	\$ 19,266	\$	43,718	\$		\$ 62,984
Term loan	\$ _	\$	54,163	\$	_	\$ 54,163
Finance lease obligations	_		5		_	5
Warrant liability—Forward Purchase Warrants	_		_		566	566
Total liabilities	\$ _	\$	54,168	\$	566	\$ 54,734

The following table sets forth the fair value of the Company's assets and liabilities at December 31, 2022 (dollars in thousands):

	Level 1	Level 2		Level 3		Totals	
Cash and cash equivalents	\$ 20,742	\$		\$		\$	20,742
Marketable securities—available for sale debt securities	_		48,182				48,182
Total assets	\$ 20,742	\$	48,182	\$	_	\$	68,924
		-		-		-	
Term loan	\$ _	\$	54,567	\$	_	\$	54,567
Finance lease obligations	_		11		_		11
Warrant liability—Forward Purchase Warrants	 _				633		633
Total liabilities	\$ _	\$	54,578	\$	633	\$	55,211

Level 1 and Level 2 of the Fair Value Hierarchy

As of March 31, 2023 and December 31, 2022, the carrying amounts of the Company's cash, cash equivalents and restricted cash approximate their fair values due to their short maturities and have been classified as Level 1 of the fair value hierarchy. The fair value of the term loan and finance lease obligations approximates their carrying value. The fair value is determined based on observable inputs on the price of the term loan in the market and has been classified as Level 2 of the fair value hierarchy. The fair value of the Company's AFS debt securities is determined based on valuations provided by external investment managers who obtain them from a variety of industry standard data providers and has been classified as Level 2 of the fair value hierarchy. Refer to Note 4 for additional information regarding the fair value of the Company's marketable securities.

Level 3 of the Fair Value Hierarchy

The Company's liability related to the Forward Purchase Warrants is measured at fair value on a recurring basis and is classified as Level 3 within the fair value hierarchy. There were no other assets or liabilities measured at fair value on a recurring basis at March 31, 2023 and December 31, 2022.

Warrant liability—Forward Purchase Warrants

As discussed in Note 2(y), 833,333 Forward Purchase Warrants were issued pursuant to the Forward Purchase Agreement dated January 13, 2021 between Crescent and Old LiveVox. Upon consummation of the Merger, the Company concluded that the Forward Purchase Warrants do not meet the derivative scope exception and are accounted for as derivative liabilities. The Forward Purchase Warrants are classified as Level 3 fair value measurement. The Company employed a Black-Scholes option pricing model specific to the contractual terms of the Forward Purchase Warrants to determine their fair value at each reporting period, with changes in fair value recognized in the consolidated statements of operations and comprehensive loss. Inherent in the options pricing model are assumptions related to current stock price, expected share price volatility, expected life, risk-free interest rate and dividend yield. The stock price is based on the closing price of the Company's Class A common

Notes to the Consolidated Financial Statements (Unaudited)

stock on Nasdaq as of the valuation date. The exercise price is based on the terms of the warrant agreement. The volatility input is estimated using the implied volatility of the Public Warrants and the volatility of the Company's peer companies. The expected life of the Forward Purchase Warrants is based on the time from valuation date to the contractual expiration date. The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of issuance for zero-coupon U.S. Treasury notes with maturities corresponding to the expected five-year term. The dividend rate is based on the historical rate, which the Company anticipates to remain at zero. Future change in these assumptions could result in a material change to the fair value of the Forward Purchase Warrants, and such changes will be recorded in the consolidated statements of operations and comprehensive loss.

The following table provides quantitative information regarding assumptions used in the Black Scholes option-pricing model to determine the fair value of the Forward Purchase Warrants:

	March 31, 2023		cember 31, 2022
Stock price	\$ 3.09	\$	2.97
Exercise price	\$ 11.50	\$	11.50
Contractual term	3.20 years		3.50 years
Expected volatility	75.00%		77.50%
Risk-free rate	3.80%		4.20%
Dividend yield	0.00%		0.00%

Changes in the Level 3 Fair Value Measurement

The changes in fair value of the Level 3 liabilities are as follows (dollars in thousands):

	March	31, 2023	December 31, 2022		
Balance, beginning of period	\$	633	\$	767	
Changes in fair value of warrant liability		(67)		(134)	
Balance, end of period	\$	566	\$	633	

During the three months ended March 31, 2023 and 2022, the gain recognized due to decrease in the fair value of warrant liability was \$0.1 million and \$0.4 million, respectively, and was recorded within other expense, net in the consolidated statements of operations and comprehensive loss.

20. Basic and Diluted Loss Per Share

Basic net loss per share is calculated by dividing net loss by the weighted average number of shares of Class A common stock outstanding during the period, including net issuance of shares upon vesting stock-based payment awards and excluding unvested stock-based payment awards and shares withheld to cover employees' withholding taxes upon vesting of stock-based payment awards. Diluted net loss per share is computed giving effect to all potentially dilutive shares of Class A common stock, including Class A common stock issuable upon vesting of stock-based payment awards and contingent earnout shares. Basic and diluted loss per share was the same for each period presented as the inclusion of all potential Class A common stock outstanding would have been antidilutive.

The computation of loss per share and weighted average shares of the Company's common stock outstanding for the three months ended March 31, 2023 and 2022 are as follows (in thousands, except per share data):

Notes to the Consolidated Financial Statements (Unaudited)

	Three Months Ended March 31,				
		2023		2022	
Numerator:					
Loss attributable to common stockholders—basic and diluted	\$	(8,469)	\$	(12,987)	
Denominator:					
Weighted average shares outstanding—basic and diluted		92,842		91,478	
Loss per share:					
Basic and diluted	\$	(0.09)	\$	(0.14)	

The following outstanding common stock equivalents were either considered antidilutive or were contingently issuable upon the resolution of their contingencies, and therefore, excluded from the computation of diluted net loss per share attributable to common stockholders for the periods presented (in thousands):

	Three Months Ended March 31,			
	2023	2022		
Earn-Out Shares (1)	5,000	5,000		
Lock-Up Shares (2)	2,544	2,544		
Finders Agreement Shares (3)	1,644	1,644		
Warrants to purchase common stock	13,333	13,333		
Shares withheld to cover employees' withholding taxes upon vesting of RSUs	108	_		
Unvested RSUs	7,034	5,482		
Unvested PSUs	1,537	1,737		
Total	31,200	29,740		

- (1) As additional consideration payable to the LiveVox Stockholder, the Company issued5,000,000 shares of Class A common stock (the "Earn-Out Shares") held in an escrow account to be released only if the price of Class A common stock trading on Nasdaq exceeds certain thresholds during the seven-year period beginning June 18, 2021. No contingent consideration shares were issued or released during the three months ended March 31, 2023 and 2022.
- (2) Represents 2,543,750 shares of converted Class A common stock held by the SPAC sponsor and certain independent directors (the "Lock-Up Shares") immediately following the closing, which were placed in an escrow account to be subject to release only if the price of Class A common stock trading on Nasdaq exceeds certain thresholds during the seven-year period beginning June 18, 2021. No contingent consideration shares were issued or released during the three months ended March 31, 2023 and 2022.
- (3) Represents 1,643,750 shares of Class A common stock (the "Finders Agreement Shares") to be issued only if the price of Class A common stock trading on Nasdaq exceeds certain thresholds during the seven-year period beginning June 18, 2021, pursuant to the terms of the Finders Agreement.No contingent consideration shares were issued during the three months ended March 31, 2023 and 2022.

21. Commitments and Contingencies

Commitments

As of March 31, 2023 and December 31, 2022, \$54.8 million and \$54.9 million of the term loan principal was outstanding, respectively. The term loan is due December 31, 2025. See Note 9 for more information.

Contingencies

Notes to the Consolidated Financial Statements (Unaudited)

The Company is subject to the possibility of various gain or loss contingencies arising in the ordinary course of business that will ultimately be resolved depending on future events. The Company considers the likelihood of loss or impairment of an asset, or the incurrence of a liability, as well as the ability to reasonably estimate the amount of loss, in determining loss contingencies. An estimated loss contingency is accrued when information available prior to issuance of the consolidated financial statements indicates that it is probable that an asset has been impaired or a liability has been incurred at the date of the consolidated financial statements, and the amount or range of loss can be reasonably estimated. Legal costs are expensed as incurred. Gain contingencies are not recognized until they are realized or realizable.

Indemnification Agreements

The Company has entered into indemnification agreements with its directors, officers and certain employees that will require us, among other things, to indemnify them against certain liabilities that may arise by reason of their status or service as directors, officers or employees. As of March 31, 2023 and December 31, 2022, there were no claims that the Company is aware of that could have a material effect on its consolidated balance sheets, consolidated statements of operations and comprehensive loss, or consolidated statements of cash flows.

Litigation and Claims

From time to time and in the ordinary course of business, the Company may be subject to various claims, charges, investigations, and litigation. As of the date of issuance of these consolidated financial statements, the Company is not a party to any claims that would have a material adverse effect on its business operations or financial position.

22. Restructuring

2023 Restructuring Plan

On January 13, 2023, the Company authorized a new restructuring plan (the "2023 Restructuring Plan"). Management, with the oversight and guidance of the Company's board of directors, determined to implement the 2023 Restructuring Plan following a review of the Company's business, operating expenses and the macroeconomic environment. The 2023 Restructuring Plan is intended to reduce the Company's cost structure and improve its operational efficiency.

The 2023 Restructuring Plan includes a reduction of approximately 98 employees, comprising approximately 16% of the Company's global workforce. Restructuring charges incurred as a result of the 2023 Restructuring Plan totaled \$3.2 million, and consist primarily of employee severance and termination benefits. The unpaid balance of restructuring charges was \$0.2 million as of March 31, 2023, which the Company expects to pay off by the end of the second quarter of 2023.

2022 Restructuring Plan

On August 1, 2022, the Company initiated a restructuring plan (the "2022 Restructuring Plan"), following a review of the Company's business, operating expenses and the macroeconomic environment. The purpose of the 2022 Restructuring Plan is to reduce the Company's cost structure and improve its operational efficiency.

The 2022 Restructuring Plan includes a reduction of approximately 3% of the Company's global workforce. Restructuring charges incurred as a result of the 2022 Restructuring Plan totaled \$0.9 million, and consist primarily of employee severance and termination benefits. The unpaid balance of restructuring charges was \$0.3 million as of March 31, 2023, which the Company expects to pay off by the end of 2023.

Restructuring Charges

The following table summarizes (a) restructuring charges incurred as a result of the 2023 Restructuring Plan and the 2022 Restructuring Plan, by major type of cost and by financial statement line item, and (b) the changes in the liability for restructuring charges, included within accrued expenses on the consolidated balance sheets (dollars in thousands):

Notes to the Consolidated Financial Statements (Unaudited)

	202	2 Restruc	cturir	ng Plan	2023 Restructuring Plan								
	Empl severan termin bene	ce and ation		Subtotal	seve ter	mployee rance and mination enefits	ter	Non-lease contract mination costs	Other costs		Subtotal	Aggr	egate Total
Liability as of December 31, 2021	\$		\$		\$		\$		\$ —	\$		\$	_
Restructuring charges													
Cost of revenue		400		400		_		_	_		_		400
Sales and marketing expense		147		147		_		_	_		_		147
General and administrative expense		_		_		_		_	_		_		_
Research and development expense		5		5		_		_	_		_		5
Total restructuring charges		552		552		_		_	_		_		552
Cash payments		(552)		(552)		_		_	_		_		(552)
Liability as of December 31, 2022	\$	_	\$	_	\$	_	\$	_	\$ —	\$	_	\$	_
Restructuring charges	·								-				<u></u>
Cost of revenue		_		_		1,110		63	2		1,175		1,175
Sales and marketing expense		242		242		1,350		_	_		1,350		1,592
General and administrative expense		_		_		43		_	_		43		43
Research and development expense		101		101		658		_	_		658		759
Total restructuring charges		343		343		3,161		63	2		3,226		3,569
Cash payments		_		_		(2,991)		(37)	(2))	(3,030)		(3,030)
Liability as of March 31, 2023	\$	343	\$	343	\$	170	\$	26	<u>\$</u>	\$	196	\$	539

23. Subsequent Events

The Company evaluated subsequent events and transactions that occurred after the balance sheet date up to the date that the consolidated financial statements were issued. Based on this review, the Company did not identify any subsequent events that would have required adjustment or disclosure in the consolidated financial statements.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Throughout this section, unless otherwise noted, the "Company," "LiveVox," "we," "us," and "our" refers to LiveVox Holdings, Inc., and its subsidiaries, collectively. You should read the following discussion and analysis of our financial condition and results of operations in conjunction with our unaudited interim consolidated financial statements and related notes thereto included in Part I, Item 1 of this Quarterly Report on Form 10-Q (this "Quarterly Report"). In addition to historical information, the following discussion and analysis contains forward-looking statements that involve risks, uncertainties and assumptions. Our actual results may differ materially from those anticipated in these forward-looking statements as a result of certain factors, including, but not limited to, those set forth in the section entitled "Item 1A. Risk Factors" in our Annual Report on Form 10-K ("Annual Report") filed with the SEC on March 2, 2023.

Overview

We enable next-generation cloud contact center functionality through a cloud contact-center-as-a-service ("CCaaS") platform that we provide for enterprises, business process outsourcers ("BPOs") and collections agencies. Our CCaaS platform provides customers with a scalable, cloud-based architecture and pre-integrated artificial intelligence ("AI") capabilities to support enterprise-grade deployments of our solutions including omnichannel customer connectivity, customer relationship management ("CRM") and workforce engagement management ("WEM"). Our omnichannel product offerings enable our customers to connect with their customers via their channel of choice, including human voice, virtual agents powered by AI, email, text or web chat. Our platform features a native CRM which unifies disparate, department-level systems of record to present contact center agents with a single view of its customers without displacing or replacing existing CRMs or other systems of record. Our WEM offerings include a lightweight yet fully-featured product that meets the needs of smaller or less mature contact center operations as well as seamless integration with WEM products from other providers.

We typically sell our products to customers under one- to three-year subscription contracts that stipulate a minimum amount of monthly usage and associated revenue with the ability for the customer to consume excess usage above the minimum contract amount, all recognized on a monthly basis following deployment to the customer. Excess usage revenue is deemed to be specific to the month in which the usage occurs, since the minimum usage commitments reset at the beginning of each month. Subscription revenue accounted for 97% and 98% for the three months ended March 31, 2023 and 2022, respectively, of our total revenue with the remainder consisting of professional services and other non-recurring revenue derived from the implementation of our products.

Restructuring Events

On August 1, 2022, the Company initiated a restructuring plan (the "2022 Restructuring Plan") to reduce the Company's cost structure and improve its operational efficiency. The 2022 Restructuring Plan includes a reduction of approximately 3% of the Company's global workforce. Restructuring charges incurred as a result of the 2022 Restructuring Plan totaled \$0.9 million, and consist primarily of employee severance and termination benefits. The unpaid balance of restructuring charges was \$0.3 million as of March 31, 2023, which the Company expects to pay off by the end of fiscal 2023.

On January 13, 2023, the Company authorized a restructuring plan (the "2023 Restructuring Plan"). Management, with the oversight and guidance of the Company's board of directors, determined to implement the 2023 Restructuring Plan following a review of the Company's business, operating expenses and the macroeconomic environment. The 2023 Restructuring Plan is intended to reduce the Company's cost structure and improve its operational efficiency. The 2023 Restructuring Plan includes a reduction of approximately 98 employees, comprising approximately 16% of the Company's global workforce. Restructuring charges incurred as a result of the 2023 Restructuring Plan totaled \$3.2 million, and consist primarily of employee severance and termination benefits. The unpaid balance of restructuring charges was \$0.2 million as of March 31, 2023, which the Company expects to pay off by the end of the second quarter of fiscal 2023.

Please see Note 22 to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report for additional information about this plan.

LiveVox's Segments

The Company has determined that its Chief Executive Officer ("CEO") is its chief operating decision maker. The Company's CEO reviews financial information presented on a consolidated basis for purposes of assessing performance and making decisions on how to allocate resources. Accordingly, the Company has determined that it operates in a single reportable segment.

Key Operating and Non-GAAP Financial Performance Metrics

In addition to measures of financial performance presented in our consolidated financial statements, we monitor the key metrics set forth below to help us evaluate growth trends, establish budgets, measure the effectiveness of our sales and marketing efforts and assess operational efficiencies.

Annualized Recurring Revenue

Annualized Recurring Revenue ("ARR") is calculated as the sum of the most recent quarter of (a) recurring subscription amounts and (b) recurring platform usage charges for all customers, multiplied by four. Professional services and revenue of a one-time nature are excluded.

ARR in the three months ended March 31, 2023 was \$143.0 million, which is \$17.3 million or 14% higher than the \$125.7 million of ARR in the three months ended March 31, 2022.

LTM Net Revenue Retention Rate

We believe that our Last Twelve Months ("LTM") Net Revenue Retention Rate provides us and investors with insight into our ability to retain and grow revenue from our customers and is a meaningful measure of the long-term value of our customer relationships. We calculate LTM Net Revenue Retention Rate by dividing the recurring revenue recognized during the most recent LTM period, provided, however, that recurring revenue from a customer in the most recent LTM period is excluded from the calculation if recurring revenue was not recognized from that customer in the preceding LTM period. Customers who cease using our products during the most recent LTM period are included in the calculation. For example, LTM Net Revenue Retention for the 12-month period ending March 2023 includes recurring revenue from all customers for whom revenue was recognized from April 2021 to March 2022 regardless of whether such customers increased, decreased, or stopped their use of our products during that period (i.e., old customers), but excludes recurring revenue from all customers who began using our services from April 2022 to March 2023 (i.e., new customers). We define monthly recurring revenue as recurring monthly contract and excess usage revenue, which we calculate separately from one-time, non-recurring revenue by month by customer. We consider all contract and excess usage revenue, which represents 97% of our revenue, to be recurring revenue as all of our contracts provide for a minimum commitment amount. We consider professional services revenue and one-time adjustments, which are booked on a one-time, non-recurring revenue. Professional services and other one-time adjustments are generally not material to the result of the calculation. However, one-time non-recurring revenue is important with respect to timing as we bill installation and non-standard statement of work fees immediately and recognize the revenue as the work is completed, which is generally in advance of the beginning of recurring revenue which is when we recogn

The following table shows our LTM Net Revenue Retention Rate for the periods presented:

_	Twelve	Months E	nded March 31,		Twelve Months	Ended December 31,	
	2023		2022		2022	2021	
LTM Net Revenue Retention Rate	113	%	113	%	113 %	105	%

Our LTM Net Revenue Retention Rate reflects the expansion over time of our existing customers as they add new products and additional units of service. A much higher percentage of the product revenue from our customers is contracted on our per minute pricing model with a minimum commitment as compared to our per agent pricing model with minimum commitments for both agents and units of service.

Our LTM Net Revenue Retention Rate was 113% in the twelve months ended March 31, 2023, which is consistent with 113% in the twelve months ended March 31, 2022. Recurring revenue expansion in the last twelve months ended March 31, 2023 was approximately 100 basis points lower than the last twelve months ended March 31, 2022 which was offset by favorable Shrink and Churn.

Our LTM Net Revenue Retention Rate increased by 8 percentage points, to 113% in the twelve months ended December 31, 2021 from 105% in the twelve months ended December 31, 2021 primarily as a result of improvement in the excess usage revenue component of the Net Revenue Retention calculation (i.e. less of an unfavorable impact) due to tapering of the impact of COVID-19. Secondarily, the contracted component of the Net Revenue Retention rate for existing customers improved. In addition, monthly minimum contract revenue for all customers grew by 20% from fiscal 2021 to fiscal 2022.

Adjusted EBITDA

In addition to net loss presented in accordance with generally accepted accounting principle ("GAAP"), we monitor Adjusted EBITDA, a Non-GAAP financial measure, to analyze our financial results and believe that it is useful to investors, as a supplement to

U.S. GAAP measures, in evaluating our ongoing operational performance and enhancing an overall understanding of our past financial performance. We believe that Adjusted EBITDA helps illustrate underlying trends in our business that could otherwise be masked by the effect of the income or expenses that we exclude from Adjusted EBITDA. Furthermore, we use this measure to establish budgets and operational goals for managing our business and evaluating our performance. We also believe that Adjusted EBITDA provides an additional tool for investors to use in comparing our recurring core business operating results over multiple periods with other companies in our industry.

Adjusted EBITDA should not be considered in isolation from, or as a substitute for, financial information prepared in accordance with U.S. GAAP, and our calculation of Adjusted EBITDA may differ from that of other companies in our industry. We compensate for the inherent limitations associated with using Adjusted EBITDA through disclosure of these limitations, presentation of our consolidated financial statements in accordance with U.S. GAAP and reconciliation of Adjusted EBITDA to the most directly comparable U.S. GAAP measure, net loss. We calculate Adjusted EBITDA as net loss before (a) depreciation and amortization, (b) long-term equity incentive bonus, (c) stock-based compensation expense, (d) interest income or expense, net, (e) change in the fair value of warrant liability, (f) other income or expense, net, (g) benefit from or provision for income taxes, and (h) other items that do not directly affect what we consider to be our core operating performance.

The following table shows a reconciliation of net loss to Adjusted EBITDA for the periods presented (dollars in thousands):

	Three Months Ended March 31, (unaudite			
		2023		2022
Net loss	\$	(8,469)	\$	(12,987)
Non-GAAP adjustments:				
Depreciation and amortization (1)		1,043		1,347
Long-term equity incentive bonus and stock-based compensation expense (2)		2,649		2,479
Interest expense, net		1,096		750
Change in the fair value of warrant liability		(67)		(392)
Other income, net		(70)		(64)
Acquisition and financing related fee and expense (3)		_		10
Provision for income taxes		480		544
Restructuring cost (4)		3,569		_
Other non-recurring expenses (3)(5)		594		_
Adjusted EBITDA	\$	825	\$	(8,313)

(1) Depreciation and amortization expenses included in our results of operations are as follows (dollars in thousands):

	Thr	Three Months Ended March 31, (unaudited)			
		023	202	22	
Cost of revenue	\$	308	\$	609	
Sales and marketing expense		593		611	
General and administrative expense		115		91	
Research and development expense		27		36	
Total depreciation and amortization	\$	1,043	\$	1,347	

(2) There were no long-term equity incentive bonus during the periods presented. Stock-based compensation expenses included in our results of operations are as follows (dollars in thousands):

	1 h	Three Months Ended March 31, (unaudited)				
		2023		2022		
Cost of revenue	\$	108	\$	312		
Sales and marketing expense		319		607		
General and administrative expense		1,377		660		
Research and development expense		845		900		
Total stock-based compensation expenses	\$	2,649	\$	2,479		

- (3) Included in general and administrative expense for all periods presented.
- (4) Restructuring costs relate to charges incurred as a result of the 2023 Restructuring Plan initiated in January 2023 and the 2022 Restructuring Plan initiated in August 2022. These restructuring costs included in our results of operations are as follows (dollars in thousands):

	Thr	Three Months Ended March 31, (unaudited)			
	2	023		2022	
Cost of revenue	\$	1,175	\$	_	
Sales and marketing expense		1,592		_	
General and administrative expense		43		_	
Research and development expense		759		_	
Total restructuring costs	\$	3,569	\$	_	

(5) Other non-recurring expenses consist primarily of impairment loss on right-of-use asset associated with the closure of certain underutilized physical offices within the United States.

Non-GAAP Gross Profit and Non-GAAP Gross Margin Percentage

In addition to gross profit presented in accordance with GAAP, we also monitor non-GAAP gross profit and non-GAAP gross margin percentage. Management uses Non-GAAP gross profit and Non-GAAP gross margin percentage to evaluate operating performance and to determine resource allocation among our various product offerings. We believe Non-GAAP gross profit and Non-GAAP gross margin percentage allow for better comparison of our financial results with those of our competitors. We also believe Non-GAAP gross profit and Non-GAAP gross margin percentage provide useful information to investors and others to understand and evaluate our operating results in the same manner as our management and board of directors.

Non-GAAP gross profit and Non-GAAP gross margin percentage should not be considered in isolation from, or as a substitute for, U.S. GAAP measures. Non-GAAP gross profit and Non-GAAP gross margin percentage may not be comparable to similarly titled measures of other companies because other companies may not calculate Non-GAAP gross profit and Non-GAAP gross margin percentage or similarly titled measures in the same manner as we do. U.S. GAAP defines gross profit as revenue less cost of revenue. Cost of revenue includes all expenses associated with our various product offerings as more fully described under the caption "—Components of Results of Operations —Cost of Revenue—" below. We define Non-GAAP gross profit as gross profit after adding back the following items:

- · depreciation and amortization;
- · long-term equity incentive bonus and stock-based compensation expense; and
- · restructuring cost

We add back depreciation and amortization, long-term equity incentive bonus and stock-based compensation expense, and restructuring cost because they are one-time or non-cash items. We eliminate the impact of these one-time or non-cash items because we do not consider them indicative of our core operating performance. Their exclusion facilitates comparisons of our operating performance on a period-to-period basis. Therefore, we believe showing Non-GAAP gross margin to remove the impact of these one-time or non-cash expenses is helpful to investors in assessing our gross profit and gross margin performance in a way that is similar to how management assesses our performance.

We calculate Non-GAAP gross margin percentage by dividing Non-GAAP gross profit by revenue, expressed as a percentage of revenue.

The following table shows a reconciliation of gross profit to Non-GAAP gross profit for the periods presented (dollars in thousands):

	Three Mon	ths Ended	March 31,	(unaudited)	
	2023			2022	
Gross profit	\$ 23,604		\$	18,461	
Depreciation and amortization	308			609	
Long-term equity incentive bonus and stock-based compensation expense	108			312	
Restructuring cost	1,175			_	
Non-GAAP gross profit	\$ 25,195		\$	19,382	
Gross margin %	64.0	%		57.5	%
Non-GAAP gross margin %	68.3	%		60.4	%

Components of Results of Operations

Revenue

We derive revenue by providing products under a variety of pricing models. Our recently released AI Virtual Agent product and our historical Voice product are provided under a usage-based pricing model with prices calculated on a per-minute basis with a contracted minimum commitment in accordance with the terms of the underlying pricing agreements. Voice is our predominant source of revenue. Other revenue sources are derived from products under the following pricing models:

- 1) a per "unit of measure" with a minimum commitment (e.g., Speech IQ);
- 2) the combination of per agent and per "unit of measure" models with minimum contracted commitments for each (e.g., SMS, email, U-CRM services);
- 3) a per agent pricing model with a minimum agent commitment (e.g., U-Script, U-Ticket, U-Chat, U-Quality Management, U-Screen Capture, U-CSAT, Advanced Analytics, Hosted PBX services); and
 - 4) a per agent pricing model with a minimum agent commitment with a monthly maximum commitment (e.g., Attempt Supervisor, Advanced Analytics).

Outside of Voice, our pricing models detailed above are relatively new to the market and are not yet material to our business from a financial perspective.

Cost of Revenue

Our cost of revenue consists of personnel costs and associated costs such as travel, information technology and stock-based compensation for Implementation and Training Services, Customer Care, Technical Support, Professional Services, User Acceptance Quality Assurance, Technical Operations and VoIP services to our customers. Other costs of revenue include non-cash costs associated with depreciation and amortization including acquired technology, charges from telecommunication providers for communications, data center costs and costs to providers of cloud communication services, software, equipment maintenance and support costs to maintain service delivery operations.

In the fourth quarter of fiscal 2021, we completed a major strategic milestone when our data center transitioned from a model based on maintaining a co-location facility with our own capital equipment to a 100% cloud strategy based on monthly recurring charges for capacity added in generally small step function increments. As a result, by the end of fiscal 2021 we eliminated our capital expenditures for data center equipment, fully depreciated our old co-location facility and increased our data center costs for our cloud provisioning. A primary factor in our improvement in gross margin from the first quarter of fiscal 2022 has been the optimization of public cloud data center costs and the rationalization and redeployment of duplicative resources. We expect feature release efficiencies for our cloud operations as research and development resources eliminate the release effort associated with our co-location deployment.

As our business grows, we expect to realize economies of scale in our cost of revenue. We use the LiveVox platform to facilitate data-driven innovations to identify and facilitate efficiency improvement to our implementation, customer care and support, and technical operations teams. Additionally, our research and development priorities include ease of implementation, reliability and ease of use objectives that reduce costs and result in economies of scale relative to revenue growth.

Operating Expenses

We classify our operating expenses as sales and marketing, general and administrative, and research and development.

Sales and Marketing. Sales and marketing expenses consist primarily of salaries and related expenses, including stock-based compensation, for personnel in sales and marketing, sales commissions, channel special program incentive funds ("SPIFF") and channel commissions, travel costs, as well as marketing pipeline management, content delivery, programs, campaigns, lead generation, and allocated overhead. We believe it is important to continue investing in sales and marketing to continue to generate revenue growth, and we expect sales and marketing expenses to increase in absolute dollars and fluctuate as a percentage of revenue as we continue to support our growth initiatives.

General and Administrative. General and administrative expenses consist primarily of salary and related expenses, including stock-based compensation, for management, finance and accounting, legal, information systems and human resources personnel, professional fees, compliance costs, other corporate expenses and allocated overhead. We expect that general and administrative expenses will fluctuate in absolute dollars from period to period but decline as a percentage of revenue over time.

Research and Development. Research and development expenses consist primarily of salary and related expenses, including stock-based compensation, for LiveVox personnel as well as limited outsourced software development resources related to the identification and development of improvements, and expanded features for our products, as well as quality assurance, testing, product management and allocated overhead. Research and development costs are expensed as incurred. We have not performed research and development for internal-use software that would meet the qualifications for capitalization. We believe it is important to continue investing in research and development to continue to expand and improve our products and generate future revenue growth, and we expect research and development expenses to increase in absolute dollars and fluctuate as a percentage of revenue as we continue to support our growth initiatives.

Results of Operations

Comparison of the three months ended March 31, 2023 and 2022

The following tables summarize key components of our results of operations for the three months ended March 31, 2023 and 2022 (in thousands, except per share data):

		Three Months Ended March 31 (unaudited)				
	2023		2022			
Revenue	\$ 36,86	66 \$	32,093			
Cost of revenue	13,26	2	13,632			
Gross profit	23,60	4	18,461			
Operating expenses						
Sales and marketing expense	13,48	0	14,652			
General and administrative expense	9,17	1	7,468			
Research and development expense	7,98	3	8,490			
Total operating expenses	30,63	4	30,610			
Loss from operations	(7,03	0)	(12,149)			
Interest expense, net	1,09	6	750			
Change in the fair value of warrant liability	(6	7)	(392)			
Other income, net	(7	0)	(64)			
Total other expense, net	95	9	294			
Pre-tax loss	(7,98	9)	(12,443)			
Provision for income taxes	48	0	544			
Net loss	\$ (8,46	9) \$	(12,987)			
Net loss per share—basic and diluted	\$ (0.0	9) \$	(0.14)			
Weighted average shares outstanding—basic and diluted	92,84	2	91,478			

Revenue

	Thre	e Months End	ed March 31	, (unaudited)			
		2023		2022	\$ Change	% Chan	ge
Revenue	\$	36,866	\$	32,093	\$ 4,773	14.9	%

Revenue increased by \$4.8 million, or 14.9%, to \$36.9 million in the three months ended March 31, 2023 from \$32.1 million in the three months ended March 31, 2022, primarily due to 13% growth in contracted revenue driven by the acquisition of new customers and upsells to our existing customer base.

Cost of revenue

	Three Mont	hs Ende	ed March 31,	(unaudited)				
	2023			2022		\$ Change	% Cha	ıge
Cost of revenue	\$ 13,262		\$	13,632		\$ (370)	(2.7)	%
% of revenue	36.0	0/0		42.5	0/0			

Cost of revenue decreased by \$0.4 million, or 2.7%, to \$13.3 million in the three months ended March 31, 2023 from \$13.6 million in the three months ended March 31, 2022. The decrease was attributable primarily to a decrease in personnel costs of \$1.3 million resulting from decreased headcount, which was partially offset by an increase in employee severance and termination costs of \$1.0 million associated with the 2023 Restructuring Plan. In addition, stock-based compensation expenses decreased by \$0.2 million. In connection with the 2023 Restructuring Plan, any unvested Restricted Stock Units ("RSUs") and Performance-based Restricted Stock Units ("PSUs") awards granted under the 2021 Equity Incentive Plan (the "2021 Plan") to employees terminated in restructuring event were forfeited without consideration, and the stock-based compensation expenses recognized prior to the cancellation were reversed in the three months ended March 31, 2023.

Gross profit

		Three Mon	ths Ende	d March 31,	(unaudited)				
		2023			2022		\$ Change	% Cha	ange
Gross profit	\$	23,604		\$	18,461		\$ 5,143	27.9	%
Gross margin percentage	margin 64.0 %			57.5	%				

Gross profit increased by \$5.1 million, or 27.9%, to \$23.6 million in the three months ended March 31, 2023 from \$18.5 million in the three months ended March 31, 2022. The increase in gross profit was a result of increased revenue of \$4.8 million, decreased personnel costs of \$1.3 million and decreased stock-based compensation expenses of \$0.2 million, which were partially offset by increased employee severance and termination costs of \$1.0 million, described above.

Sales and marketing expense

		Three Mon	ths Ende	ed March 31,	(unaudited)					
	2023			2022			\$ Change		% Change	
Sales and marketing expense	\$	13,480		\$	14,652		\$	(1,172)	(8.0)	%
% of revenue		36.6	%		45.7	%				

Sales and marketing expense decreased by \$1.2 million, or 8.0%, to \$13.5 million in the three months ended March 31, 2023 from \$14.7 million in the three months ended March 31, 2022. The decrease was attributable primarily to a decrease in personnel costs of \$1.6 million resulting from decreased headcount, which was partially offset by an increase in employee severance and termination

costs of \$1.4 million associated with the 2023 Restructuring Plan. In addition, stock-based compensation expenses decreased by \$0.3 million. In connection with the 2023 Restructuring Plan, any unvested RSUs and PSUs granted under the 2021 Plan to employees terminated in restructuring event were forfeited without consideration, and the stock-based compensation expenses recognized prior to the cancellation were reversed in the three months ended March 31, 2023. Furthermore, marketing, promotions and tradeshow expenses decreased by \$0.7 million.

General and administrative expense

		Three Mont	ths Ende	ed March 31,	(unaudited)					
	2023			2022			\$ Change		% Change	
General and administrative expense	\$	9,171		\$	7,468		\$	1,703	22.8	%
% of revenue		24 9	%		23.3	%				

General and administrative expense increased by \$1.7 million, or 22.8%, to \$9.2 million in the three months ended March 31, 2023 from \$7.5 million in the three months ended March 31, 2022. The increase was attributable primarily to impairment loss on right-of-use asset of \$0.5 million associated with the closure of certain underutilized physical offices within the United States. In addition, stock-based compensation expenses increased by \$0.7 million and personnel costs increased by \$0.3 million as a result of increased headcount.

Research and development expense

	Three Mon	ths Ende	d March 31,	(unaudited)					
	2023			2022		\$ 0	Change	% Chan	ıge
Research and development expense	\$ 7,983		\$	8,490		\$	(507)	(6.0)	%
% of revenue	 21.7	%		26.5	%				

Research and development expense decreased by \$0.5 million, or 6.0%, to \$8.0 million in the three months ended March 31, 2023 from \$8.5 million in the three months ended March 31, 2022. The decrease was attributable primarily to a decrease in personnel costs of \$0.6 million resulting from decreased headcount, which was partially offset by an increase in employee severance and termination costs of \$0.7 million associated with the 2023 Restructuring Plan. In addition, stock-based compensation expenses decreased by \$0.1 million. In connection with the 2023 Restructuring Plan, any unvested RSUs and PSUs granted under the 2021 Plan to employees terminated in restructuring event were forfeited without consideration, and the stock-based compensation expenses recognized prior to the cancellation were reversed in the three months ended March 31, 2023. Furthermore, software expenses decreased by \$0.3 million and consulting and contracted services decreased by \$0.2 million.

Interest expense, net

	Three Mont	hs Ende	d March 31,	unaudited)					
	2023			2022		\$ (Change	% Cha	ange
Interest expense, net	\$ 1,096		\$	750		\$	346	46.1	%
% of revenue	 3.0	0/2		2.3	0/0				

Interest expense, net increased by \$0.3 million, or 46.1%, to \$1.1 million in the three months ended March 31, 2023 from \$0.8 million in the three months ended March 31, 2022. The increase was attributable primarily to increased interest expense of \$0.4 million associated with higher interest rates in the three months ended March 31, 2023. The increase in interest expense was partially offset by increase in interest income of \$0.1 million associated with the marketable securities which we invested since the fourth quarter of fiscal 2021.

Change in the fair value of warrant liability

	Three Mon	ths Ended	d March 31,	(unaudited)					
	2023			2022		\$ (Change	% Chan	ge
Change in the fair value of warrant liability	\$ (67)		\$	(392)		\$	325	(82.9)	%
% of revenue	 (0.2)	0/0		(1.2)	0/2				

Gain recognized due to change in the fair value of warrant liability decreased by \$0.3 million, or 82.9%. The decrease was attributable primarily to a decrease in the fair value of Forward Purchase Warrants of \$0.1 million in the three months ended March 31, 2023 compared to decrease in fair value of \$0.4 million in the three months ended March 31, 2022. For more information, see Note 19 of the Company's consolidated financial statements included in Part I, Item 1 of this Quarterly Report.

Liquidity and Capital Resources

Sources of Cash

LiveVox's consolidated financial statements have been prepared assuming the Company will continue as a going concern for the 12-month period from the date of issuance of the consolidated financial statements, which contemplates the realization of assets and the settlement of liabilities and commitments in the normal course of business. The Company's main sources of liquidity include:

- Available-for-sale ("AFS") debt securities, which are all classified as short-term securities to fund current operations and may be liquidated at the Company's discretion if the need arises. The Company held AFS debt securities of \$43.7 million and \$48.2 million as of March 31, 2023 and December 31, 2022, respectively. Please see Note 4 to the Company's consolidated financial statements included in Part I, Item 1 of this Quarterly Report for more information;
- The term loan and revolving credit facility that the Company entered into with PNC Bank on November 7, 2016 (as amended, the "Credit Facility"), which has been amended several times. On March 31, 2023, the Company entered into the Eighth Amendment to the Credit Facility (the "Eighth Amendment"). As of the Eighth Amendment effective date, the term loan commitment is \$54.9 million, the revolver commitment is \$5.0 million and the letter of credit sublimit is \$1.5 million. Please see Note 9 and 10 to the Company's consolidated financial statements included in Part I, Item 1 of this Quarterly Report for more information.

Cash Requirements

LiveVox's cash requirements within the next 12 months consist primarily of operating and administrative activities including employee related expenses and general, operating and overhead expenses, current maturities of the Company's term loan, operating and finance leases and other obligations.

LiveVox's long-term cash requirements consist of various contractual obligations and commitments, including:

- Term loan The Company has contractual obligations under its term loan to make principal and interest payments. Please see Note 9 to the Company's consolidated financial statements included in Part I, Item 1 of this Quarterly Report for a discussion of the contractual obligations under the Company's term loan and the timing of principal maturities. The principal amount is due December 31, 2025;
- Operating and finance lease obligations The Company leases its corporate headquarters and worldwide offices under operating leases, and finance computer and networking equipment and software purchases for its co-location data centers under finance leases. Please see Note 8 to the Company's consolidated financial statements included in Part I, Item 1 of this Quarterly Report for further detail of the Company's obligations under operating and finance leases and the timing of expected future lease payments;
- Other liabilities These include other long-term liabilities reflected in the Company's consolidated balance sheets as of March 31, 2023, including obligations associated with certain employee and non-employee incentive plans, Forward Purchase Warrants, unrecognized tax benefits and various long-term liabilities, which have some inherent uncertainty in the timing of these payments.

Future capital requirements will depend on many factors, including the Company's customer growth rate, customer retention, timing and extent of development efforts, the expansion of sales and marketing activities, the introduction of new and enhanced services offerings, the continuing market acceptance of the Company's services, effective integration of acquisition activities, if any, and maintaining the Company's bank credit facility. Additionally, the duration and extent of the impact from the current macroeconomic and geopolitical conditions continues to depend on future developments that cannot be accurately predicted at this

time. While those factors have caused operational difficulties, and may continue to create challenges for the Company's performance, they have not, thus far, had a substantial net impact on the Company's liquidity position.

The Company believes the cash generated by operating cash flows and debt will be sufficient to meet the Company's anticipated cash requirements for at least the next 12 months from the date of this Quarterly Report and beyond, while maintaining sufficient liquidity for normal operating purposes.

Acquisition Opportunities

The Company believes that there may be opportunity for further consolidation in LiveVox's industry. From time to time, the Company evaluates potential strategic opportunities, including acquisitions of other providers of cloud-based services. The Company has been in, and from time to time may engage in, discussions with counterparties in respect of various potential strategic acquisition and investment transactions. Some of these transactions could be material to the Company's business and, if completed, could require significant commitments of capital, result in increased leverage or dilution and/or subject the Company to unexpected liabilities. In connection with evaluating potential strategic acquisition and investment transactions, the Company may incur significant expenses for the evaluation and due diligence investigation of these potential transactions.

Comparison of cash flows for the three months ended March 31, 2023 and 2022

The following table summarizes key components of our cash flows for the three months ended March 31, 2023 and 2022 (dollars in thousands):

	Thre	ee Months Ended	March 31,	(unaudited)
		2023		2022
Net cash used in operating activities	\$	(5,657)	\$	(14,626)
Net cash provided by (used in) investing activities		4,786		(256)
Net cash used in financing activities		(642)		(146)
Effect of foreign currency translation		37		(97)
Net decrease in cash, cash equivalents and restricted cash	\$	(1,476)	\$	(15,125)

Net cash used in operating activities

Cash flows used in operating activities in the three months ended March 31, 2023 decreased by \$9.0 million to cash outflows of \$5.7 million from cash outflows of \$14.6 million during the three months ended March 31, 2022. The decrease to net cash used in operating activities was primarily attributable to a decrease of \$4.5 million in net loss and an increase of \$1.1 million in non-cash adjustments to net loss. The increase in non-cash items was primarily attributable to an increase in impairment loss on right-of-use asset of \$0.5 million associated with the closure of certain underutilized physical offices within the United States, a decrease in gain recognized of \$0.3 million due to change in the fair value of warrant liability, and an increase in credit loss expense of \$0.4 million due to the adoption of Accounting Standards Update ("ASU") No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments and all subsequent updates (collectively, the "ASC 326") on January 1, 2023. Net cash used in operating activities also included an increase of \$3.4 million in cash from operating assets and liabilities, primarily due to the timing of cash payments to vendors and cash receipts from customers.

Net cash provided by (used in) investing activities

Cash flows provided by (used in) investing activities in the three months ended March 31, 2023 increased by \$5.0 million to cash inflows of \$4.8 million from cash outflows of \$0.3 million during the three months ended March 31, 2022. The increase to net cash provided by (used in) investing activities was primarily attributable to an increase of \$6.1 million in proceeds from sale of AFS debt securities and an increase of \$5.5 million in proceeds from maturities and principal paydowns of AFS debt securities, partially offset by an increase of \$7.1 million in purchases of AFS debt securities.

Net cash used in financing activities

Cash flows used in financing activities in the three months ended March 31, 2023 increased by \$0.5 million to cash outflows of \$0.6 million from cash outflows of \$0.1 million during the three months ended March 31, 2022. The increase to net cash used in financing activities was primarily attributable to payment of principal of \$0.4 million under a structured payable arrangement related to the Company's commercial insurance policy on directors and officers, payment of original issue discount of \$0.3 million related to the Eighth Amendment to the Credit Facility that the Company entered into on March 31, 2023, and payment of employees' withholding taxes of \$0.3 million on net share settlement of share-based awards under the 2021 Plan. These payments were partially

offset by the drawdown on the revolving Credit Facility of \$0.3 million on March 31, 2023 to pay for the original issue discount related to the Eighth Amendment, and the net transfer from LiveVox TopCo of \$0.2 million to pay for the bonuses to the Company's Chief Financial Officer in connection with the consummation of the Merger.

Critical Accounting Estimates

Management's discussion and analysis of our financial condition and results of operations is based on our consolidated financial statements included in Part I, Item 1 of this Quarterly Report, which have been prepared in accordance with U.S. GAAP.

The preparation of these consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reporting periods. Significant items subject to such estimates and assumptions include, but are not limited to, the determination of the useful lives of long-lived assets, period of benefit of deferred sales commissions, allowances for credit losses, fair value of marketable securities, fair value of goodwill and long-lived assets, fair value of incentive awards, fair value of warrants, establishing standalone selling price, valuation of deferred tax assets, income tax uncertainties and other contingencies. Management periodically evaluates such estimates and they are adjusted prospectively based upon such periodic evaluation. Actual results could differ from those estimates, and such differences could be material to the Company's consolidated financial position and results of operations, requiring adjustment to these balances in future periods.

While our significant accounting policies are more fully described in the notes to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report, we believe that the following accounting estimates are critical to our business operations and understanding of our financial results. We consider an accounting judgment, estimate or assumption to be critical when (a) the estimate or assumption is complex in nature or requires a high degree of subjectivity and judgment and (b) the use of different judgments, estimates and assumptions could have a material impact on our consolidated financial statements.

Impairment of long-lived assets, including intangible assets

Long-lived assets to be held and used are reviewed for impairment whenever events or changes in circumstances indicate that the related carrying amount may not be recoverable. When required, impairment losses on assets to be held and used are recognized based on the fair value of the asset and long-lived assets to be disposed of are reported at the lower of the carrying amount or fair value. No impairment losses have been recognized in any of the periods presented.

We perform our annual impairment review of goodwill on October 1 of each year, and when a triggering event occurs between annual impairment tests. In testing for goodwill impairment, the Company has the option to first assess qualitative factors to determine if it is more likely than not that the fair value of the Company's single reporting unit is less than its carrying amount, including goodwill, or bypass the qualitative assessment and proceed directly to the quantitative impairment test to determine if the fair value of the reporting unit exceeds its carrying amount. If the fair value is determined to be less than the carrying value, an impairment charge is recorded for the amount by which the reporting unit's carrying amount exceeds its fair value, limited to the total amount of goodwill allocated to that reporting unit. No impairment losses have been recognized in any of the periods presented.

Intangible assets, consisting of acquired developed technology, corporate name, customer relationships and workforce, are reviewed for impairment whenever events or changes in circumstances indicate an asset's carrying value may not be recoverable. No impairment losses have been recognized in any of the periods presented.

Allowance for credit losses for marketable securities and accounts receivable

On January 1, 2023, the Company adopted ASC 326 and applied to its AFS debt securities and accounts receivable. ASC 326 replaces the existing incurred loss impairment model with a current expected credit loss impairment model. It also eliminates the concept of other-than-temporary impairment and requires credit losses related to AFS debt securities to be recorded through an allowance for credit losses rather than as a reduction in the amortized cost basis of the securities.

At each reporting date, the Company evaluates the amortized cost of AFS debt securities compared to their fair value to determine whether an AFS debt security is impaired, and, if the fair value is below amortized cost basis, whether the decline in fair value has resulted from credit losses or other factors. In making this assessment, the Company considers factors such as the extent to which fair value is less than amortized cost basis, any changes to the rating of the security by a rating agency, and any adverse conditions specifically related to the security. Credit related unrealized losses are recognized as an allowance for credit losses in the consolidated balance sheets with a corresponding charge in other income (expense), net, in the consolidated statements of operations and comprehensive loss. Non-credit related unrealized losses are recorded in other comprehensive income (loss), as applicable, net of applicable taxes.

Trade accounts receivable are stated net of any write-offs and the allowance for credit losses, at the amount the Company expects to collect. The Company performs ongoing credit evaluations of its customers and generally does not require collateral unless a customer has previously defaulted. The Company maintains an allowance for credit losses, which represents the best estimate of

lifetime expected credit losses against the existing accounts receivable, inclusive of unbilled receivables, based on certain factors including the age of the receivable balance, historical write-off experience, past collection experience with the customer, credit quality of the customer, current economic conditions, and reasonable and supportable forecasts of future economic conditions. If the financial condition of the Company's customers were to deteriorate, adversely affecting their ability to make payments, additional allowances would be required. Accounts receivable are charged off against the allowance for credit losses after all means of collection have been exhausted and the potential for recovery is considered remote. Recoveries of accounts receivable previously written off are recorded as income when received.

While management believes it has exercised prudent judgment and applied reasonable assumptions, there can be no assurance that in the future, changes in economic conditions or other factors would not cause changes in the financial health of our marketable security issuers or our customers. If the financial health deteriorates, the timing and amount of payments received could be impacted and therefore, could result in a change to our estimated credit losses.

Revenue Recognition

The Company recognizes revenue in accordance with ASC 606, Revenue from Contracts with Customers.

The Company derives substantially all of its revenue by providing cloud-based contact center products under a usage-based model. The Company's performance obligations are satisfied over time as the customer has continuous access to its hosted technology platform solutions through its data centers and simultaneously receives and consumes the benefits and the Company performs its services. Other immaterial ancillary revenue is derived from call recording, local caller identification packages, performance/speech analytics, text messaging services and professional services billed monthly on primarily usage-based fees, and to a lesser extent, fixed fees. Professional services, which represents approximately 3% of revenue, are billed on a fixed-price or on a time and material basis and the revenue is recognized over time as the services are rendered.

The Company has service-level agreements with customers warranting defined levels of uptime reliability and performance. If the services do not meet certain criteria, fees are subject to adjustment or refund representing a form of variable consideration. The Company records reductions to revenue for these estimated customer credits at the time the related revenue is recognized. These customer credits are estimated based on current and historical customer trends, and communications with its customers. Such customer credits have not been significant to date.

For contracts with multiple performance obligations (e.g., including various combinations of services), the Company allocates the contract price to each performance obligation based on its relative standalone selling price ("SSP"). The Company generally determines SSP based on the prices charged to customers. In instances where SSP is not directly observable, the Company determines the SSP using information that generally includes market conditions or other observable inputs.

Income Taxes

The Company accounts for income taxes using the asset and liability approach. Deferred tax assets and liabilities are recognized for the future tax consequences arising from the temporary differences between the tax basis of an asset or liability and its reported amount in the consolidated financial statements, as well as from net operating loss and tax credit carryforwards. Deferred tax amounts are determined by using the tax rates expected to be in effect when the taxes will be paid or refunds received, as provided for under currently enacted tax law. A valuation allowance is provided for deferred tax assets that, based on available evidence, are not expected to be realized.

The Company recognizes the effect of income tax positions only if those positions are more likely than not to be sustained in a court of last resort. Recognized income tax positions are measured at the largest amount that is greater than 50% likely of being realized. Changes in recognition or measurement are reflected in the period in which the change in judgment occurs. The Company does not believe its consolidated financial statements include any uncertain tax positions. It is the Company's policy to recognize interest and penalties accrued on any unrecognized tax benefit as a component of income tax expense.

Judgment is required in assessing the future tax consequences of events that have been recognized in our consolidated financial statements or tax returns. Variations in the actual outcome of these future tax consequences could materially impact our consolidated financial statements.

Stock-Based Compensation

Management Incentive Units

During 2019, LiveVox TopCo, LLC ("LiveVox TopCo"), the sole stockholder of the Company prior to the Merger, established a Management Incentive Unit program whereby the LiveVox TopCo board of directors has the power and discretion to approve the issuance of Class B Units of LiveVox TopCo that represent management incentive units ("MIUs") to any manager, director, employee, officer or consultant of the Company or its subsidiaries. Vesting begins on the date of issuance, and the MIUs vest ratably over five years with 20% of the MIUs vesting on each anniversary of a specified vesting commencement date, subject to the grantee's

continued employment with the Company on the applicable vesting date. Vesting of the MIUs will accelerate upon consummation of a "sale of the company", which is defined in the LiveVox TopCo limited liability company agreement. The Company recognizes stock-based compensation expense based on the service condition on a straight-line basis over the requisite service period of five years, reduced for actual forfeited MIUs. Stock-based compensation expense for MIUs is measured based on the grant date fair value of the award using a Monte Carlo simulation. Assumptions used in the Monte Carlo simulation are holding period, expected share price volatility, discount for lack of marketability, and risk-free interest rate. While the Company believes that the assumptions used in these calculations are reasonable, differences in actual experience or changes in assumptions could materially affect the expense related to the Company's MIUs.

2021 Equity Incentive Plan

On June 16, 2021, the stockholders of the Company approved the 2021 Plan, which became effective upon the closing of the Merger on June 18, 2021. The Company grants RSUs and PSUs to employees, executives, directors, and eligible consultants of the Company. RSUs are subject to service conditions only and typically vest over periods ranging from one to six years based on the grantee's role in the Company. PSUs, which are granted to certain key employees, vest either based on the achievement of predetermined market conditions, or based on both service and market conditions. All RSUs and PSUs will be settled in shares of Class A common stock and are classified as equity awards. Equity-classified awards are recognized as stock-based compensation expense over an employee's requisite service period or a nonemployee's vesting period on the basis of the grant-date fair value. Generally, the Company recognizes stock-based compensation expense of PSUs subject to graded market vesting on a tranche-by-tranche basis (i.e., the accelerated attribution method). The fair value of the RSUs is estimated using the closing price of the Company's Class A common stock on Nasdaq on the measurement date. The fair value of the PSUs at each measurement date is estimated using a Monte Carlo simulation. The key inputs used in the Monte Carlo simulation are stock price, expected share price volatility, expected life, risk-free interest rate, and vesting hurdles. While the Company believes that the assumptions used in these calculations are reasonable, differences in actual experience or changes in assumptions could materially affect the expense related to the Company's 2021 Plan.

Acquisitions

The Company evaluates acquisitions of assets and other similar transactions to assess whether or not the transaction should be accounted for as a business combination or asset acquisition by first applying a screen test to determine if substantially all of the fair value of the gross assets acquired is concentrated in a single identifiable asset or group of similar identifiable assets. If the screen is met, the transaction is accounted for as an asset acquisition. If the screen is not met, further determination is required as to whether or not the Company has acquired inputs and processes that have the ability to create outputs which would meet the definition of a business. Significant judgment is required in the application of the screen test to determine whether an acquisition is a business combination or an acquisition of assets.

If an acquisition is determined to be a business combination, the assets acquired and liabilities assumed are recorded at their respective estimated fair values at the date of the acquisition. Any excess of the purchase price over the estimated fair values of the identifiable net assets acquired is recorded as goodwill.

If an acquisition is determined to be an asset acquisition, the cost of the asset acquisition, including transaction costs, are allocated to identifiable assets acquired and liabilities assumed based on a relative fair value basis. If the cost of the asset acquisition is less than the fair value of the net assets acquired, no gain is recognized in earnings. The excess fair value of the acquired net assets acquired over the consideration transferred is allocated on a relative fair value basis to the identifiable net assets (excluding non-qualifying assets).

Determining estimated fair value requires a significant amount of judgment and estimates. If our assumptions change or errors are determined in our calculations, the fair value could materially change resulting in a change in our goodwill or identifiable net assets acquired.

Public and Forward Purchase Warrants

Immediately following the Merger, the Company assumed 833,333 Forward Purchase Warrants ("Forward Purchase Warrants") and 12,499,995 public warrants ("Public Warrants") (collectively the "Warrants") that had been previously issued by Crescent. Each whole Warrant entitles the holder to purchase one share of the Company's Class A common stock at a price of \$11.50 per share, subject to adjustments.

Upon consummation of the Merger, the Company concluded that (a) the Public Warrants meet the derivative scope exception for contracts in the Company's own stock and are recorded in stockholders' equity and (b) the Forward Purchase Warrants do not meet the derivative scope exception and are recorded as liabilities on the consolidated balance sheets at fair value upon the Merger, with subsequent changes in the fair value recognized in the consolidated statements of operations and comprehensive loss at each reporting date. The Forward Purchase Warrants are classified as Level 3 fair value measurement and the fair value is measured using a Black-Scholes option pricing model. Inherent in options pricing models are assumptions related to current stock price, exercise price,

expected share price volatility, expected life, risk-free interest rate and dividend yield. While the Company believes that the assumptions used in these calculations are reasonable, changes in assumptions could materially affect the liabilities related to the Warrants.

Recently Adopted Accounting Pronouncements

See Note 2 to our consolidated financial statements included in Part I, Item 1 of this Quarterly Report for recently adopted accounting pronouncements and recently issued accounting pronouncements not yet adopted as of the balance sheet date included in this Quarterly Report.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Concentration risk

Financial instruments that potentially subject the Company to significant concentrations of credit risk consist primarily of cash and cash equivalents, marketable securities and accounts receivable. Risks associated with cash and cash equivalents and marketable securities are mitigated using what the Company considers creditworthy institutions. The Company performs ongoing credit evaluations of its customers' financial condition. Substantially all of LiveVox's assets are in the United States.

As of March 31, 2023 and December 31, 2022, no single issuer represented more than 10% of our marketable securities.

As of March 31, 2023 and December 31, 2022, no single customer represented more than 10% of our total accounts receivable. For the three months ended March 31, 2023 and 2022, no single customer represented more than 10% of our revenue.

The Company relies on third parties for telecommunication, bandwidth, and colocation services that are included in cost of revenue.

As of March 31, 2023, two vendors accounted for approximately 34% of our accounts payable. No other single vendor exceeded 10% of our accounts payable at March 31, 2023. At December 31, 2022, one vendor accounted for approximately 38% of our accounts payable. No other single vendor exceeded 10% of our accounts payable at December 31, 2022. We believe there could be a material impact on future operating results should a relationship with an existing significant supplier cease.

Interest rate sensitivity

The term loan portion of the Credit Facility is subject to interest rate risk. Effective the Eighth Amendment entered into on March 31, 2023, the LIBOR benchmark interest rate was replaced with an adjusted SOFR benchmark interest rate. As amended, borrowings under the Credit Facility will accrue interest at a per annum rate, at the Company's option, based on either (a) a base rate (as defined in the agreement governing the Credit Facility) plus a margin of 4.0% or (b) an adjusted term SOFR (based on one- or three-month interest periods) plus a margin of 5.0%. The Company elected a LIBOR rate at March 31, 2023 and the interest rate for the term loan portion of the Credit Facility was 9.6%. Interest changes affect the fair value of the term loan but the impact on our financial position, cash flows and results of operations was not significant in any period.

Foreign exchange risk

The Company reports its results in U.S. dollars, which is its reporting currency. The functional currency of the Company's foreign subsidiaries is their local currency. We also have international sales that are denominated in foreign currencies. For these international subsidiaries and customers, the monetary assets and liabilities are translated into U.S. dollars at the current exchange rate as of the balance sheet date, and all non-monetary assets and liabilities are translated into U.S. dollars at historical exchange rates. Revenue and expenses are translated using average rates in effect on a monthly basis. The resulting translation gain and loss adjustments are recorded directly as a separate component of stockholders' equity (accumulated other comprehensive loss), unless there is a sale or complete liquidation of the underlying foreign investments, or the adjustment is inconsequential.

We experience fluctuations in transaction gains or losses from remeasurement of monetary assets and liabilities that are denominated in currencies other than the functional currency of the entities in which they are recorded. Exchange gains and losses resulting from foreign currency transactions were not significant in any period and are reported in other income (expense), net in the consolidated statements of operations and comprehensive loss.

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO"), we evaluated the effectiveness of our disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended ("Exchange Act"), as of March 31, 2023.

Based on management's evaluation, our CEO and CFO concluded that, as of March 31, 2023, our disclosure controls and procedures were designed at a reasonable assurance level and were effective to provide reasonable assurance that the information required to be disclosed by the Company in the reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our CEO and CFO, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control over Financial Reporting

There have been no changes in our internal control over financial reporting during our most recent fiscal quarter ended March 31, 2023 that have materially affected or are reasonably likely to materially affect our internal control over financial reporting.

Limitations on Effectiveness of Controls and Procedures

Our disclosure controls and procedures and internal control over financial reporting are designed to provide reasonable assurance of achieving their objectives as specified above. Our management recognizes that any control system, no matter how well designed and operated, is based upon certain judgments and assumptions and can provide only reasonable, not absolute, assurance that its objectives will be met. In addition, the design of a control system must reflect the fact that there are resource constraints and that our management is required to apply its judgment in evaluating the benefits of possible controls and procedures relative to their costs.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

The Company is currently, and from time to time may become, involved in legal or regulatory proceedings arising in the ordinary course of its business, including tort claims, employment disputes and commercial contract disputes. Although the outcome of such claims cannot be predicted with certainty, as of the date of this Quarterly Report, we were not a party to any litigation or regulatory proceeding that would reasonably be expected to be material to LiveVox's business, results of operations, financial condition or cash flows. Please read Note 21 to the consolidated financial statements included in Part I, Item 1 of this Quarterly Report, which is incorporated by reference herein.

ITEM 1A. RISK FACTORS

Factors that could cause our actual results to differ materially from those results in this Quarterly Report are described in our Annual Report. The occurrence of one or more of these factors, alone or in combination with other events or circumstances, may have a material adverse effect on our business, financial condition, results of operations and cash flows. Additional risk factors not presently known to us or that we currently deem immaterial may also impair our business, results of operations, financial condition or cash flows. As of the date of this Quarterly Report, there have been no material changes to the risk factors previously disclosed in the Annual Report, which risk factor section in the Annual Report is hereby incorporated in this Quarterly Report by reference.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

ITEM 4. MINE SAFETY DISCLOSURES

Not applicable.

ITEM 5. OTHER INFORMATION

None

ITEM 6. EXHIBITS

The following exhibits are filed as part of, or incorporated by reference into, this Quarterly Report.

Exhibit No.	Description of Exhibits
3.1	Second Amended and Restated Certificate of Incorporation of LiveVox Holdings, Inc. (filed as Exhibit 3.1 to the Current Report on Form 8-K of the Company on June 24, 2021 and incorporated herein by reference).
3.2	Second Amended and Restated Bylaws of LiveVox Holdings, Inc. (filed as Exhibit 3.2 to the Current Report on Form 8-K of the Company on June 24, 2021 and incorporated herein by reference).
10.1*	Eighth Amendment to Credit Agreement, dated as of March 31, 2023, by and among the Company, the other loan parties party thereto, each lender party thereto and PNC Bank, National Association, as administrative agent for the lenders party thereto.
31.1*	Certification of the Chief Executive Officer, pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2*	Certification of the Chief Financial Officer, pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1**	Certification of the Chief Executive Officer, pursuant to Rule 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2**	Certification of the Chief Financial Officer, pursuant to Rule 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS*	Inline XBRL Instance Document - the instance document does not appear in the Interactive Data File because its XBRL tags are embedded within the Inline XBRL document
101.SCH*	Inline XBRL Taxonomy Extension Schema Document
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase Document
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase Document
	53

101.LAB* Inline XBRL Taxonomy Extension Label Linkbase Document
101.PRE* Inline XBRL Taxonomy Extension Presentation Linkbase Document

104 Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101).

Filed herewith.Furnished herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Quarterly Report to be signed on its behalf by the undersigned thereunto duly authorized.

	LiveV	/ox Holdings, Inc.
Date: May 9, 2023	By:	/s/ John DiLullo
		John DiLullo Chief Executive Officer and Director (Principal Executive Officer)
Date: May 9, 2023	By:	/s/ Gregg Clevenger
		Gregg Clevenger Executive Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)

EIGHTH AMENDMENT TO CREDIT AGREEMENT

This Eighth Amendment to Credit Agreement (this "Amendment") is made on March 31, 2023, by and among LIVEVOX INTERMEDIATE LLC, a Delaware limited liability company (f/k/a LiveVox Holdings, Inc.) ("Parent"), LIVEVOX, INC., a Delaware corporation ("Borrower"), LIVEVOX INTERNATIONAL, INC., a Delaware corporation ("LiveVox International Guarantor"), ENGAGE HOLDINGS, LLC, an Ohio limited liability company d/b/a Businessphone.com ("Engage Guarantor"), SPEECHIQ, LLC, an Ohio limited liability company ("SpeechIQ Guarantor" and, together with Engage Guarantor and LiveVox International Guarantor, the "Guarantors"), the financial institutions party hereto (collectively, the "Lenders" and, individually, a "Lender") and PNC BANK, NATIONAL ASSOCIATION ("PNC"), a national banking association, as administrative agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and permitted assigns in such capacity, "Agent").

BACKGROUND

WHEREAS, on November 7, 2016, among others, Parent, Borrower, Lenders and Agent entered into, inter alia, a certain Credit Agreement (as amended by that certain (i) First Amendment to Credit Agreement, dated as of February 28, 2018, (ii) Second Amendment to Credit Agreement, dated as of May 6, 2019, (iii) Third Amendment and Waiver to Credit Agreement, dated as of December 16, 2019, (iv) Fourth Amendment to Credit Agreement, dated as of December 20, 2019, (v) Fifth Amendment to Credit Agreement, dated as of January 13, 2021, (vi) Sixth Amendment to Credit Agreement, dated as of February 5, 2021, and (vii) Seventh Amendment to Credit Agreement, dated as of August 2, 2021) (and as further amended, restated, amended and restated, modified, renewed, extended, replaced or substituted from time to time, the "Credit Agreement") to reflect certain financing arrangements between the parties thereto. The Credit Agreement and all Loan Documents executed in connection therewith are collectively referred to as the "Financing Agreements." All capitalized terms used herein and not otherwise defined herein shall have the meaning ascribed thereto in the Credit Agreement. In the case of a direct conflict between the provisions of the Credit Agreement and the provisions of this Amendment, the provisions hereof shall prevail;

WHEREAS, Borrower has requested and Agent and Lenders have agreed, subject to the terms and conditions of this Amendment, to amend the Credit Agreement.

NOW, THEREFORE, with the foregoing background hereinafter deemed incorporated by reference herein and made a part hereof, the parties hereto, intending to be legally bound, promise and agree as follows:

- 1. <u>Amendments</u>. Upon the Eighth Amendment Effective Date (as defined below), the Credit Agreement is hereby amended to delete the bold, red stricken text (indicated textually in the same manner as the following example: stricken text) and to add the bold, blue double-underlined text (indicated textually in the same manner as the following example: double-underlined text), each as set forth in Exhibit A hereto.
 - 2. Representations and Warranties. Each Loan Party hereby:
- (a) represents and warrants that such Loan Party has all requisite power and authority to enter into this Amendment and the Loan Documents to which it is a party executed in connection herewith. The execution, delivery, and performance by each Loan Party of this Amendment and the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of such Loan Party. The execution, delivery, and performance by

074658.16087/130783604v.3

each Loan Party of this Amendment and the Loan Documents to which it is a party do not and will not (i) violate any material provision of federal, state, or local law or regulation applicable to any Loan Party or its Subsidiaries, the Governing Documents of any Loan Party or its Subsidiaries, or any order, judgment, or decree of any court or other Governmental Authority binding on any Loan Party or its Subsidiaries, (ii) conflict with, result in a breach of, or constitute (with due notice or lapse of time or both) a default under any Material Contract of any Loan Party or its Subsidiaries except to the extent that any such conflict, breach or default could not individually or in the aggregate reasonably be expected to have a Material Adverse Effect, (iii) result in or require the creation or imposition of any Lien of any nature whatsoever upon any assets of any Loan Party, other than Permitted Liens, or (iv) require any approval of any holder of Equity Interests of a Loan Party or any approval or consent of any Person under any Material Contract of any Loan Party, other than consents or approvals that have been obtained and that are still in force and effect and except, in the case of Material Contracts, for consents or approvals, the failure to obtain the same could not individually or in the aggregate reasonably be expected to cause a Material Adverse Effect;

- (b) reaffirms all representations and warranties made to Agent and Lenders under the Credit Agreement and all of the other Financing Agreements and confirms that all are true and correct in all material respects (except for any representation and warranty which, by its terms, is qualified as to materiality or a Material Adverse Effect, in which case such representations and warranties shall be true and correct in all respects) as of the date hereof as if made on and as of the date hereof, except for representations and warranties which related exclusively to an earlier date, which shall be true and correct in all material respects (except for any representation and warranty which, by its terms, is qualified as to materiality or a Material Adverse Effect, in which case such representations and warranties shall be true and correct in all respects) on and as of such earlier date;
- (c) represents and warrants that no Default or Event of Default has occurred and is continuing under any of the Financing Agreements; and
- (d) represents and warrants that this Amendment and, if any, all assignments, instruments, documents, and agreements executed and delivered in connection herewith are valid, binding and enforceable in accordance with its or their respective terms, except as such enforceability may be limited by any applicable bankruptcy, insolvency, moratorium or similar laws affecting creditors' rights generally.
- 3. <u>Conditions Precedent/Effectiveness Conditions</u>. This Amendment shall be effective upon the date of satisfaction of all of the following conditions precedent (the "<u>Eighth Amendment Effective Date</u>"). All documents, instruments and information, if any, required to be delivered hereunder shall be in form and substance reasonably satisfactory to Agent and Agent's counsel:
 - (a) **Loan Documents**. Agent shall have received each of the following:
 - (i) this Amendment fully executed by Borrower, Parent and the Guarantors; and
 - (ii) the Third Amended and Restated Fee Letter fully executed by Borrower.
- (b) Fees. To the extent invoiced at least one (1) Business Day prior to the date hereof, Borrower shall have paid all Lender Group Expenses incurred in connection with the transactions evidenced by this Amendment.

- (c) Secretary's Certificate; Organizational Documents; Good Standing Certificates. Agent shall have received (i) a certificate of each Loan Party, dated as of the Eighth Amendment Effective Date, with appropriate insertions and attachments, including the certificate of incorporation/formation or other similar organizational document of each Loan Party certified by the relevant authority of the jurisdiction of organization of such Loan Party (to the extent such certification is available in the relevant jurisdiction), the bylaws, operating agreements or other similar organizational document of each Loan Party and the relevant board resolutions or written consents of each Loan Party, and (ii) a good standing certificate, or comparable certificate (to the extent available) for any jurisdiction outside of the United States, as applicable, for each Loan Party from its jurisdiction of organization.
- (d) **No Material Litigation**. There are no actions, suits, or proceedings pending or, to the knowledge of the Loan Parties, after due inquiry, threatened in writing against any Loan Party that either individually or in the aggregate could reasonably be expected to result in a Material Adverse Effect.
- (e) Lien Searches. Agent shall have received the results of a recent lien search on each Loan Party and such results shall be reasonably satisfactory to Agent.
- (f) **KYC**. Agent shall have received all information reasonably requested by Agent to complete all necessary "know your customer" regulatory compliance checks and background checks with respect to the Loan Parties, and the results thereof shall be reasonably satisfactory to Agent in its sole discretion.
- (g) Certificate of Beneficial Ownership. Agent and each Lender shall have received, in form and substance reasonably acceptable to Agent and each Lender, an executed Certificate of Beneficial Ownership.

4. Reference to and Effect on the Loan Documents.

- (a) Notwithstanding anything to the contrary contained herein or in the Credit Agreement, (i) each LIBOR Rate Loan (as defined in the Credit Agreement as in effect prior to the Eighth Amendment Effective Date (the "Existing Credit Agreement")) outstanding on the Eighth Amendment Effective Date (each, an "Existing LIBOR Loan") shall remain outstanding as such until the expiration of the Interest Period applicable to such Existing LIBOR Loan, in accordance with, and subject to all of the terms and conditions of, the Existing Credit Agreement, and (ii) interest on each such Existing LIBOR Loan shall continue to accrue to, and shall be payable on, each interest payment date applicable thereto until the Interest Period for such Existing LIBOR Loan ends, in each case in accordance with Section 2.14 of the Existing Credit Agreement. From and after the Eighth Amendment Effective Date, (x) the Borrower shall not be permitted to request that any Lender fund, and no Lender shall fund, any LIBOR Rate Loan, (y) no Existing LIBOR Loan may be continued as a LIBOR Rate Loan and (z) each Existing LIBOR Loan may be converted to a Term SOFR Rate Loan or a Base Rate Loan in accordance with the Credit Agreement.
- (b) Upon and after the effectiveness of this Amendment, each reference in the Credit Agreement to "this Agreement", "hereunder", "hereof" or words of like import referring to the Credit Agreement shall mean and be a reference to the Credit Agreement as modified and amended hereby.
- (c) Except as specifically amended above, the Credit Agreement and all Loan Documents are and shall continue to be in full force and effect and are hereby in all respects ratified and confirmed and shall constitute the legal, valid, binding and enforceable obligations of the Loan Parties to Agent and the Lenders.

- (d) The execution, delivery and effectiveness of this Amendment shall not, except as expressly provided for herein, operate as a waiver of any right, power or remedy of Agent and/or the Lenders under the Credit Agreement or any of the Loan Documents, nor constitute a waiver of any provision of the Credit Agreement or any of the Loan Documents.
- (e) To the extent that any terms and conditions in any of the Loan Documents shall contradict or be in conflict with any terms or conditions of the Credit Agreement, after giving effect to this Amendment, such terms and conditions are hereby deemed modified or amended accordingly to reflect the terms and conditions of the Credit Agreement as modified or amended hereby.
- 5. <u>Ratification</u>. Each Loan Party hereby restates, ratifies and reaffirms each and every term and condition set forth in the Credit Agreement, as amended hereby, and the Loan Documents effective as of the Eighth Amendment Effective Date.
- 6. <u>Estoppel</u>. To induce Agent and the Lenders to enter into this Amendment and to continue to make advances to Borrower under the Credit Agreement, each Loan Party hereby acknowledges and agrees that, as of the date hereof, there exists no right of offset, defense, counterclaim or objection in favor of Borrower as against Agent or any Lender with respect to the Obligations.
- 7. <u>Integration</u>. This Amendment, together with the Credit Agreement and all other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the date hereof.
- 8. <u>Severability of Provisions</u>. Each provision of this Amendment shall be severable from every other provision of this Amendment for the purpose of determining the legal enforceability of any specific provision.

9. <u>Miscellaneous</u>.

- (a) Third Party Rights. No rights are intended to be created hereunder for the benefit of any third party donee, creditor, or incidental beneficiary.
- (b) <u>Section Headings</u>. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Amendment.
- (c) <u>Interpretation</u>. Neither this Amendment nor any uncertainty or ambiguity herein shall be construed against the Lender Group or Parent or Borrower, whether under any rule of construction or otherwise. On the contrary, this Amendment has been reviewed by all parties and shall be construed and interpreted according to the ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.
- (d) <u>Counterparts; Electronic Execution</u>. This Amendment may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together, shall constitute but one and the same Amendment. Delivery of an executed counterpart of this Amendment by telefacsimile or other electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Amendment. Any party delivering an executed counterpart of this Amendment by telefacsimile or other electronic method of transmission also shall deliver an original executed counterpart of this Amendment

but the failure to deliver an original executed counterpart shall not affect the validity, enforceability, and binding effect of this Amendment.

(e) <u>Choice of Law. Section 12</u> of the Credit Agreement is hereby incorporated herein by reference.

IN WITNESS WHEREOF, the parties have caused this Amendment to be executed and delivered by their duly authorized officers as of the date first above written.

PARENT: LIVEVOX INTERMEDIATE LLC

By: /s/ Gregg Clevenger Name: Gregg Clevenger Title: Chief Financial Officer

BORROWER: LIVEVOX, INC.

By: /s/ Gregg Clevenger Name: Gregg Clevenger Title: Chief Financial Officer

GUARANTORS: LIVEVOX INTERNATIONAL, INC.

By: /s/ Gregg Clevenger Name: Gregg Clevenger Title: Chief Financial Officer

SPEECHIQ, LLC

By: /s/ John DiLullo Name: John DiLullo

Title: President, Chief Executive Officer and Secretary

ENGAGE HOLDINGS, LLC

By: /s/ John DiLullo Name: John DiLullo

Title: President, Chief Executive Officer and Secretary

Signature Page to Eighth Amendment to Credit Agreement

AGENT AND LENDER:

PNC BANK, NATIONAL ASSOCIATION

By: /s/ Eric Klein Name: Eric Klein Title: Vice President

Signature Page to Eighth Amendment to Credit Agreement

LENDER: BANK HAPOALIM B.M.

By: /s/ Tal Hofman Name: Tal Hofman

Title: VP

By: /s/ Gal Defes Name: Gal Defes Title: SVP

Signature Page to Eighth Amendment to Credit Agreement

EXHIBIT A

Credit Agreement

(See attached)

074658.16087/130783604v.3

CREDIT AGREEMENT

by and among

PNC BANK, NATIONAL ASSOCIATION,

as Administrative Agent,

THE LENDERS THAT ARE PARTIES HERETO

as the Lenders,

LIVEVOX INTERMEDIATE LLC (f/k/a LiveVox Holdings, Inc.)

as Parent

LIVEVOX, INC.

as Borrower, and

the Guarantors party hereto

Dated as of November 7, 2016

074658.16087/130783594v.4

TABLE OF CONTENTS

Page

DEFINI	TIONS AND CONSTRUCTION- 1
1.1.	Definitions 1
1.2.	Accounting Terms 1
1.3.	Code 2
1.4.	Construction 2
1.5.	Time References 3
1.6.	Schedules and Exhibits 3
<u>1.7.</u>	<u>Limited Condition Acquisitions</u> <u>3</u>
<u>1.8.</u>	Benchmark Replacement Notification 3
<u>1.9.</u>	Conforming Changes Relating to Term SOFR Rate 4
LOANS	AND TERMS OF PAYMENT: 34
2.1.	Revolving Loans 34
2.2.	Procedures for Requesting Revolving Loans; Procedures for Selection of Applicable Interest Rates for All Loans 45
2.3.	Term Loan 67
2.4.	Swing Loans 78
2.5.	Disbursement of Loan Proceeds 89
2.6.	Making and Settlement of Loans 910
2.7.	Maximum Loans 4112
2.8.	Manner and Repayment of Loans 4112
2.9.	Defaulting Lenders +112
2.10.	Statement of Account 4415
2.11.	Letters of Credit 4416
2.12.	Application of Payments; Termination of Commitments; Prepayments 2223
2.13.	Payment of Obligations 2728
2.14.	Interest Rates and Letter of Credit Fees: Rates, Payments, and Calculations 2829
2.15.	Fees 2930
2.16.	Special Provisions Applicable to LIBOR Alternate Rate of Interest 3031
2.17.	Capital Requirements 3136
CONDI	ΓΙΟΝS; TERM OF AGREEMENT . 3337
3.1.	Conditions Precedent to the Initial Extension of Credit 3337

3.2. Conditions Precedent to all Extensions of Credit 3338

1.

2.

3.

3.3. Maturity 3338 3.4. Effect of Maturity 3338 3.5. Early Termination by Borrower 3438 3.6. Conditions Subsequent 3438 4. REPRESENTATIONS AND WARRANTIES: 3439 4.1. Due Organization and Qualification; Subsidiaries 3539 4.2. Due Authorization; No Conflict 3540 4.3. Governmental Consents 3640 4.4. Binding Obligations; Perfected Liens 3640 4.5. Title to Assets; No Encumbrances 3641 4.6. Litigation <u>3641</u> 4.7. Compliance with Laws 3741 4.8. No Material Adverse Effect 3742 4.9. Solvency 3742 4.10. Employee Benefits 3742 4.11. Environmental Condition 3842 4.12. Complete Disclosure 3842 4.13. Material Contracts 3843 4.14. Patriot Act 3943 4.15. Indebtedness 3943 4.16. Payment of Taxes 3944 4.17. Margin Stock <u>3944</u> 4.18. Governmental Regulation 4044 4.19. Anti-Terrorism Laws; Anti-Corruption Laws 4044 4.20. Employee and Labor Matters 4045 4.21. Parent as a Holding Company 4145 4.22. SpeechIQ Acquisition; Engage Acquisition 4145 4.23. Merger Documentation Representations 4145 4.24. Leases 4145 4.25. Hedge Agreements 4146 4.26. Certificate of Beneficial Ownership 4146 5. AFFIRMATIVE COVENANTS: 4146 5.1. Financial Statements, Reports, Certificates 4246 5.2. Reporting 4246 5.3. Existence 4246 5.4. Maintenance of Properties 4247

```
5.5.
           Taxes <u>4247</u>
      5.6. Insurance \frac{4247}{}
      5.7. Inspection <u>4348</u>
      5.8. Compliance with Laws 4448
      5.9. Environmental 4448
      5.10. Disclosure Updates 4449
      5.11. Formation of Subsidiaries 4549
      5.12. Further Assurances 4550
      5.13. Lender Meetings 4651
      5.14. Bank Products 4651
      5.15. Certificate of Beneficial Ownership and Other Additional Information 4651
            Amendments to SpeechIQ Acquisition Documentation or Engage Acquisition Documentation 4751
      5.17. Amendments to Merger Documentation 4751
      5.18. Engage PPP Loan. 4751
6. NEGATIVE COVENANTS: 4752
      6.1. Indebtedness 4752
      6.2. Liens 47<u>52</u>
      6.3. Restrictions on Fundamental Changes 4752
      6.4. Disposal of Assets 4853
      6.5. Nature of Business 4853
      6.6. Prepayments and Amendments 4853
      6.7. Restricted Payments 5054
      6.8. Accounting Methods 5155
      6.9. Investments <u>5155</u>
      6.10. Transactions with Affiliates 5155
      6.11. Use of Proceeds 5357
      6.12. Limitation on Issuance of Equity Interests 5358
      6.13. Parent as Holding Company 5458
      6.14. OFAC; Patriot Act 5458
      6.15. Amendments to SpeechIQ Acquisition Documentation or Engage Acquisition Documentation 5458
      6.16. Amendments to Management Incentive Documentation 5459
```

6.17. Amendments to Merger Documentation 5459

```
FINANCIAL COVENANTS: 5459
   EVENTS OF DEFAULT: 5660
      8.1. Payments <u>5660</u>
      8.2. Covenants <u>5761</u>
      8.3. Judgments <del>576</del>1
      8.4. Voluntary Bankruptcy, etc. 5761
      8.5. Involuntary Bankruptcy, etc 5761
      8.6. Default Under Other Agreements 5862
      8.7. Representations, etc 5862
      8.8. Guaranty <del>58</del>62
      8.9. Security Documents <u>5862</u>
      8.10. Loan Documents 5862
      8.11. Change of Control 5962
      8.12. Seizures <del>5963</del>
9. RIGHTS AND REMEDIES: 5963
      9.1. Rights and Remedies 5963
      9.2. Remedies Cumulative 6064
      9.3. [Reserved] 6064
10. WAIVERS; INDEMNIFICATION: 6264
      10.1. Demand; Protest; etc 6264
      10.2. The Lender Group's Liability for Collateral 6264
      10.3. Indemnification 6265
11. NOTICES: 6466
12. CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION 6567
13. ASSIGNMENTS AND PARTICIPATIONS; SUCCESSORS: 6769
      13.1. Assignments and Participations 6769
      13.2. Successors <del>72</del>74
14. AMENDMENTS; WAIVERS. 7274
      14.1. Amendments and Waivers 7274
      14.2. Replacement of Certain Lenders 7476
      14.3. No Waivers; Cumulative Remedies 7476
15. AGENT; THE LENDER GROUP: 7577
      15.1. Appointment and Authorization of Agent 7577
      15.2. Delegation of Duties 7678
      15.3. Liability of Agent 7678
```

```
15.4. Reliance by Agent 7678
            Notices of Default or Event of Default 7678
      15.5.
      15.6.
            Credit Decision 7779
      15.7. Costs and Expenses; Indemnification 7780
      15.8. Agent in Individual Capacity 7880
      15.9. Successor Agent 7880
      15.10. Lender in Individual Capacity 7981
      15.11. Collateral Matters 7981
      15.12. Restrictions on Actions by Lenders; Sharing of Payments 8183
      15.13. Agency for Perfection 8284
      15.14. Payments by Agent to the Lenders 8284
      15.15. Concerning the Collateral and Related Loan Documents 8284
      15.16. Financial Examination Reports; Confidentiality; Disclaimers by Lenders; Other Reports and Information 8284
      15.17. Several Obligations; No Liability 8385
16. WITHHOLDING TAXES: 8486
      16.1. Payments 8486
      16.2. Exemptions <u>8587</u>
      16.3. Reductions 8688
      16.4. Refunds 8789
17. GENERAL PROVISIONS: 8789
      17.1. Effectiveness 8789
      17.2.
            Section Headings 8790
            Interpretation 8890
      17.4. Severability of Provisions 8890
      17.5. Bank Product Providers 8890
      17.6. Debtor-Creditor Relationship 8991
      17.7. Counterparts; Electronic Execution 8991
      17.8. Revival and Reinstatement of Obligations; Certain Waivers 8991
      17.9. Confidentiality 9092
      17.10. Survival 9193
      17.11. Patriot Act 9294
      17.12. Integration <u>9294</u>
```

EXHIBITS AND SCHEDULES

Exhibit 2.3 Form of Term Loan Note
Exhibit 2.4(a) Form of Swing Loan Note
Exhibit A-1 Form of Assignment and Acceptance
Exhibit C-1 Form of Compliance Certificate
Exhibit C-2 Form of Early Election Certificate
Exhibit I-1 Form of IP Reporting Certificate
Exhibit P-1 Form of Perfection Certificate
Exhibit 3.1(o) Form of Financial Condition Certificate

Exhibit 2.1 Form of Revolving Loan Note

Schedule A-1 Schedule C-1 Commitments
Schedule C-2 Competitors
Schedule I-1 Ineligible Institutions
Schedule P-1 Permitted Investments

Schedule P-2 Schedule R-1 Schedule 3.1 Schedule 3.6 Conditions Precedent Conditions Subsequent

Schedule 4.1(b) Capitalization of Borrower Schedule 4.1(c) Capitalization of Borrower's Sub

Schedule 4.1(c) Capitalization of Borrower's Subsidiaries

Schedule 4.6 Litigation
Schedule 4.11 Environmental Matters

Schedule 4.11 Environmental Matters
Schedule 4.13 Material Contracts
Schedule 4.14 Permitted Indebtedness
Schedule 4.16 Payment of Taxes

Schedule 5.1 Financial Statements, Reports, Certificates

Schedule 5.1 (d) Billing Systems
Schedule 5.2 Collateral Reporting
Schedule 6.5 Nature of Business

CREDIT AGREEMENT

THIS CREDIT AGREEMENT (this "Agreement"), is entered into as of November 7, 2016, by and among the lenders identified on the signature pages hereof (each of such lenders, together with its successors and permitted assigns, is referred to hereinafter as a "Lender", as that term is hereinafter further defined), PNC BANK, NATIONAL ASSOCIATION, a national banking association, as administrative agent for each member of the Lender Group and the Bank Product Providers (in such capacity, together with its successors and assigns in such capacity, "Agent"), LIVEVOX INTERMEDIATE LLC, a Delaware limited liability company (f/k/a LiveVox Holdings, Inc.) ("Parent"), LIVEVOX, INC., a Delaware corporation ("Borrower"), ENGAGE HOLDINGS, LLC, an Ohio limited liability company ("Engage Guarantor"), LIVEVOX INTERNATIONAL, INC., a Delaware corporation ("LiveVox International Guarantor"), SPEECHIQ, LLC, an Ohio limited liability company ("SpeechIQ Guarantor"), and together with Engage Guarantor and LiveVox International Guarantor, the "Guarantors").

The parties agree as follows:

1. DEFINITIONS AND CONSTRUCTION.

- 1.1. **<u>Definitions</u>**. Capitalized terms used in this Agreement shall have the meanings specified therefor on <u>Schedule 1.1</u>.
- 1.2. <u>Accounting Terms</u>. All accounting terms not specifically defined herein shall be construed in accordance with GAAP; <u>provided</u>, that if Borrower notifies Agent that Borrower requests an amendment to any provision hereof to eliminate the effect of any Accounting Change occurring after the Closing Date or in the application thereof on the operation of such provision (or if Agent notifies Borrower that the Required Lenders request an amendment to any provision hereof for such purpose), regardless of whether any such notice is given before or after such Accounting Change or in the application thereof, then Agent and Borrower agree that they will negotiate in good faith amendments to the provisions of this Agreement that are directly affected by such Accounting Change with the intent of having the respective positions of the Lenders and Borrower after such Accounting Change conform as nearly as possible to their respective positions as of the date of this Agreement and, until any such amendments have been agreed upon and agreed to by the Required Lenders, the provisions in this Agreement shall be calculated as if no such Accounting Change had occurred; provided, further, that all obligations of any Person that are or would have been treated as operating leases for purposes of GAAP prior to the issuance by the Financial Accounting Standards Board on February 25, 2016 of an Accounting Standards Update (the "ASU") shall continue to be accounted for as operating leases for purposes of all financial definitions, calculations and covenants for purposes of this Agreement (other than for purposes of the delivery of financial statements prepared in accordance with GAAP), whether or not such operating lease obligations were in effect on such date, notwithstanding the fact that such obligations are required in accordance with the ASU (on a prospective or retroactive basis or otherwise) to be treated as capitalized or finance lease obligations in accordance with GAAP. When used herein, the term "financial statements" shall include the notes and schedules thereto. Whenever the term "Parent" is used in respect of a financial covenant or a related definition, it shall be understood to mean Parent and its Subsidiaries on a consolidated basis, unless the context clearly requires otherwise. Notwithstanding anything to the contrary contained herein, (a) all financial statements delivered hereunder shall be prepared, and all financial covenants contained herein shall be calculated, without giving effect to any election under the Statement of Financial Accounting Standards No. 159 (or any similar accounting principle) permitting a Person to value its financial liabilities or Indebtedness at the fair value thereof, and (b) the term "unqualified opinion" as used herein to

074658.16087/130783594v.4

refer to opinions or reports provided by accountants shall mean an opinion or report that is (i) unqualified, and (ii) does not include any explanation, supplemental comment, or other comment concerning the ability of the applicable Person to continue as a going concern or concerning the scope of the audit.

- 1.3. <u>Code</u>. Any terms used in this Agreement that are defined in the Code shall be construed and defined as set forth in the Code unless otherwise defined herein; <u>provided</u>, that to the extent that the Code is used to define any term herein and such term is defined differently in different Articles of the Code, the definition of such term contained in Article 9 of the Code shall govern.
- 1.4. Construction. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, references to the plural include the singular, references to the singular include the plural, the terms "includes" and "including" are not limiting, and the term "or" has, except where otherwise indicated, the inclusive meaning represented by the phrase "and/or." The words "hereof," "herein," "hereby," "hereunder," and similar terms in this Agreement or any other Loan Document refer to this Agreement or such other Loan Document, as the case may be, as a whole and not to any particular provision of this Agreement or such other Loan Document, as the case may be. Section, subsection, clause, schedule, and exhibit references herein are to this Agreement unless otherwise specified. Any reference in this Agreement or in any other Loan Document to any agreement, instrument, or document shall include all alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements, thereto and thereof, as applicable (subject to any restrictions on such alterations, amendments, changes, extensions, modifications, renewals, replacements, substitutions, joinders, and supplements set forth herein). The words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties. Any reference herein or in any other Loan Document to the satisfaction, repayment, or payment in full of the Obligations shall mean (a) the payment or repayment in full in immediately available funds of (i) the principal amount of, and interest accrued and unpaid with respect to, all outstanding Loans, together with the payment of any premium applicable to the repayment of the Loans, (ii) all Lender Group Expenses that have accrued and are unpaid regardless of whether demand has been made therefor, (iii) all fees or charges that have accrued hereunder or under any other Loan Document (including the Letter of Credit Fee and the Unused Line Fee) and are unpaid, (b) in the case of contingent reimbursement obligations with respect to Letters of Credit, providing Letter of Credit Collateralization, (c) in the case of obligations with respect to Bank Products (other than Hedge Obligations that, at such time, are allowed by the applicable Bank Product Provider to remain outstanding without being required to be repaid or cash collateralized), providing Bank Product Collateralization, (d) the receipt by Agent of cash collateral in order to secure any other contingent Obligations for which a claim or demand for payment has been made on or prior to such time or in respect of matters or circumstances known to Agent or a Lender at such time that are reasonably expected to result in any loss, cost, damage, or expense (including attorneys' fees and legal expenses), such cash collateral to be in such amount as Agent reasonably determines is appropriate to secure such contingent Obligations, (e) the payment or repayment in full in immediately available funds of all other outstanding Obligations (including the payment of any termination amount then applicable (or which would or could become applicable as a result of the repayment of the other Obligations) under Hedge Agreements provided by Hedge Providers) other than (i) unasserted contingent indemnification Obligations, (ii) any Bank Product Obligations (other than Hedge Obligations) that, at such time, are allowed by the applicable Bank Product Provider to remain outstanding without being required to be repaid or cash collateralized, and (iii) any Hedge Obligations that, at such time, are allowed by the applicable Hedge Provider to remain outstanding without being required to be repaid or cash collateralized, and (f) the termination of all of the Commitments of the Lenders. Any reference herein to any Person shall be construed to

include such Person's successors and assigns. Any requirement of a writing contained herein or in any other Loan Document shall be satisfied by the transmission of a Record. Any officer of a Loan Party executing any Loan Document or any certificate or other document made or delivered pursuant hereto or thereto, so executes or certifies in his/her capacity as a an officer on behalf of the applicable Loan Party and not in any individual capacity.

- 1.5. **Time References**. Unless the context of this Agreement or any other Loan Document clearly requires otherwise, all references to time of day refer to the time of day in New York, New York. For purposes of the computation of a period of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each means "to and including"; <u>provided</u> that, with respect to a computation of fees or interest payable to Agent or any Lender, such period shall in any event consist of at least one full day.
- 1.6. Schedules and Exhibits. All of the schedules and exhibits attached to this Agreement shall be deemed incorporated herein by reference.
- 1.7. Limited Condition Acquisitions. Notwithstanding anything herein to the contrary, with respect to any Limited Condition Acquisition, at the Borrower's option, any Leverage Ratio, Recurring Revenue Ratio, Fixed Charge Coverage Ratio or Dollar threshold required for consummation of such Limited Condition Acquisition shall be determined, and any representation and warranty, Default or Event of Default blocker shall be tested, as of the date the definitive acquisition agreement for such Limited Condition Acquisition is entered into and calculated as if such Limited Condition Acquisition was consummated on such date so long as no Specified Event of Default has occurred and is continuing on such date; provided, that, if the Borrower has made such an election, in connection with the calculation of any ratio or basket permitted hereunder, during the period from and after such date and prior to the earliest of (i) the date on which such Limited Condition Acquisition is consummated, (ii) the definitive agreement for such Limited Condition Acquisition is terminated and (iii) the date that is 180 days after the definitive agreement for such Limited Condition Acquisition is executed, any ratio shall be calculated (x) on a pro forma basis assuming such Limited Condition Acquisition and other pro forma events in connection therewith have been consummated; provided further, that, in any event, no Specified Event of Default shall have occurred and be continuing on the date such Limited Condition Acquisition is consummated.
- 1.8. **Benchmark Replacement Notification**. Section 2.16(b) of this Agreement provides a mechanism for determining an alternate rate of interest in the event that the Term SOFR Rate is no longer available or in certain other circumstances. Agent does not warrant or accept any responsibility for and shall not have any liability with respect to, the administration, submission or any other matter related to the Term SOFR Rate or with respect to any alternative or successor rate thereto, or replacement rate therefor.
- 1.9. Conforming Changes Relating to Term SOFR Rate. With respect to the Term SOFR Rate, Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document; provided that, with respect to any such amendment effected, Agent shall provide notice to the Borrower and the Lenders of each such amendment implementing such Conforming Changes reasonably promptly after such amendment becomes effective.

2. LOANS AND TERMS OF PAYMENT.

2.1. **Revolving Loans**.

(a) Revolving Loans. Subject to the terms and conditions set forth in this Agreement, each Lender, severally and not jointly, will make Revolving Loans to Borrower in aggregate amounts outstanding at any time equal to such Lender's Pro Rata Share of an amount equal to, as of any date of determination, (i) the Maximum Revolver Amount minus (ii) the aggregate amount of all Revolving Loans and Swing Loans then outstanding plus the Maximum Undrawn Amount of all Letters of Credit then outstanding. The Revolving Loans shall be evidenced by one or more secured promissory notes (collectively, the "Revolving Loan Note") substantially in the form attached hereto as Exhibit 2.1. Notwithstanding anything to the contrary contained in the foregoing or otherwise in this Agreement, (x) the outstanding aggregate principal amount of Swing Loans and the Revolving Loans at any one time outstanding plus the Maximum Undrawn Amount of all Letters of Credit then outstanding shall not exceed the Maximum Revolver Amount and (y) Agent shall have the right (but not the obligation) to establish from time to time Bank Product Reserves against the Maximum Revolving Loan Amount. Amounts borrowed pursuant to this Section 2.1 may be repaid and, subject to the terms and conditions of this Agreement, reborrowed at any time during the term of this Agreement. The outstanding principal amount of the Revolving Loans, Swing Loans and all other extensions of credit hereunder, together with interest accrued and unpaid thereon, shall be due and payable on the Maturity Date or, if earlier, on the date on which they are declared due and payable pursuant to the terms of this Agreement.

(b) **Protective Advances**.

- (i) Any contrary provision of this Agreement or any other Loan Document notwithstanding, at any time (after the occurrence and during the continuance of a Default or an Event of Default, or any of the other applicable conditions precedent set forth in Section 3 are not satisfied), Agent hereby is authorized by Borrower and the Lenders, from time to time, in Agent's sole discretion, to make Revolving Loans to, or for the benefit of, Borrower, on behalf of the Revolving Lenders, that Agent, in its Permitted Discretion, deems necessary or desirable (1) to preserve or protect the Collateral, or any portion thereof, or (2) to enhance the likelihood of repayment of the Obligations (other than the Bank Product Obligations) (the Revolving Loans described in this Section 2.1(b)(i) shall be referred to as "Protective Advances"). Notwithstanding the foregoing, the aggregate amount of all Protective Advances outstanding at any one time, when taken together with the aggregate amount of all Bank Product Reserves then outstanding, shall not exceed \$2,500,000 (the "Maximum Applicable Reserve").
- (ii) Each Protective Advance shall be deemed to be a Revolving Loan hereunder, except that no Protective Advance shall be eligible to be a LIBORTerm SOFR Rate Loan and, prior to Settlement therefor, all payments on the Protective Advances shall be payable to Agent solely for its own account. The Protective Advances shall be repayable on demand, secured by Agent's Liens, constitute Obligations hereunder, and bear interest at the rate applicable from time to time to Revolving Loans that are Base Rate Loans. The provisions of this Section 2.1(b) are for the exclusive benefit of Agent and the Lenders and are not intended to benefit Borrower (or any other Loan Party) in any way.

2.2. Procedures for Requesting Revolving Loans; Procedures for Selection of Applicable Interest Rates for All Loans

(a) Borrower may notify Agent in writing prior to 1:00 p.m. on a Business Day of Borrower's request to incur, on that day, a Revolving Loan hereunder. Should any

amount required to be paid as interest hereunder, or as fees or other charges under this Agreement or any other Loan Document or with respect to any other Obligation, become due, same shall be deemed a request for a Revolving Loan maintained as a Base Rate Loan as of the date such payment is due, in the amount required to pay in full such interest, fee, charge or Obligation, and such request shall be irrevocable.

- (b) Notwithstanding the provisions of subsection (a) above, in the event Borrower desires to obtain a LIBORTerm SOFR Rate Loan for any Loan (other than a Swing Loan or Protective Advance), Borrower shall give Agent written notice by no later than 1:00 p.m. on the day which is three (3) Business Days prior to the date such LIBORTerm SOFR Rate Loan is to be borrowed, specifying (i) the date of the proposed borrowing (which shall be a Business Day), (ii) the type of borrowing and the amount of such Loan to be borrowed, which amount shall be in a minimum amount of \$1,000,000 and in integral multiples of \$50,000 thereafter, and (iii) the duration of the first Interest Period therefor. Interest Periods for LIBORTerm SOFR Rate Loans shall be for one, two, or three months; provided that, if an Interest Period would end on a day that is not a Business Day, it shall end on the next succeeding Business Day unless such day falls in the next succeeding calendar month in which case the Interest Period shall end on the next preceding Business Day. No LIBORTerm SOFR Rate Loan shall be made available to Borrower during the continuance of a Default or an Event of Default. After giving effect to each requested LIBORTerm SOFR Rate Loan, including those which are converted from a Base Rate Loan under Section 2.2(e), there shall not be outstanding more than five (5) LIBORTerm SOFR Rate Loans, in the aggregate.
- (c) Each Interest Period of a <u>LIBOR Term SOFR</u> Rate Loan shall commence on the date such <u>LIBOR Term SOFR</u> Rate Loan is made and shall end on such date as Borrower may elect as set forth in subsection (b)(iii) above, provided that the exact length of each Interest Period shall be determined in accordance with the practice of the interbank market for offshore Dollar deposits and no Interest Period shall end after the Maturity Date.
- (d) Borrower shall elect the initial Interest Period applicable to a LIBORTerm SOFR Rate Loan by its notice of borrowing given to Agent pursuant to Section 2.2(b) or by its notice of conversion given to Agent pursuant to Section 2.2(e), as the case may be. Borrower shall elect the duration of each succeeding Interest Period by giving irrevocable written notice to Agent of such duration not later than 1:00 p.m. on the day which is three (3) Business Days prior to the last day of the then current Interest Period applicable to such LIBORTerm SOFR Rate Loan. If Agent does not receive timely notice of the Interest Period elected by Borrower, Borrower shall be deemed to have elected to convert such LIBORTerm SOFR Rate Loan to a Base Rate Loan as of the last day of the Interest Period applicable to such LIBORTerm SOFR Rate Loan, subject to Section 2.2(e) below.
- (e) Provided that no Default or Event of Default shall have occurred and be continuing, Borrower may, on the last Business Day of the then current Interest Period applicable to any outstanding <u>LIBORTerm SOFR</u> Rate Loan, or on any Business Day with respect to Base Rate Loans, convert any such loan into a loan of another type in the same aggregate principal amount provided that any conversion of a <u>LIBORTerm SOFR</u> Rate Loan shall be made only on the last Business Day of the then current Interest Period applicable to such <u>LIBORTerm SOFR</u> Rate Loan. If Borrower desires to convert a loan, Borrower shall give Agent written notice by no later than 1:00 p.m. (i) on the day which is three (3) Business Days prior to the date on which such conversion from a Base Rate Loan to a <u>LIBORTerm SOFR</u> Rate Loan, or (ii) on the day which is one (1) Business Day prior to the date on which such conversion is to occur (which date shall be the last Business Day of the Interest Period for the applicable <u>LIBORTerm SOFR</u> Rate Loan to a Base Rate Loan, specifying, in each case, the date of such conversion, the

loans to be converted and if the conversion is to a <u>LIBORTerm SOFR</u> Rate Loan, the duration of the first Interest Period therefor. <u>In the case of the renewal of a Term SOFR Rate Loan at the end of an Interest Period, the first day of the new Interest Period shall be the last day of the preceding Interest Period, without duplication in payment of interest for such day.</u>

- (f) At its option and upon written notice given prior to 1:00 p.m. at least three (3) Business Days prior to the date of such prepayment, Borrower may, subject to Section 2.2(g) hereof, prepay the LIBORTerm SOFR Rate Loans in whole at any time or in part from time to time with accrued interest on the principal being prepaid to the date of such repayment. Borrower shall specify the date of prepayment of Loans which are LIBORTerm SOFR Rate Loans and the amount of such prepayment. In the event that any prepayment of a LIBORTerm SOFR Rate Loan is required or permitted on a date other than the last Business Day of the then current Interest Period with respect thereto, Borrower shall indemnify Agent and Lenders therefor in accordance with Section 2.2(g) hereof.
- (g) Borrower shall indemnify Agent and Lenders and hold Agent and Lenders harmless from and against any and all losses or expenses that Agent and/or Lenders may sustain or incur as a consequence of any prepayment, conversion of or any default by Borrower in the payment of the principal of or interest on any LIBORTerm SOFR Rate Loan or failure by Borrower to complete a borrowing of, a prepayment of or conversion of or to a LIBORTerm SOFR Rate Loan after notice thereof has been given, including, but not limited to, any interest payable by Agent or a Lender to lenders of funds obtained by it in order to make or maintain its LIBORTerm SOFR Rate Loans hereunder. A certificate as to any additional amounts payable pursuant to the foregoing sentence submitted by Agent or any Lender to Borrower shall be conclusive absent manifest error.
- (h) Notwithstanding any other provision hereof, if any applicable law, treaty, regulation or directive, or any change therein or in the interpretation or application thereof, including without limitation any Change in Law, shall make it unlawful for Agent or any Lender (for purposes of this subsection (h), the term "Lender" shall include Agent, any Lender and the office or branch where any Agent or any Lender or any Person controlling Agent or such Lender makes or maintains any LIBORTerm SOFR Rate Loans) to make or maintain its LIBORTerm SOFR Rate Loans, the obligation of Lenders (or such affected Lender) to make LIBORTerm SOFR Rate Loans hereunder shall forthwith be cancelled and Borrower shall, if any affected LIBORTerm SOFR Rate Loans are then outstanding, promptly upon request from Agent, either pay all such affected LIBORTerm SOFR Rate Loans or convert such affected LIBORTerm SOFR Rate Loans into loans of another type. If any such payment or conversion of any LIBORTerm SOFR Rate Loan is made on a day that is not the last day of the Interest Period applicable to such LIBORTerm SOFR Rate Loan, Borrower shall pay Agent, upon Agent's request, such amount or amounts set forth in clause (g) above. A certificate as to any additional amounts payable pursuant to the foregoing sentence submitted by Agent to Borrower shall be conclusive absent manifest error.
- (i) Anything to the contrary contained herein notwithstanding, neither Agent nor any Lender, nor any of their participants, is required to actually acquire Term SOFR deposits to fund or otherwise match fund any Obligation as to which interest accrues based on the Term SOFR Rate. The provisions set forth herein shall apply as if each Lender or its participants had match funded any Obligation as to which interest is accruing based on the Term SOFR Rate by acquiring SOFR deposits for each Interest Period in the amount of the Term SOFR Rate Loans.
- 2.3. <u>Term Loan</u>. On the Closing Date, each Lender with a Term Loan Commitment (severally, not jointly or jointly and severally) made a term loan to Borrower in an aggregate principal amount equal to \$35,000,000 (the "<u>Closing Date Term Loan</u>"). On the First

Amendment Effective Date, each Lender with a Term Loan Commitment (severally, not jointly or jointly and severally) made an additional term loan to Borrower in an amount equal to such Lender's Term Loan Commitment in an aggregate principal amount equal to \$10,350,000 (the "First Amendment Term Loan"). As of the Third Amendment Effective Date, the outstanding principal balance of the Closing Date Term Loan and the First Amendment Term Loan, collectively, was \$43,706,250. Subject to the term and conditions of this Agreement, on the Third Amendment Effective Date, each Lender with a Term Loan Commitment (severally, not jointly or jointly and severally) made an additional term loan to Borrower in an amount equal to such Lender's Term Loan Commitment in an aggregate principal amount equal to \$13,900,000 (the "Third Amendment Term Loan"), and together with the Closing Date Term Loan and First Amendment Term Loan, collectively, the "Term Loan"). As of the Seventh Amendment Effective Date, the Term Loan shall be, with respect to principal, subject to acceleration during the existence of an Event of Default under this Agreement or termination of this Agreement, be payable in installments as follows:

Installment Due Date	Installment Amount
September 30, 2021	\$140,235.00
December 31, 2021	\$140,235.00
March 31, 2022	\$140,235.00
June 30, 2022	\$140,235.00
September 30, 2022	\$140,235.00
December 31, 2022	\$140,235.00
March 31, 2023	\$140,235.00
June 30, 2023	\$280,470.00
September 30, 2023	\$280,470.00
December 31, 2023	\$280,470.00
March 31, 2024	\$280,470.00
June 30, 2024	\$490,823.00
September 30, 2024	\$490,823.00
December 31, 2024	\$490,823.00
March 31, 2025	\$490,823.00
June 30, 2025	\$701,176.00
September 30, 2025 and the last day of each fiscal quarter thereafter	\$701,176.00
Maturity Date	The remaining unpaid principal and accrued interest thereon

The Term Loan shall be evidenced by one or more secured promissory notes (collectively, the "Amended and Restated Term Note"). The Term Loan may consist of Base Rate Loans or LIBORTerm SOFR Rate Loans, or a combination thereof, as Borrower may request; and in the event that Borrowers desire Borrower desires to obtain or extend any portion of the Term Loan as a LIBORTerm SOFR Rate Loan or to convert any portion of the Term Loan from a Base Rate Loan to a LIBORTerm SOFR Rate Loan, Borrower shall comply with the notification requirements set forth in Sections 2.2(b) and/or (e) and the provisions of Sections 2.2(b) through (h) shall apply. The Third Amendment Term Loan shall be initially made as a LIBOR Rate Loan, and combined with the existing LIBOR Rate Loan for each of the Closing Date Term Loan and the First Amendment Term Loan, with an Interest Period ending as of December 31, 2019.

2.4. Swing Loans.

- Lenders and Agent for administrative convenience, Agent, Lenders holding Revolving Commitments and Swing Loan Lender agree that in order to facilitate the administration of this Agreement, Swing Loan Lender may, at its election and option made in its sole discretion cancelable at any time for any reason whatsoever, make swing loan advances ("Swing Loans") available to Borrower as provided for in this Section 2.4 at any time or from time to time after the Closing Date to, but not including, the Maturity Date, in an aggregate principal amount up to but not in excess of the Maximum Swing Loan Advance Amount, provided that the outstanding aggregate principal amount of Swing Loans and the Revolving Loans at any one time outstanding shall not exceed the Maximum Revolving Loan Amount less the Maximum Undrawn Amount of all outstanding Letters of Credit. All Swing Loans shall be Base Rate Loans only. Borrower may borrow (at the option and election of Swing Loan Lender), repay and reborrow (at the option and election of Swing Loan Lender) Swing Loans and Swing Loan Lender may make Swing Loans as provided in this Section 2.4 during the period between Settlement Dates. All Swing Loans shall be evidenced by a secured promissory note (the "Swing Loan Note") substantially in the form attached hereto as Exhibit 2.4(a). Swing Loan Lender's agreement to make Swing Loans under this Agreement is cancelable at any time for any reason whatsoever and the making of Swing Loans by Swing Loan Lender from time to time shall not create any duty or obligation, or establish any course of conduct, pursuant to which Swing Loan Lender shall thereafter be obligated to make Swing Loans in the future.
- (b) Upon either (i) any request by Borrower for a Revolving Loan made pursuant to Section 2.2(a) hereof or (ii) the occurrence of any deemed request by Borrower for a Revolving Loan pursuant to the provisions of the last sentence of Section 2.2(a) hereof, Swing Loan Lender may elect, in its sole discretion, to have such request or deemed request treated as a request for a Swing Loan, and may advance same day funds to Borrower as a Swing Loan; provided that notwithstanding anything to the contrary provided for herein, Swing Loan Lender may not make Swing Loans if Swing Loan Lender has been notified by Agent or by Required Lenders that one or more of the applicable conditions set forth in Section 3.2 of this Agreement have not been satisfied or the Revolving Commitments have been terminated for any reason.
- (c) Upon the making of a Swing Loan (whether before or after the occurrence of a Default or an Event of Default and regardless of whether a Settlement has been requested with respect to such Swing Loan), each Lender holding a Revolving Commitment shall be deemed, without further action by any party hereto, to have unconditionally and irrevocably purchased from Swing Loan Lender, without recourse or warranty, an undivided interest and participation in such Swing Loan in proportion to its Pro Rata Share thereof. Swing Loan Lender or Agent may, at any time, require the Lenders holding Revolving Commitments to fund such

participations by means of a Settlement as provided for in <u>Section 2.6(d)</u> below. From and after the date, if any, on which any Lender holding a Revolving Commitment is required to fund, and funds, its participation in any Swing Loans purchased hereunder, Agent shall promptly distribute to such Lender its Pro Rata Share of all payments of principal and interest and all proceeds of Collateral received by Agent in respect of such Swing Loan; provided that no Lender holding a Revolving Commitment shall be obligated in any event to make Revolving Loans in an amount in excess of the amount of its Revolving Commitment minus its Participation Commitment (taking into account any reallocations under <u>Section 2.9</u>) with respect to the outstanding Swing Loans and the Maximum Undrawn Amount of all outstanding Letters of Credit.

2.5. Disbursement of Loan Proceeds. All Loans shall be disbursed from whichever office or other place Agent may designate from time to time and, together with any and all other Obligations of Borrower to Agent, Lenders, Swing Loan Lender and Issuing Bank, shall be charged to the Loan Account on Agent's books. The proceeds of each Loan requested by Borrower or deemed to have been requested by Borrower under Section 2.2(a), 2.6(b) or 2.11(d) hereof shall, (a) with respect to requested Revolving Loans, to the extent Lenders make such Revolving Loans in accordance with Section 2.2(a), 2.6(b) or 2.11(d) hereof and with respect to Swing Loans made upon any request by Borrower for a Revolving Loan to the extent Swing Loan Lender makes such Swing Loan in accordance with Section 2.4(b) hereof, be made available to Borrower on the day so requested by way of credit to the Funding Account in immediately available federal funds or other immediately available funds or, (b) with respect to Revolving Loans deemed to have been requested by Borrower or Swing Loans made upon any deemed request for a Revolving Loan by Borrower, be disbursed to Agent to be applied to the outstanding Obligations giving rise to such deemed request. Prior to the Maturity Date, Borrower may use the Revolving Loans and Swing Loans by borrowing, prepaying and reborrowing, all in accordance with the terms and conditions hereof. Agent is authorized to make the Loans, and Issuing Bank is authorized to issue the Letters of Credit, under this Agreement based upon telephonic or other instructions received from anyone purporting to be an Authorized Person or, without instructions, if pursuant to Section 2.2(a), 2.6(b) or 2.11(d). Borrower agrees to establish and maintain the Funding Account with PNC for the purpose of receiving the proceeds of the Loans requested by Borrower and made by Agent, Swing Loan Lender or the Lenders hereunder shall be made to the Funding Account.

2.6. Making and Settlement of Loans.

- (a) Each borrowing of Revolving Loans shall be advanced according to the Pro Rata Shares of Lenders holding the Revolving Commitments (subject to any contrary terms of <u>Section 2.9</u>). The Term Loan shall be advanced according to the Pro Rata Shares of Lenders holding the Term Loan Commitments. Each borrowing of Swing Loans shall be advanced by Swing Loan Lender alone.
- (b) Promptly after receipt by Agent of a request or a deemed request for a Revolving Loan pursuant to Section 2.2(a) and, with respect to Revolving Loans, to the extent Agent elects not to provide a Swing Loan or the making of a Swing Loan would result in the aggregate amount of all outstanding Swing Loans exceeding the maximum amount permitted in Section 2.4(a), Agent shall notify Lenders holding the Revolving Commitments of its receipt of such request specifying the information provided by Borrower and the apportionment among Lenders of the requested Revolving Loan as determined by Agent in accordance with the terms hereof. Each Lender shall remit the principal amount of each Revolving Loan to Agent such that Agent is able to, and Agent shall, to the extent the applicable Lenders have made funds available to it for such purpose and subject to Section 3.2, fund such Revolving Loan to Borrower in

Dollars and immediately available funds to the Payment Office prior to the close of business, on the applicable borrowing date; <u>provided</u> that if any applicable Lender fails to remit such funds to Agent in a timely manner, Agent may elect in its sole discretion to fund with its own funds the Revolving Loan of such Lender on such borrowing date, and such Lender shall be subject to the repayment obligation in <u>Section 2.6(c)</u> hereof.

- Commitment that such Lender will not make the amount which would constitute its applicable Pro Rata Share of the requested Revolving Loan available to Agent, Agent may (but shall not be obligated to) assume that such Lender has made such amount available to Agent on such date in accordance with Section 2.6(b) and may, in reliance upon such assumption, make available to Borrower a corresponding amount. In such event, if a Lender has not in fact made its applicable Pro Rata Share of the requested Revolving Loan available to Agent, then the applicable Lender and Borrower severally agree to pay to Agent on demand such corresponding amount with interest thereon, for each day from and including the date such amount is made available to Borrower through but excluding the date of payment to Agent, at (i) in the case of a payment to be made by such Lender, the greater of (A) (x) the daily average Federal Funds Effective Rate (computed on the basis of a year of 360 days) during such period as quoted by Agent, times (y) such amount or (B) a rate determined by Agent in accordance with banking industry rules on interbank compensation, and (ii) in the case of a payment to be made by Borrower, the Revolving Interest Rate for Base Rate Loans. If such Lender pays its Pro Rata Share of the applicable Revolving Loan to Agent, then the amount so paid shall constitute such Lender's Revolving Loan. Any payment by Borrower shall be without prejudice to any claim Borrower may have against a Lender holding a Revolving Commitment that shall have failed to make such payment to Agent. A certificate of Agent submitted to any Lender or Borrower with respect to any amounts owing under this paragraph (c) shall be conclusive, in the absence of manifest error.
- (d) Agent, on behalf of Swing Loan Lender and itself, shall demand settlement (a "Settlement") of all or any Swing Loans and Protective Advances with Lenders holding the Revolving Commitments on at least a weekly basis, or on any more frequent date that Agent elects or that Swing Loan Lender at its option exercisable for any reason whatsoever may request, by notifying Lenders holding the Revolving Commitments of such requested Settlement by facsimile, telephonic or electronic transmission no later than 3:00 p.m. on the date of such requested Settlement (the "Settlement Date"). Subject to any contrary provisions of Section 2.9, each Lender holding a Revolving Commitment shall transfer the amount of such Lender's Pro Rata Share of the outstanding principal amount (plus interest accrued thereon to the extent requested by Agent) of the applicable Swing Loan and/or Protective Advance with respect to which Settlement is requested by Agent, to such account of Agent as Agent may designate not later than 5:00 p.m. on such Settlement Date if requested by Agent by 3:00 p.m., otherwise not later than 5:00 p.m. on the next Business Day. Settlements may occur at any time notwithstanding that the conditions precedent to making Revolving Loans set forth in Section 3.2 have not been satisfied or the Revolving Commitments shall have otherwise been terminated at such time. All amounts so transferred to Agent shall be applied against the amount of outstanding Swing Loans or Protective Advances, as applicable, and, when so applied shall constitute Revolving Loans of such Lenders accruing interest as Base Rate Loans. If any such amount is not transferred to Agent by any Lender holding a Revolving Commitment on such Settlement Date, Agent shall be entitled to recover such amount on demand from such Lender together with interest thereon as specified in Section 2.6(c).
- (e) If any Lender or Participant (a "Benefited Lender") shall at any time receive any payment of all or part of its Loans, or interest thereon, or receive any Collateral in respect thereof (whether voluntarily or involuntarily or by set-off) in a greater proportion than

any such payment to and Collateral received by any other Lender, if any, in respect of such other Lender's Loans, or interest thereon, and such greater proportionate payment or receipt of Collateral is not expressly permitted hereunder, such Benefited Lender shall purchase for cash from the other Lenders a participation in such portion of each such other Lender's Loans, or shall provide such other Lender with the benefits of any such Collateral, or the proceeds thereof, as shall be necessary to cause such Benefited Lender to share the excess payment or benefits of such Collateral or proceeds ratably with each of the other Lenders; <u>provided, however</u>, that if all or any portion of such excess payment or benefits is thereafter recovered from such Benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that each Lender so purchasing a portion of another Lender's Loans may exercise all rights of payment (including rights of set-off) with respect to such portion as fully as if such Lender were the direct holder of such portion, and the obligations owing to each such purchasing Lender in respect of such participation and such purchasing Lender in respect of such participation secured by the Collateral, and the obligations owing to each such purchasing Lender in respect of such participation and such purchased portion of any other Lender's Loans shall be part of the Obligations secured by the Collateral.

2.7. <u>Maximum Loans</u>. The aggregate balance of Revolving Loans plus Swing Loans outstanding at any time shall not exceed the Maximum Revolving Loan Amount less the aggregate Maximum Undrawn Amount of all issued and outstanding Letters of Credit.

2.8. Manner and Repayment of Loans.

- (a) The Revolving Loans and Swing Loans shall be due and payable in full on the Maturity Date subject to earlier prepayment as herein provided. The Term Loan shall be due and payable as provided in Section 2.3 hereof, subject to mandatory prepayments as herein provided. Notwithstanding the foregoing, all Loans shall be subject to earlier repayment upon (x) acceleration during the existence of an Event of Default under this Agreement or (y) termination of this Agreement. Each payment (including each prepayment) by Borrower on account of the principal of and interest on the Loans (other than the Term Loan) shall, unless otherwise provided herein during an Application Event, be applied, first to the outstanding Swing Loans and next, in accordance with the Pro Rata Shares of the Lenders, to the outstanding Revolving Loans (subject to any contrary provisions of Section 2.9). Each payment (including each prepayment) by Borrower on account of the principal of and interest on the Term Loan shall be applied to the Term Loan in accordance with the Pro Rata Shares of the Lenders.
- (b) All payments of principal, interest and other amounts payable hereunder, or under any of the other Loan Documents shall be made without deduction, set-off or counterclaim and shall be made to Agent at the Payment Office not later than 1:00 p.m. on the due date therefor in Dollars in federal funds or other funds immediately available to Agent. Anything to the contrary contained herein notwithstanding, any payment item shall be deemed received by Agent only if it is received at the Payment Office on a Business Day on or before 1:00 p.m. If any payment item is received at the Payment Office on a non-Business Day or after 1:00 p.m. on a Business Day (unless Agent, in its sole discretion, elects to credit it on the date received), it shall be deemed to have been received by Agent as of the opening of business on the immediately following Business Day. Agent shall have the right to effectuate payment of any and all Obligations due and owing hereunder by charging the Loan Account or by making Loans as provided in Section 2.2 hereof. The receipt of any payment item by Agent shall not be required to be considered a payment on account unless such payment item is a wire transfer of immediately available federal funds made to the Payment Office or unless and until such payment item is honored when presented for payment. Should any payment item not be honored

when presented for payment, then Borrower shall be deemed not to have made such payment and interest shall be calculated accordingly.

2.9. **Defaulting Lenders**.

- (a) Notwithstanding anything to the contrary contained herein, in the event any Lender is a Defaulting Lender, all rights and obligations hereunder of such Defaulting Lender and of the other parties hereto shall be modified to the extent of the express provisions of this Section 2.9 so long as such Lender is a Defaulting Lender.
- (b) (i) except as otherwise expressly provided for in this Section 2.9, Revolving Loans shall be made pro rata from Lenders holding Revolving Commitments which are not Defaulting Lenders based on their respective Pro Rata Shares, and no Pro Rata Share of any Lender or any Pro Rata Share of any Revolving Loans required to be advanced by any Lender shall be increased as a result of any Lender being a Defaulting Lender. Amounts received in respect of principal of any type of Revolving Loans shall be applied to reduce such type of Revolving Loans of each Lender (other than any Defaulting Lender) holding a Revolving Commitment in accordance with their Pro Rata Shares; provided, that, Agent shall not be obligated to transfer to a Defaulting Lender any payments received by Agent for Defaulting Lender's benefit, nor shall a Defaulting Lender be entitled to the sharing of any payments hereunder (including any principal, interest or fees). Amounts payable to a Defaulting Lender shall instead be paid to or retained by Agent. Agent may hold and, in its discretion, re-lend to Borrower the amount of such payments received or retained by it for the account of such Defaulting Lender.
 - (ii) Unused Line Fees shall cease to accrue in favor of such Defaulting Lender.
- (iii) if any Swing Loans are outstanding or any Letters of Credit (or drawings under any Letter of Credit for which Issuing Bank has not been reimbursed) are outstanding or exist at the time any such Lender holding a Revolving Commitment becomes a Defaulting Lender, then:
- (A) Defaulting Lender's Participation Commitment in the outstanding Swing Loans and of the Maximum Undrawn Amount of all outstanding Letters of Credit shall be reallocated among Non-Defaulting Lenders holding Revolving Commitments in proportion to the respective Pro Rata Shares of such Non-Defaulting Lenders to the extent (but only to the extent) that (x) such reallocation does not cause the aggregate sum of outstanding Revolving Loans made by any such Non-Defaulting Lender holding a Revolving Commitment plus such Lender's reallocated Participation Commitment in the outstanding Swing Loans plus such Lender's reallocated Participation Commitment in the aggregate Maximum Undrawn Amount of all outstanding Letters of Credit to exceed the Revolving Commitment of any such Non-Defaulting Lender, and (y) no Default or Event of Default has occurred and is continuing at such time;
- (B) if the reallocation described in clause (A) above cannot, or can only partially, be effected, Borrower shall within one Business Day following notice by Agent (1) first, prepay any outstanding Swing Loans that cannot be reallocated, and (2) second, deliver Letter of Credit Collateralization corresponding to such Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit (after giving effect to any partial reallocation pursuant to clause (A) above) for so long as such Obligations are outstanding:

- (C) if Borrower delivers Letter of Credit Collateralization for any portion of such Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit pursuant to clause (B) above, Borrower shall not be required to pay any Letter of Credit Fees with respect to such Defaulting Lender's Pro Rata Share of the Maximum Undrawn Amount of all Letters of Credit during the period such Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit are subject to such Letter of Credit Collateralization;
- (D) if Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit is reallocated pursuant to clause (A) above, then the Letter of Credit Fees payable to Lenders holding Revolving shall be adjusted and reallocated to Non-Defaulting Lenders holding Revolving Commitments in accordance with such reallocation; and
- (E) if all or any portion of such Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit is neither reallocated nor subject to Letter of Credit Collateralization pursuant to clauses (A) or (B) above, then, without prejudice to any rights or remedies of Issuing Bank or any other Lender hereunder, all Letter of Credit Fees payable with respect to such Defaulting Lender's Pro Rata Share of the Maximum Undrawn Amount of all Letters of Credit shall be payable to the Issuing Bank (and not to such Defaulting Lender) until (and then only to the extent that) such Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit is reallocated and/or subject to Letter of Credit Collateralization; and
- (iv) so long as any Lender holding a Revolving Commitment is a Defaulting Lender, Swing Loan Lender shall not be required to fund any Swing Loans and Issuing Bank shall not be required to issue, amend or increase any Letter of Credit, unless Swing Loan Lender or Issuing Bank, as applicable is satisfied that the related exposure and Defaulting Lender's Participation Commitment in the Maximum Undrawn Amount of all Letters of Credit and all Swing Loans (after giving effect to any such issuance, amendment, increase or funding) will be fully allocated to Non-Defaulting Lenders holding Revolving Commitments and/or Letter of Credit Collateralization delivered for such Letters of Credit will be provided by Borrower in accordance with clause (A) and (B) above, and participating interests in any newly made Swing Loan or any newly issued or increased Letter of Credit shall be allocated among Non-Defaulting Lenders in a manner consistent with Section 2.9(b)(iii)(A) above (and such Defaulting Lender shall not participate therein).
- (c) A Defaulting Lender shall be deemed not to be a "Lender" for purposes of voting on any matters (including the granting of any consents or waivers) with respect to this Agreement and the other Loan Documents solely to the extent such matters (i) relate to Revolving Loans and Revolving Commitments and (ii) except for the matters expressly set forth in this Section 2.9, do not result in treatment of the Revolving Loans held by such Defaulting Lender in a manner materially and adversely different from the treatment of the Revolving Loans held by non-Defaulting Lenders.
- (d) Other than as expressly set forth in this <u>Section 2.9</u>, the rights and obligations of a Defaulting Lender (including the obligation to indemnify Agent) and the other parties hereto shall remain unchanged. Nothing in this <u>Section 2.9</u> shall be deemed to release any Defaulting Lender from its obligations under this Agreement and the other Loan Documents, shall alter such obligations, shall operate as a waiver of any default by such Defaulting Lender hereunder, or shall prejudice any rights which Borrower, Agent or any Lender may have against any Defaulting Lender as a result of any default by such Defaulting Lender hereunder.

- (e) In the event that Agent, Borrower, Issuing Bank and Swing Loan Lender agree in writing that a Defaulting Lender has adequately remedied all matters that caused such Lender to be a Defaulting Lender, then Agent will so notify the parties hereto, and the Participation Commitments of Lenders holding Revolving Commitments (including such cured Defaulting Lender) of the Swing Loans and Maximum Undrawn Amount of all outstanding Letters of Credit shall be reallocated to reflect the inclusion of such Lender's Revolving Commitment, and on such date such Lender shall purchase at par such of the Revolving Loans of the other Lenders as Agent shall determine may be necessary in order for such Lender to hold such Revolving Loans in accordance with its Pro Rata Share.
- (f) If Swing Loan Lender or Issuing Bank has a good faith belief that any Lender holding a Revolving Commitment has defaulted in fulfilling its obligations under one or more other agreements in which such Lender commits to extend credit, Swing Loan Lender shall not be required to fund any Swing Loans and Issuing Bank shall not be required to issue, amend or increase any Letter of Credit, unless Swing Loan Lender or Issuing Bank, as the case may be, shall have entered into arrangements with Borrower or such Lender, satisfactory to Swing Loan Lender or Issuing Bank, as the case may be, to defease any risk to it in respect of such Lender hereunder.
- (g) All Loans (other than Swing Loans and Protective Advances) shall be made by the Lenders contemporaneously and in accordance with their Pro Rata Shares. It is understood that (i) no Lender shall be responsible for any failure by any other Lender to perform its obligation to make any Loan hereunder, nor shall any Commitment of any Lender be increased or decreased as a result of any failure by any other Lender to perform its obligations hereunder, and (ii) no failure by any Lender to perform its obligations hereunder shall excuse any other Lender from its obligations hereunder.
- 2.10. Statement of Account. Agent shall maintain, in accordance with its customary procedures, a loan account (the "Loan Account") in the name of Borrower in which shall be recorded the date and amount of each Loan made by Agent, Lenders and Swing Loan Lender and the date and amount of each payment in respect thereof; provided, however, the failure by Agent to record the date and amount of any Loan shall not adversely affect Agent, any Lender or Swing Loan Lender. Each month, Agent shall send to Borrower a statement showing the accounting for the Loans made, payments made or credited in respect thereof, and other transactions between Agent, Lenders, Swing Loan Lender and Borrower during such month. The monthly statements shall be deemed correct and binding upon Borrower in the absence of manifest error and shall constitute an account stated between Agent, Lenders, Swing Loan Lender and Borrower unless Agent receives a written statement of Borrower's specific exceptions thereto within thirty (30) days after such statement is received by Borrower. The records of Agent with respect to the Loan Account shall be conclusive evidence absent manifest error of the amounts of Loans and other charges thereto and of payments applicable thereto.

2.11. Letters of Credit.

(a) Generally.

(i) Subject to the terms and conditions hereof, Issuing Bank shall issue or cause the issuance of standby letters of credit denominated in Dollars ("Letters of Credit") for the account of Borrower except to the extent that the issuance thereof would then cause the sum of (A) the outstanding Revolving Loans plus (B) the outstanding Swing Loans, plus (C) the Maximum Undrawn Amount of all outstanding Letters of Credit, plus (D) the Maximum Undrawn Amount of the Letter of Credit to be issued to exceed the Maximum Revolving Loan Amount); provided, however, that the Maximum Undrawn Amount of all

outstanding Letters of Credit shall not exceed in the aggregate at any time the Letter of Credit Sublimit. All disbursements or payments related to Letters of Credit shall be deemed to be Revolving Loans made as Base Rate Loans and shall bear interest at the Revolving Interest Rate for Base Rate Loans. Letters of Credit that have not been drawn upon shall not bear interest (but fees shall accrue in respect of outstanding Letters of Credit as provided in Section 2.14(b) hereof).

(ii) Notwithstanding any provision of this Agreement to the contrary, Issuing Bank shall not be under any obligation to issue any Letter of Credit if (A) any order, judgment or decree of any Governmental AuthorityBody or arbitrator shall by its terms purport to enjoin or restrain Issuing Bank from issuing any Letter of Credit, or any law applicable to Issuing Bank or any request or directive (whether or not having the force of law) from any Governmental AuthorityBody with jurisdiction over Issuing Bank shall prohibit, or request that Issuing Bank refrain from, the issuance of letters of credit generally or the Letter of Credit in particular or shall impose upon Issuing Bank with respect to the Letter of Credit any restriction, reserve or capital requirement (for which Issuing Bank is not otherwise compensated hereunder) not in effect on the date of this Agreement, or shall impose upon Issuing Bank any unreimbursed loss, cost or expense which was not applicable on the date of this Agreement, and which Issuing Bank in good faith deems material to it, (B) the issuance of the Letter of Credit would violate one or more policies of Issuing Bank applicable to letters of credit generally or (C) the proceeds of such Letter of Credit would be made available to any Person, (1) to fund any activity or business of or with any Sanctioned Person or in any country or territory that, at the time of such issuance, is the subject of any Sanctions or (2) in any manner that would immediately (after the expiration of any notice, cure or similar period) result in a violation of any Sanctions by any person party to or bound by the terms of this Agreement.

(b) Issuance of Letters of Credit.

- (i) Borrower may request Issuing Bank to issue or cause the issuance of a Letter of Credit by delivering to Issuing Bank, with a copy to Agent at the Payment Office, prior to 1:00 p.m., at least five (5) Business Days prior to the proposed date of issuance, such Issuing Bank's form of Letter of Credit Application (the "Letter of Credit Application") completed to the satisfaction of Agent and Issuing Bank; and, such other certificates, documents and other papers and information as Agent or Issuing Bank may reasonably request. Issuing Bank shall not issue any requested Letter of Credit if such Issuing Bank has received notice from Agent or any Lender that one or more of the applicable conditions set forth in Section 3.2 of this Agreement have not been satisfied or the commitments of Lenders to make Revolving Loans hereunder have been terminated for any reason.
- (ii) Each Letter of Credit shall, among other things, (A) provide for the payment of sight drafts, or other written demands for payment, and (B) have an expiry date not later than twelve (12) months after such Letter of Credit's date of issuance and in no event later than the Maturity Date. Each standby Letter of Credit shall be subject either to the Uniform Customs and Practice for Documentary Credits as most recently published by the International Chamber of Commerce at the time a Letter of Credit is issued or the International Standby Practices (International Chamber of Commerce Publication Number 590), or any subsequent revision thereof at the time a standby Letter of Credit is issued, as determined by Issuing Bank.
 - (iii) Agent shall use its reasonable efforts to notify Lenders of the request by Borrower for a Letter of Credit hereunder.
- (c) Requirements For Issuance of Letters of Credit. Borrower shall authorize and direct any Issuing Bank to name Borrower as the "Applicant" or "Account Party"

of each Letter of Credit. If Agent is not the Issuing Bank of any Letter of Credit, Borrower shall authorize and direct Issuing Bank to deliver to Agent all instruments, documents, and other writings and property received by Issuing Bank pursuant to the Letter of Credit and to accept and rely upon Agent's instructions and agreements with respect to all matters arising in connection with the Letter of Credit, the application therefor.

(d) Disbursements, Reimbursement.

- (i) Immediately upon the issuance of each Letter of Credit, each Lender holding a Revolving Commitment shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from Issuing Bank a participation in each Letter of Credit and each drawing thereunder in an amount equal to such Lender's Pro Rata Share of the Maximum Undrawn Amount of such Letter of Credit (as in effect from time to time) and the amount of such drawing, respectively.
- (ii) In the event of any request for a drawing under a Letter of Credit by the beneficiary or transferee thereof, Issuing Bank will promptly notify Agent and Borrower. Regardless of whether Borrower shall have received such notice, Borrower shall reimburse (such obligation to reimburse Issuing Bank shall sometimes be referred to as a "Reimbursement Obligation") Issuing Bank prior to 12:00 Noon, on each date that an amount is paid by Issuing Bank under any Letter of Credit (each such date, a "Drawing Date") in an amount equal to the amount so paid by Issuing Bank. In the event Borrower fails to reimburse Issuing Bank for the full amount of any drawing under any Letter of Credit by 12:00 Noon, on the Drawing Date, Issuing Bank will promptly notify Agent and each Lender holding a Revolving Commitment thereof, and Borrower shall be automatically deemed to have requested that a Revolving Loan maintained as a Base Rate Loan be made by Lenders to be disbursed on the Drawing Date under such Letter of Credit, and Lenders holding the Revolving Commitments shall be unconditionally obligated to fund such Revolving Loan (all whether or not the conditions specified in Section 3.2 are then satisfied or the commitments of Lenders to make Revolving Loans hereunder have been terminated for any reason) as provided for in Section 2.11(d)(iii) immediately below. Any notice given by Issuing Bank pursuant to this Section 2.11(d)(iii) may be oral if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.
- (iii) Each Lender holding a Revolving Commitment shall upon any notice pursuant to Section 2.11(d)(ii) make available to Issuing Bank through Agent at the Payment Office an amount in immediately available funds equal to its Pro Rata Share (subject to any contrary provisions of Section 2.9) of the amount of the drawing, whereupon the participating Lenders shall (subject to Section 2.11(d) (iv)) each be deemed to have made a Revolving Loan maintained as a Base Rate Loan to Borrower in that amount. If any Lender holding a Revolving Commitment so notified fails to make available to Agent, for the benefit of Issuing Bank, the amount of such Lender's Pro Rata Share of such amount by 2:00 p.m. on the Drawing Date, then interest shall accrue on such Lender's obligation to make such payment, from the Drawing Date to the date on which such Lender makes such payment (A) at a rate per annum equal to the Federal Funds Effective Rate during the first three (3) days following the Drawing Date and (B) at a rate per annum equal to the rate applicable to Revolving Loans maintained as a Base Rate Loan on and after the fourth day following the Drawing Date. Agent and Issuing Bank will promptly give notice of the occurrence of the Drawing Date, but failure of Agent or Issuing Bank to give any such notice on the Drawing Date or in sufficient time to enable any Lender holding a Revolving Commitment to effect such payment on such date shall not relieve such Lender from its obligations under this Section 2.11(d)(iii), provided that such Lender shall not be obligated to pay interest as provided in Section 2.11(d)(iii)(A) and (B) until and commencing from the date of receipt of notice from Agent or Issuing Bank of a drawing.

- (iv) With respect to any unreimbursed drawing that is not converted into a Revolving Loan maintained as a Base Rate Loan to Borrower in whole or in part as contemplated by Section 2.11(d)(ii), because of Borrower's failure to satisfy the conditions set forth in Section 3.2 hereof (other than any notice requirements) or for any other reason, Borrower shall be deemed to have incurred from Agent a borrowing (each a "Letter of Credit Borrowing") in the amount of such drawing. Such Letter of Credit Borrowing shall be due and payable on demand (together with interest) and shall bear interest at the rate per annum applicable to a Revolving Loan maintained as a Base Rate Loan. Each applicable Lender's payment to Agent pursuant to Section 2.11(d)(iii) shall be deemed to be a payment in respect of its participation in such Letter of Credit Borrowing and shall constitute a "Participation Advance" from such Lender in satisfaction of its Participation Commitment in respect of the applicable Letter of Credit under this Section 2.11(d).
- (v) Each applicable Lender's Participation Commitment in respect of the Letters of Credit shall continue until the last to occur of any of the following events: (A) Issuing Bank ceases to be obligated to issue or cause to be issued Letters of Credit hereunder; (B) no Letter of Credit issued or created hereunder remains outstanding and uncancelled; and (C) all Persons (other than Borrower) have been fully reimbursed for all payments made under or relating to Letters of Credit.

(e) Repayment of Participation Advances.

- (i) Upon (and only upon) receipt by Agent for the account of Issuing Bank of immediately available funds from Borrower (A) in reimbursement of any payment made by Issuing Bank or Agent under the Letter of Credit with respect to which any Lender has made a Participation Advance to Agent, or (B) in payment of interest on such a payment made by Issuing Bank or Agent under such a Letter of Credit, Agent will pay to each Lender holding a Revolving Commitment, in the same funds as those received by Agent, the amount of such Lender's Pro Rata Share of such funds, except Agent shall retain the amount of the Pro Rata Share of such funds of any Lender holding a Revolving Commitment that did not make a Participation Advance in respect of such payment by Agent (and, to the extent that any of the other Lenders holding the Revolving Commitment have funded any portion such Defaulting Lender's Participation Advance in accordance with the provisions of Section 2.9, Agent will pay over to such Non-Defaulting Lenders a pro rata portion of the funds so withheld from such Defaulting Lender).
- (ii) If Issuing Bank or Agent is required at any time to return to Borrower, or to a trustee, receiver, liquidator, custodian, or any official in any insolvency proceeding, any portion of the payments made by Borrower to Issuing Bank or Agent pursuant to Section 2.11(e) in reimbursement of a payment made under the Letter of Credit or interest or fee thereon, each applicable Lender shall, on demand of Agent, forthwith return to Issuing Bank or Agent the amount of its Pro Rata Share of any amounts so returned by Issuing Bank or Agent plus interest at the Federal Funds Effective Rate.
- (f) **Documentation**. Borrower agrees to be bound by the terms of the Letter of Credit Application and by Issuing Bank's interpretations of any Letter of Credit issued on behalf of Borrower and by Issuing Bank's written regulations and customary practices relating to letters of credit, though Issuing Bank's interpretations may be different from Borrower's own. In the event of a conflict between the Letter of Credit Application and this Agreement, this Agreement shall govern. It is understood and agreed that, except in the case of gross negligence, bad faith or willful misconduct (as determined by a court of competent jurisdiction in a final non-appealable judgment), Issuing Bank shall not be liable for any error, negligence and/or mistakes,

whether of omission or commission, in following Borrower's instructions or those contained in the Letters of Credit or any modifications, amendments or supplements thereto.

- (g) **Determination to Honor Drawing Request.** In determining whether to honor any request for drawing under any Letter of Credit by the beneficiary thereof, Issuing Bank shall be responsible only to determine that the documents and certificates required to be delivered under such Letter of Credit have been delivered and that they comply on their face with the requirements of such Letter of Credit and that any other drawing condition appearing on the face of such Letter of Credit has been satisfied in the manner so set forth.
- (h) Nature of Participation and Reimbursement Obligations. The obligation of each Lender holding a Revolving Commitment in accordance with this Agreement to make the Revolving Loans or Participation Advances as a result of a drawing under a Letter of Credit, and the obligations of Borrower to reimburse Issuing Bank upon a draw under a Letter of Credit, shall be absolute, unconditional and irrevocable, and shall be performed strictly in accordance with the terms of this Section 2.11(h) under all circumstances, including the following circumstances:
- (i) any set-off, counterclaim, recoupment, defense or other right which such Lender or Borrower, as the case may be, may have against Issuing Bank, Agent, Borrower or Lender, as the case may be, or any other Person for any reason whatsoever;
- (ii) the failure of Borrower or any other Person to comply, in connection with a Letter of Credit Borrowing, with the conditions set forth in this Agreement for the making of a Revolving Loan, it being acknowledged that such conditions are not required for the making of a Letter of Credit Borrowing and the obligation of Lenders to make Participation Loans under Section 2.11(d);
 - (iii) any lack of validity or enforceability of any Letter of Credit;
- (iv) any claim of breach of warranty that might be made by Borrower, Agent, Issuing Bank or any Lender against the beneficiary of a Letter of Credit, or the existence of any claim, set-off, recoupment, counterclaim, cross-claim, defense or other right which Borrower, Agent, Issuing Bank or any Lender may have at any time against a beneficiary, any successor beneficiary or any transferee of any Letter of Credit or assignee of the proceeds thereof (or any Persons for whom any such transferee or assignee may be acting), Issuing Bank, Agent or any Lender or any other Person, whether in connection with this Agreement, the transactions contemplated herein or any unrelated transaction (including any underlying transaction between Borrower or any Subsidiaries or Affiliates of Borrower and the beneficiary for which any Letter of Credit was procured);
- (v) the lack of power or authority of any signer of (or any defect in or forgery of any signature or endorsement on) or the form of or lack of validity, sufficiency, accuracy, enforceability or genuineness of any draft, demand, instrument, certificate or other document presented under or in connection with any Letter of Credit, or any fraud or alleged fraud in connection with any Letter of Credit, or the transport of any property or provision of services relating to a Letter of Credit, in each case even if Issuing Bank or any of Issuing Bank's Affiliates has been notified thereof;
- (vi) payment by Issuing Bank under any Letter of Credit against presentation of a demand, draft or certificate or other document which is forged or does not fully comply with the terms of such Letter of Credit (provided that the foregoing shall not excuse Issuing Bank from any obligation under the terms of any applicable Letter of Credit to require

the presentation of documents that on their face appear to satisfy any applicable requirements for drawing under such Letter of Credit prior to honoring or paying any such draw);

- (vii) the solvency of, or any acts or omissions by, any beneficiary of any Letter of Credit, or any other Person having a role in any transaction or obligation relating to a Letter of Credit, or the existence, nature, quality, quantity, condition, value or other characteristic of any property or services relating to a Letter of Credit;
- (viii) any failure by Issuing Bank or any of Issuing Bank's Affiliates to issue any Letter of Credit in the form requested by Borrower, unless Agent and Issuing Bank have each received written notice from Borrower of such failure within three (3) Business Days after Issuing Bank shall have furnished Agent and Borrower a copy of such Letter of Credit and such error is material and no drawing has been made thereon prior to receipt of such notice;
 - (ix) the occurrence of any Material Adverse Effect;
 - (x) any breach of this Agreement or any other Loan Document by any party thereto;
 - (xi) the occurrence or continuance of an insolvency proceeding with respect to any Loan Party;
 - (xii) the fact that a Default or an Event of Default shall have occurred and be continuing;
- (xiii) the fact that the Maturity Date shall have occurred or this Agreement or the obligations of Lenders or Agent to make Loans have been terminated; and
 - (xiv) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing.

(i) Liability for Acts and Omissions.

(i) As between Borrower and Issuing Bank, Swing Loan Lender, Agent and Lenders, Borrower assumes all risks of the acts and omissions of, or misuse of the Letters of Credit by, the respective beneficiaries of such Letters of Credit. In furtherance and not in limitation of the foregoing, Issuing Bank shall not be responsible for: (A) the form, validity, sufficiency, accuracy, genuineness or legal effect of any document submitted by any party in connection with the application for an issuance of any such Letter of Credit, even if it should in fact prove to be in any or all respects invalid, insufficient, inaccurate, fraudulent or forged (even if Issuing Bank or any of its Affiliates shall have been notified thereof); (B) the validity or sufficiency of any instrument transferring or assigning or purporting to transfer or assign any such Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason; (C) the failure of the beneficiary of any such Letter of Credit, or any other party to which such Letter of Credit may be transferred, to comply fully with any conditions required in order to draw upon such Letter of Credit or any other claim of Borrower against any beneficiary of such Letter of Credit, or any such transferee, or any dispute between or among Borrower and any beneficiary of any Letter of Credit or any such transferee; (D) errors, omissions, interruptions or delays in transmission or delivery of any messages, by mail, cable, facsimile, telex or otherwise, whether or not they be in cipher; (E) errors in interpretation of technical terms; (F) any loss or delay in the transmission or otherwise of any document required in order to make a drawing under any such Letter of Credit or of the proceeds thereof; (G) the misapplication by the beneficiary of any such Letter of Credit

of the proceeds of any drawing under such Letter of Credit; or (H) any consequences arising from causes beyond the control of Issuing Bank, including any governmental acts, and none of the above shall affect or impair, or prevent the vesting of, any of Issuing Bank's rights or powers hereunder. Nothing in the preceding sentence shall relieve Issuing Bank from liability for Issuing Bank's gross negligence, bad faith or willful misconduct (as determined by a court of competent jurisdiction in a final non-appealable judgment) in connection with actions or omissions described in such clauses (A) through (H) of such sentence. In no event shall Issuing Bank or Issuing Bank's Affiliates be liable to Borrower or any other or any other Loan Party for any indirect, consequential, incidental, punitive, exemplary or special damages or expenses (including without limitation attorneys' fees), or for any damages resulting from any change in the value of any property relating to a Letter of Credit.

- (ii) Without limiting the generality of the foregoing, Issuing Bank and each of its Affiliates: (A) may rely on any oral or other communication believed in good faith by Issuing Bank or such Affiliate to have been authorized or given by or on behalf of the applicant for a Letter of Credit; (B) may honor any presentation if the documents presented appear on their face substantially to comply with the terms and conditions of the relevant Letter of Credit; (C) may honor a previously dishonored presentation under a Letter of Credit, whether such dishonor was pursuant to a court order, to settle or compromise any claim of wrongful dishonor, or otherwise, and shall be entitled to reimbursement to the same extent as if such presentation had initially been honored, together with any interest paid by Issuing Bank or its Affiliates; (D) may honor any drawing that is payable upon presentation of a statement advising negotiation or payment, upon receipt of such statement (even if such statement indicates that a draft or other document is being delivered separately), and shall not be liable for any failure of any such draft or other document to arrive, or to conform in any way with the relevant Letter of Credit; (E) may pay any paying or negotiating bank claiming that it rightfully honored under the laws or practices of the place where such bank is located; and (F) may settle or adjust any claim or demand made on Issuing Bank or its Affiliate in any way related to any order issued at the applicant's request to an arriver, a letter of guarantee or of indemnity issued to a steamship agent or carrier or any document or instrument of like import (each an "Order") and honor any drawing in connection with any Letter of Credit that is the subject of such Order, notwithstanding that any drafts or other documents presented in connection with such Letter of Credit fail to conform in any way with such Letter of Credit.
- (iii) In furtherance and extension and not in limitation of the specific provisions set forth above, any action taken or omitted by Issuing Bank under or in connection with the Letters of Credit issued by it or any documents and certificates delivered thereunder, if taken or omitted in good faith and without gross negligence, bad faith or willful misconduct, (in each case, as determined by a court of competent jurisdiction in a final non-appealable judgment), shall not put Issuing Bank under any resulting liability to Borrower, any other Loan Party, Agent or any Lender.
- (j) If by reason of (x) any Change in Law, or (y) compliance by Issuing Bank or any other member of the Lender Group with any direction, request, or requirement (irrespective of whether having the force of law) of any Governmental Authority Body or monetary authority including, Regulation D of the Board of Governors as from time to time in effect (and any successor thereto):
- (i) any reserve, deposit, or similar requirement is or shall be imposed or modified in respect of any Letter of Credit issued or caused to be issued hereunder or hereby, or

(ii) there shall be imposed on Issuing Bank or any other member of the Lender Group any other condition regarding any Letter of Credit,

and the result of the foregoing is to increase, directly or indirectly, the cost (other than Taxes, which shall be governed by Section 16 of this Agreement) to Issuing Bank or any other member of the Lender Group of issuing, making, participating in, or maintaining any Letter of Credit or to reduce the amount receivable in respect thereof, then, and in any such case, Agent may, at any time within a reasonable period after the additional cost is incurred or the amount received is reduced, notify Borrower, and Borrower shall pay within 30 days after demand therefor, such amounts as Agent may specify to be necessary to compensate Issuing Bank or any other member of the Lender Group for such additional cost or reduced receipt, together with interest on such amount from the date of such demand until payment in full thereof at the rate then applicable to Base Rate Loans hereunder; provided, that (A) Borrower shall not be required to provide any compensation pursuant to this Section 2.11(j) for any such amounts incurred more than 180 days prior to the date on which the demand for payment of such amounts is first made to Borrower, and (B) if an event or circumstance giving rise to such amounts is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof. The determination by Agent of any amount due pursuant to this Section 2.11(j), as set forth in a certificate setting forth the calculation thereof in reasonable detail, shall, in the absence of manifest or demonstrable error, be final and conclusive and binding on all of the parties hereto

2.12. Application of Payments; Termination of Commitments; Prepayments.

(a) Apportionment and Application.

- (i) So long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all principal and interest payments received by Agent shall be apportioned ratably among the Lenders (according to the unpaid principal balance of the Obligations to which such payments relate held by each Lender) and all payments of fees and expenses received by Agent (other than fees or expenses that are for Agent's separate account or for the separate account of Issuing Bank) shall be apportioned ratably among the Lenders having a Pro Rata Share of the type of Commitment or Obligation to which a particular fee or expense relates. Subject to Section 2.12(a)(iv), Section 2.12(c)(ii), and Section 2.12(d), all payments to be made hereunder by Borrower shall be remitted to Agent and all such payments, and all proceeds of Collateral received by Agent, shall be applied, so long as no Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, to reduce the balance of the Revolving Loans outstanding and, thereafter, to Borrower (to be wired to the Funding Account) or such other Person entitled thereto under applicable law.
- (ii) At any time that an Application Event has occurred and is continuing and except as otherwise provided herein with respect to Defaulting Lenders, all Distributions shall be applied as follows:
- (A) <u>first</u>, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to Agent under the Loan Documents, until paid in full,
 - (B) second, to pay any fees or premiums then due to Agent under the Loan Documents until paid in full,
 - (C) third, to pay interest due in respect of all Protective Advances until paid in full,

- (D) fourth, to pay the principal of all Protective Advances until paid in full,
- (E) <u>fifth</u>, ratably, to pay any Lender Group Expenses (including cost or expense reimbursements) or indemnities then due to any of the Lenders or relating to the Loans under the Loan Documents, until paid in full,
- paid in full.
- (F) <u>sixth</u>, ratably, to pay any fees or premiums then due to any of the Lenders under the Loan Documents until

(K) eleventh, to pay any other Obligations owing to Lenders other than Obligations owed to Defaulting

- (G) seventh, to pay interest accrued in respect of the Swing Loans until paid in full,
- (H) eighth, to pay the principal of all Swing Loans until paid in full,
- (I) <u>ninth</u>, ratably, to pay interest accrued in respect of the remaining Loans until paid in full,
- (J) tenth, ratably (i) to pay the principal of all remaining Loans (other than the Term Loan) until paid in full, (ii) to Agent, to be held by Agent, for the benefit of Issuing Bank (and for the ratable benefit of each of the Lenders that have an obligation to pay to Agent, for the account of Issuing Bank, a share of each unreimbursed drawing under each Letter of Credit), as cash collateral in an amount up to 105% of the Maximum Undrawn Amount of all then outstanding Letters of Credit (to the extent permitted by applicable law, such cash collateral shall be applied to the reimbursement of any unreimbursed drawings under Letters of Credit as and when such disbursement occurs and, if a Letter of Credit expires undrawn, the cash collateral held by Agent in respect of such Letter of Credit shall, to the extent permitted by applicable law, be reapplied pursuant to this Section 2.12(a)(ii), beginning with tier (A) hereof), (iii) ratably, to the Bank Product Providers based upon amounts then certified by the applicable Bank Product Provider to Agent (in form and substance satisfactory to Agent) to be due and payable to such Bank Product Providers on account of Bank Product Obligations, and (iv) to pay the outstanding principal balance of the Term Loan (in the inverse order of the maturity of the installments due thereunder) until the Term Loan is paid in full,
- Lenders.
- (L) <u>twelfth</u>, ratably, to pay any Obligations owing to Defaulting Lenders, and
- (M) thirteenth, to Borrower (to be wired to the Funding Account) or such other Person entitled thereto under applicable law.
- (iii) Agent promptly shall distribute to each Lender, pursuant to the applicable wire instructions received from each Lender in writing, such funds as it may be entitled to receive, subject to a Settlement delay as provided in Section 2.6.
- (iv) In each instance, so long as no Application Event has occurred and is continuing, Section 2.12(a)(i) shall not apply to any payment made by Borrower to Agent and specified by Borrower to be for the payment of specific Obligations then due and payable (or prepayable) under any provision of this Agreement or any other Loan Document.

- (v) For purposes of <u>Section 2.12(a)(ii)</u>, "paid in full" of a type of Obligation means payment in cash or immediately available funds of all amounts owing on account of such type of Obligation, including interest accrued after the commencement of any Insolvency Proceeding, default interest, interest on interest, and expense reimbursements, irrespective of whether any of the foregoing would be or is allowed or disallowed in whole or in part in any Insolvency Proceeding.
- (vi) In the event of a direct conflict between the priority provisions of this <u>Section 2.12</u> and any other provision contained in this Agreement or any other Loan Document, it is the intention of the parties hereto that such provisions be read together and construed, to the fullest extent possible, to be in concert with each other. In the event of any actual, irreconcilable conflict that cannot be resolved as aforesaid, if the conflict relates to the provisions of <u>Section 2.9</u> and this <u>Section 2.12</u>, then the provisions of <u>Section 2.9</u> shall control and govern, and if otherwise, then the terms and provisions of this Section 2.12 shall control and govern.

(b) Termination of Commitments.

- (i) **Revolver Commitments**. The Revolver Commitments shall terminate on the Maturity Date or such earlier date as determined by Agent pursuant to <u>Section 9.1</u>.
 - (ii) Term Loan Commitments. The Term Loan Commitments shall terminate upon the making of the Term Loan.

(c) Optional Prepayments.

- (i) **Revolving Loans**. Borrower may prepay the principal of any Revolving Loan at any time in whole or in part, without premium or penalty.
- (ii) **Term Loan**. Borrower may, upon at least 5 Business Days prior written notice to Agent, prepay the principal of the Term Loan, in whole or in part. If prior to an Application Event any Lender holding the Term Loan waives in writing its right to any prepayment of the Term Loan described in the foregoing sentence, Agent will remit such amount when received, ratably, to all other Lenders holding the Term Loan until the Term Loan held by such Lenders have been paid in full in cash. Each prepayment made pursuant to this Section 2.12(c)(ii) shall be accompanied by the payment of accrued interest to the date of such payment on the amount prepaid. Each such prepayment shall be applied against the remaining installments of principal due on the Term Loan in the inverse order of maturity (for the avoidance of doubt, any amount that is due and payable on the Maturity Date shall constitute an installment); provided, however, that notwithstanding the foregoing if no Default or Event of Default exists at the time of such prepayment, Borrower may designate any such optional prepayment to be applied to future installments in direct order of maturity (to be applied first to the installment due on the date closest to the date of such prepayment); provided further that no more than 4-quarterly installments in direct order of maturity may be so designated in any 4-quarter period. Any such notice of prepayment will be irrevocable, provided, however, that notwithstanding the foregoing, Borrower may rescind one such notice during the term of this Agreement so long as: (A) the notice of rescission is in writing and signed by Borrower; (B) the notice of rescission is received by Agent at least 2 Business Days prior to the date set forth as the date of any such prepayment in such notice; (C) no Default or Event of Default exists at the time such notice of rescission is received by Agent; and (D) no other notice of rescission has at any time been delivered by Borrower.

(d) Mandatory Prepayments.

- (i) **Overadvance**. If, at any time the aggregate balance of Loans (other than the Term Loan) plus the Maximum Undrawn Amount of all issued and outstanding Letters of Credit exceeds the Maximum Revolver Amount, then Borrower shall promptly, but in any event, within 1 Business Day prepay the Obligations in accordance with <u>Section 2.12(e)(i)</u> in an amount equal to the amount of such excess.
- (ii) **Dispositions**. Within 3 Business Days of the date of receipt by Parent or any of its Subsidiaries of the Net Cash Proceeds of any voluntary or involuntary sale or disposition by Parent or any of its Subsidiaries of assets (including casualty losses or condemnations but excluding sales or dispositions which qualify as Permitted Dispositions under clauses (a), (b), (c), (d), (e), (f), (i), (j), (k), (l), (m), or (n) of the definition of Permitted Dispositions), Borrower shall prepay the outstanding Obligations in accordance with Section 2.12(e)(ii) in an amount equal to 100% of such Net Cash Proceeds (including condemnation awards and payments in lieu thereof) received by such Person in connection with such sales or dispositions; provided that, so long as (A) no Default or Event of Default shall have occurred and is continuing or would result therefrom, (B) Borrower shall have given Agent prior written notice of Borrower's intention to apply such monies to the costs of replacement of the properties or assets that are the subject of such sale or disposition or the cost of purchase or construction of other assets useful in the business of Parent or its Subsidiaries, (C) the monies are held in a Deposit Account in which Agent has a perfected first-priority security interest, and (D) Parent or its Subsidiaries, as applicable, complete such replacement, purchase, or construction within 180 days after the initial receipt of such monies, then the Loan Party whose assets were the subject of such disposition (or within such 180 day period have entered into a binding contractual arrangement to complete such replacement, purchase, or construction, so long as such contractual arrangement is reasonably satisfactory to Agent and such replacement, purchase, or construction is completed within 270 days after the initial receipt of such proceeds) shall have the option to apply such monies to the costs of replacement of the assets that are the subject of such sale or disposition or the cost of purchase or construction of other assets useful in the business of parent Parent or its Subsidiaries (or binding contractual obligations) unless and to the extent that such applicable period shall have expired without such replacement, purchase, or construction being made or completed, in which case, any amounts remaining in the Deposit Account referred to in clause (C) above shall be paid to Agent and applied in accordance with Section 2.12(e)(ii). Nothing contained in this Section 2.12(d)(ii) shall permit Parent or any of its Subsidiaries to sell or otherwise dispose of any assets other than in accordance with Section 6.4.

(iii) [Reserved].

(iv) **Indebtedness**. Within 1 Business Day of the date of incurrence by Parent or any of its Subsidiaries of any Indebtedness (other than Permitted Indebtedness), Borrower shall prepay the outstanding principal amount of the Obligations in accordance with Section 2.12(e)(ii) in an amount equal to 100% of the Net Cash Proceeds received by such Person in connection with such incurrence. The provisions of this Section 2.12(e)(iv) shall not be deemed to be implied consent to any such incurrence otherwise prohibited by the terms of this Agreement.

(v) [Reserved].

(vi) **Excess Cash Flow**. Within 10 days of delivery to Agent of audited annual financial statements pursuant to <u>Section 5.1</u>, commencing with the delivery to Agent of the financial statements for Parent's fiscal year ended December 31, 2017 but excluding the delivery to the Agent of the financial statements for Parent's fiscal years ended December 31,

2018 and December 31, 2021 or, if such financial statements are not delivered to Agent on the date such statements are required to be delivered pursuant to Section 5.1, within 10 days after the date such statements were required to be delivered to Agent pursuant to Section 5.1, Borrower shall prepay the outstanding principal amount of the Obligations in accordance with Section 2.12(e) in an amount equal to (1) the Applicable Percentage of the Excess Cash Flow of Parent and its Subsidiaries for such fiscal year, minus (2) the aggregate amount of all voluntary prepayments in respect of the outstanding principal balance of the Term Loan made by Borrower during such fiscal year, minus (3) the amount of all voluntary prepayments in respect of the outstanding balance of Revolving Loans to the extent accompanied by a reduction of the Revolving Loan Commitment with respect thereto; provided, that any Excess Cash Flow payment made pursuant to this Section 2.12(d)(vi) shall exclude the portion of Excess Cash Flow that is attributable to the target of a Permitted Acquisition and that accrued prior to the closing date of such Permitted Acquisition; provided, further, that, no such prepayment shall be required pursuant to this Section 2.12(d)(vi) if the amount that would be required to be repaid is less than or equal to \$1,000,000 (the "ECF Threshold Amount"), and then, such prepayment shall only be required to the extent in excess of the ECF Threshold Amount.

(vii) [Reserved].

(e) Application of Payments.

- (i) Each prepayment pursuant to <u>Section 2.12(d)(i)</u> shall, (A) so long as no Application Event shall have occurred and be continuing, be applied, first, to the outstanding principal amount of the Loans (other than the Term Loan) until paid in full, and second, to cash collateralize the Letters of Credit in an amount equal to of the Maximum Undrawn Amount of all then outstanding Letters of Credit, and (B) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in <u>Section 2.12(a)(ii)</u>.
- (ii) Each prepayment pursuant to Section 2.12(d)(ii), 2.12(d)(vi) or 2.12(d)(vi) shall (A) so long as no Application Event shall have occurred and be continuing, be applied, first, to the outstanding principal amount of the Term Loan until paid in full, second, to the outstanding principal amount of the other Loans (with, at the election of the Revolving Lenders a corresponding permanent reduction in the Maximum Revolver Amount), until paid in full, and third, to cash collateralize the Letters of Credit in an amount equal to of the Maximum Undrawn Amount of all then outstanding Letters of Credit (with, at the election of the Revolving Lenders, a corresponding permanent reduction in the Maximum Revolver Amount), and (B) if an Application Event shall have occurred and be continuing, be applied in the manner set forth in Section 2.12(a)(ii). If, prior to an Application Event, any Lender waives in writing its right to any prepayment of the Loans or cash collateralization of the then outstanding Letters of Credit described in the foregoing subclause (A), Agent will remit such amount when received, ratably, to all other Lenders until the principal amount of the Loans held by such other Lenders have been paid in full in cash and of the Maximum Undrawn Amount of all then outstanding Letters of Credit is cash collateralized in the order set forth in subclause (A). Each such prepayment of the Term Loan shall be applied against the remaining installments of principal of the Term Loan in the inverse order of maturity (for the avoidance of doubt, any amount that is due and payable on the Maturity Date shall constitute an installment).
- 2.13. **Payment of Obligations**. Borrower promises to pay all of the Obligations (including principal, interest, premiums, if any, fees, costs, and expenses (including Lender Group Expenses)) in full on the Maturity Date or, if earlier, on the date on which the Obligations (other than the Bank Product Obligations, which are due and payable as provided therein) become due and payable pursuant to the terms of this Agreement. Borrower agrees that its obligations contained in this <u>Section 2.13</u> shall survive payment or satisfaction in full of all other

Obligations. Except to the extent provided to the contrary herein, all costs and expenses payable hereunder or under any of the other Loan Documents, all Lender Group Expenses and all amounts payable under Section 2.15(c) shall be due and payable on the earlier of (x) the first day of the month following the date on which the applicable costs, expenses, Lender Group Expenses or amounts payable under Section 2.15(c) were first incurred or (y) the date on which demand therefor is made by Agent (it being acknowledged and agreed that any charging of such costs, expenses, Lender Group Expenses or amounts due under Section 2.15(c) to the Loan Account pursuant to the provisions of the following sentence shall be deemed to constitute a demand for payment thereof for the purposes of this subclause (y); provided, that notwithstanding anything herein to the contrary, the Loan Account may not be charged for Lender Group Expenses to the extent an Overadvance would be caused thereby, unless Borrower has first been given 3 Business Days to pay such Lender Group Expenses). Borrower hereby authorizes Agent, from time to time without prior notice to Borrower, to charge to the Loan Account as a Revolving Loan made as a Base Rate Loan or, at the discretion of Swing Loan Lender, as a Swing Loan: (a) all payments with respect to any of the Obligations required hereunder (including without limitation principal payments, payments of interest, payments of Letter of Credit Fees, payments of all other fees provided for hereunder and payments under Section 10.3) as and when each such payment shall become due and payable (whether as regularly scheduled, upon or after acceleration, upon maturity or otherwise), and (b) without limiting the generality of the foregoing clause (a), (i) all amounts expended by Agent pursuant to any other Section of this Agreement, the Guaranty and Security Agreement or any other Loan Document, (ii) all expenses which Agent incurs in connection with the forwarding of Loan proceeds, and (iii) any other sums expended by Agent due to Borrower's failure to perform or comply with its obligations under this Agreement, the Guaranty and Security Agreement or any other Loan Document. All amounts referenced in this Section 2.13 shall constitute Obligations, and, if not charged to the Loan Account as a Revolving Loan maintained as a Base Rate Loan, shall be due and payable on demand by Agent or, if the Loans are then due and payable in full or an Event of Default exists under Section 8.4 or 8.5, due and payable when incurred without demand. To the extent Revolving Loans are not actually funded by the other Lenders in respect of any such amounts so charged, all such amounts so charged shall be deemed to be Revolving Loans made by and owing to Agent and Agent shall be entitled to all rights (including accrual of interest) and remedies of a Lender under this Agreement and the other Loan Documents with respect to such Revolving Loans.

2.14. Interest Rates and Letter of Credit Fees: Rates, Payments, and Calculations.

(a) Interest. Interest on Loans shall be payable in arrears on the first Business Day of each calendar month with respect to Base Rate Loans and, with respect to LIBORTerm SOFR Rate Loans, at the end of each Interest Period; provided that all accrued and unpaid interest shall be due and payable on the earlier of the Maturity Date or the date such Loans are otherwise due and payable. Interest charges shall be computed on the actual principal amount of Loans outstanding during the month at a rate per annum equal to (a) with respect to Loans other than the Term Loan and Letters of Credit, the Revolving Interest Rate, and (b) with respect to the Term Loan, the Term Loan Rate (as applicable, the "Contract Rate"). Except as expressly provided otherwise in this Agreement, any Obligations other than the Loans that are not paid when due shall accrue interest at the Revolving Interest Rate for Base Rate Loans, subject to the provision of the final sentence of this Section 2.14 regarding the Default Rate. Whenever, subsequent to the date of this Agreement, the Alternate Base Rate is increased or decreased, the applicable Contract Rate for Base Rate Loans shall be similarly changed without notice or demand of any kind by an amount equal to the amount of such change in the Alternate Base Rate during the time such change or changes remain in effect. The LIBORTerm SOFR Rate shall be adjusted with respect to LIBORTerm SOFR Rate Loans without notice or demand of any kind on the effective date of any change in the SOFR Reserve Percentage as of such

effective date. Upon the occurrence and during the continuation of any Specified Event of Default all or any portion of the Obligations (except for undrawn Letters of Credit), as determined by Agent or the Required Lenders, that have been charged to the Loan Account pursuant to the terms hereof or are otherwise then due and payable, shall bear interest at the applicable Contract Rate plus two percent (2%) per annum (as applicable, the "<u>Default Rate</u>").

- (b) Letter of Credit Fees. Borrower shall pay (i) to Agent, for the ratable benefit of Lenders holding Revolving Commitments, fees for each Letter of Credit for the period from and excluding the date of issuance of same to and including the date of expiration or termination, equal to the average daily face amount of each outstanding Letter of Credit multiplied by the Applicable LIBOR Term SOFR Rate Margin and SOFR Adjustment for Revolving Loans that are Term SOFR Rate Loans, such fees to be calculated on the basis of a 360-day year for the actual number of days elapsed and to be payable quarterly in arrears on the first day of each calendar quarter and on the Maturity Date (or if sooner, such other date as the Obligations are, or are required to be, paid in full), and (ii) to Issuing Bank, a fronting fee of one eighth of one percent (0.125%) per annum times the average daily face amount of each outstanding Letter of Credit for the period from and excluding the date of issuance of same to and including the date of expiration or termination, to be payable quarterly in arrears on the first day of each calendar quarter and on the Maturity Date (or if sooner, such other date as the Obligations are, or are required to be, paid in full (all of the foregoing fees, the "Letter of Credit Fees"). In addition, Borrower shall pay to Agent, for the benefit of Issuing Bank, any and all administrative, issuance, amendment, payment and negotiation charges with respect to Letters of Credit and all fees and expenses, in the case of each of the foregoing, as agreed upon by Issuing Bank and the Borrower, in connection with any Letter of Credit, including in connection with the opening, amendment or renewal of any such Letter of Credit and any acceptances created thereunder, all such charges, fees and expenses, if any, to be payable on demand. All such charges shall be deemed earned in full on the date when the same are due and payable hereunder and shall not be subject to rebate or pro-ration upon the termination of this Agreement for any reason. Any such charge in effect at the time of a particular transaction shall be the charge for that transaction, notwithstanding any subsequent change in Issuing Bank's prevailing charges for that type of transaction. Upon the occurrence and during the continuation of any Specified Event of Default, the Letter of Credit Fees described in clause (i) of this Section 2.14(b) shall be increased by an additional two percent (2.0%) per annum.
- (c) Computation of Interest and Fees. Interest and fees chargeable under the Loan Documents shall be computed on the basis of a year of 360 days and for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Contract Rate (or Default Rate, if applicable) during such extension.
- (d) **Maximum Charges**. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under applicable law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under applicable law: (i) the interest rates hereunder will be reduced to the maximum rate permitted under applicable law; (ii) such excess amount shall be first applied to any unpaid principal balance owed by Borrower; and (iii) if the then remaining excess amount is greater than the previously unpaid principal balance, Lenders shall promptly refund such excess amount to Borrower and the provisions hereof shall be deemed amended to provide for such permissible rate.

2.15. Fees.

- (a) Fee Letter Fees. Borrower shall pay the amounts required to be paid in the Fee Letter in the manner and at the times required by the Fee Letter.
- (b) Unused Line Fee. If, for any day in each calendar quarter during the term of this Agreement, the daily unpaid balance of the sum of Revolving Loans <u>plus</u> Swing Loans <u>plus</u> the Maximum Undrawn Amount of all outstanding Letters of Credit (the "<u>Usage Amount</u>") does not equal the Maximum Revolving Loan Amount, then Borrower shall pay to Agent, for the ratable benefit of the Lenders with a Revolving Commitment, a fee at a rate equal to the Applicable Unused Line Fee Rate on the amount by which the Maximum Revolving Loan Amount on such day exceeds such Usage Amount (the "<u>Facility Fee</u>"). Such Facility Fee shall be payable to Agent in arrears on the first Business Day of each calendar quarter with respect to each day in the previous calendar quarter and on the Maturity Date (or, if earlier, the date on which the Obligations are, or are required to be, paid in full) with respect to the portion of calendar quarter ending on such date.
- (c) **Financial Examination and Other Fees**. Borrower shall pay to Agent, financial examination and valuation fees and charges, as and when incurred or chargeable, as follows (i) a fee of \$1,000 per day, per examiner, plus out-of-pocket expenses (including travel, meals, and lodging) for each financial examination of Borrower performed by personnel employed by Agent, and (ii) the fees or charges paid or incurred by Agent (but, in any event, no less than a charge of \$1,000 per day, per Person, plus out-of-pocket expenses (including travel, meals, and lodging)) if it elects to employ the services of one or more third Persons to perform financial examinations of Parent or its Subsidiaries, to appraise the Collateral, or any portion thereof, or to assess Parent's or its Subsidiaries' business/recurring revenue valuation; provided, that so long as no Event of Default shall have occurred and be continuing, Borrower shall not be obligated to reimburse Agent for more than 1 field examination during any calendar year or more than 1 business/recurring revenue valuation during any calendar year.

2.16. Special Provisions Applicable to LIBOR Alternate Rate of Interest.

(a) The LIBOR Rate may be adjusted by Agent with respect to any Lender on a prospective basis to take into account any additional or increased costs to such Lender of maintaining or obtaining any eurodollar deposits or increased costs, in each case, due to changes in applicable law occurring subsequent to the commencement of the then applicable Interest Period, including any Changes in Law (including any changes in tax laws (except changes of general applicability in corporate income tax laws)) and changes in the reserve requirements imposed by the Board of Governors, which additional or increased costs would increase the cost of funding or maintaining loans bearing interest at the LIBOR Rate. In any such event, the affected Lender shall give Borrower and Agent written notice (which shall include a certificate setting forth in reasonable detail the basis for adjusting such LIBOR Rate and the method for determining the amount of such adjustment) of such a determination and adjustment and Agent promptly shall transmit the notice to each other Lender and, upon its receipt of the notice from the affected Lender, Borrower may, by notice to such affected Lender (A) require such Lender to furnish to Borrower a statement setting forth in reasonable detail the basis for adjusting such LIBOR Rate and the method for determining the amount of such adjustment, or (B) repay the LIBOR Rate Loans of such Lender with respect to which such adjustment is made (together with any amounts due under Section 2.2(g)). Notwithstanding the foregoing, Borrower shall not be required to compensate any Lender pursuant to this Section 2.16(a) for such additional or increased costs incurred more than 180 days prior to the date that such Lender delivers such certificate; provided, that if the change in applicable law giving rise to such additional or

increased costs is retroactive then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

- (ba) <u>Basis For Determining</u> Interest Rate Inadequate or Unfair. In the event that Agent or any Lender shall have determined that:
- (i) reasonable means do not exist for ascertaining the LIBOR Rate applicable pursuant to Section 2.2 hereof for any Interest Period Term SOFR Rate; or
- (ii) Dollar deposits in the relevant amount and for the relevant maturity are not available in the London interbank LIBOR market, with respect to an outstanding LIBOR Term SOFR Rate Loan, a proposed LIBOR Term SOFR Rate Loan, or a proposed conversion of a Base Rate Loan into a LIBOR Term SOFR Rate Loan; or
- (iii) the making, maintenance or funding of any <u>LIBORTerm SOFR</u> Rate Loan has been made impracticable or unlawful by compliance by Agent or such Lender in good faith with any Applicable Law or any interpretation or application thereof by any Governmental <u>AuthorityBody</u> or with any request or directive of any such Governmental <u>AuthorityBody</u> (whether or not having the force of law); or
- (iv) the <u>LIBORTerm SOFR</u> Rate will not adequately and fairly reflect the cost to <u>Agent or</u> such Lender of the <u>funding</u>, establishment or maintenance of any <u>LIBORTerm SOFR</u> Rate Loan, <u>during the applicable Interest Period</u>, and <u>Lenders have provided notice of such determination to Agent</u>,

then Agent shall give Borrower prompt written or telephonic notice of such determination. If such notice is given prior to a Benchmark Replacement Date (as defined below), (i) any such requested LIBORTerm SOFR Rate Loan shall be made as a Base Rate Loan, unless Borrower shall notify Agent no later than 1:00 p.m. two (2) Business Days prior to the date of such proposed borrowing, that its request for such borrowing shall be cancelled or made as an unaffected type of LIBORTerm SOFR Rate Loan, (ii) any Base Rate Loan or LIBORTerm SOFR Rate Loan, (ii) any Base Rate Loan or LIBORTerm SOFR Rate Loan, (ii) any Base Rate Loan or LIBORTerm SOFR Rate Loan, (iii) any Base Rate Loan or LIBORTerm SOFR Rate Loan, (iii) any Base Rate Loan shall be continued as or converted into a Base Rate Loan, or, if Borrower shall notify Agent, no later than 1:00 p.m. two (2) Business Days prior to the proposed conversion, shall be converted into a Base Rate Loan at the end of the applicable Interest Period, or, if Borrower shall notify Agent, no later than 1:00 p.m. two (2) Business Days prior to the last Business Day of the then current Interest Period applicable to such affected LIBORTerm SOFR Rate Loan, shall be converted into an unaffected type of LIBORTerm SOFR Rate Loan, on the last Business Day of the then current Interest Period for such affected LIBORTerm SOFR Rate Loan (or sooner, if Agent or such any Lender cannot continue to lawfully maintain such affected LIBORTerm SOFR Rate Loan). Until such notice has been withdrawn, neither Agent nor Lenders shall have any no obligation to make an affected type of LIBORTerm SOFR Rate Loan or maintain outstanding affected LIBORTerm SOFR Rate Loans and no Borrower shall have the right to convert a Base Rate Loan or an unaffected type of LIBORTerm SOFR Rate Loan into an affected type of LIBORTerm SOFR Rate Loan.

(b) Benchmark Replacement Setting

(i) Benchmark Replacement. Notwithstanding anything to the contrary herein or in any other Loan Document, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior to any setting of the then-current Benchmark, then (A) if a Benchmark Replacement is determined in accordance with clause (1) of the

definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any other Loan Document in respect of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document and (B) if a Benchmark Replacement is determined in accordance with clause (2) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any other Loan Document in respect of any Benchmark setting at or after 5:00 p.m. on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document so long as the Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Required Lenders.

- (ii) Benchmark Replacement Conforming Changes. In connection with the use, administration, adoption or implementation of a Benchmark Replacement, Agent will have the right to make Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in the other Loan Documents, any amendments implementing such Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (iii) Notices: Standards for Decisions and Determinations. Agent will promptly notify the Borrower and the Lenders of (A) the implementation of any Benchmark Replacement, and (B) the effectiveness of any Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. Agent will notify the Borrower of, (x) the removal or reinstatement of any tenor of a Benchmark pursuant to clause (iv) below and (y) the commencement of any Benchmark Unavailability Period. Any determination, decision or election that may be made by Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section, including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document except, in each case, as expressly required pursuant to this Section.
- (iv) Unavailability of Tenor of Benchmark. Notwithstanding anything to the contrary herein or in any of the other Loan Documents, at any time (including in connection with the implementation of a Benchmark Replacement), (A) if the then-current Benchmark is a term rate or based on a term rate and either (I) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by Agent in its reasonable discretion or (II) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor of such Benchmark is not or will not be representative, then Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (B) if a tenor that was removed pursuant to clause (A) above either (I) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (II) is not, or is no longer, subject to an announcement that it is not or will not be representative for a Benchmark (including a Benchmark Replacement), then Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate such previously removed tenor.

(v) Benchmark Unavailability Period. Upon Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, with respect to the Term SOFR Rate, Borrower may revoke any pending request for a Loan bearing interest based on such rate, or conversion to or continuation of Loans bearing interest based on such rate to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for a Base Rate Loan or conversion to a Base Rate Loan. During any Benchmark Unavailability Period or at any time that a tenor for the then-current Benchmark is not an Available Tenor, the component of the Alternate Base Rate based upon the then-current Benchmark or such tenor for such Benchmark, as applicable, will not be used in any determination of the Alternate Base Rate.

(vi) Certain Defined Terms. As used in this Section:

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an Interest Period pursuant to this Agreement as of such date and not including, for the avoidance of doubt, any tenor of such Benchmark that is then-removed from the definition of "Interest Period" pursuant to clause (iv) of this Section.

"Benchmark" means, initially, SOFR and the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to this Section.

"Benchmark Replacement" means, with respect to any Benchmark Transition Event, the first alternative set forth in the order below that can be determined by Agent for the applicable Benchmark Replacement Date:

(1) Daily Simple SOFR;

(2) the sum of: (a) the alternate benchmark rate that has been selected by Agent and the Borrower, giving due consideration to (x) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (y) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for U.S. dollar-denominated syndicated credit facilities at such time and (B) the related Benchmark Replacement Adjustment;

provided that, if the Benchmark Replacement as determined pursuant to clause (2) above would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents; provided further that any Benchmark Replacement shall be administratively feasible as determined by the Agent in its sole discretion.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustments, (which may be a positive or negative value or zero) that has been selected by Agent and the Borrower giving due consideration to (A) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (B)

any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for U.S. dollar-denominated syndicated credit facilities at such time.

"Benchmark Replacement Date" means a date and time determined by Agent, which date shall be no later than the earliest to occur of the following events with respect to the then-current Benchmark:

- (1) in the case of clause (1) or (2) of the definition of "Benchmark Transition Event," the later of (A) the date of the public statement or publication of information referenced therein and (B) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate or based on a term rate, all Available Tenors of such Benchmark (or such component thereof); or
- (2) in the case of clause (3) of the definition of "Benchmark Transition Event," the date determined by Agent, which date shall promptly follow the date of the public statement or publication of information referenced therein.

For the avoidance of doubt, if such Benchmark is a term rate or is based on a term rate, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (1) or (2) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (1) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide such Benchmark (or component thereof) or, if such Benchmark is a term rate or is based on a term rate, all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate or based on a term rate, any Available Tenor of such Benchmark (or such component thereof);
- (2) a public statement or publication of information by a Governmental Body having jurisdiction over the Agent, the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Federal Reserve Board, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component thereof) or, if such Benchmark is a term rate or is based on a term rate, all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide such Benchmark (or such component thereof) or, if such Benchmark is a term rate or is based on a term rate, any Available Tenor of such Benchmark (or such component thereof); or; or

(3) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) or a Governmental Body having jurisdiction over Agent announcing that such Benchmark (or such component thereof) or, if such Benchmark is a term rate or based on a term rate, all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative.

For avoidance of doubt, if such Benchmark is a term rate or is based on a term rate, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means the period (if any) (x) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any other Loan Document in accordance with this Section 2.16(b) and (y) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any other Loan Document in accordance with this Section titled "Benchmark Replacement Setting."

"Floor" means the benchmark rate floor, if any, provided in this Agreement initially (as of the execution of this Agreement, the modification, amendment or renewal of this Agreement or otherwise) with respect to the Term SOFR Rate, or if no floor is specified, zero.

"Relevant Governmental Body" means the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or any successor thereto.

"Unadjusted Benchmark Replacement" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

2.17. Capital Requirements.

(a) If, after the Closing Date, Issuing Bank or any Lender determines that (i) any Change in Law regarding capital or reserve requirements for banks or bank holding companies, or (ii) compliance by Issuing Bank or such Lender, or their respective parent bank holding companies, with any guideline, request or directive of any Governmental AuthorityBody regarding capital adequacy (whether or not having the force of law), has the effect of reducing the return on Issuing Bank's, such Lender's, or such holding companies' capital as a consequence of Issuing Bank's or such Lender's commitments hereunder to a level below that which Issuing Bank, such Lender, or such holding companies could have achieved but for such Change in Law or compliance (taking into consideration Issuing Bank's, such Lender's, or such holding companies' then existing policies with respect to capital adequacy and assuming the full utilization of such entity's capital) by any amount deemed by Issuing Bank or such Lender to be material, then Issuing Bank or such Lender may notify Borrower and Agent thereof. Following receipt of such notice, Borrower agrees to pay Issuing Bank or such Lender on demand the amount of such reduction of return of capital as and when such reduction is determined, payable within 90 days after presentation by Issuing Bank or such Lender of a statement in the amount and setting forth in reasonable detail Issuing Bank's or such Lender's calculation thereof and the assumptions upon which such calculation was based (which statement shall be deemed true and

correct absent manifest error). In determining such amount, Issuing Bank or such Lender may use any reasonable averaging and attribution methods. Failure or delay on the part of Issuing Bank or any Lender to demand compensation pursuant to this Section shall not constitute a waiver of Issuing Bank's or such Lender's right to demand such compensation; provided that Borrower shall not be required to compensate Issuing Bank or a Lender pursuant to this Section for any reductions in return incurred more than 180 days prior to the date that Issuing Bank or such Lender notifies Borrower of such Change in Law giving rise to such reductions and of such Lender's intention to claim compensation therefor; provided further that if such claim arises by reason of the Change in Law that is retroactive, then the 180-day period referred to above shall be extended to include the period of retroactive effect thereof.

- If Issuing Bank or any Lender requests additional or increased costs referred to in Section 2.11(j) or Section 2.16(a) or amounts under Section 2.17(a) or sends a notice under Section 2.16(b) relative to changed circumstances (such Issuing Bank or Lender, an "Affected Lender"), then such Affected Lender shall use reasonable efforts to promptly designate a different one of its lending offices or to assign its rights and obligations hereunder to another of its offices or branches, if (i) in the reasonable judgment of such Affected Lender, such designation or assignment would eliminate or reduce amounts payable pursuant to Section 2.11(i), Section 2.16(a) or Section 2.17(a), as applicable, or would eliminate the illegality or impracticality of funding or maintaining LIBOR Term SOFR Rate Loans and (ii) in the reasonable judgment of such Affected Lender, such designation or assignment would not subject it to any material unreimbursed cost or expense and would not otherwise be materially disadvantageous to it. Borrower agrees to pay all reasonable out-of-pocket costs and expenses incurred by such Affected Lender in connection with any such designation or assignment. If, after such reasonable efforts, such Affected Lender does not so designate a different one of its lending offices or assign its rights to another of its offices or branches so as to eliminate Borrower's obligation to pay any future amounts to such Affected Lender pursuant to Section 2.11(i), Section 2.16(a) or Section 2.17(a), as applicable, or to enable Borrower to obtain LIBOR Term SOFR Rate Loans, then Borrower (without prejudice to any amounts then due to such Affected Lender under Section 2.11(i), Section 2.16(a) or Section 2.17(a), as applicable) may, unless prior to the effective date of any such assignment the Affected Lender withdraws its request for such additional amounts under Section 2.11(j), Section 2.16(a) or Section 2.17(a), as applicable, or indicates that it is no longer unlawful or impractical to fund or maintain LIBOR Term SOFR Rate Loans, may designate a different Issuing Bank or substitute a Lender, in each case, reasonably acceptable to Agent to purchase the Obligations owed to such Affected Lender and such Affected Lender's commitments hereunder (a "Replacement Lender"), and if such Replacement Lender agrees to such purchase, such Affected Lender shall assign to the Replacement Lender its Obligations and commitments, and upon such purchase by the Replacement Lender, which such Replacement Lender shall be deemed to be "Issuing Bank" or a "Lender" (as the case may be) for purposes of this Agreement and such Affected Lender shall cease to be "Issuing Bank" or a "Lender" (as the case may be) for purposes of this Agreement.
- (c) Notwithstanding anything herein to the contrary, the protection of Sections 2.11(j), 2.16(a), and 2.17 shall be available to Issuing Bank and each Lender (as applicable) regardless of any possible contention of the invalidity or inapplicability of the law, rule, regulation, judicial ruling, judgment, guideline, treaty or other change or condition which shall have occurred or been imposed, so long as it shall be customary for issuing banks or lenders affected thereby to comply therewith. Notwithstanding any other provision herein, neither Issuing Bank nor any Lender shall demand compensation pursuant to this Section 2.17 if it shall not at the time be the general policy or practice of Issuing Bank or such Lender (as the case may be) to demand such compensation in similar circumstances under comparable provisions of other credit agreements, if any.

3. CONDITIONS; TERM OF AGREEMENT.

- 3.1. <u>Conditions Precedent to the Initial Extension of Credit</u>. The obligation of each Lender to make the initial extensions of credit provided for hereunder is subject to the fulfillment, to the satisfaction of Agent and each Lender, of each of the conditions precedent set forth on <u>Schedule 3.1</u> (the making of such initial extensions of credit by a Lender being conclusively deemed to be its satisfaction or waiver of the conditions precedent).
- 3.2. <u>Conditions Precedent to all Extensions of Credit</u>. The obligation of the Lender Group (or any member thereof) to make any Revolving Loans hereunder (or to extend any other credit hereunder) at any time shall be subject to the following conditions precedent:
- (a) the representations and warranties of Parent or its Subsidiaries contained in this Agreement or in the other Loan Documents shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) on and as of the date of such extension of credit, as though made on and as of such date (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date); and
- (b) no Default or Event of Default shall have occurred and be continuing on the date of such extension of credit, nor shall either result from the making thereof.
 - 3.3. Maturity. This Agreement shall continue in full force and effect for a term ending on the Maturity Date.
- 3.4. Effect of Maturity. On the Maturity Date, all commitments of the Lender Group to provide additional credit hereunder shall automatically be terminated and all of the Obligations immediately shall become due and payable without notice or demand and Borrower shall be required to repay all of the Obligations in full. No termination of the obligations of the Lender Group (other than payment in full of the Obligations and termination of the Commitments) shall relieve or discharge any Loan Party of its duties, obligations, or covenants hereunder or under any other Loan Document and Agent's Liens in the Collateral shall continue to secure the Obligations and shall remain in effect until all Obligations have been paid in full and the Commitments have been terminated. When all of the Obligations have been paid in full and the Lender Group's obligations to provide additional credit under the Loan Documents have been terminated irrevocably, Agent will, at Borrower's sole expense, execute and deliver any termination statements, lien releases, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, Agent's Liens and all notices of security interests and liens previously filed by Agent.
- 3.5. <u>Early Termination by Borrower</u>. Borrower has the option, at any time upon 5 Business Days prior written notice to Agent, to terminate this Agreement and terminate the Commitments hereunder by repaying to Agent all of the Obligations in full. The foregoing notwithstanding, (a) Borrower may rescind termination notices relative to proposed payments in full of the Obligations with the proceeds of third party Indebtedness if the closing for such issuance or incurrence does not happen on or before the date of the proposed termination (in which case, a new notice shall be required to be sent in connection with any subsequent termination), and (b) Borrower may extend the date of termination at any time with the consent of Agent (which consent shall not be unreasonably withheld or delayed).

3.6. <u>Conditions Subsequent</u>. The obligation of the Lender Group (or any member thereof) to continue to make Revolving Loans (or otherwise extend credit hereunder) is subject to the fulfillment, on or before the date applicable thereto, of the conditions subsequent set forth on <u>Schedule 3.6</u> (the failure by Borrower to so perform or cause to be performed such conditions subsequent as and when required by the terms thereof (unless such date is extended, in writing (including via email), by Agent, which Agent may do without obtaining the consent of the other members of the Lender Group), shall constitute an Event of Default).

4. REPRESENTATIONS AND WARRANTIES.

In order to induce the Lender Group to enter into this Agreement, each Loan Party makes the following representations and warranties to the Lender Group which shall be true, correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the Closing Date and/or the Third Amendment Effective Date, as applicable, and shall be true, correct, and complete, in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof), as of the date of the making of each Revolving Loan (or other extension of credit) made thereafter, as though made on and as of the date of such Revolving Loan (or other extension of credit) (except to the extent that such representations and warranties relate solely to an earlier date, in which case such representations and warranties shall be true and correct in all material respects (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of such earlier date) and such representations and warranties shall survive the execution and delivery of this Agreement:

4.1. <u>Due Organization and Qualification; Subsidiaries</u>.

- (a) Each Loan Party (i) is duly organized and existing and in good standing under the laws of the jurisdiction of its organization, (ii) is qualified to do business in any state where the failure to be so qualified could reasonably be expected to result in a Material Adverse Effect, and (iii) has all requisite power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to enter into the Loan Documents to which it is a party and to carry out the transactions contemplated thereby.
- (b) Set forth on Schedule 4.1(b) (as such Schedule may be updated from time to time to reflect changes resulting from transactions permitted under this Agreement) is a complete and accurate description of the authorized Equity Interests of Parent, by class, and, as of the Closing Date and/or, solely with respect to SpeechIQ Guarantor, the Third Amendment Effective Date, as applicable, a description of the number of shares of each such class that are issued and outstanding. Parent is not subject to any obligation (contingent or otherwise) to repurchase or otherwise acquire or retire any shares of its Equity Interests or any security convertible into or exchangeable for any of its Equity Interests.
- (c) Set forth on Schedule 4.1(c) (as such Schedule may be updated from time to time to reflect changes resulting from transactions permitted under this Agreement), is a complete and accurate list of the Loan Parties' direct and indirect Subsidiaries, showing: (i) the number of shares of each class of common and preferred Equity Interests authorized for each of such Subsidiaries, and (ii) the number and the percentage of the outstanding shares of each such class owned directly or indirectly by Parent. All of the outstanding Equity Interests of each such Subsidiary has been validly issued and is fully paid and non-assessable.

(d) There are no subscriptions, options, warrants, or calls relating to any shares of any Loan Party's or its Subsidiaries' Equity Interests, including any right of conversion or exchange under any outstanding security or other instrument.

4.2. <u>Due Authorization; No Conflict.</u>

- (a) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party have been duly authorized by all necessary action on the part of such Loan Party.
- (b) As to each Loan Party, the execution, delivery, and performance by such Loan Party of the Loan Documents to which it is a party do not and will not (i) violate any material provision of federal, state, or local law or regulation applicable to any Loan Party or its Subsidiaries, the Governing Documents of any Loan Party or its Subsidiaries, or any order, judgment, or decree of any court or other Governmental AuthorityBody binding on any Loan Party or its Subsidiaries, (ii) conflict with, result in a breach of, or constitute (with due notice or lapse of time or both) a default under any Material Contract of any Loan Party or its Subsidiaries except to the extent that any such conflict, breach or default could not individually or in the aggregate reasonably be expected to have a Material Adverse Effect, (iii) result in or require the creation or imposition of any Lien of any nature whatsoever upon any assets of any Loan Party, other than Permitted Liens, or (iv) require any approval of any holder of Equity Interests of a Loan Party or any approval or consent of any Person under any Material Contract of any Loan Party, other than consents or approvals that have been obtained and that are still in force and effect and except, in the case of Material Contracts, for consents or approvals, the failure to obtain the same could not individually or in the aggregate reasonably be expected to cause a Material Adverse Effect.
- 4.3. Governmental Consents. The execution, delivery, and performance by each Loan Party of the Loan Documents to which such Loan Party is a party and the consummation of the transactions contemplated by the Loan Documents do not and will not require any registration with, consent, or approval of, or notice to, or other action with or by, any Governmental AuthorityBody, other than registrations, consents, approvals, notices, or other actions that have been obtained and that are still in force and effect and except for filings and recordings with respect to the Collateral to be made, or otherwise delivered to Agent for filing or recordation, as of the Closing Date and/or the Third Amendment Effective Date and/or the Seventh Amendment Effective Date, as applicable.

4.4. Binding Obligations; Perfected Liens.

- (a) Each Loan Document has been duly executed and delivered by each Loan Party that is a party thereto and is the legally valid and binding obligation of such Loan Party, enforceable against such Loan Party in accordance with its respective terms, except as enforcement may be limited by equitable principles or by bankruptcy, insolvency, reorganization, moratorium, or similar laws relating to or limiting creditors' rights generally.
- (b) Agent's Liens are validly created, perfected (other than (i) in respect of motor vehicles that are subject to a certificate of title, (ii) money, (iii) letter-of-credit rights (other than supporting obligations), (iv) commercial tort claims that are in the aggregate less than \$50,000, and (v) any Deposit Accounts and Securities Accounts not subject to a Control Agreement as permitted by Section 7(k)(iv) of the Guaranty and Security Agreement, and subject only to the filing of financing statements, the recordation of the Copyright Security Agreement and the recordation of the Mortgages, in each case, in the appropriate filing offices), and are first

priority Liens, subject only to Liens permitted pursuant to clauses (d), (e), (f), (g), (h), (i), (j), (k), (l), (n), (q), (r) of the definition of Permitted Liens.

4.5. **Title to Assets; No Encumbrances**. Each of the Loan Parties and its Subsidiaries has (a) good, sufficient and legal title to (in the case of fee interests in Real Property), (b) valid leasehold interests in (in the case of leasehold interests in real or personal property), and (c) good and marketable title to (in the case of all other personal property), all of their respective assets reflected in their most recent financial statements delivered pursuant to Section 5.1, in each case except for assets disposed of since the date of such financial statements to the extent permitted hereby. All of such assets are free and clear of Liens except for Permitted Liens.

4.6. Litigation

- (a) There are no actions, suits, or proceedings pending or, to the knowledge of the Loan Parties, after due inquiry, threatened in writing against a Loan Party or any of its Subsidiaries that either individually or in the aggregate could reasonably be expected to result in a Material Adverse Effect.
- (b) <u>Schedule 4.6(b)</u> sets forth a complete and accurate description, with respect to each of the actions, suits, or proceedings with asserted liabilities in excess of, or that could reasonably be expected to result in liabilities in excess of, \$50,000 that, as of the Closing Date and/or, solely with respect to SpeechIQ Guarantor, the Third Amendment Effective Date, as applicable, is pending or, to the knowledge of the Loan Parties, threatened in writing against a Loan Party or any of its Subsidiaries, of (i) the parties to such actions, suits, or proceedings, (ii) the nature of the dispute that is the subject of such actions, suits, or proceedings, (iii) the procedural status, as of the Closing Date and/or, solely with respect to SpeechIQ Guarantor, the Third Amendment Effective Date, as applicable, with respect to such actions, suits, or proceedings, and (iv) whether any liability of the Loan Parties' and their Subsidiaries in connection with such actions, suits, or proceedings is covered by insurance.
- 4.7. Compliance with Laws. No Loan Party nor any of its Subsidiaries (a) is in violation of any applicable laws, rules, regulations, executive orders, or codes (including Environmental Laws) that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect, or (b) is subject to or in default with respect to any final judgments, writs, injunctions, decrees, rules or regulations of any court or any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.
- 4.8. No Material Adverse Effect. All historical financial statements relating to Parent and its Subsidiaries that have been delivered by any of the Loan Parties to Agent have been prepared in accordance with GAAP (except, in the case of unaudited financial statements, for the lack of footnotes and being subject to year-end audit adjustments) and present fairly in all material respects, the Parent's and its Subsidiaries' consolidated financial condition as of the date thereof and results of operations for the period then ended. Since December 31, 2015, no event, circumstance, or change has occurred that has or could reasonably be expected to result in a Material Adverse Effect with respect to the Loan Parties and their Subsidiaries.

4.9. Solvency.

(a) The Loan Parties, taken as a whole, are Solvent.

- (b) No transfer of property is being made by any Loan Party and no obligation is being incurred by any Loan Party in connection with the transactions contemplated by this Agreement or the other Loan Documents with the intent to hinder, delay, or defraud either present or future creditors of such Loan Party.
- 4.10. Employee Benefits. No Loan Party, none of its Subsidiaries, nor any of their respective ERISA Affiliates maintains, contributes to or has an obligation to contribute to any Benefit Plan.
- 4.11. Environmental Condition. Except as set forth on Schedule 4.11, (a) to the Loan Parties' knowledge, no Loan Party's nor any of its Subsidiaries' properties or assets has ever been used by a Loan Party, its Subsidiaries, or by previous owners or operators in the disposal of, or to produce, store, handle, treat, release, or transport, any Hazardous Materials, where such disposal, production, storage, handling, treatment, release or transport was in violation, in any material respect, of any applicable Environmental Law and reasonably could be expected to result in a Material Adverse Effect, (b) to the Loan Parties' knowledge, after due inquiry, no Loan Party's nor any of its Subsidiaries' properties or assets has ever been designated or identified in any manner pursuant to any environmental protection statute as a Hazardous Materials disposal site which designation or identification reasonably could be expected to result in a Material Adverse Effect, (c) no Loan Party nor any of its Subsidiaries has received any written notice that a Lien arising under any Environmental Law has attached to any revenues or to any Real Property owned or operated by a Loan Party or its Subsidiaries, and (d) no Loan Party nor any of its Subsidiaries nor any of their respective facilities or operations is subject to any outstanding written order, consent decree, or settlement agreement with any Person relating to any Environmental Law or Environmental Liability that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect.
- 4.12. Complete Disclosure. All factual information (taken as a whole) (other than forward looking information and projections and information of a general economic nature and general information about any Loan Party's industry) furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender (including all information contained in the Schedules hereto or in the other Loan Documents) for purposes of or in connection with this Agreement, the other Loan Documents, or any transaction contemplated herein or therein is, and all other such factual information taken as a whole (other than forward-looking information and projections and information of a general economic nature and general information about any Loan Party's industry) hereafter furnished by or on behalf of a Loan Party or its Subsidiaries in writing to Agent or any Lender will be, true and accurate, in all material respects, on the date as of which such information is dated or certified and not incomplete by omitting to state any fact necessary to make such information (taken as a whole) not misleading in any material respect at such time in light of the circumstances under which such information was provided. The Projections delivered to Agent as of the Closing Date represent, and as of the date on which any other Projections are delivered to Agent, such additional Projections represent, the Loan Parties' good faith estimate, on the date such Projections are delivered, of the Loan Parties' and their Subsidiaries' future performance for the periods covered thereby based upon assumptions believed by the Loan Parties' to be reasonable at the time of the delivery thereof to Agent (it being understood that such Projections are subject to significant uncertainties and contingencies, many of which are beyond the control of the Loan Parties and their Subsidiaries, and no assurances can be given that such Projections will be realized, and although reflecting the Loan Parties' good faith estimate, projections or forecasts based on methods and assumptions which the Loan Parties believed to be reasonable at the time such Projections were prepared, are not to be viewed as facts, and that actual results during the period or periods covered by the Projections may differ materially from projected or estimated results).

- 4.13. Material Contracts. Set forth on Schedule 4.13 (as updated from time to time) is a reasonably detailed description of the Material Contracts (other than the Advisory Agreement and the Applicable Affiliate Transaction Documents) of each Loan Party and its Subsidiaries; provided, however, that the Loan Parties may amend Schedule 4.13 to add additional Material Contracts so long as such amendment occurs by written notice to Agent at the time that the Loan Parties provide their quarterly financial statements pursuant to Section 5.1. Except for matters which, either individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, each Material Contract (other than those that have expired at the end of their normal terms) (a) is in full force and effect and is binding upon and enforceable against the applicable Loan Party or its Subsidiary (except as enforcement may be limited by equitable principles or by bankruptcy, insolvency, reorganization, moratorium, or similar laws relating to or limiting creditors' rights in any jurisdiction generally) and, to the best of the Loan Parties' knowledge, after due inquiry, each other Person that is a party thereto in accordance with its terms, (b) has not been otherwise amended or modified (other than amendments or modifications permitted by Section 6.7), and (c) is not in default due to the action or inaction of the applicable Loan Party or its Subsidiary in a manner that could not reasonably be expected to result in: (A) a Default or an Event of Default or (B) a Material Adverse Effect.
- 4.14. **Patriot Act**. The Loan Parties, each of their respective Subsidiaries and each of their respective Affiliates are in compliance with (a) the Trading with the Enemy Act, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B Chapter V, as amended) and any other enabling legislation or executive order relating thereto, (b) the Patriot Act and (c) other Federal or State laws relating to "know your customer" and anti-money laundering rules and regulations.
- 4.15. <u>Indebtedness</u>. Set forth on <u>Schedule 4.14</u> is a true and complete list of all Indebtedness of (a) each Loan Party (other than SpeechIQ Guarantor) and each of its Subsidiaries outstanding immediately prior to the Closing Date that is to remain outstanding immediately after giving effect to the closing hereunder on the Closing Date and such Schedule accurately sets forth the aggregate principal amount of such Indebtedness as of the Closing Date and (b) SpeechIQ Guarantor and its Subsidiaries outstanding immediately prior to the Third Amendment Effective Date that is to remain outstanding immediately after giving effect to the closing hereunder on the Third Amendment Effective Date and such Schedule accurately sets forth the aggregate principal amount of such Indebtedness as of the Third Amendment Effective Date.
- 4.16. Payment of Taxes. Except as set forth in Schedule 4.16 and as otherwise permitted under Section 5.5, all federal and other material tax returns and reports of each Loan Party and its Subsidiaries required to be filed by any of them have been timely filed, and all taxes shown on such tax returns to be due and payable and all assessments, fees and other governmental charges upon a Loan Party and its Subsidiaries and upon their respective assets, income, businesses and franchises that are due and payable have been paid when due and payable. Each Loan Party and each of its Subsidiaries have made adequate provision in accordance with GAAP for all taxes not yet due and payable. No Loan Party knows of any proposed tax assessment against a Loan Party or any of its Subsidiaries that is not being actively contested by such Loan Party or such Subsidiary diligently, in good faith, and by appropriate proceedings; provided such reserves or other appropriate provisions, if any, as shall be required in conformity with GAAP shall have been made or provided therefor.
- 4.17. <u>Margin Stock</u>. No Loan Party nor any of its Subsidiaries is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of the loans made to Borrower will be used to purchase or carry any Margin Stock or to extend credit to others for the

purpose of purchasing or carrying any Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors.

- 4.18. **Governmental Regulation**. No Loan Party nor any of its Subsidiaries is subject to regulation under the Federal Power Act or the Investment Company Act of 1940 or under any other federal or state statute or regulation which may limit its ability to incur Indebtedness or which may otherwise render all or any portion of the Obligations unenforceable. No Loan Party nor any of its Subsidiaries is a "registered investment company" or a company "controlled" by a "registered investment company" or a "principal underwriter" of a "registered investment company" as such terms are defined in the Investment Company Act of 1940.
- 4.19. Anti-Terrorism Laws; Anti-Corruption Laws. On the date specified in item (d) on Schedule 3.6 (as may be extended pursuant to Section 3.6) and at all times thereafter, the Loan Parties' policies and procedures are sufficient to ensure compliance by the Loan Parties, their Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions. The Loan Parties, their Subsidiaries and their respective officers and directors and, to the knowledge of the Loan Parties, their respective employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. None of any Loan Party, any Subsidiary or, to the knowledge of any Loan Party, any of their respective directors, officers, agents or employees that will act in any capacity in connection with or benefit from the credit facility established hereby is a Sanctioned Person. No Loan, use of proceeds or other transaction contemplated by this Agreement will violate Anti-Corruption Laws or applicable Sanctions.
- 4.20. Employee and Labor Matters. There is (i) no unfair labor practice complaint pending or, to the knowledge of the Loan Parties', threatened against Parent or its Subsidiaries before any Governmental AuthorityBody and no grievance or arbitration proceeding pending or to the knowledge of the Loan Parties' threatened against Parent or its Subsidiaries which arises out of or under any collective bargaining agreement and that could reasonably be expected to result in a material liability, (ii) no strike, labor dispute, slowdown, stoppage or similar action or grievance pending or to the knowledge of the Loan Parties' threatened in writing against Parent or its Subsidiaries that could reasonably be expected to result in a material liability, or (iii) to the knowledge of the Loan Parties', after due inquiry, no union representation question existing with respect to the employees of Parent or its Subsidiaries. None of Parent or its Subsidiaries and no union organizing activity taking place with respect to any of the employees of Parent or its Subsidiaries. None of Parent or its Subsidiaries has incurred any liability or obligation under the Worker Adjustment and Retraining Notification Act or similar state law, which remains unpaid or unsatisfied. The hours worked and payments made to employees of Parent or its Subsidiaries have not been in violation of the Fair Labor Standards Act or any other applicable legal requirements, except to the extent such violations could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect. All material payments due from Parent or its Subsidiaries on account of wages and employee health and welfare insurance and other benefits have been paid or accrued as a liability on the books of Parent or its Subsidiaries, except where the failure to do so could not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- 4.21. **Parent as a Holding Company**. Parent is a holding company and does not have any material liabilities (other than liabilities arising under the Loan Documents, the Advisory Agreement, the Acquisition Documents to which it is a party), own any material assets (other than the Equity Interests of Borrower and its other Subsidiaries) or engage in any operations, activities or business (other than the ownership of Borrower and its other Subsidiaries and activities incidental to the direct and indirect ownership of its Subsidiaries, activities incidental to the maintenance of its corporate existence, activities related to the debt and equity financing of

its acquisition of the Borrower and the consummation of the transactions in connection therewith, and as otherwise expressly permitted by the Loan Documents).

- 4.22. **SpeechIQ Acquisition; Engage Acquisition**. As of the Third Amendment Effective Date, to the knowledge of Borrower, the SpeechIQ Acquisition Agreement Representations are true and correct in all material respects (or, if qualified by materiality under the terms thereof, in all respects). As of the Sixth Amendment Effective Date, to the knowledge of Borrower, the Engage Acquisition Agreement Representations are true and correct in all material respects (or, if qualified by materiality under the terms thereof, in all respects).
- 4.23. <u>Merger Documentation Representations</u>. As of the Fifth Amendment Effective Date, to the knowledge of the Loan Parties, the Merger Documentation Representations are true and correct in all material respects (or, if qualified by materiality under the terms thereof, in all respects), except for Merger Documentation Representations which relate exclusively to an earlier date, which shall be true and correct in all material respects (or, if qualified by materiality under the terms thereof, in all respects) on and as of such earlier date.
- 4.24. <u>Leases</u>. Each Loan Party and its Subsidiaries enjoy peaceful and undisturbed possession under all leases material to their business and to which they are parties or under which they are operating, and, subject to Permitted Protests, all of such material leases are valid and subsisting and no material default by the applicable Loan Party or its Subsidiaries exists under any of them.
- 4.25. <u>Hedge Agreements</u>. On each date that any Hedge Agreement is executed by any Hedge Provider, Borrower and each other Loan Party satisfy all eligibility, suitability and other requirements under the Commodity Exchange Act (7 U.S.C. § 1, et seq., as in effect from time to time) and the Commodity Futures Trading Commission regulations.
- 4.26. <u>Certificate of Beneficial Ownership</u>. The Certificate of Beneficial Ownership executed and delivered to Agent and Lenders for each Borrower on or prior to the Third Amendment Effective Date, as updated from time to time in accordance with this Agreement, is accurate, complete and correct as of the Third Amendment Effective Date and as of the date any such update is delivered.

5. AFFIRMATIVE COVENANTS.

The Loan Parties covenant and agree that, until termination of all of the Commitments and payment in full of the Obligations:

5.1. Financial Statements, Reports, Certificates. Each Loan Party (as applicable) (a) will deliver to Agent, with copies to each Lender, each of the financial statements, reports, and other items set forth on Schedule 5.1 no later than the times specified therein, (b) agrees that no Subsidiary of a Loan Party will have a fiscal year different from that of Parent, (c) agrees to maintain a system of accounting that enables the Loan Parties to produce financial statements in accordance with GAAP, and (d) except as set forth on Schedule 5.1(d), agrees that it will, and will cause each other Loan Party to, maintain in all material respects its billing systems/practices and shall only make material modifications thereto with notice to, and with the consent of, Agent, which consent shall not be unreasonably withheld; provided that, notwithstanding anything herein to the contrary, in the case of documents required to be delivered pursuant to this Section 5.1 and included in materials otherwise filed with the SEC, each Loan Party shall be deemed to satisfy the requirements of this Section 5.1 to the extent (x) such documents are included in materials filed with the SEC in compliance with the reporting requirements under the Exchange Act, and (y) such documents are delivered electronically to Agent and if so delivered.

shall be deemed to have been delivered on the date on which such Loan Party posts such documents, or provides a link thereto, on such Loan Party's or Public Holdco's website on the internet at such Loan Party's website address.https://investors.livevox.com.

- 5.2. **Reporting.** The applicable Loan Parties will deliver to Agent (and if so requested by Agent, with copies for each Lender) each of the reports set forth on <u>Schedule 5.2</u> at the times specified therein.
- 5.3. Existence. Except as otherwise permitted under Section 6.3 or Section 6.4, each Loan Party will, and will cause each of their Subsidiaries to, at all times maintain and preserve in full force and effect such Person's existence and good standing in its jurisdiction of organization and, except as could not reasonably be expected to result in a Material Adverse Effect, good standing with respect to all other jurisdictions in which it is qualified to do business and any rights, franchises, permits, licenses, accreditations, authorizations, or other approvals material to their businesses; provided, however, that no Loan Party or any of its Subsidiaries shall be required to preserve any such right or franchise, licenses or permits if such Person's board of directors (or similar governing body) shall determine that the preservation thereof is no longer desirable in the conduct of the business of such Person, and that the loss thereof is not disadvantageous in any material respect to such Person or to the Lenders.
- 5.4. Maintenance of Properties. Each Loan Party will, and will cause each of its Subsidiaries to, maintain and preserve all of its tangible property that are necessary or useful in the proper conduct of its business in good working order and condition, ordinary wear, tear, casualty, and condemnation and Permitted Dispositions excepted (and except where the failure to do so could not reasonably be expected to result in a Material Adverse Effect) and comply with the material provisions of all material leases to which it is a party as lessee, so as to prevent the loss or forfeiture thereof, unless such provisions are the subject of a Permitted Protest.
- 5.5. Taxes. Each Loan Party will, and will cause each of its Subsidiaries to, pay in full before delinquency or before the expiration of any extension period all material governmental assessments and taxes imposed, levied, or assessed against any Loan Party or its Subsidiaries, or any of their respective assets or in respect of any of its income, businesses, or franchises to be paid in full, before delinquency or before the expiration of any extension period, except to the extent that the validity of such governmental assessment or tax is the subject of a Permitted Protest.
- 5.6. Insurance. The Loan Parties will, and will cause each of their Subsidiaries to, at the Loan Parties' expense, maintain insurance respecting each of the Loan Parties' and their Subsidiaries' assets wherever located, covering liabilities, losses or damages as are customarily are insured against by other Persons engaged in same or similar businesses and similarly situated and located. The Loan Parties also shall maintain (with respect to each of the Loan Parties and their Subsidiaries) business interruption, general liability, product liability insurance (to the extent in effect on the Seventh Amendment Effective Date), director's and officer's liability insurance, as well as insurance against larceny, embezzlement and criminal misappropriation. All such policies of insurance shall be with financially sound and reputable insurance companies acceptable to Agent (it being agreed that, as of the Closing Date, Travelers Property Cas Co. of America, AXIS Insurance Company, and Arch Insurance Group are acceptable to Agent) and in such amounts as is carried generally in accordance with sound business practice by companies in similar businesses similarly situated and located and, in any event, in amount, adequacy, and scope reasonably satisfactory to Agent (it being agreed that the amount, adequacy, and scope of the policies of insurance of the Loan Parties in effect as of the Seventh Amendment Effective Date are acceptable to Agent). All property insurance policies covering the Collateral are to be made payable to Agent for the benefit of Agent and the Lenders, as their interests may appear, in

case of loss, pursuant to a standard loss payable endorsement with a standard noncontributory "lender" or "secured party" clause and are to contain such other provisions as Agent may reasonably require to fully protect the Lenders' interest in the Collateral and to any payments to be made under such policies. All certificates of property and general liability insurance are to be delivered to Agent, with the loss payable (but only in respect of Collateral) and additional insured endorsements in favor of Agent and shall provide for not less than 30 days (10 days in the case of non-payment) prior written notice to Agent of the exercise of any right of cancellation. If the Loan Parties fail to maintain such insurance, Agent may arrange for such insurance, but at the Loan Parties' expense and without any responsibility on Agent's part for obtaining the insurance, the solvency of the insurance companies, the adequacy of the coverage, or the collection of claims. The Loan Parties shall give Agent prompt notice of any loss exceeding \$250,000 covered by its or its Subsidiaries' casualty or business interruption insurance. Upon the occurrence and during the continuance of an Event of Default, Agent shall have the sole right to file claims under any property and general liability insurance policies in respect of the Collateral, to receive, receipt and give acquittance for any payments that may be payable thereunder, and to execute any and all endorsements, receipts, releases, assignments, reassignments or other documents that may be necessary to effect the collection, compromise or settlement of any claims under any such insurance policies.

5.7. **Inspection**.

- (a) The Loan Parties will, and will cause each of their Subsidiaries to, permit Agent and its duly authorized representatives or agents to visit any of its properties and inspect any of its assets or books and records, to examine and make copies of its books and records, and to discuss its affairs, finances, and accounts with, and to be advised as to the same by, its officers and employees (provided an authorized representative of the Loan Parties shall be given a reasonable opportunity to be present) at such reasonable times and intervals as Agent may designate and, so long as no Default or Event of Default has occurred and is continuing, with reasonable prior notice to the Loan Parties and during regular business hours; provided, however, that so long as no Event of Default exists, the Loan Parties will only be obligated to reimburse Agent and Agent's employees or agents (and no other Lender or such other Lender's employees or agents) for costs and expenses arising in connection with any such inspection. Any Lender or its duly authorized representatives or agents may accompany Agent or any of its duly authorized representatives or agents at such Lender's expense.
- (b) The Loan Parties will, and will cause each of their Subsidiaries to, permit Agent and each of its duly authorized representatives or agents to conduct appraisals and valuations at such reasonable times and intervals as Agent may designate.
- 5.8. <u>Compliance with Laws</u>. The Loan Parties will, and will cause each of their Subsidiaries to, comply with the requirements of all applicable laws, rules, regulations, and orders of any Governmental <u>AuthorityBody</u>, other than laws, rules, regulations, and orders the noncompliance with which, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect. The Loan Parties will maintain in effect and enforce policies and procedures designed to ensure compliance by the Loan Parties, their Subsidiaries and the respective directors, officers, employees and agents of the foregoing with Anti-Corruption Laws and applicable Sanctions.

- 5.9. **Environmental**. The Loan Parties will, and will cause each of their Subsidiaries to,
- (a) Keep any property either owned or operated by Parent or its Subsidiaries free of any Environmental Liens or post bonds or other financial assurances sufficient to satisfy the obligations or liability evidenced by such Environmental Liens,
- (b) Comply, in all material respects, with Environmental Laws and provide to Agent documentation of such compliance which Agent reasonably requests where failure to do so reasonably could be expected to result in a Material Adverse Effect,
- (c) Promptly notify Agent of any release of which any Loan Party has knowledge of a Hazardous Material in any reportable quantity from or onto property owned or operated by Parent or its Subsidiaries, which release could reasonably be expected to result in a Material Adverse Effect and take any Remedial Actions required to abate said release or otherwise to come into compliance, in all material respects, with applicable Environmental Law in each case where the failure to do so could reasonably be expected to result in a Material Adverse Effect, and
- (d) Promptly, but in any event within 10 days of its receipt thereof, provide Agent with written notice of any of the following: (i) notice that an Environmental Lien has been filed against any of the real or personal property of Parent or its Subsidiaries, (ii) commencement of any Environmental Action or written notice that an Environmental Action will be filed against Parent, or their Subsidiaries, which Environmental Action, if adversely resolved, could reasonably be expected to result in a Material Adverse Effect and (iii) notice of a violation, citation, or other administrative order relating to Environmental Laws, which, if adversely resolved, could reasonably be expected to result in a Material Adverse Effect.
- 5.10. **Disclosure Updates**. The Loan Parties will, promptly and in no event later than 5 Business Days after obtaining knowledge thereof, notify Agent if any written information, exhibit, or report (other than projections and other forward looking statements) furnished to Agent or the Lenders contained, at the time it was furnished, any untrue statement of a material fact or omitted to state any material fact necessary to make the statements contained therein (taken as a whole) not misleading in any material respect, in light of the circumstances in which made (it being understood that with respect to projections and other forward-looking statements, the same are subject to the proviso in the last sentence of Section 4.16). The foregoing to the contrary notwithstanding, any notification pursuant to the foregoing provision will not cure or remedy the effect of the prior untrue statement of a material fact or omission of any material fact nor shall any such notification have the effect of amending or modifying this Agreement or any of the Schedules hereto.
- 5.11. Formation of Subsidiaries. The Loan Parties will, at the time that any Loan Party forms any direct or indirect Subsidiary (other than a Subsidiary which is a CFC or an Excluded Domestic Subsidiary) or acquires any direct or indirect Subsidiary (other than a Subsidiary which is a CFC or an Excluded Domestic Subsidiary) after the Closing Date, within 20 days of such formation or acquisition (or such later date as permitted by Agent in its sole discretion) (a) cause such new Subsidiary to provide to Agent a joinder to the Guaranty and Security Agreement, together with such other security agreements (including mortgages with respect to any Real Property owned in fee of such new Subsidiary with a fair market value greater than \$1,000,000), as well as appropriate financing statements (and with respect to all property subject to a mortgage, fixture filings), all in form and substance reasonably satisfactory to Agent (including being sufficient to grant Agent a first priority Lien (subject to Permitted Liens) in and to the assets of such newly formed or acquired Subsidiary); provided, that the

joinder to the Guaranty and Security Agreement, and such other security agreements shall not be required to be provided to Agent with respect to any Subsidiary of a Loan Party that is a CFC or an Excluded Domestic Subsidiary if providing such agreements would result in material adverse tax consequences to the Loan Parties or the costs of the Loan Parties of providing such guaranty or such security agreement are unreasonably excessive (as determined by Agent in consultation with Borrower) in relation to the benefits to Agent and Lenders of the security or guarantee thereby, (b) provide, or cause the applicable Loan Party to provide, to Agent a pledge agreement (or an addendum to the Guaranty and Security Agreement) and appropriate certificates and powers or financing statements, pledging all of the direct or beneficial ownership interest in such new Subsidiary in form and substance reasonably satisfactory to Agent; provided, that only 65% of the total outstanding voting Equity Interests of any first tier Subsidiary of any Loan Party that is a CFC or an Excluded Domestic Subsidiary (and none of the Equity Interests of any Subsidiary of such CFC or an Excluded Domestic Subsidiary) shall be required to be pledged if pledging a greater amount would result in material adverse tax consequences or the costs of the Loan Parties of providing such pledge are unreasonably excessive (as determined by Agent in consultation with Borrower) in relation to the benefits to Agent and Lenders of the pledge thereby (which pledge, if reasonably requested by Agent, shall be governed by the laws of the jurisdiction of such Subsidiary), and (c) provide to Agent all other documentation, including one or more opinions of counsel reasonably satisfactory to Agent, which, in its opinion, is appropriate with respect to the execution and delivery of the applicable documentation referred to above (including policies of title insurance or other documentation with respect to all Real Property owned in fee and subject to a Mortgage). Any document, agreement, or instrument executed or issued pursuant to this <u>Section 5.11</u> shall constitute a Loan Document.

5.12. Further Assurances. The Loan Parties will, at any time upon the reasonable request of Agent, execute or deliver to Agent any and all financing statements, fixture filings, security agreements, pledges, assignments, mortgages, deeds of trust, opinions of counsel, and all other documents (the "Additional Documents") that Agent may reasonably request in form and substance reasonably satisfactory to Agent, to create, perfect, and continue perfected or to better perfect Agent's Liens in all of the assets of Parent and its Subsidiaries (whether now owned or hereafter arising or acquired, tangible or intangible, real or personal but excluding assets that are expressly excluded from the Collateral pursuant to the terms of the Security Agreement), to create and perfect Liens in favor of Agent in any Real Property acquired by any Loan Party with a fair market value in excess of \$1,000,000, and in order to fully consummate all of the transactions contemplated hereby and under the other Loan Documents; provided that the foregoing shall not apply to any Subsidiary of a Loan Party that is a CFC or Excluded Domestic Subsidiary if providing such documents would result in material adverse tax consequences. To the maximum extent permitted by applicable law, if any Loan Party refuses or fails to execute or deliver any reasonably requested Additional Documents within a reasonable period of time following the request to do so, each Loan Party authorizes Agent to execute any such Additional Documents in the applicable Loan Party's or its Subsidiary's name, as applicable, and authorizes Agent to file such executed Additional Documents in any appropriate filing office. In furtherance of, and not in limitation of, the foregoing, each Loan Party shall take such actions as Agent may reasonably request from time to time to ensure that the Obligations are guarantied by the Guarantors and are secured by substantially all of the assets of Parent and its Subsidiaries, including all of the outstanding capital Equity Interests of Borrower and each Loan Party's Subsidiaries (subject to exceptions and limitations contained in the Loan Documents with respect to CFCs or Excluded Domestic Subsidiaries).

5.13. Lender Meetings. The Loan Parties will, within 120 days after the close of each fiscal year of Parent, at the request of Agent or of the Required Lenders and upon reasonable prior notice, hold a meeting (at a mutually agreeable location and time or, at the option of Agent, by conference call) with all Lenders who choose to attend such meeting at which meeting shall

be reviewed the financial results of the previous fiscal year and the financial condition of Parent and its Subsidiaries and the projections presented for the current fiscal year of Parent.

- 5.14. **Bank Products**. If on or before the 90th day after the Closing Date, the Loan Parties have failed to deliver to Agent duly executed Control Agreements with respect to each Deposit Account of the Loan Parties maintained with Wells Fargo, the Loan Parties shall, within 150 days of the Closing Date, (a) close all such accounts with Wells Fargo that are not Excluded Deposit and Securities Accounts (as defined in the Guaranty and Security Agreement), and provide evidence thereof to Agent in form reasonably satisfactory to Agent, and (b) establish their primary depository and treasury management relationships with PNC or one or more of its Affiliates and will maintain such depository and treasury management relationships with PNC or one or more of its Affiliates, the Loan Parties shall use commercially reasonable efforts to so establish their primary depository and treasury management relationships with PNC prior to the 135th day after the Closing Date.
- 5.15. Certificate of Beneficial Ownership and Other Additional Information. The applicable Loan Parties will provide to Agent and the Lenders: (i) promptly, upon request, confirmation of the accuracy of the information set forth in the most recent Certificate of Beneficial Ownership provided to the Agent and Lenders; (ii) (x) promptly, upon request and (y) concurrently with the monthly deliveries required under Schedule 5.1, a new Certificate of Beneficial Ownership, in form and substance reasonably acceptable to Agent and each Lender, when the individual(s) to be identified as a Beneficial Owner have changed; and (iii) such other information and documentation as may reasonably be requested by Agent or any Lender from time to time for purposes of compliance by Agent or such Lender with applicable laws (including without limitation the Patriot Act and other "know your customer" and anti-money laundering rules and regulations), and any policy or procedure implemented by Agent or such Lender to comply therewith.
- 5.16. <u>Amendments to SpeechIQ Acquisition Documentation or Engage Acquisition Documentation</u>. Promptly following execution thereof, copies of any amendment, supplement, waiver or other modification of the SpeechIQ Acquisition Documentation or the Engage Acquisition Documentation.
- 5.17. <u>Amendments to Merger Documentation</u>. Promptly following execution thereof, copies of any amendment, supplement, waiver or other modification of the Merger Agreement, each Support Agreement (as defined in the Merger Agreement) and the Sponsor Support Agreement (as defined in the Merger Agreement).
 - 5.18. **Engage PPP Loan**. The Loan Parties hereby covenant and agree that:
 - (a) the Engage PPP Loan shall not be secured by any Collateral;
 - (b) the Engage PPP Loan shall not be guaranteed by any Loan Party;
- (c) to the knowledge of the Loan Parties, the Engage Guarantor shall have used the proceeds of the Engage PPP Loan solely for purposes permitted under the PPP Program and all regulations promulgated thereunder and all other applicable laws and shall not have used the proceeds of the Engage PPP Loan for any purpose prohibited under the PPP Program or any other applicable law; and

(d) the Loan Parties shall provide evidence that the Engage Sellers have established and entered into the Engage PPP Loan Escrow Agreement.

6. NEGATIVE COVENANTS.

Each Loan Party covenants and agrees that, until termination of all of the Commitments and payment in full of the Obligations:

- 6.1. <u>Indebtedness</u>. The Loan Parties will not, and will not permit any of their Subsidiaries to create, incur, assume, suffer to exist, guarantee, or otherwise become or remain, directly or indirectly, liable with respect to any Indebtedness, except for Permitted Indebtedness.
- 6.2. <u>Liens</u>. The Loan Parties will not, and will not permit any of their Subsidiaries to create, incur, assume, or suffer to exist, directly or indirectly, any Lien on or with respect to any of its assets, of any kind, whether now owned or hereafter acquired, or any income or profits therefrom, except for Permitted Liens.
 - 6.3. **Restrictions on Fundamental Changes.** The Loan Parties will not, and will not permit any of their Subsidiaries to,
- (a) other than in order to consummate a Permitted Acquisition, enter into any merger, consolidation, reorganization, or recapitalization, or reclassify its Equity Interests, except for (i) any merger between Loan Parties, provided, that a Domestic Subsidiary (other than an Excluded Domestic Subsidiary) must be the surviving entity of any such merger to which Borrower is a party, and shall provide Agent at least 20 Business Days' notice of such merger and execute all documentation reasonably requested by Agent to evidence such Person's assumption of Borrower's Obligations, and no merger may occur between Parent and another Loan Party, (ii) any merger between a Loan Party and a Subsidiary of such Loan Party that is not a Loan Party so long as such Loan Party is the surviving entity of any such merger, and (iii) any merger between Subsidiaries of Parent that are not Loan Parties,
- (b) liquidate, wind up, or dissolve itself (or suffer any liquidation or dissolution), except for (i) the liquidation or dissolution of non-operating Subsidiaries of Loan Parties with nominal assets and nominal liabilities, (ii) the liquidation or dissolution of a Loan Party (other than Parent or Borrower) or any of its wholly-owned Subsidiaries so long as all of the assets (including any interest in any Equity Interests) of such liquidating or dissolving Loan Party or Subsidiary are transferred to a Loan Party that is not liquidating or dissolving, or (iii) the liquidation or dissolution of a Subsidiary of Parent that is not a Loan Party (other than any such Subsidiary the Equity Interests of which (or any portion thereof) is subject to a Lien in favor of Agent) so long as all of the assets of such liquidating or dissolving Subsidiary are transferred to a Subsidiary of Parent that is not liquidating or dissolving, or
- (c) suspend or cease operating a substantial portion of its or their business, except as permitted pursuant to clauses (a) or (b) above or in connection with a transaction permitted under <u>Section 6.4</u>.
- 6.4. <u>Disposal of Assets</u>. Other than Permitted Dispositions or transactions expressly permitted by <u>Sections 6.3</u> or <u>6.9</u>, the Loan Parties will not, and will not permit any of their Subsidiaries to convey, sell, lease, license, assign, transfer, or otherwise dispose of (or enter into an agreement to convey, sell, lease, license, assign, transfer, or otherwise dispose of) any of their assets.

- 6.5. <u>Nature of Business</u>. The Loan Parties will not, and will not permit any of their Subsidiaries to make any change in the nature of its or their principal business as described in <u>Schedule 6.5</u> or acquire any properties or assets that are not reasonably related to the conduct of such business activities; <u>provided</u>, that the foregoing shall not prevent Parent and its Subsidiaries from engaging in any business that is reasonably related, complementary or ancillary to its or their business.
 - 6.6. **Prepayments and Amendments.** The Loan Parties will not, and will not permit any of their Subsidiaries to,
 - (a) Except in connection with Refinancing Indebtedness permitted by Section 6.1,
- (i) optionally prepay, redeem, defease, purchase, or otherwise acquire any Indebtedness of Parent or its Subsidiaries, other than (A) the Obligations in accordance with this Agreement, (B) Permitted Intercompany Advances, (C) with respect to any Permitted Disposition, the amount of any Permitted Indebtedness (not to exceed for Parent and its Subsidiaries, with amounts prepaid pursuant to clause (E) below \$750,000 in the aggregate of any fiscal year) secured by any Permitted Lien on the asset subject to such Permitted Disposition that is required to be, and is, repaid in connection with such Permitted Dispositions, (D) Indebtedness that is subordinated in right of payment to the Obligations to the extent the prepayment is expressly permitted at the time under the terms of the applicable subordination agreement pursuant to which Agent is party or the prepayment is funded solely with the identifiable proceeds of a Qualified Equity Contribution substantially contemporaneously with the prepayment, (E) Capital Lease Obligations (not to exceed for Parent and its Subsidiaries with amounts prepaid pursuant to clause (C) above, \$750,000 in its aggregate in any fiscal year), (F) solely with respect to the (i) the SpeechIQ Incentive Payment, (ii) the SpeechIQ Holdback, (iii) the Teckst Incentive Payment, (iv) the Teckst Holdback or (v) the Engage Earn-Out, payments to the extent permitted by clause (ii) or (iii) below or (G) the Engage PPP Loan so long as the payment is funded using the proceeds of the escrow arrangement in effect on the Sixth Amendment Effective Date;
- (ii) make any payment on account of Indebtedness that has been contractually subordinated in right of payment to the Obligations if such payment is not permitted at such time under the subordination terms and conditions; or
- (iii) make any payment on account of (i) the SpeechIQ Incentive Payment, (ii) the SpeechIQ Holdback, (iii) the Teckst Incentive Payment, (iv) the Teckst Holdback or (v) the Engage Earn-Out, unless, in each case, (A) no Default or Event of Default shall have occurred and be continuing or would result therefrom and, in the case of any payment on account of subclauses (i) through (v), (B) immediately prior and immediately after giving effect to each such payment, Liquidity was and would remain \$2,000,000 or more; provided, that any payments made pursuant to this clause (iii) may be made without regard to the restrictions set forth in sub-clauses (A) and (B) so long as any such payments are funded with (x) cash proceeds of direct or indirect common equity contributions in Borrower by Sponsor which are substantially simultaneously applied to make such payment or (y) any issuance of equity interests that are not otherwise prohibited under this Agreement, in the case of clause (y) only, in form and substance reasonably satisfactory to Agent (Agent hereby acknowledges and agrees that the proceeds of the transactions contemplated by the Merger Documentation will satisfy this clause (y). If payments in connection with this clause (iii) are restricted as a result of the occurrence of the events described in sub-clauses (A) or (B), or otherwise deferred in accordance with the SpeechIQ Acquisition Agreement, the Teckst Acquisition Agreement or the Engage Earn-Out, as applicable, such payments shall be deferred and any payments made in connection therewith

shall be permitted solely when the occurrence of the events in sub-clauses (A) or (B) have been cured or waived by Agent and Lenders.

- (b) Directly or indirectly, amend, modify, or change any of the terms or provisions of
- (i) any agreement, instrument, document, indenture, or other writing evidencing or concerning Permitted Indebtedness in any manner materially adverse to the interest of Agent or any Lender or which, by virtue of such amendment, modification or change would render such otherwise Permitted Indebtedness prohibited under Section 6.1 of this Agreement, other than (A) the Obligations in accordance with this Agreement, (B) Permitted Intercompany Advances, and (C) Indebtedness permitted under clauses (c), (f), (h), (j) and (k) of the definition of Permitted Indebtedness,
- (ii) the Advisory Agreement in any manner that would be adverse to the interests of the Agent or the Lenders; provided, however, in connection with the consummation of an IPO, the Advisory Agreement may be amended to remove the requirement to pay the annual management fee and instead pay the Permitted Term Out Fee upon the occurrence of such IPO,
- (iii) any Material Contract (other than the Advisory Agreement) except to the extent that such amendment, modification or change could not, individually or in the aggregate, reasonably be expected to be materially adverse to the interests of the Lenders, or
- (iv) the Governing Documents of any Loan Party or any of its Subsidiaries if the effect thereof, either individually or in the aggregate, could reasonably be expected to be materially adverse to the interests of the Lenders.
- 6.7. **Restricted Payments**. The Loan Parties will not, and will not permit any of their Subsidiaries or Parent to make any Restricted Payment; <u>provided</u>, that, so long as it is permitted by law, and so long as no Default or Event of Default shall have occurred and be continuing or would result therefrom,
- (a) Parent may make distributions to former employees, officers, or directors of Parent (or any spouses, ex-spouses, or estates of any of the foregoing) on account of repurchases or redemptions of Equity Interests of Parent held by such Persons; provided, that the aggregate amount of all such repurchases and such redemptions made by Parent from and after the Seventh Amendment Effective Date plus the amount of Indebtedness outstanding under clause (l) of the definition of Permitted Indebtedness, does not exceed \$3,000,000 in the aggregate, and (ii) immediately prior and immediately after giving effect to such distribution and repurchase, Liquidity was and would remain \$5,000,000 or more,
- (b) Parent may make distributions to former employees, officers, or directors of Parent (or any spouses, ex-spouses, or estates of any of the foregoing), solely in the form of forgiveness of Indebtedness of such Persons owing to Parent on account of repurchases of the Equity Interests of Parent held by such Persons; <u>provided</u> that such Indebtedness was incurred by such Persons solely to acquire Equity Interests of Parent, and
- (c) the Borrower and any of its Subsidiaries may make <u>Dividends dividends</u>, <u>distributions or payments up</u> to Parent (which Parent may pay to any direct or indirect parent company of Parent so long as such entity is a member of the consolidated, combined or similar tax group of Borrower and is liable under applicable law for taxes imposed on the income of Borrower) to permit Parent (or any such direct or indirect parent company) to pay, for any

taxable period for which Borrower and its Subsidiaries are members of a consolidated, combined or similar income tax group for U.S. federal and/or applicable state or local income tax purposes (a "<u>Tax Group</u>") of which Parent (or any direct or indirect parent company of Parent) is the common parent, any consolidated, combined or similar income Taxes of such Tax Group that are due and payable by Parent (or such direct or indirect parent company of Parent) for such taxable period, but only to the extent attributable to the taxable income of Borrower and its Subsidiaries.

- 6.8. <u>Accounting Methods</u>. The Loan Parties will not, and will not permit any of their Subsidiaries to modify or change its fiscal year or its method of accounting (other than as may be required to conform to GAAP).
- 6.9. <u>Investments</u>. The Loan Parties will not, and will not permit any of their Subsidiaries or Parent to, directly or indirectly, make or acquire any Investment or incur any liabilities (including contingent obligations) for or in connection with any Investment except for Permitted Investments.
- 6.10. <u>Transactions with Affiliates</u>. The Loan Parties will not, and will not permit any of their Subsidiaries to, directly or indirectly, enter into or permit to exist any transaction with any Affiliate of Parent or any of its Subsidiaries except for:
- (a) transactions (other than the payment of management, consulting, monitoring, or advisory fees) between Parent or its Subsidiaries, on the one hand, and any Affiliate of Parent or its Subsidiaries, on the other hand, so long as such transactions (i) are fully disclosed to Agent and each Lender prior to the consummation thereof, if they involve one or more payments by Parent or its Subsidiaries in excess of \$750,000 for any single transaction or series of related transactions, and (ii) are no less favorable, taken as a whole, to Parent or its Subsidiaries, as applicable, than would be obtained in an arm's length transaction with a non-Affiliate,
- (b) so long as it has been approved by Parent's or its applicable Subsidiary's board of directors (or comparable governing body) in accordance with applicable law, any indemnity provided for the benefit of executive officers and directors (or equivalent position with respect to Parent's Subsidiaries) of Parent or the applicable Subsidiary,
- (c) so long as it has been approved by the applicable Loan Party's or their applicable Subsidiary's board of directors (or comparable governing body) in accordance with applicable law, the payment of reasonable fees, compensation, or employee benefit arrangements to employees, officers, and outside directors of the Loan Parties and their Subsidiaries in the ordinary course of business and consistent with industry practice,
 - (d) transactions permitted by Section 6.3 or Section 6.7, or any Permitted Intercompany Advance,
- (e) the payment, pursuant to the Advisory Agreement, of (i) management, consulting, monitoring, and advisory fees to Sponsor or its Affiliates in an aggregate amount not to exceed the amount set forth in the Advisory Agreement as in effect on the Closing Date in any calendar year of Parent, so long as (x) no Default or Event of Default has occurred and is continuing or would result therefrom, and (y) after taking into account all such payments to be made on any date, the Loan Parties would have Liquidity of at least \$2,500,000; provided, that if at any time any such management, consulting, monitoring or advisory fees to Sponsor or its Affiliates are not permitted to be paid as a result of the failure to satisfy either of the conditions set forth in subclauses (x) or (y) of this Section 6.10(e), then (1) such amounts shall continue to accrue, and (2) any such amounts that have accrued but which were not permitted to be paid may

be paid in any subsequent quarter so long as each of the conditions set forth in subclauses (x) or (y) of this Section 6.10(e) are satisfied at the time of the making of such payments, and such accrued amount paid does not exceed the amount permitted in clause (i) above in any calendar year, and (ii) reasonable out-of-pocket expenses of Sponsor or its Affiliates,

- (f) the payment pursuant to the Advisory Agreement of transaction fees charged by Sponsor for providing services to Parent and its Subsidiaries in connection with a Permitted Acquisition so long as (i) such transaction fees are paid at or substantially concurrent with the closing of such Permitted Acquisition, (ii) no Default or Event of Default has occurred and is continuing or would result therefrom, (iii) after taking into account all such payments to be made on any date, the Loan Parties would have Liquidity of at least \$2,500,000, and (iv) such transaction fees do not exceed an amount equal to the amount required pursuant to the Advisory Agreement as in effect on the Closing Date for any Permitted Acquisition (provided that such amount may be increased by an amount equal to the proceeds of any Qualified Equity Contribution substantially contemporaneously with the closing of such Permitted Acquisition which are not used to finance a portion of the Purchase Price for such Permitted Acquisition),
- (g) if the Advisory Agreement is amended in accordance with the terms of this Agreement to require payment of the Permitted Term-Out Fee, payment of the Permitted Term Out Fee as required by the Advisory Agreement,
- (h) the following transactions among the Loan Parties and their Subsidiaries, in each case so long as they are, in the ordinary course of business, no less favorable to a Loan Party than would be obtained in an arm's length transaction, and in conformance with the legal requirement of the various tax laws of the jurisdictions in which the Loan Parties and their Subsidiaries operate: (i) transfer pricing charges among the Loan Parties and their Subsidiaries in respect of the sale of software products, hardware products, and the provision of related professional services to end users; and (ii) internal research and development services, maintenance support services, back-office charges and matters ancillary thereto,
- (i) reasonable out-of-pocket expenses (including in connection with service on the board or directors of Parent) pursuant to any financial advisory, financing, underwriting, or placement agreement or in respect of other investment banking activities, including in connection with acquisitions or divestitures that are permitted by this Agreement, or
- (j) reasonable out-of-pocket expenses with respect to services provided by Sponsor in relation to Permitted Acquisitions following the Closing Date;
- (k) agreements for the non-exclusive licensing of intellectual property among Parent and its Subsidiaries for the purpose of the licensee thereof operating its business (and not for the purpose of sale or licensing to third parties), and agreements for the assignment of intellectual property from Parent or any of its Subsidiaries to any Loan Party;
- (l) agreements or arrangements with such Person (or any of its Subsidiaries) identified in writing by Borrower to Agent and Lenders prior to the Closing Date with respect to co-marketing and cross marketing of software or similar products and sharing of services (which may include, but is not limited to, sharing telecom and data center infrastructure, financial reporting and other back office functions) so long as collectively such arrangements are reasonably expected to result in a net increase of revenue or a net decrease of expenses to Parent and its Subsidiaries (taken as a whole); and
 - (m) transactions among Loan Parties not otherwise prohibited pursuant to the terms of this Agreement.

Notwithstanding <u>Section 6.7</u> hereof, to the extent any of the payments that are permitted to be made pursuant to this <u>Section 6.10</u> are in respect of obligations of Parent, any Subsidiary of Parent shall be permitted to make a distribution to Parent in the amount of any such obligation so as to provide Parent with a source of funding for such payment.

- 6.11. Use of Proceeds. The Loan Parties will not, and will not permit any of their Subsidiaries to use the proceeds of any Loan made hereunder for any purpose other than (a) with respect to the Closing Date Term Loan on the Closing Date, (i) to repay, in full, the outstanding principal, accrued interest, and accrued fees and expenses owing to the Existing Lenders, and (ii) to pay the fees, costs, and expenses incurred in connection with this Agreement, the other Loan Documents, and the transactions contemplated hereby and thereby, in each case, as set forth in the Closing Date Disbursement Letter, (b) with respect to the Third Amendment Term Loan on the Third Amendment Closing Date, to pay the fees, costs and expenses incurred in connection with the SpeechIQ Acquisition Documentation, and the transactions contemplated thereby, as set forth in the Third Amendment Disbursement Letter and (c) at all other times, consistent with the terms and conditions hereof, with respect to all Loans for their lawful and permitted purposes (including that no part of the proceeds of the Loans made to Borrower will be used in violation of any applicable law or to purchase or carry any Margin Stock or to extend credit to others for the purpose of purchasing or carrying any such Margin Stock or for any purpose that violates the provisions of Regulation T, U or X of the Board of Governors). The Borrower will not request any Loan, and no Loan Party shall use, and each shall procure that their Subsidiaries and the respective directors, officers, employees and agents of each of the foregoing shall not use, the proceeds of any Loan directly or, to the knowledge of any Loan Party, indirectly (x) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (y) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person or in any Sanctioned Country, to the extent such activities, business or transaction would be prohibited by Sanctions if conducted by a corporation incorporated in the United States or a European Union member state or (z) in any manner that would immediately (after the expiration of any notice, cure or similar period) result in the violation of any Sanctions applicable to any party hereto.
- 6.12. <u>Limitation on Issuance of Equity Interests</u>. Except for the issuance or sale of Qualified Equity Interests by Parent or Equity Interests of any Subsidiary of Parent to Parent or another Subsidiary of Parent, the Loan Parties will not, and will not permit any of their Subsidiaries to issue or sell or enter into any agreement or arrangement for the issuance or sale of any of its Equity Interests.
- 6.13. Parent as Holding Company. Parent shall not incur any liabilities (other than liabilities arising under the Loan Documents, the Advisory Agreement and in connection with its acquisition of the Borrower), own or acquire any assets (other than the Equity Interests of Borrower and its other Subsidiaries) or engage itself in any operations or business, except in connection with its ownership of Borrower and its other Subsidiaries and its rights and obligations under the Loan Documents and activities incidental to the direct and indirect ownership of its Subsidiaries, activities incidental to the maintenance of its corporate existence, activities related to any debt and equity financing of its acquisition of the Borrower and the consummation of the transactions related thereto.
- 6.14. **OFAC; Patriot Act**. No Loan Party shall, and no Loan Party shall permit any of its Subsidiaries to fail to comply with the laws, regulations and executive orders referred to in <u>Section 4.19</u> hereof.

- 6.15. <u>Amendments to SpeechIQ Acquisition Documentation or Engage Acquisition Documentation</u>. Borrower shall not amend, supplement or otherwise modify (pursuant to a waiver or otherwise) the terms and conditions of the SpeechIQ Acquisition Documentation or the Engage Acquisition Agreement, in each case, in a manner materially adverse to the interests of the Lenders.
- 6.16. <u>Amendments to Management Incentive Documentation</u>. The Loan Parties will not, and will not permit any of their Subsidiaries or Parent to, amend, supplement or otherwise modify (pursuant to a waiver or otherwise) the terms and conditions of the Management Incentive Documentation in a manner materially adverse to the interests of the Lender Group.
- 6.17. <u>Amendments to Merger Documentation</u>. The Loan Parties will not, and will not permit any of their Subsidiaries or Parent to, amend, supplement or otherwise modify (pursuant to a waiver or otherwise) the terms and conditions of the Merger Documentation in a manner materially adverse to the interests of the Lender Group without the prior written consent of the Lender Group (such consent not to be unreasonably withheld, conditioned or delayed).

7. FINANCIAL COVENANTS.

Each of Loan Party covenants and agrees that, until termination of all of the Commitments and payment in full of the Obligations, the Loan Parties will:

Minimum Recurring Revenues. With respect to each measurement period specified below, achieve Recurring Revenues, measured on a quarter-end basis, of at least the required amount set forth in the following table for the applicable four-quarter period ending on the date set forth opposite thereto:

Applicable Amount	Applicable Period End
\$86,010,000	March 31, 2020
\$87,408,000	June 30, 2020
\$88,300,000	September 30, 2020
\$91,703,000	December 31, 2020
\$95,032,000	March 31, 2021
\$98,653,000	June 30, 2021
\$102,635,000	September 30, 2021
\$106,436,000	December 31, 2021
\$111,184,000	March 31, 2022
\$117,825,000	June 30, 2022

\$124,635,000	September 30, 2022
\$130,983,000	December 31, 2022
\$ 138,262,000 126,960,000	March 31, 2023
\$ 145,986,000 129,586,000	June 30, 2023
\$ 153,907,000 <u>131,259,000</u>	September 30, 2023
\$ 162,419,000 <u>132,480,000</u>	December 31, 2023
\$ 169,940,000 <u>134,786,000</u>	March 31, 2024
\$ 177,922,000 <u>137,522,000</u>	June 30, 2024
\$ 179,128,000 140,362,000	September 30, 2024
\$ 182,721,000 <u>142,817,000</u>	December 31, 2024
\$ 191,183,000 145,267,000	March 31, 2025
\$ 200,162,000 147,721,000	June 30, 2025
\$ 209,370,000 150,419,000	September 30, 2025
\$ 219,265,000 153,011,000	December 31, 2025 and the last day of each calendar quarter thereafter

(b) **Minimum Liquidity**. At all times maintain Liquidity of at least \$2,000,000.

8. EVENTS OF DEFAULT.

Any one or more of the following events shall constitute an event of default (each, an "Event of Default") under this Agreement:

8.1. Payments. If Borrower fails to pay when due and payable, or when declared due and payable, (a) all or any portion of the Obligations consisting of interest, fees, or charges due the Lender Group, reimbursement of Lender Group Expenses, or other amounts (other than any portion thereof constituting principal) constituting Obligations (including any portion thereof that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), and such failure continues for a period of 3 Business Days, (b) all or any portion of the principal of the Loans, or (c) any amount payable to Issuing Bank in reimbursement of any drawing under a Letter of Credit;

8.2. **Covenants**. If any Loan Party or any of its Subsidiaries:

- (a) fails to perform or observe any covenant or other agreement contained in any of (i) Sections 3.6, 5.3 (solely if a Loan Party is not in good standing in its jurisdiction of organization), 5.6, 5.7 (solely if a Loan Party refuses to allow Agent or its representatives or agents to visit any Loan Party's properties, inspect its assets or books or records, examine and make copies of its books and records, or discuss such Loan Party's affairs, finances, and accounts with officers and employees of such Loan Party), 5.10, 5.11, 5.13 or 5.14 of this Agreement, (ii) Section 6 of this Agreement, (iii) Section 7 of the Guaranty and Security Agreement;
- (b) fails to perform or observe any covenant or other agreement contained in any of <u>Sections 5.1</u> or <u>5.2</u> of this Agreement and such failure continues for a period of 5 days after the date such covenant was to have been performed or observed;
- (c) fails to perform or observe any covenant or other agreement contained in any of <u>Sections 5.3</u> (other than if a Loan Party is not in good standing in its jurisdiction of organization), <u>5.4</u>, <u>5.5</u>, <u>5.8</u>, and <u>5.12</u> of this Agreement and such failure continues for a period of 10 days after the earlier of (i) the date on which such failure shall first become known to any officer of a Loan Party or (ii) the date on which written notice thereof is given to Borrower or any other applicable Loan Party by Agent; or
- (d) fails to perform or observe any covenant or other agreement contained in this Agreement, or in any of the other Loan Documents, in each case, other than any such covenant or agreement that is the subject of another provision of this Section 8 (in which event such other provision of this Section 8 shall govern), and such failure continues for a period of 30 days after the earlier of (i) the date on which such failure shall first become known to any officer of a Loan Party or (ii) the date on which written notice thereof is given to Borrower or any other applicable Loan Party by Agent;
- 8.3. <u>Judgments</u>. If one or more judgments, orders, or awards for the payment of money involving an aggregate amount of \$1,000,000, or more (except to the extent fully covered (other than to the extent of customary deductibles) by insurance pursuant to which the insurer has not denied coverage) is entered or filed against a Loan Party or any of its Subsidiaries, or with respect to any of their respective assets, and either (a) there is a period of 30 consecutive days at any time after the entry of any such judgment, order, or award during which (1) the same is not discharged, satisfied, vacated, or bonded pending appeal, or (2) a stay of enforcement thereof is not in effect, or (b) enforcement proceedings are commenced upon such judgment, order, or award;
 - 8.4. Voluntary Bankruptcy, etc. If an Insolvency Proceeding is commenced by a Loan Party or any of its Subsidiaries;
- 8.5. Involuntary Bankruptcy, etc. If an Insolvency Proceeding is commenced against a Loan Party or any of its Subsidiaries and any of the following events occur: (a) such Loan Party or such Subsidiary consents to the institution of such Insolvency Proceeding against it, (b) the petition commencing the Insolvency Proceeding is not dismissed within 60 calendar days of the date of the filing thereof, (d) an interim trustee is appointed to take possession of all or any substantial portion of the properties or assets of, or to operate all or any substantial portion of the business of, such Loan Party or its Subsidiary, or (e) an order for relief shall have been issued or entered therein;

- 8.6. **Default Under Other Agreements**. If there is (a) a default in one or more agreements to which a Loan Party or any of its Subsidiaries is a party with one or more third Persons relative to a Loan Party's or any of its Subsidiaries' Indebtedness involving an aggregate amount of \$1,500,000 or more, and such default (i) occurs at the final maturity of the obligations thereunder, or (ii) results in a right by such third Person, after the giving of any required notices and the termination of any grace periods irrespective of whether exercised, to accelerate the maturity of such Loan Party's or its Subsidiary's obligations thereunder, (b) a default under or an involuntary early termination of one or more Bank Product Agreements of which a Loan Party or any of its Subsidiaries is a party, or (c) any default by any Loan Party under any Material Contract, which default continues for more than the applicable cure period, if any, with respect thereto and is not timely waived in writing by the other parties thereto;
- 8.7. Representations, etc. If any warranty, representation, certificate, statement, or Record made herein or in any other Loan Document or delivered in writing to Agent or any Lender in connection with this Agreement or any other Loan Document proves to be untrue in any material respect (except that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof) as of the date of issuance or making or deemed making thereof;
- 8.8. **Guaranty**. If the obligation of any Guarantor under the guaranty contained in the Guaranty and Security Agreement is limited or terminated by operation of law or by such Guarantor (other than in accordance with the terms of this Agreement);
- 8.9. Security Documents. If the Guaranty and Security Agreement or any other Loan Document that purports to create a Lien, shall, for any reason, fail or cease to create a valid and perfected and, except to the extent of Permitted Liens which are non-consensual Permitted Liens, permitted purchase money Liens or the interests of lessors under Capital Leases (subject to any specific exceptions to perfection set forth in this Agreement and the Security Agreement), first priority Lien on the Collateral covered thereby, except (a) as a result of a disposition of the applicable Collateral in a transaction permitted under this Agreement, or (b) as the result of an action or failure to act on the part of Agent;
- 8.10. <u>Loan Documents</u>. The validity or enforceability of any Loan Document shall at any time for any reason (other than solely as the result of an action or failure to act on the part of Agent) be declared to be null and void, or a proceeding shall be commenced by a Loan Party or its Subsidiaries, or by any Governmental <u>AuthorityBody</u> having jurisdiction over a Loan Party or its Subsidiaries, seeking to establish the invalidity or unenforceability thereof, or a Loan Party or its Subsidiaries shall deny that such Loan Party or its Subsidiaries has any liability or obligation purported to be created under any Loan Document;
 - 8.11. Change of Control. A Change of Control shall occur, whether directly or indirectly; or
- 8.12. <u>Seizures</u>. Any: (a) portion of the Collateral having a value in excess of \$1,000,000 (except to the extent fully covered (other than to the extent of customary deductibles) by insurance pursuant to which the insurer has not denied coverage or to the extent compensated by such Governmental <u>AuthorityBody</u> at the time of such action or pursuant to which such Governmental <u>AuthorityBody</u> has confirmed in writing that it shall provide such compensation) shall be seized, subject to garnishment or taken by a Governmental <u>AuthorityBody</u> and such seizure, garnishment or taking is not reversed or vacated within thirty (30) days thereafter; or (b) the title and rights of any Loan Party with respect to Collateral having a value in excess of \$1,000,000 (except to the extent fully covered (other than to the extent of customary deductibles) by insurance pursuant to which the insurer has not denied coverage or to the extent compensated

by such Governmental AuthorityBody at the time of such action or pursuant to which such Governmental AuthorityBody has confirmed in writing that it shall provide such compensation) shall have become the subject matter of claim, litigation, suit, garnishment or other proceeding taken by a Governmental AuthorityBody for the seizure of such Collateral, whether pursuant to the indictment of any Loan Party under any criminal statute or otherwise, and either (i) there is a period of thirty (30) consecutive days at any time after the entry of a judgment or order with respect thereto which (A) the same is not discharged, satisfied, vacated, or bonded pending appeal, or (B) a stay of enforcement thereof is not in effect, or (ii) enforcement proceedings are commenced upon such judgment or order.

9. RIGHTS AND REMEDIES.

- 9.1. **Rights and Remedies.** Upon the occurrence and during the continuation of an Event of Default, Agent may, and, at the instruction of the Required Lenders, shall (in each case under clauses (a) or (b) by written notice to Borrower), in addition to any other rights or remedies provided for hereunder or under any other Loan Document or by applicable law, do any one or more of the following:
- (a) (i) declare the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents to be immediately due and payable, whereupon the same shall become and be immediately due and payable and Borrower shall be obligated to repay all of such Obligations in full, without presentment, demand, protest, or further notice or other requirements of any kind, all of which are hereby expressly waived by Borrower, and (ii) direct Borrower to provide (and Borrower agrees that upon receipt of such notice it will provide) Letter of Credit Collateralization to Agent to be held as security for Borrower's reimbursement obligations for drawings that may subsequently occur under issued and outstanding Letters of Credit;
- (b) declare the Commitments terminated, whereupon the Commitments shall immediately be terminated together with (i) any obligation of any Revolving Lender to make Revolving Loans, (ii) the obligation of the Swing Loan Lender to make Swing Loans, and (iii) the obligation of Issuing Bank to issue Letters of Credit; and
- (c) exercise all other rights and remedies available to Agent or Lenders under the Loan Documents, under applicable law, or in equity; provided, that, with respect to any Event of Default resulting solely from failure of Borrower to comply with the financial covenants as set forth in Section 7, neither Agent nor Required Lenders may exercise the foregoing remedies in this Section 9.1 until the date that is 16 Business Days after the earlier of (A) the date on which the Compliance Certificate is required to be delivered for the applicable fiscal quarter or month pursuant to Section 5.1 and (B) the date on which such Compliance Certificate is actually delivered (it being understood and agreed that no Lender shall have any obligation to make Revolving Loans or otherwise extend credit to Borrower during such 16 Business Day period).

The foregoing to the contrary notwithstanding, upon the occurrence of any Event of Default described in Section 8.4 or Section 8.5, in addition to the remedies set forth above, without any notice to Borrower or any other Person or any act by the Lender Group, the Commitments shall automatically terminate and the Obligations (other than the Bank Product Obligations), inclusive of the principal of, and any and all accrued and unpaid interest and fees in respect of, the Loans and all other Obligations (other than the Bank Product Obligations), whether evidenced by this Agreement or by any of the other Loan Documents, shall automatically become and be immediately due and payable and Borrower shall automatically be obligated to repay all of such Obligations in full (including Borrower being obligated to provide (and Borrower agrees that it

will provide) (1) Letter of Credit Collateralization to Agent to be held as security for Borrower's reimbursement obligations in respect of drawings that may subsequently occur under issued and outstanding Letters of Credit, and (2) Bank Product Collateralization to be held as security for Borrower's or its Subsidiaries' obligations in respect of outstanding Bank Products), without presentment, demand, protest, or notice or other requirements of any kind, all of which are expressly waived by Parent and Borrower.

9.2. **Remedies Cumulative.** The rights and remedies of the Lender Group under this Agreement, the other Loan Documents, and all other agreements shall be cumulative. The Lender Group shall have all other rights and remedies not inconsistent herewith as provided under the Code, by law, or in equity. No exercise by the Lender Group of one right or remedy shall be deemed an election, and no waiver by the Lender Group of any Event of Default shall be deemed a continuing waiver. No delay by the Lender Group shall constitute a waiver, election, or acquiescence by it.

9.3. [Reserved].

10. WAIVERS; INDEMNIFICATION.

- 10.1. **Demand; Protest; etc.** Borrower waives demand, protest, notice of protest, notice of default or dishonor, notice of payment and nonpayment, nonpayment at maturity, release, compromise, settlement, extension, or renewal of documents, instruments, chattel paper, and guarantees at any time held by the Lender Group on which Borrower may in any way be liable.
- 10.2. The Lender Group's Liability for Collateral. Borrower hereby agrees that: (a) so long as Agent complies with its obligations, if any, under the Code, the Lender Group shall not in any way or manner be liable or responsible for: (i) the safekeeping of the Collateral, (ii) any loss or damage thereto occurring or arising in any manner or fashion from any cause, (iii) any diminution in the value thereof, or (iv) any act or default of any carrier, warehouseman, bailee, forwarding agency, or other Person, and (b) all risk of loss, damage, or destruction of the Collateral shall be borne by Borrower other than any thereof resulting from the gross negligence, bad faith, or willful misconduct of any Lender-Related Person as finally determined by a court of competent jurisdiction.
- 10.3. Indemnification. Each Loan Party shall pay, indemnify, defend, and hold the Agent-Related Persons, the Lender-Related Persons, and each Participant (each, an "Indemnified Person") harmless (to the fullest extent permitted by law) from and against any and all claims, demands, suits, actions, investigations, proceedings, liabilities, fines, costs, penalties, and damages (excluding with respect to disputes solely between the Lenders, but including disputes between Agent and any Lender), and all reasonable fees and disbursements of attorneys, experts, or consultants and all other reasonable costs and expenses actually incurred in connection therewith or in connection with the enforcement of this indemnification (as and when they are incurred and irrespective of whether suit is brought), at any time asserted against, imposed upon, or incurred by any of them (a) in connection with or as a result of or related to the execution and delivery (provided that Borrower shall not be liable for costs and expenses (including attorneys' fees) of any Lender (other than PNC) incurred in advising, structuring, drafting, reviewing, administering or syndicating the Loan Documents), enforcement, performance, or administration (including any restructuring or workout with respect hereto) of this Agreement, any of the other Loan Documents, or the transactions contemplated hereby or thereby or the monitoring of Parent's and its Subsidiaries' compliance with the terms of the Loan Documents (provided, that the indemnification in this clause (a) shall not extend to (i) disputes solely between or among the Lenders that do not involve any acts or omissions of any Loan Party, or (ii) disputes solely

between or among the Lenders and their respective Affiliates that do not involve any acts or omissions of any Loan Party; it being understood and agreed that the indemnification in this clause (a) shall extend to Agent (but not the Lenders) relative to disputes between or among Agent on the one hand, and one or more Lenders, or one or more of their Affiliates, on the other hand, or (iii) any Taxes or any costs attributable to Taxes, which shall be governed by Section 16), (b) with respect to any actual or prospective investigation, litigation, or proceeding related to this Agreement, any other Loan Document, the making of any Loans or issuance of any Letters of Credit hereunder, or the use of the proceeds of the Loans or the Letters of Credit provided hereunder (irrespective of whether any Indemnified Person is a party thereto), or any act, omission, event, or circumstance in any manner related thereto, including, without limitation, any threatened or actual imposition of fines or penalties, or disgorgement of benefits, for violation by any Loan Party or Affiliate thereof of any applicable law relating to anti-terrorism, money-laundering or international trade compliance, and (c) in connection with or arising out of any presence or release of Hazardous Materials at, on, under, to or from any assets or properties owned, leased or operated by Borrower or any of its Subsidiaries or any Environmental Actions, Environmental Liabilities or Remedial Actions related in any way to any such assets or properties of Borrower or any of its Subsidiaries (each and all of the foregoing, the "Indemnified Liabilities"). The foregoing to the contrary notwithstanding, Borrower shall have no obligation to any Indemnified Person under this Section 10.3 with respect to any Indemnified Liability that a court of competent jurisdiction finally determines to have resulted from the bad faith, gross negligence or willful misconduct of such Indemnified Person or its officers, directors, advisors, representatives, employees, attorneys, or agents. This provision shall survive the termination of this Agreement and the repayment in full of the Obligations. If any Indemnified Person makes any payment to any other Indemnified Person with respect to an Indemnified Liability as to which Borrower was required to indemnify the Indemnified Person receiving such payment, the Indemnified Person making such payment is entitled to be indemnified and reimbursed by Borrower with respect thereto. WITHOUT LIMITATION, THE FOREGOING INDEMNITY SHALL APPLY TO EACH INDEMNIFIED PERSON WITH RESPECT TO INDEMNIFIED LIABILITIES WHICH IN WHOLE OR IN PART ARE CAUSED BY OR ARISE OUT OF ANY NEGLIGENT ACT OR OMISSION OF SUCH INDEMNIFIED PERSON OR OF ANY OTHER PERSON EXCEPT TO THE EXTENT THAT SUCH ACT CONSTITUTES BAD FAITH, GROSS NEGLIGENCE OR WILLFUL MISCONDUCT.

11. NOTICES.

Unless otherwise provided in this Agreement, all notices or demands relating to this Agreement or any other Loan Document shall be in writing and (except for financial statements and other informational documents which may be sent by first-class mail, postage prepaid) shall be personally delivered or sent by registered or certified mail (postage prepaid, return receipt requested), overnight courier, electronic mail (at such email addresses as a party may designate in accordance herewith), or telefacsimile. In the case of notices or demands to Parent, Borrower or Agent, as the case may be, they shall be sent to the respective address set forth below:

If to Parent or Borrower:

LIVEVOX HOLDINGS, INC.

e/o Golden Gate Private Equity, Inc.

One Embarcadero Center, 39th Floor 655 Montgomery Street, Suite 1000.

San Francisco, California 94111

Attn: Stephen D. Oetgen and Greg Mason

Fax No. (415) 983-2701 Chief Financial Officer and General Counsel

Email: soctgen@goldengatecaplegal@livevox.com gmason@goldengatecapfinance@livevox.com

with copies to: KIRKI

KIRKLAND & ELLIS LLP 555 California Street

San Francisco WILSON SONSINI GOODRICH & ROSATI, P.C.

650 Page Mill Road

Palo Alto, California 9470894304 Attn: Samantha Good, Esq. Andy Hirsch Fax No. (415) 439-1500650.493.6811 Email: sgood@kirklandahirsch@wsgr.com

GOLDEN GATE PRIVATE EQUITY, INC.

One Embarcadero Center, 39th Floor San Francisco, California 94111

Attn: Stephen D. Oetgen and Greg Mason

Fax No. (415) 983-2701

Email: soetgen@goldengatecap.com gmason@goldengatecap.com

If to Agent: PNC BANK, NATIONAL ASSOCIATION

100 Pine Street, Suite 1500 San Francisco, California 94111 Attn: Christopher Duranto Fax No. (415) 733-1623

Email: christopher.duranto@pnc.com

with copies to: BLANK ROME, LLP

2029 Century Park East, 6th Floor Los Angeles, California 90067

Attn: Danielle V. GarciaBlank Rome LLP

1271 Avenue of the Americas New York, New York 10020 Attn: Michael Kim, Esq.

Fax No. (424917) 239-3394332-3780

Email: dgarcia Michael.kim ablankrome.com

Any party hereto may change the address at which they are to receive notices hereunder, by notice in writing in the foregoing manner given to the other party. All notices or demands sent in accordance with this <u>Section 11</u>, shall be deemed received on the earlier of the date of actual receipt or 3 Business Days after the deposit thereof in the mail; <u>provided</u>, that (a) notices sent by overnight courier service shall be deemed to have been given when received, (b) notices by facsimile shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next Business Day for the recipient) and (c) notices by electronic mail

shall be deemed received upon the sender's receipt of an acknowledgment from the intended recipient (such as by the "return receipt requested" function, as available, return email or other written acknowledgment).

12. CHOICE OF LAW AND VENUE; JURY TRIAL WAIVER; JUDICIAL REFERENCE PROVISION.

- (a) THE VALIDITY OF THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS (UNLESS EXPRESSLY PROVIDED TO THE CONTRARY IN ANOTHER LOAN DOCUMENT IN RESPECT OF SUCH OTHER LOAN DOCUMENT), THE CONSTRUCTION, INTERPRETATION, AND ENFORCEMENT HEREOF AND THEREOF, THE RIGHTS OF THE PARTIES HERETO AND THERETO WITH RESPECT TO ALL MATTERS ARISING HEREUNDER OR THEREUNDER OR THERETO, AND ANY CLAIMS, CONTROVERSIES OR DISPUTES ARISING HEREUNDER OR THEREUNDER OR RELATED HERETO OR THERETO SHALL, IN ACCORDANCE WITH SECTION 5-1401 OF THE GENERAL OBLIGATIONS LAW OF THE STATE OF NEW YORK, BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.
- (b) THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS SHALL BE TRIED AND LITIGATED ONLY IN THE STATE AND, TO THE EXTENT PERMITTED BY APPLICABLE LAW, FEDERAL COURTS LOCATED IN THE STATE OF NEW YORK; PROVIDED, THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE AGENT ELECTS TO BRING SUCH ACTION OR WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. EACH OF PARENT AND BORROWER AND EACH MEMBER OF THE LENDER GROUP WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT EACH MAY HAVE TO ASSERT THE DOCTRINE OF FORUM NON CONVENIENS OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 12(b).
- (c) TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH OF PARENT AND BORROWER AND EACH MEMBER OF THE LENDER GROUP HEREBY WAIVE THEIR RESPECTIVE RIGHTS, IF ANY, TO A JURY TRIAL OF ANY CLAIM, CONTROVERSY, DISPUTE OR CAUSE OF ACTION DIRECTLY OR INDIRECTLY BASED UPON OR ARISING OUT OF ANY OF THE LOAN DOCUMENTS OR ANY OF THE TRANSACTIONS CONTEMPLATED THEREIN, INCLUDING CONTRACT CLAIMS, TORT CLAIMS, BREACH OF DUTY CLAIMS, AND ALL OTHER COMMON LAW OR STATUTORY CLAIMS (EACH A " <u>CLAIM</u>"). EACH OF PARENT AND BORROWER AND EACH MEMBER OF THE LENDER GROUP REPRESENT THAT EACH HAS REVIEWED THIS WAIVER AND EACH KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH LEGAL COUNSEL. IN THE EVENT OF LITIGATION, A COPY OF THIS AGREEMENT MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.
- (d) EACH OF PARENT AND BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY SUBMITS TO THE EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS LOCATED IN THE STATE OF NEW YORK, IN

ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN DOCUMENTS, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT AGENT MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST ANY LOAN PARTY OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.

(e) NO CLAIM MAY BE MADE BY ANY LOAN PARTY AGAINST THE AGENT, THE SWING LENDER, ANY OTHER LENDER, ISSUING BANK, OR ANY AFFILIATE, DIRECTOR, OFFICER, EMPLOYEE, COUNSEL, REPRESENTATIVE, AGENT, OR ATTORNEY-IN-FACT OF ANY OF THEM FOR ANY SPECIAL, INDIRECT, CONSEQUENTIAL, PUNITIVE OR EXEMPLARY DAMAGES OR LOSSES IN RESPECT OF ANY CLAIM FOR BREACH OF CONTRACT OR ANY OTHER THEORY OF LIABILITY ARISING OUT OF OR RELATED TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR ANY ACT, OMISSION, OR EVENT OCCURRING IN CONNECTION THEREWITH, AND EACH LOAN PARTY HEREBY WAIVES, RELEASES, AND AGREES NOT TO SUE UPON ANY CLAIM FOR SUCH DAMAGES, WHETHER OR NOT ACCRUED AND WHETHER OR NOT KNOWN OR SUSPECTED TO EXIST IN ITS FAVOR.

13. ASSIGNMENTS AND PARTICIPATIONS; SUCCESSORS.

13.1. Assignments and Participations.

- (a) (i) Subject to the conditions set forth in clause (a)(ii) below, any Lender may assign and delegate all or any portion of its rights and duties under the Loan Documents (including the Obligations owed to it and its Commitments) to one or more assignees so long as such prospective assignee is an Eligible Transferee (each, an "Assignee"), with the prior written consent (such consent not be unreasonably withheld or delayed) of:
- (A) Borrower; <u>provided</u>, that no consent of Borrower shall be required (1) if an Event of Default has occurred and is continuing, or (2) in connection with an assignment to a Person that is a Lender or an Affiliate (other than natural persons) of a Lender (other than any Defaulting Lender or its Affiliates); <u>provided further</u>, that Borrower shall be deemed to have consented to a proposed assignment unless it objects thereto by written notice to Agent within 5 Business Days after having received notice thereof; and
 - (B) Agent, Swing Loan Lender, and Issuing Bank.
 - (ii) Assignments shall be subject to the following additional conditions:
- (A) no assignment may be made to an Ineligible Institution, a Competitor or to a natural person; <u>provided</u>, that Assignments may be made to any Competitor so long as a Specified Event of Default has occurred and is continuing for 30 consecutive days,
- (B) no assignment may be made to a Loan Party or an Affiliate of a Loan Party other than to Sponsor Affiliated Entities; provided, that any such Sponsor

Affiliated Entity that is or becomes a Lender hereunder shall be subject to the provisions of Section 13.1(k) hereof,

- (C) the amount of the Commitments and the other rights and obligations of the assigning Lender hereunder and under the other Loan Documents subject to each such assignment (determined as of the date the Assignment and Acceptance with respect to such assignment is delivered to Agent) shall be in a minimum amount (unless waived by Agent) of \$5,000,000 (except such minimum amount shall not apply to (I) an assignment or delegation by any Lender to any other Lender, an Affiliate of any Lender, or a Related Fund of such Lender or (II) a group of new Lenders, each of which is an Affiliate of each other or a Related Fund of such new Lender to the extent that the aggregate amount to be assigned to all such new Lenders is at least \$5,000,000),
- (D) each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement,
- (E) the parties to each assignment shall execute and deliver to Agent an Assignment and Acceptance; <u>provided</u>, that Borrower and Agent may continue to deal solely and directly with the assigning Lender in connection with the interest so assigned to an Assignee until written notice of such assignment, together with payment instructions, addresses, and related information with respect to the Assignee, have been given to Borrower and Agent by such Lender and the Assignee,
- (F) unless waived by Agent, the assigning Lender or Assignee has paid to Agent, for Agent's separate account, a processing fee in the amount of \$3,500, and
- (G) the assignee, if it is not a Lender, shall deliver to Agent an Administrative Questionnaire in a form approved by Agent (the "<u>Administrative Questionnaire</u>").
- (b) From and after the date that Agent receives the executed Assignment and Acceptance and, if applicable, payment of the required processing fee, (i) the Assignee thereunder shall be a party hereto and, to the extent that rights and obligations hereunder have been assigned to it pursuant to such Assignment and Acceptance, shall be a "Lender" and shall have the rights and obligations of a Lender under the Loan Documents, and (ii) the assigning Lender shall, to the extent that rights and obligations hereunder and under the other Loan Documents have been assigned by it pursuant to such Assignment and Acceptance, relinquish its rights (except with respect to Section 10.3) and be released from any future obligations under this Agreement (and in the case of an Assignment and Acceptance covering all or the remaining portion of an assigning Lender's rights and obligations under this Agreement and the other Loan Documents, such Lender shall cease to be a party hereto and thereto); provided, that nothing contained herein shall release any assigning Lender from obligations that survive the termination of this Agreement, including such assigning Lender's obligations under Section 15 and Section 17.9(a).
- (c) By executing and delivering an Assignment and Acceptance, the assigning Lender thereunder and the Assignee thereunder confirm to and agree with each other and the other parties hereto as follows: (i) other than as provided in such Assignment and Acceptance, such assigning Lender makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness, sufficiency or value of this Agreement or any other Loan Document furnished pursuant hereto, (ii) such assigning Lender makes no representation or warranty and assumes no responsibility with respect to the

financial condition of the Loan Parties or the performance or observance by the Loan Parties of any of their obligations under this Agreement or any other Loan Document furnished pursuant hereto, (iii) such Assignee confirms that it has received a copy of this Agreement, together with such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance, (iv) such Assignee will, independently and without reliance upon Agent, such assigning Lender or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement, (v) such Assignee appoints and authorizes Agent to take such actions and to exercise such powers under this Agreement and the other Loan Documents as are delegated to Agent, by the terms hereof and thereof, together with such powers as are reasonably incidental thereto, and (vi) such Assignee agrees that it will perform all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

- (d) Immediately upon Agent's receipt of the required processing fee, if applicable, and delivery of notice to the assigning Lender pursuant to Section 13.1(b), this Agreement shall be deemed to be amended to the extent, but only to the extent, necessary to reflect the addition of the Assignee and the resulting adjustment of the Commitments arising therefrom. The Commitment allocated to each Assignee shall reduce such Commitments of the assigning Lender *pro tanto*.
- (e) Any Lender may at any time sell to one or more commercial banks, financial institutions, or other Persons not constituting Competitors (except to the extent such assignment would be permitted hereunder) (a "Participant") participating interests in all or any portion of its Obligations, its Commitment, and the other rights and interests of that Lender (the "Originating Lender") hereunder and under the other Loan Documents; provided, that (i) the Originating Lender shall remain a "Lender" for all purposes of this Agreement and the other Loan Documents and the Participant receiving the participating interest in the Obligations, the Commitments, and the other rights and interests of the Originating Lender hereunder shall not constitute a "Lender" hereunder or under the other Loan Documents and the Originating Lender's obligations under this Agreement shall remain unchanged, (ii) the Originating Lender shall remain solely responsible for the performance of such obligations, (iii) Borrower, Agent, and the Lenders shall continue to deal solely and directly with the Originating Lender in connection with the Originating Lender's rights and obligations under this Agreement and the other Loan Documents, (iv) no Lender shall transfer or grant any participating interest under which the Participant has the right to approve any amendment to, or any consent or waiver with respect to, this Agreement or any other Loan Document, except to the extent such amendment to, or consent or waiver with respect to this Agreement or of any other Loan Document would (A) extend the final maturity date of the Obligations hereunder in which such Participant is participating, (B) reduce the interest rate applicable to the Obligations hereunder in which such Participant is participating, (C) release all or substantially all of the Collateral or guaranties (except to the extent expressly provided herein or in any of the Loan Documents) supporting the Obligations hereunder in which such Participant is participating, (D) postpone the payment of, or reduce the amount of, the interest or fees payable to such Participant through such Lender (other than a waiver of default interest), or (E) decreases the amount or postpones the due dates of scheduled principal repayments or prepayments or premiums payable to such Participant through such Lender, (v) no participation shall be sold to a natural person, (vi) no participation shall be sold to a Loan Party or an Affiliate of a Loan Party other than to Sponsor Affiliated Entities; provided, that any such Person that is or becomes a Participant hereunder shall be subject to the provisions of Section 13.1(k) hereof as if it were an assignee rather than a participant, and (vii) all amounts payable by Borrower hereunder shall be determined as if such Lender had not sold such participation, except that, if amounts outstanding under this Agreement are due and unpaid, or shall have been declared or shall have become due and payable upon the occurrence of

an Event of Default, each Participant shall be deemed to have the right of set off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement. The rights of any Participant only shall be derivative through the Originating Lender with whom such Participant participates and no Participant shall have any rights under this Agreement or the other Loan Documents or any direct rights as to the other Lenders, Agent, Borrower, the Collateral, or otherwise in respect of the Obligations. No Participant shall have the right to participate directly in the making of decisions by the Lenders among themselves.

- (f) In connection with any such assignment or participation or proposed assignment or participation or any grant of a security interest in, or pledge of, its rights under and interest in this Agreement, a Lender may, subject to the provisions of <u>Section 17.9</u>, disclose all documents and information which it now or hereafter may have relating to Parent and its Subsidiaries and their respective businesses.
- (g) Any other provision in this Agreement notwithstanding, any Lender may at any time create a security interest in, or pledge, all or any portion of its rights under and interest in this Agreement in favor of any Federal Reserve Bank in accordance with Regulation A of the Federal Reserve Bank or U.S. Treasury Regulation 31 CFR §203.24, and such Federal Reserve Bank may enforce such pledge or security interest in any manner permitted under applicable law.
- (h) Agent (as a non-fiduciary agent on behalf of Borrower) shall maintain, or cause to be maintained, a register (the "Register") on which it enters the name and address of each Lender as the registered owner of the Term Loan (and the principal amount thereof and stated interest thereon) held by such Lender (each, a "Registered Loan"). Other than in connection with an assignment by a Lender of all or any portion of its portion of the Term Loan to an Affiliate of such Lender or a Related Fund of such Lender (i) a Registered Loan (and the registered note, if any, evidencing the same) may be assigned or sold in whole or in part only by registration of such assignment or sale on the Register (and each registered note shall expressly so provide) and (ii) any assignment or sale of all or part of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by registration of such assignment or sale on the Register, together with the surrender of the registered note, if any, evidencing the same duly endorsed by (or accompanied by a written instrument of assignment or sale duly executed by) the holder of such registered note, whereupon, at the request of the designated assignee(s) or transferee(s), one or more new registered notes in the same aggregate principal amount shall be issued to the designated assignee(s) or transferee(s). Prior to the registration of assignment or sale of any Registered Loan (and the registered note, if any evidencing the same), Borrower shall treat the Person in whose name such Registered Loan (and the registered note, if any, evidencing the same) is registered as the owner thereof for the purpose of receiving all payments thereon and for all other purposes, notwithstanding notice to the contrary. In the case of any assignment by a Lender of all or any portion of its Term Loan to an Affiliate of such Lender or a Related Fund of such Lender, and which assignment is not recorded in the Register, the assigning Lender, on behalf of Borrower, shall maintain a re
- (i) In the event that a Lender sells participations in the Registered Loan, such Lender, as a non-fiduciary agent on behalf of Borrower, shall maintain (or cause to be maintained) a register on which it enters the name of all participants in the Registered Loans held by it (and the principal amount (and stated interest thereon) of the portion of such Registered Loans that is subject to such participations) (the "Participant Register"). A Registered Loan (and the Registered Note, if any, evidencing the same) may be participated in whole or in part only by registration of such participation on the Participant Register (and each registered note shall

expressly so provide). Any participation of such Registered Loan (and the registered note, if any, evidencing the same) may be effected only by the registration of such participation on the Participant Register.

- (j) Agent shall make a copy of the Register (and each Lender shall make a copy of its Participant Register in the extent it has one) available for review by Borrower from time to time as Borrower may reasonably request.
- (k) (i) Each Sponsor Affiliated Entity, in connection with any (A) consent (or decision not to consent) to any amendment, modification, waiver, consent or other action with respect to any of the terms of any Loan Document or in respect of any consent, waiver, vote, or other action in connection with an Insolvency Proceeding involving a Loan Party, (B) other action on any matter related to any Loan Document or occurring in relation to any Insolvency Proceeding involving a Loan Party, or (C) direction to Agent or any Lender to undertake any action (or refrain from taking any action) with respect to or under any Loan Document, agrees that, except with respect to any amendment, modification, waiver, consent or other action with respect to the matters governed by Section 14.1(a)(i) through (iv) that affect such Lender or that adversely and disproportionately affects such Sponsor Affiliated Entity in any material respect as compared to other Lenders, shall be deemed to have voted (and, at the request of Agent, shall so vote, and each Sponsor Affiliated Entity, as a Lender, hereby grants (and each Sponsor Affiliated Entity that is not a Lender on the Closing Date will be deemed to have granted on the date such holder executes and assignment and acceptance agreement) to Agent an irrevocable proxy and the right to vote) its interest as a Lender without discretion in such proportion as the allocation of voting with respect to such matter by Lenders who are not Sponsor Affiliated Entities. Borrower, each Sponsor Affiliated Entity, as a Lender (and each Sponsor Affiliate that becomes a Lender by its acceptance of an assignment) agrees that if a case under the Bankruptcy Code is commenced by or against Borrower or any other Loan Party, in connection with a vote under a Plan under Bankruptcy Code, Agent may vote the claims of the Sponsor Affiliated Entities to accept such Plan if the Plan has been accepted by other Lenders holding at least two-thirds in amount and more than one half in number of the allowed Lender claims of such class that vote (excluding the claims of Sponsor Affiliate Entities) and the Agent may otherwise vote the claims of the Sponsor Affiliate Entities to reject such Plan (provided, that such right to vote shall not, without the prior approval of such Sponsor Affiliate Entity, be exercised in a manner that would affect such Sponsor Affiliate Entity in a manner that is disproportionately adverse as compared to the treatment of the claims of the other Lenders in such class). Subject to clause (ii) below, each Sponsor Affiliated Entity hereby irrevocably appoints Agent (such appointment being coupled with an interest) as such Sponsor Affiliated Entity's attorney-in-fact, with full authority in the place and stead of such Sponsor Affiliated Entity and in the name of such Sponsor Affiliated Entity, from time to time in Agent's discretion to take any action and to execute any instrument that Agent may deem reasonably necessary to carry out the provisions of this clause (i). Agent is irrevocably authorized by each Lender to withdraw any vote submitted by such Lender in contravention of the procedures set forth herein. The provisions of this Section 13.1(k)(i) will not apply during such time as all of the Obligations are held by Sponsor Affiliated Entities.
- (ii) Notwithstanding anything to the contrary in this Agreement, (A) no Sponsor Affiliated Entity shall have any right to receive any information or material prepared by Agent or any Lender or any communication by or among Agent or one or more Lenders, except to the extent such information or materials have been made available to Borrower or its representatives and (B) no Sponsor Affiliated Entity (x) shall have any right to attend (including by telephone) any meeting or discussions (or portion thereof) among Agent and any Lender to which representatives of Borrower are not then present or (y) make or bring (or participate in, other than as a passive participant in or recipient of its pro rata benefits of) any claim, in its

capacity as a Lender, against Agent or any other member of the Lender Group with respect to any duties or obligations or alleged duties or obligations of such Agent or any other member of the Lender Group under the Loan Documents other than a claim as a result of a material breach of the express terms of the Loan Documents by or as arising as a result of the bad faith, willful misconduct or gross negligence of such Agent or other member of the Lender Group (provided that any such claim brought but later determined by a court of competent jurisdiction by a final and nonappealable order to not have been a material breach by or the bad faith, willful misconduct or gross negligence of such Agent or other member of the Lender Group shall be deemed to be a breach by such Sponsor Affiliated Entity that made or brought such claim). The provisions of this Section 13.1(k)(ii) will not apply during such time as all of the Obligations are held by Sponsor Affiliated Entities.

13.2. <u>Successors</u>. This Agreement shall bind and inure to the benefit of the respective successors and assigns of each of the parties; <u>provided</u>, that Borrower may not assign this Agreement or any rights or duties hereunder without the Lenders' prior written consent and any prohibited assignment shall be absolutely void ab initio. No consent to assignment by the Lenders shall release Borrower from its Obligations. A Lender may assign this Agreement and the other Loan Documents and its rights and duties hereunder and thereunder pursuant to <u>Section 13.1</u> and, except as expressly required pursuant to <u>Section 13.1</u>, no consent or approval by Borrower is required in connection with any such assignment.

14. AMENDMENTS; WAIVERS.

14.1. Amendments and Waivers.

- (a) No amendment, waiver or other modification of any provision of this Agreement or any other Loan Document (other than Bank Product Agreements or the Fee Letter), and no consent with respect to any departure by any Loan Party or its Subsidiary therefrom, shall be effective unless the same shall be in writing and signed by the Required Lenders (or by Agent at the written request of the Required Lenders) and the Loan Parties that are party thereto and then any such waiver or consent shall be effective, but only in the specific instance and for the specific purpose for which given; provided, that no such waiver, amendment, or consent shall, unless in writing and signed by all of the Lenders directly affected thereby and all of the Loan Parties that are party thereto, do any of the following:
 - (i) increase the amount of or extend the expiration date of any Commitment of such affected Lender;
- (ii) reduce the principal of any loan or other extension of credit hereunder, or reduce any fees or other amounts payable hereunder or under any other Loan Document (except in connection with the waiver of applicability of <u>Section 2.6(c)</u> (which waiver shall be effective with the written consent of the Required Lenders);
- (iii) amend, modify, or eliminate this Section or any provision of this Agreement providing for consent or other action by all Lenders; or
- (iv) amend, modify, or eliminate any of the provisions of <u>Section 13.1</u> with respect to assignments to, or participations with, Persons who are Sponsor Affiliated Entities.

- (b) No amendment, waiver, modification, or consent shall amend, modify, waive, or eliminate,
- (i) the definition of, or any of the terms or provisions of, the Fee Letter, without the written consent of Agent and Borrower (and shall not require the written consent of any of the Lenders),
- (ii) any provision of <u>Section 15</u> pertaining to Agent, or any other rights or duties of Agent under this Agreement or the other Loan Documents, without the written consent of Agent, Borrower, and the Required Lenders;
- (c) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to Issuing Bank, or any other rights or duties of Issuing Bank under this Agreement or the other Loan Documents, without the written consent of Issuing Bank, Agent, Borrower, and the Required Lenders;
- (d) No amendment, waiver, modification, elimination, or consent shall amend, modify, or waive any provision of this Agreement or the other Loan Documents pertaining to Swing Loan Lender, or any other rights or duties of Swing Loan Lender under this Agreement or the other Loan Documents, without the written consent of Swing Loan Lender, Agent, Borrower, and the Required Lenders; and
- (e) Anything in this <u>Section 14.1</u> to the contrary notwithstanding, (i) any amendment, modification, elimination, waiver, consent, termination, or release of, or with respect to, any provision of this Agreement or any other Loan Document that relates only to the relationship of the Lender Group among themselves, and that does not affect the rights or obligations of Parent or Borrower, shall not require consent by or the agreement of any Loan Party; <u>provided</u>, that any such amendment, modification, elimination, waiver, consent, termination or release shall be effectuated pursuant to an agreement among the Required Lenders and the Loan Parties shall not be required to take any action or otherwise modify their conduct with respect thereto, and (ii) any amendment, waiver, modification, elimination, or consent of or with respect to any provision of this Agreement or any other Loan Document may be entered into without the consent of, or over the objection of, any Defaulting Lender.
- (f) Anything in this <u>Section 14.1</u> to the contrary notwithstanding, if Agent and Borrower shall have jointly identified an obvious error or any error or omission of a technical or immaterial nature, in each case, in any provision of this Agreement or the other the Loan Documents, then Agent and Borrower shall be permitted to amend such provision and such amendment shall become effective without any further action or consent of any other party to any Loan Document if the same is not objected to in writing by the Required Lenders within three (3) Business Days following receipt of notice thereof.
- (g) Nothing in this <u>Section 14.1</u> shall be construed to prohibit the amendment of any schedule hereto or to any other Loan Document which is expressly permitted to be amended pursuant to written notice provided to Agent by Borrower or the applicable Loan Party.

14.2. Replacement of Certain Lenders.

(a) If (i) any action to be taken by the Lender Group or Agent hereunder requires the consent, authorization, or agreement of all Lenders affected thereby and if such action has received the consent, authorization, or agreement of the Required Lenders but not of all Lenders or all Lenders affected thereby, or (ii) any Lender makes a claim for

compensation under <u>Section 16</u>, then Borrower or Agent, upon at least 5 Business Days prior irrevocable notice, may permanently replace any Lender that failed to give its consent, authorization, or agreement (a "<u>Non-Consenting Lender</u>") or any Lender that made a claim for compensation (a "<u>Tax Lender</u>") with one or more Replacement Lenders, and the Non-Consenting Lender or Tax Lender, as applicable, shall have no right to refuse to be replaced hereunder. Such notice to replace the Non-Consenting Lender or Tax Lender, as applicable, shall specify an effective date for such replacement, which date shall not be later than 15 Business Days after the date such notice is given.

- (b) Prior to the effective date of such replacement, the Non-Consenting Lender or Tax Lender, as applicable, and each Replacement Lender shall execute and deliver an Assignment and Acceptance, subject only to the Non-Consenting Lender or Tax Lender, as applicable, being repaid in full its share of the outstanding Obligations (without any premium or penalty of any kind whatsoever, but including (i) all interest, fees and other amounts that may be due in payable in respect thereof, and (ii) an assumption of its Pro Rata Share of participations in the Letters of Credit). If the Non-Consenting Lender or Tax Lender, as applicable, shall refuse or fail to execute and deliver any such Assignment and Acceptance prior to the effective date of such replacement, Agent may, but shall not be required to, execute and deliver such Assignment and Acceptance in the name or and on behalf of the Non-Consenting Lender or Tax Lender, as applicable, shall be deemed to have executed and delivered such Assignment and Acceptance. The replacement of any Non-Consenting Lender or Tax Lender, as applicable, shall be made in accordance with the terms of Section 13.1. Until such time as one or more Replacement Lenders shall have acquired all of the Obligations, the Commitments, and the other rights and obligations of the Non-Consenting Lender or Tax Lender, as applicable, hereunder and under the other Loan Documents, the Non-Consenting Lender or Tax Lender, as applicable, shall remain obligated to make the Non-Consenting Lender's or Tax Lender's, as applicable, Pro Rata Share of Revolving Loans and to purchase a participation in each Letter of Credit, in an amount equal to its Pro Rata Share of participations in such Letters of Credit.
- 14.3. No Waivers; Cumulative Remedies. No failure by Agent or any Lender to exercise any right, remedy, or option under this Agreement or any other Loan Document, or delay by Agent or any Lender in exercising the same, will operate as a waiver thereof. No waiver by Agent or any Lender will be effective unless it is in writing, and then only to the extent specifically stated. No waiver by Agent or any Lender on any occasion shall affect or diminish Agent's and each Lender's rights thereafter to require strict performance by Parent and Borrower of any provision of this Agreement. Agent's and each Lender's rights under this Agreement and the other Loan Documents will be cumulative and not exclusive of any other right or remedy that Agent or any Lender may have.

15. AGENT: THE LENDER GROUP.

15.1. Appointment and Authorization of Agent. Each Lender hereby designates and appoints PNC as its agent under this Agreement and the other Loan Documents and each Lender hereby irrevocably authorizes (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to designate, appoint, and authorize) Agent to execute and deliver each of the other Loan Documents on its behalf and to take such other action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to Agent by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Agent agrees to act as agent for and on behalf of the Lenders (and the Bank Product Providers) on the conditions contained in this Section 15. Any provision to the contrary

contained elsewhere in this Agreement or in any other Loan Document notwithstanding, Agent shall not have any duties or responsibilities, except those expressly set forth herein or in the other Loan Documents, nor shall Agent have or be deemed to have any fiduciary relationship with any Lender (or Bank Product Provider), and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent. Without limiting the generality of the foregoing, the use of the term "agent" in this Agreement or the other Loan Documents with reference to Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any applicable law. Instead, such term is used merely as a matter of market custom, and is intended to create or reflect only a representative relationship between independent contracting parties. Each Lender hereby further authorizes (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent to act as the secured party under each of the Loan Documents that create a Lien on any item of Collateral. Except as expressly otherwise provided in this Agreement, Agent shall have and may use its sole discretion with respect to exercising or refraining from exercising any discretionary rights or taking or refraining from taking any actions that Agent expressly is entitled to take or assert under or pursuant to this Agreement and the other Loan Documents. Without limiting the generality of the foregoing, or of any other provision of the Loan Documents that provides rights or powers to Agent, Lenders agree that Agent shall have the right to exercise the following powers as long as this Agreement remains in effect: (a) maintain, in accordance with its customary business practices, ledgers and records reflecting the status of the Obligations, the Collateral, payments and proceeds of Collateral, and related matters, (b) execute or file any and all financing or similar statements or notices, amendments, renewals, supplements, documents, instruments, proofs of claim, notices and other written agreements with respect to the Loan Documents, (c) make Revolving Loans, for itself or on behalf of Lenders, as provided in the Loan Documents, (d) exclusively receive, apply, and distribute payments and proceeds of the Collateral as provided in the Loan Documents, (e) open and maintain such bank accounts and cash management arrangements as Agent deems necessary and appropriate in accordance with the Loan Documents for the foregoing purposes, (f) perform, exercise, and enforce any and all other rights and remedies of the Lender Group with respect to Parent or its Subsidiaries, the Obligations, the Collateral, or otherwise related to any of same as provided in the Loan Documents, and (g) incur and pay such Lender Group Expenses as Agent may deem necessary or appropriate for the performance and fulfillment of its functions and powers pursuant to the Loan Documents.

- 15.2. <u>Delegation of Duties</u>. Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys in fact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. Agent shall not be responsible for the negligence or misconduct of any agent or attorney in fact that it selects as long as such selection was made without gross negligence or willful misconduct.
- 15.3. <u>Liability of Agent</u>. None of the Agent-Related Persons shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except for its own gross negligence or willful misconduct), or (b) be responsible in any manner to any of the Lenders (or Bank Product Providers) for any recital, statement, representation or warranty made by Parent or any of its Subsidiaries or Affiliates, or any officer or director thereof, contained in this Agreement or in any other Loan Document, or in any certificate, report, statement or other document referred to or provided for in, or received by Agent under or in connection with, this Agreement or any other Loan Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document, or for any failure of Parent or its Subsidiaries or any other party to any Loan Document to perform its obligations hereunder or thereunder. No Agent-Related Person shall be under any obligation to any Lenders

(or Bank Product Providers) to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or to inspect the books and records or properties of Parent or its Subsidiaries.

- 15.4. Reliance by Agent. Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telegram, telefacsimile or other electronic method of transmission, telex or telephone message, statement or other document or conversation believed by it to be genuine and correct and to have been signed, sent, or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to Borrower or counsel to any Lender), independent accountants and other experts selected by Agent. Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless Agent shall first receive such advice or concurrence of the Lenders as it deems appropriate and until such instructions are received, Agent shall act, or refrain from acting, as it deems advisable. If Agent so requests, it shall first be indemnified to its reasonable satisfaction by the Lenders (and, if it so elects, the Bank Product Providers) against any and all liability and expense that may be incurred by it by reason of taking or continuing to take any such action. Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of the Required Lenders and such request and any action taken or failure to act pursuant thereto shall be binding upon all of the Lenders (and Bank Product Providers).
- 15.5. Notices of Default or Event of Default. Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default, except with respect to defaults in the payment of principal, interest, fees, and expenses required to be paid to Agent for the account of the Lenders and, except with respect to Events of Default of which Agent has actual knowledge, unless Agent shall have received written notice from a Lender or Borrower referring to this Agreement, describing such Default or Event of Default, and stating that such notice is a "notice of default." Agent promptly will notify the Lenders of its receipt of any such notice or of any Event of Default of which Agent has actual knowledge. If any Lender obtains actual knowledge of any Event of Default, such Lender promptly shall notify the other Lenders and Agent of such Event of Default. Each Lender shall be solely responsible for giving any notices to its Participants, if any. Subject to Section 15.4, Agent shall take such action with respect to such Default or Event of Default as may be requested by the Required Lenders in accordance with Section 9; provided, that unless and until Agent has received any such request, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable.
- 15.6. <u>Credit Decision</u>. Each Lender (and Bank Product Provider) acknowledges that none of the Agent-Related Persons has made any representation or warranty to it, and that no act by Agent hereinafter taken, including any review of the affairs of Parent and its Subsidiaries or Affiliates, shall be deemed to constitute any representation or warranty by any Agent-Related Person to any Lender (or Bank Product Provider). Each Lender represents (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to represent) to Agent that it has, independently and without reliance upon any Agent-Related Person and based on such due diligence, documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of Borrower or any other Person party to a Loan Document, and all applicable bank regulatory laws relating to the transactions contemplated hereby, and made its own decision to enter into this Agreement and to extend credit to Borrower. Each Lender also represents (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to represent) that it will, independently and without reliance upon any Agent-Related Person and based on such documents and information as it shall deem appropriate at the time,

continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of Borrower or any other Person party to a Loan Document. Except for notices, reports, and other documents expressly herein required to be furnished to the Lenders by Agent, Agent shall not have any duty or responsibility to provide any Lender (or Bank Product Provider) with any credit or other information concerning the business, prospects, operations, property, financial and other condition or creditworthiness of Borrower or any other Person party to a Loan Document that may come into the possession of any of the Agent-Related Persons. Each Lender acknowledges (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that Agent does not have any duty or responsibility, either initially or on a continuing basis (except to the extent, if any, that is expressly specified herein) to provide such Lender (or Bank Product Provider) with any credit or other information with respect to Borrower, its Affiliates or any of their respective business, legal, financial or other affairs, and irrespective of whether such information came into Agent's or its Affiliates' or representatives' possession before or after the date on which such Lender became a party to this Agreement (or such Bank Product Provider entered into a Bank Product Agreement).

- 15.7. Costs and Expenses; Indemnification. Agent may incur and pay Lender Group Expenses to the extent Agent reasonably deems necessary or appropriate for the performance and fulfillment of its functions, powers, and obligations pursuant to the Loan Documents, including court costs, attorneys' fees and expenses, fees and expenses of financial accountants, advisors, consultants, and appraisers, costs of collection by outside collection agencies, auctioneer fees and expenses, and costs of security guards or insurance premiums paid to maintain the Collateral, whether or not Borrower is obligated to reimburse Agent or Lenders for such expenses pursuant to this Agreement or otherwise. Agent is authorized and directed to deduct and retain sufficient amounts from payments or proceeds of the Collateral received by Agent to reimburse Agent for such out-of-pocket costs and expenses prior to the distribution of any amounts to Lenders (or Bank Product Providers). In the event Agent is not reimbursed for such costs and expenses by Parent or its Subsidiaries, each Lender hereby agrees that it is and shall be obligated to pay to Agent such Lender's ratable thereof. Each of the Lenders, on a ratable basis, shall indemnify and defend the Agent-Related Persons (to the extent not reimbursed by or on behalf of Borrower and without limiting the obligation of Borrower to do so) from and against any and all Indemnified Liabilities; provided, that no Lender shall be liable for the payment to any Agent-Related Person of any portion of such Indemnified Liabilities resulting solely from such Person's gross negligence or willful misconduct nor shall any Lender be liable for the obligations of any Defaulting Lender in failing to make a Revolving Loan or other extension of credit hereunder. Without limitation of the foregoing, each Lender shall reimburse Agent upon demand for such Lender's ratable share of any costs or out of pocket expenses (including attorneys, accountants, advisors, and consultants fees and expenses) incurred by Agent in connection with the preparation, execution, delivery, administration, modification, amendment, or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement or any other Loan Document to the extent that Agent is not reimbursed for such expenses by or on behalf of Borrower. The undertaking in this Section shall survive the payment of all Obligations hereunder and the resignation or replacement of Agent.
- 15.8. Agent in Individual Capacity. PNC and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, provide Bank Products to, acquire Equity Interests in, and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with Parent and its Subsidiaries and Affiliates and any other Person party to any Loan Document as though PNC were not Agent hereunder, and, in each case,

without notice to or consent of the other members of the Lender Group. The other members of the Lender Group acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, pursuant to such activities, PNC or its Affiliates may receive information regarding Parent or its Affiliates or any other Person party to any Loan Documents that is subject to confidentiality obligations in favor of Parent or such other Person and that prohibit the disclosure of such information to the Lenders (or Bank Product Providers), and the Lenders acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver Agent will use its reasonable best efforts to obtain), Agent shall not be under any obligation to provide such information to them. The terms "Lender" and "Lenders" include PNC in its individual capacity.

- 15.9. Successor Agent. Agent may resign as Agent (a) upon 30 days (10 days if an Event of Default has occurred and is continuing) prior written notice to the Lenders (unless such notice is waived by the Required Lenders) and Borrower (unless such notice is waived by Borrower) and without any notice to the Bank Product Providers or (b) as provided in Section 15.11(c)(iv)(E). If Agent resigns under this Agreement, the Required Lenders shall be entitled, with (so long as no Event of Default has occurred and is continuing) the consent of Borrower (such consent not to be unreasonably withheld, delayed, or conditioned), appoint a successor Agent for the Lenders (and the Bank Product Providers). If, at the time that Agent's resignation is effective, it is acting as Issuing Bank or the Swing Loan Lender, such resignation shall also operate to effectuate its resignation as Issuing Bank or the Swing Loan Lender, as applicable, and it shall automatically be relieved of any further obligation to issue Letters of Credit or to make Swing Loans. If no successor Agent is appointed prior to the effective date of the resignation of Agent, Agent may appoint, after consulting with the Lenders and Borrower, a successor Agent. If Agent has materially breached or failed to perform any material provision of this Agreement or of applicable law, the Required Lenders may agree in writing to remove and replace Agent with a successor Agent from among the Lenders with (so long as no Event of Default has occurred and is continuing) the consent of Borrower (such consent not to be unreasonably withheld, delayed, or conditioned). In any such event, upon the acceptance of its appointment as successor Agent hereunder, such successor Agent shall succeed to all the rights, powers, and duties of the retiring Agent and the term "Agent" shall mean such successor Agent and the retiring Agent's appointment, powers, and duties as Agent shall be terminated. After any retiring Agent's resignation hereunder as Agent, the provisions of this Section 15 shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement. If no successor Agent has accepted appointment as Agent by the date which is 30 days following a retiring Agent's notice of resignation, the retiring Agent's resignation shall nevertheless thereupon become effective and the Lenders shall perform all of the duties of Agent hereunder until such time, if any, as the Lenders appoint a successor Agent as provided for above.
- 15.10. Lender in Individual Capacity. Any Lender and its respective Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, provide Bank Products to, acquire Equity Interests in and generally engage in any kind of banking, trust, financial advisory, underwriting, or other business with Parent and its Subsidiaries and Affiliates and any other Person party to any Loan Documents as though such Lender were not a Lender hereunder without notice to or consent of the other members of the Lender Group (or the Bank Product Providers). The other members of the Lender Group acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge) that, pursuant to such activities, such Lender and its respective Affiliates may receive information regarding Parent or its Affiliates or any other Person party to any Loan Documents that is subject to confidentiality obligations in favor of Parent or such other Person and that prohibit the disclosure of such information to the Lenders, and the Lenders acknowledge (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to acknowledge)

that, in such circumstances (and in the absence of a waiver of such confidentiality obligations, which waiver such Lender will use its reasonable best efforts to obtain), such Lender shall not be under any obligation to provide such information to them.

15.11. Collateral Matters.

The Lenders hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent to release any Lien on any Collateral (i) upon the termination of the Commitments and payment and satisfaction in full by Borrower of all of the Obligations, (ii) constituting property being sold or disposed of if a release is required or desirable in connection therewith and if Borrower certifies to Agent that the sale or disposition is permitted under Section 6.4 (and Agent may rely conclusively on any such certificate, without further inquiry), (iii) constituting property in which Parent or its Subsidiaries owned no interest at the time Agent's Lien was granted nor at any time thereafter, (iv) constituting property leased or licensed to Parent or its Subsidiaries under a lease or license that has expired or is terminated in a transaction permitted under this Agreement, or (v) in connection with a credit bid or purchase authorized under this Section 15.11. The Loan Parties and the Lenders hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to authorize) Agent, based upon the instruction of the Required Lenders, to (a) consent to credit bid or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any sale thereof conducted under the provisions of the Bankruptcy Code, including Section 363 of the Bankruptcy Code, (b) credit bid or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any sale or other disposition thereof conducted under the provisions of the Code, including pursuant to Sections 9-610 or 9-620 of the Code, or (c) credit bid or purchase (either directly or indirectly through one or more entities) all or any portion of the Collateral at any other sale or foreclosure conducted or consented to by Agent in accordance with applicable law in any judicial action or proceeding or by the exercise of any legal or equitable remedy. In connection with any such credit bid or purchase, (i) the Obligations owed to the Lenders and the Bank Product Providers shall be entitled to be, and shall be, credit bid on a ratable basis (with Obligations with respect to contingent or unliquidated claims being estimated for such purpose if the fixing or liquidation thereof would not impair or unduly delay the ability of Agent to credit bid or purchase at such sale or other disposition of the Collateral and, if such contingent or unliquidated claims cannot be estimated without impairing or unduly delaying the ability of Agent to credit bid at such sale or other disposition, then such claims shall be disregarded, not credit bid, and not entitled to any interest in the Collateral that is the subject of such credit bid or purchase) and the Lenders and the Bank Product Providers whose Obligations are credit bid shall be entitled to receive interests (ratably based upon the proportion of their Obligations credit bid in relation to the aggregate amount of Obligations so credit bid) in the Collateral that is the subject of such credit bid or purchase (or in the Equity Interests of the any entities that are used to consummate such credit bid or purchase) and (ii) Agent, based upon the instruction of the Required Lenders, may accept non-cash consideration, including debt and equity securities issued by any entities used to consummate such purchase and in connection therewith Agent may reduce the Obligations owed to the Lenders and the Bank Product Providers based upon the value of such non-cash consideration. Except as provided above, Agent will not execute and deliver a release of any Lien on any Collateral without the prior written authorization of (y) if the release is of all or substantially all of the Collateral, all of the Lenders (without requiring the authorization of the Bank Product Providers), or (z) otherwise, the Required Lenders (without requiring the authorization of the Bank Product Providers). Upon request by Agent or Borrower at any time, the Lenders will (and if so requested, the Bank Product Providers will) confirm in writing Agent's authority to release any such Liens on particular types or items of Collateral pursuant to this Section 15.11; provided, that (1) anything to the contrary contained in any of the Loan Documents notwithstanding, Agent shall not be

required to execute any document or take any action necessary to evidence such release on terms that, in Agent's opinion, could expose Agent to liability or create any obligation or entail any consequence other than the release of such Lien without recourse, representation, or warranty, and (2) such release shall not in any manner discharge, affect, or impair the Obligations or any Liens (other than those expressly released) upon (or obligations of Borrower in respect of) any and all interests retained by Borrower, including, the proceeds of any sale, all of which shall continue to constitute part of the Collateral. Each Lender further hereby irrevocably authorize (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to irrevocably authorize) Agent, at its option and in its sole discretion, to subordinate any Lien granted to or held by Agent under any Loan Document to the holder of any Permitted Lien on such property if such Permitted Lien secures Permitted Purchase Money Indebtedness.

- (b) Agent shall have no obligation whatsoever to any of the Lenders (or the Bank Product Providers) (i) to verify or assure that the Collateral exists or is owned by Parent or its Subsidiaries or is cared for, protected, or insured or has been encumbered, (ii) to verify or assure that Agent's Liens have been properly or sufficiently or lawfully created, perfected, protected, or enforced or are entitled to any particular priority, (iii) to impose, maintain, increase, reduce, implement, or eliminate any particular reserve hereunder or to determine whether the amount of any reserve is appropriate or not, or (iv) to exercise at all or in any particular manner or under any duty of care, disclosure or fidelity, or to continue exercising, any of the rights, authorities and powers granted or available to Agent pursuant to any of the Loan Documents, it being understood and agreed that in respect of the Collateral, or any act, omission, or event related thereto, subject to the terms and conditions contained herein, Agent may act in any manner it may deem appropriate, in its sole discretion given Agent's own interest in the Collateral in its capacity as one of the Lenders and that Agent shall have no other duty or liability whatsoever to any Lender (or Bank Product Provider) as to any of the foregoing, except as otherwise expressly provided herein.
- (c) Agent and Lenders agree that all Reorganization Securities to be distributed under a Plan will be remitted to Agent and promptly distributed to the holders of the Obligations according to the priorities set forth in Section 2.12(a)(ii).

15.12. Restrictions on Actions by Lenders; Sharing of Payments.

- (a) Each of the Lenders agrees that it shall not, without the express written consent of Agent, and that it shall, to the extent it is lawfully entitled to do so, upon the written request of Agent, set off against the Obligations, any amounts owing by such Lender to Parent Borrower or its Subsidiaries or any deposit accounts of Parent or its Subsidiaries now or hereafter maintained with such Lender. Each of the Lenders further agrees that it shall not, unless specifically consented to or requested to do so in writing by Agent, (i) take or cause to be taken any action to enforce any Loan Document against Borrower or any Guarantor, including, without limitation, the commencement of any legal or equitable proceedings, (ii) to take or cause to be taken any action to foreclose any Lien on any security interest in any of the Collateral or otherwise enforce any security interest in, any of the Collateral, including, without limitation, seeking relief from the automatic stay, seeking adequate protection of its interests in the Collateral, (iii) directly or indirectly, provide, offer to provide, or support any financing to any Loan Party under Section 364 of the Bankruptcy Code secured by a Lien senior to or *pari passu* with the Liens securing the Obligations or (iv) make any objection to any action or inaction taken by Agent during an Insolvency Proceeding.
- (b) Notwithstanding anything herein to the contrary, Sponsor Affiliated Entities may contribute their Obligations hereunder to Parent in exchange for Qualified Equity Interests in Parent and, in connection therewith, immediately and without further action by any

Person, such Obligations shall immediately be deemed satisfied and cancelled, and the applicable Sponsor Affiliated Entity shall execute any and all documents reasonably required by Agent to reflect such contribution and cancellation (including without limitation an undertaking to remain obligated under <u>Section 15.7</u> as if such Sponsor Affiliated Entity remained a Lender with a Pro Rata Share equal to such Obligations with respect to actions or other matters arising on or prior to the date of such exchange).

- 15.13. Agency for Perfection. Agent hereby appoints each other Lender (and each Bank Product Provider) as its agent (and each Lender hereby accepts (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to accept) such appointment) for the purpose of perfecting Agent's Liens in assets which, in accordance with Article 8 or Article 9, as applicable, of the Code can be perfected by possession or control. Should any Lender obtain possession or control of any such Collateral, such Lender shall notify Agent thereof, and, promptly upon Agent's request therefor shall deliver possession or control of such Collateral to Agent or in accordance with Agent's instructions.
- 15.14. **Payments by Agent to the Lenders**. All payments to be made by Agent to the Lenders (or Bank Product Providers) shall be made by bank wire transfer of immediately available funds pursuant to such wire transfer instructions as each party may designate for itself by written notice to Agent. Concurrently with each such payment, Agent shall identify whether such payment (or any portion thereof) represents principal, premium, fees, or interest of the Obligations.
- 15.15. Concerning the Collateral and Related Loan Documents. Each member of the Lender Group authorizes and directs Agent to enter into this Agreement and the other Loan Documents. Each member of the Lender Group agrees (and by entering into a Bank Product Agreement, each Bank Product Provider shall be deemed to agree) that any action taken by Agent in accordance with the terms of this Agreement or the other Loan Documents relating to the Collateral and the exercise by Agent of its powers set forth therein or herein, together with such other powers that are reasonably incidental thereto, shall be binding upon all of the Lenders (and such Bank Product Provider).
- 15.16. Financial Examination Reports; Confidentiality; Disclaimers by Lenders; Other Reports and Information. By becoming a party to this Agreement, each Lender:
- (a) is deemed to have requested that Agent furnish such Lender, promptly after it becomes available, a copy of each financial examination report respecting Parent or its Subsidiaries (each, a "Report") prepared by or at the request of Agent, and Agent shall so furnish each Lender with such Reports,
- (b) expressly agrees and acknowledges that Agent does not (i) make any representation or warranty as to the accuracy of any Report, and (ii) shall not be liable for any information contained in any Report,
- (c) expressly agrees and acknowledges that the Reports are not comprehensive audits or examinations, that Agent or other party performing any financial examination will inspect only specific information regarding Parent and its Subsidiaries and will rely significantly upon Parent's and its Subsidiaries' books and records, as well as on representations of Borrower's personnel,
- (d) agrees to keep all Reports and other material, non-public information regarding Parent and its Subsidiaries and their operations, assets, and existing and contemplated business plans in a confidential manner in accordance with Section 17.9, and

- (e) without limiting the generality of any other indemnification provision contained in this Agreement, agrees: (i) to hold Agent and any other Lender preparing a Report harmless from any action the indemnifying Lender may take or fail to take or any conclusion the indemnifying Lender may reach or draw from any Report in connection with any loans or other credit accommodations that the indemnifying Lender has made or may make to Borrower, or the indemnifying Lender's participation in, or the indemnifying Lender's purchase of, a loan or loans of Borrower, and (ii) to pay and protect, and indemnify, defend and hold Agent, and any such other Lender preparing a Report harmless from and against, the claims, actions, proceedings, damages, costs, expenses, and other amounts (including, attorneys' fees and costs) incurred by Agent and any such other Lender preparing a Report as the direct or indirect result of any third parties who might obtain all or part of any Report through the indemnifying Lender.
- (f) In addition to the foregoing, (w) if Agent receives notice of a Permitted Acquisition or of a transaction between Parent or its Subsidiaries permitted under Section 6.10(a) or (x) any Lender may from time to time request of Agent in writing that Agent provide to such Lender a copy of any report or document provided by Parent or its Subsidiaries to Agent that has not been contemporaneously provided by Parent or such Subsidiary to such Lender, and, upon receipt of such request, Agent promptly shall provide a copy of same to such Lender, (y) to the extent that Agent is entitled, under any provision of the Loan Documents, to request additional reports or information from Parent or its Subsidiaries, any Lender may, from time to time, reasonably request Agent to exercise such right as specified in such Lender's notice to Agent, whereupon Agent promptly shall request of Borrower the additional reports or information reasonably specified by such Lender, and, upon receipt thereof from Parent or such Subsidiary, Agent promptly shall provide a copy of same to such Lender, and (z) any time that Agent renders to Borrower a statement regarding the Loan Account, Agent shall send a copy of such statement to each Lender.
- 15.17. Several Obligations; No Liability. Notwithstanding that certain of the Loan Documents now or hereafter may have been or will be executed only by or in favor of Agent in its capacity as such, and not by or in favor of the Lenders, any and all obligations on the part of Agent (if any) to make any credit available hereunder shall constitute the several (and not joint) obligations of the respective Lenders on a ratable basis, according to their respective Commitments, to make an amount of such credit not to exceed, in principal amount, at any one time outstanding, the amount of their respective Commitments. Nothing contained herein shall confer upon any Lender any interest in, or subject any Lender to any liability for, or in respect of, the business, assets, profits, losses, or liabilities of any other Lender. Each Lender shall be solely responsible for notifying its Participants of any matters relating to the Loan Documents to the extent any such notice may be required, and no Lender shall have any obligation, duty, or liability to any Participant of any other Lender. Except as provided in Section 15.7, no member of the Lender Group shall have any liability for the acts of any other member of the Lender Group. No Lender shall be responsible to Borrower or any other Person for any failure by any other Lender (or Bank Product Provider) to fulfill its obligations to make credit available hereunder, nor to advance for such Lender (or Bank Product Provider) hereunder or in connection with the financing contemplated herein.

6. WITHHOLDING TAXES.

16.1. Payments.

(a) All payments made by Borrower hereunder or under any note or other Loan Document will be made without setoff, counterclaim, or other defense. In addition, all such payments will be made free and clear of, and without deduction or withholding for, any

present or future Indemnified Taxes, and in the event any deduction or withholding of Indemnified Taxes is required, Borrower shall comply with the next sentence of this Section 16.1. If any Indemnified Taxes are so levied or imposed, Borrower agrees to pay the full amount of such Indemnified Taxes and such additional amounts as may be necessary so that every payment of all amounts due under this Agreement, any note, or Loan Document, including any amount paid pursuant to this Section 16.1 after withholding or deduction for or on account of any Indemnified Taxes, will not be less than the amount provided for herein provided, however, that Borrower shall not be required to increase any such amounts payable to Agent or any Lender (i) on account of Taxes that would not have been imposed but for such Lender changing its residence, place of organization, principal place of business or its branch or lending office participating in the transactions set forth herein; provided, however, this clause (i) shall not apply to the extent the amount such Lender would be entitled to receive under such circumstances does not exceed the amount such Lender would have been entitled to receive in the absence of the change or if the change was made at the request of any Loan Party; or (ii) if the increase in such amount payable results from Agent's or such Lender's own bad faith, willful misconduct or gross negligence (as finally determined by a court of competent jurisdiction). Borrower will furnish to Agent as promptly as possible after the date the payment of any Indemnified Tax imposed with respect to any Loan Document is due pursuant to applicable law, certified copies of tax receipts evidencing such payment by Borrower. Borrower agrees to pay any present or future stamp, value added or documentary taxes or any other excise or property taxes, charges, or similar levies ("Other Taxes") that arise from any payment made hereunder or from the execution, delivery, performance, recordation, or filing of, or otherwise with respect to t

(b) The Loan Parties shall jointly and severally indemnify each Indemnified Person (as defined in Section 10.3) (collectively a "Tax Indemnitee") for the full amount of Indemnified Taxes and Other Taxes arising in connection with this Agreement or any other Loan Document or breach thereof by any Loan Party (including, without limitation, any Indemnified Taxes and Other Taxes imposed or asserted on, or attributable to, amounts payable under this Section 16; provided, however, that for purposes of any Indemnified Taxes or Other Taxes imposed or asserted on, or attributable to, amounts payable under this Section 16, each Tax Indemnitee agrees to be subject to the definition of Excluded Taxes as though it were a Lender) imposed on, or paid by, such Tax Indemnitee and all reasonable costs and expenses related thereto (including fees and disbursements of attorneys and other tax professionals), as and when they are incurred and irrespective of whether suit is brought, whether or not such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental AuthorityBody; provided, however, that Borrower shall not be required to indemnify any Tax Indemnitee with respect to (i) any Taxes described in Section 16.1(a)(i); or (ii) if the increase in such amount payable results from such Tax Indemnitee's own bad faith, willful misconduct or gross negligence (as finally determined by a court of competent jurisdiction). The obligations of the Loan Parties under this Section 16 shall survive the termination of this Agreement, the resignation and replacement of the Agent, and the repayment of the Obligations.

16.2. **Exemptions**.

(a) If a Lender or Participant is entitled to claim an exemption or reduction from United States withholding tax, such Lender or Participant agrees with and in favor of Agent, to deliver to Agent upon a request by Agent with respect thereto (or, in the case of a Participant, to the Lender granting the participation only) one of the following before receiving its first payment under this Agreement:

- (i) if such Lender or Participant is entitled to claim an exemption from United States withholding tax pursuant to the portfolio interest exception, (A) a statement of the Lender or Participant, signed under penalty of perjury, that it is not a (I) a "bank" as described in Section 881(c)(3)(A) of the IRC, (II) a 10% shareholder of Borrower (within the meaning of Section 871(h)(3)(B) of the IRC), or (III) a controlled foreign corporation related to Borrower within the meaning of Section 864(d)(4) of the IRC, and (B) a properly completed and executed IRS Form W-8BEN, IRS Form W-8BEN-E, or Form W-8IMY (with proper attachments), as applicable;
- (ii) if such Lender or Participant is entitled to claim an exemption from, or a reduction of, withholding tax under a United States tax treaty, a properly completed and executed copy of IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable;
- (iii) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding tax because it is effectively connected with a United States trade or business of such Lender, a properly completed and executed copy of IRS Form W-8ECI;
- (iv) if such Lender or Participant is entitled to claim that interest paid under this Agreement is exempt from United States withholding tax because such Lender or Participant serves as an intermediary, a properly completed and executed copy of IRS Form W-8IMY (with proper attachments); or
- (v) a properly completed and executed copy of any other form or forms, including IRS Form W-9, as may be required under the IRC or other laws of the United States as a condition to exemption from, or reduction of, United States withholding or backup withholding tax.
- (b) Each Lender or Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to promptly notify Agent (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.
- (c) If a Lender or Participant claims an exemption from withholding tax in a jurisdiction other than the United States, such Lender or such Participant agrees with and in favor of Agent, to deliver to Agent (or, in the case of a Participant, to the Lender granting the participation only) any such form or forms, as may be required under the laws of such jurisdiction as a condition to exemption from, or reduction of, foreign withholding or backup withholding tax before receiving its first payment under this Agreement, but only if such Lender or such Participant is legally able to deliver such forms, provided, that nothing in this Section 16.2(c) shall require a Lender or Participant to disclose any information that it deems to be confidential (including without limitation, its tax returns). Each Lender and each Participant shall provide new forms (or successor forms) upon the expiration or obsolescence of any previously delivered forms and to promptly notify Agent (or, in the case of a Participant, to the Lender granting the participation only) of any change in circumstances which would modify or render invalid any claimed exemption or reduction.
- (d) If a Lender or Participant claims exemption from, or reduction of, withholding tax and such Lender or Participant sells, assigns, grants a participation in, or otherwise transfers all or part of the Obligations of Borrower to such Lender or Participant, such Lender or Participant agrees to notify Agent (or, in the case of a sale of a participation interest, to the Lender granting the participation only) of the percentage amount in which it is no longer the

beneficial owner of Obligations of Borrower to such Lender or Participant. To the extent of such percentage amount, Agent will treat such Lender's or such Participant's documentation provided pursuant to Section 16.2(a) or 16.2(c) as no longer valid. With respect to such percentage amount, such Participant or Assignee may provide new documentation, pursuant to Section 16.2(a) or 16.2(c), if applicable. Borrower agrees that each Participant shall be entitled to the benefits of this Section 16 with respect to its participation in any portion of the Commitments and the Obligations so long as such Participant complies with the obligations set forth in this Section 16 with respect thereto.

(e) If a payment made to Agent, any Lender or any other recipient of any payment to be made by or on account of any Obligation, Letters of Credit or Loan Documents of Borrower hereunder (each, a "Recipient") would be subject to U.S. federal withholding Tax imposed under FATCA if such Recipient fails to comply with the applicable reporting requirements of FATCA, such Recipient shall deliver to Borrower and Agent (A) a certification of foreign status signed by the chief financial officer, principal accounting officer, treasurer or controller and (B) other documentation reasonably requested by Borrower or Agent sufficient for Agent or Borrower to comply with their obligations under FATCA and to determine if such Recipient is exempt from withholding under FATCA.

16.3. **Reductions**.

- (a) If a Lender or a Participant is entitled to a reduction in the applicable withholding tax, Agent (or, in the case of a Participant, to the Lender granting the participation) may withhold from any interest payment to such Lender or such Participant an amount equivalent to the applicable withholding tax after taking into account such reduction. If the forms or other documentation required by Section 16.2(a) or 16.2(c) are not delivered to Agent (or, in the case of a Participant, to the Lender granting the participation) or Borrower, then Agent (or, in the case of a Participant, to the Lender granting the participation) may withhold from any interest payment to such Lender or such Participant not providing such forms or other documentation an amount equivalent to the applicable withholding tax.
- (b) If the IRS or any other Governmental AuthorityBody of the United States or other jurisdiction asserts a claim that Agent (or, in the case of a Participant, to the Lender granting the participation) did not properly withhold tax from amounts paid to or for the account of any Lender or any Participant due to a failure on the part of the Lender or any Participant (because the appropriate form was not delivered, was not properly executed, or because such Lender failed to notify Agent (or such Participant failed to notify the Lender granting the participation) of a change in circumstances which rendered the exemption from, or reduction of, withholding tax ineffective, or for any other reason) such Lender shall indemnify and hold Agent harmless (or, in the case of a Participant, such Participant shall indemnify and hold the Lender granting the participation), as tax or otherwise, including penalties and interest, and including any taxes imposed by any jurisdiction on the amounts payable to Agent (or, in the case of a Participant, to the Lender granting the participation only) under this Section 16, together with all costs and expenses (including attorneys' fees and expenses). The obligation of the Lenders and the Participants under this subsection shall survive the payment of all Obligations and the resignation or replacement of Agent.
- 16.4. **Refunds**. If Agent or a Lender determines, in its sole discretion, that it has received a refund of any Indemnified Taxes to which Borrower has paid additional amounts pursuant to this <u>Section 16</u>, so long as no Default or Event of Default has occurred and is continuing, it shall pay over such refund to Borrower (but only to the extent of payments made, or additional amounts paid, by Borrower under this <u>Section 16</u> with respect to Indemnified Taxes

giving rise to such a refund), net of all out-of-pocket expenses of Agent or such Lender and without interest (other than any interest paid by the applicable Governmental AuthorityBody with respect to such a refund); provided, that Borrower, upon the request of Agent or such Lender, agrees to repay the amount paid over to Borrower (plus any penalties, interest or other charges, imposed by the applicable Governmental AuthorityBody, other than such penalties, interest or other charges imposed as a result of the willful misconduct or gross negligence of Agent hereunder) to Agent or such Lender in the event Agent or such Lender is required to repay such refund to such Governmental AuthorityBody. Notwithstanding anything in this Agreement to the contrary, this Section 16 shall not be construed to require Agent or any Lender to make available its tax returns (or any other information which it deems confidential) to Borrower or any other Person.

17. GENERAL PROVISIONS.

- 17.1. **Effectiveness**. This Agreement shall be binding and deemed effective when executed by Parent, Borrower, Agent, and each Lender whose signature is provided for on the signature pages hereof.
- 17.2. <u>Section Headings</u>. Headings and numbers have been set forth herein for convenience only. Unless the contrary is compelled by the context, everything contained in each Section applies equally to this entire Agreement.
- 17.3. <u>Interpretation</u>. Neither this Agreement nor any uncertainty or ambiguity herein shall be construed against the Lender Group or Parent or Borrower, whether under any rule of construction or otherwise. On the contrary, this Agreement has been reviewed by all parties and shall be construed and interpreted according to the ordinary meaning of the words used so as to accomplish fairly the purposes and intentions of all parties hereto.
- 17.4. <u>Severability of Provisions</u>. Each provision of this Agreement shall be severable from every other provision of this Agreement for the purpose of determining the legal enforceability of any specific provision.
- 17.5. Bank Product Providers. Each Bank Product Provider in its capacity as such shall be deemed a third party beneficiary hereof and of the provisions of the other Loan Documents for purposes of any reference in a Loan Document to the parties for whom Agent is acting. Agent hereby agrees to act as agent for such Bank Product Providers and, by virtue of entering into a Bank Product Agreement, the applicable Bank Product Provider shall be automatically deemed to have appointed Agent as its agent and to have accepted the benefits of the Loan Documents. It is understood and agreed that the rights and benefits of each Bank Product Provider under the Loan Documents consist exclusively of such Bank Product Provider's being a beneficiary of the Liens and security interests (and, if applicable, guarantees) granted to Agent and the right to share in payments and collections out of the Collateral as more fully set forth herein. In addition, each Bank Product Provider, by virtue of entering into a Bank Product Agreement, shall be automatically deemed to have agreed that Agent shall have the right, but shall have no obligation, to establish, maintain, relax, or release reserves in respect of the Bank Product Obligations and that if reserves are established there is no obligation on the part of Agent to determine or insure whether the amount of any such reserve is appropriate or not. In connection with any such distribution of payments or proceeds of Collateral, Agent shall be entitled to assume no amounts are due or owing to any Bank Product Provider unless such Bank Product Provider has provided a written certification (setting forth a reasonably detailed calculation) to Agent as to the amounts that are due and owing to it and such written certification is received by Agent a reasonable period of time prior to the making of such distribution. Agent shall have no obligation to calculate the amount due and payable with respect to any Bank

Products, but may rely upon the written certification of the amount due and payable from the applicable Bank Product Provider. In the absence of an updated certification, Agent shall be entitled to assume that the amount due and payable to the applicable Bank Product Provider is the amount last certified to Agent by such Bank Product Provider as being due and payable (less any distributions made to such Bank Product Provider on account thereof). Borrower may obtain Bank Products from any Bank Product Provider, although Borrower is not required to do so. Borrower acknowledges and agrees that no Bank Product Provider has committed to provide any Bank Products and that the providing of Bank Products by any Bank Product Provider is in the sole and absolute discretion of such Bank Product Provider. Notwithstanding anything to the contrary in this Agreement or any other Loan Document, no provider or holder of any Bank Product shall have any voting or approval rights hereunder (or be deemed a Lender) solely by virtue of its status as the provider or holder of such agreements or products or the Obligations owing thereunder, nor shall the consent of any such provider or holder be required (other than in their capacities as Lenders, to the extent applicable) for any matter hereunder or under any of the other Loan Documents, including as to any matter relating to the Collateral or the release of Collateral or Guarantors.

- 17.6. **Debtor-Creditor Relationship**. The relationship between the Lenders and Agent, on the one hand, and the Loan Parties, on the other hand, is solely that of creditor and debtor. No member of the Lender Group has (or shall be deemed to have) any fiduciary relationship or duty to any Loan Party arising out of or in connection with the Loan Documents or the transactions contemplated thereby, and there is no agency or joint venture relationship between the members of the Lender Group, on the one hand, and the Loan Parties, on the other hand, by virtue of any Loan Document or any transaction contemplated therein.
- 17.7. Counterparts; Electronic Execution. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together, shall constitute but one and the same Agreement. Delivery of an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission shall be equally as effective as delivery of an original executed counterpart of this Agreement. Any party delivering an executed counterpart of this Agreement by telefacsimile or other electronic method of transmission also shall deliver an original executed counterpart of this Agreement but the failure to deliver an original executed counterpart shall not affect the validity, enforceability, and binding effect of this Agreement. The foregoing shall apply to each other Loan Document mutatis mutandis.

17.8. Revival and Reinstatement of Obligations; Certain Waivers.

(a) If any member of the Lender Group or any Bank Product Provider repays, refunds, restores, or returns in whole or in part, any payment or property (including any proceeds of Collateral) previously paid or transferred to such member of the Lender Group or such Bank Product Provider in full or partial satisfaction of any Obligation or on account of any other obligation of any Loan Party under any Loan Document or any Bank Product Agreement, because the payment, transfer, or the incurrence of the obligation so satisfied is asserted or declared to be void, voidable, or otherwise recoverable under any law relating to creditors' rights, including provisions of the Bankruptcy Code relating to fraudulent transfers, preferences, or other voidable or recoverable obligations or transfers (each, a "Voidable Transfer"), or because such member of the Lender Group or Bank Product Provider elects to do so on the reasonable advice of its counsel in connection with a claim that the payment, transfer, or incurrence is or may be a Voidable Transfer, then, as to any such Voidable Transfer, or the amount thereof that such member of the Lender Group or Bank Product Provider elects to repay, restore, or return (including pursuant to a settlement of any claim in respect thereof), and as to all

reasonable costs, expenses, and attorneys' fees of such member of the Lender Group or Bank Product Provider related thereto, (i) the liability of the Loan Parties with respect to the amount or property paid, refunded, restored, or returned will automatically and immediately be revived, reinstated, and restored and will exist and (ii) Agent's Liens securing such liability shall be effective, revived, and remain in full force and effect, in each case, as fully as if such Voidable Transfer had never been made. If, prior to any of the foregoing, (A) Agent's Liens shall have been released or terminated or (B) any provision of this Agreement shall have been terminated or cancelled, Agent's Liens, or such provision of this Agreement, shall be reinstated in full force and effect and such prior release, termination, cancellation or surrender shall not diminish, release, discharge, impair or otherwise affect the obligation of any Loan Party in respect of such liability or any Collateral securing such liability.

(b) Anything to the contrary contained herein notwithstanding, if Agent or any Lender accepts a guaranty of only a portion of the Obligations pursuant to any guaranty, Borrower hereby waives its right under Section 2822(a) of the California Civil Code or any similar laws of any other applicable jurisdiction to designate the portion of the Obligations satisfied by the applicable guarantor's partial payment.

17.9. **Confidentiality**.

(a) Agent and Lenders each individually (and not jointly or jointly and severally) agree that material, non-public information regarding Parent and its Subsidiaries, their operations, assets, and existing and contemplated business plans ("Confidential Information") shall be treated by Agent and the Lenders in a confidential manner, and shall not be disclosed by Agent and the Lenders to Persons who are not parties to this Agreement, except: (i) to attorneys for and other advisors, accountants, auditors, and consultants to any member of the Lender Group and to employees, directors and officers of any member of the Lender Group (the Persons in this clause (i), "Lender Group Representatives") on a "need to know" basis in connection with this Agreement, the other Loan Documents and the transactions contemplated hereby or thereby and on a confidential basis; provided, that any such Person shall have agreed to receive such information such as the terms are the terms of the contemplated hereby or thereby and on a confidential basis; provided, that any such Person shall have agreed to receive such information such as the terms of the contemplated hereby or thereby and on a confidential basis; provided, that any such Person shall have agreed to receive such information such as the contemplated hereby or thereby and on a confidential basis; provided, that any such Person shall have agreed to receive such information such as the contemplated hereby or the contemplated her of this Section 17.9, (ii) to Subsidiaries and Affiliates of any member of the Lender Group (including the Bank Product Providers), provided that any such Subsidiary or Affiliate shall have agreed to receive such information hereunder subject to the terms of this Section 17.9 or is otherwise obligated to maintain the confidentiality of such Confidential Information in accordance with such member of the Lender Group's policies and procedures, (iii) as may be required by any government agency or other regulatory authority with legal authority to regulate the business and operations of any Lender or any Affiliate of a Lender so long as (if legally permitted and reasonably practicable) such authorities are informed of the confidential nature of such information, (iv) as may be required by statute, decision, or judicial or administrative order, rule, or regulation; provided that (x) prior to any disclosure under this clause (iv), the disclosing party agrees to provide Borrower with prior notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior notice to Borrower pursuant to the terms of the applicable statute, decision, or judicial or administrative order, rule, or regulation and (y) any disclosure under this clause (iv) shall be limited to the portion of the Confidential Information as may be required by such statute, decision, or judicial or administrative order, rule, or regulation, (v) as may be agreed to in advance in writing by Borrower, (vi) as requested or required by any Governmental Authority Body pursuant to any subpoena or other legal process, provided, that, (x) prior to any disclosure under this clause (vi) the disclosing party agrees to provide Borrower with prior written notice thereof, to the extent that it is practicable to do so and to the extent that the disclosing party is permitted to provide such prior written notice to Borrower pursuant to the terms of the subpoena or other legal process and (y) any disclosure under this clause (vi) shall be limited to the portion of the Confidential Information as may be required by such Governmental

AuthorityBody pursuant to such subpoena or other legal process, (vii) as to any such information that is or becomes generally available to the public (other than as a result of prohibited disclosure by Agent or the Lenders or the Lender Group Representatives), (viii) in connection with any assignment, participation or pledge of any Lender's interest under this Agreement, provided that prior to receipt of Confidential Information any such assignee, participant, or pledgee shall have agreed in writing to receive such Confidential Information either subject to the terms of this Section 17.9 or pursuant to confidentiality requirements substantially similar to those contained in this Section 17.9 (and such Person may disclose such Confidential Information to Persons employed or engaged by them as described in clause (i) above), (ix) in connection with any litigation or other adversary proceeding involving parties hereto which such litigation or adversary proceeding involves claims related to the rights or duties of such parties under this Agreement or the other Loan Documents; provided, that, prior to any disclosure to any Person (other than any Loan Party, Agent, any Lender, any of their respective counsel) under this clause (ix) with respect to litigation involving any Person (other than any Loan Party, Agent, any Lender, any of their respective Affiliates, or their respective counsel), the disclosing party agrees to provide Borrower with prior written notice thereof, and (x) in connection with, and to the extent reasonably necessary for, the exercise of any secured creditor remedy under this Agreement or under any other Loan Document.

- (b) Anything in this Agreement to the contrary notwithstanding, Agent may disclose information concerning the terms and conditions of this Agreement and the other Loan Documents to loan syndication and pricing reporting services or in its marketing or promotional materials, with such information to consist of deal terms and other information customarily found in such publications or marketing or promotional materials and may otherwise use the name, logos, and other insignia of Borrower or the other Loan Parties and the Commitments provided hereunder in any "tombstone" or other advertisements, on its website or in other marketing materials of the Agent.
- (c) The Loan Parties hereby acknowledge that Agent or its Affiliates may make available to the Lenders materials or information provided by or on behalf of Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks, SyndTrak or another similar electronic system (the "Platform") and certain of the Lenders may be "public-side" Lenders (i.e., Lenders that do not wish to receive material non-public information with respect to the Loan Parties or their securities) (each, a "Public Lender"). The Loan Parties shall be deemed to have authorized Agent and its Affiliates and the Lenders to treat Borrower Materials marked "PUBLIC" or otherwise at any time filed with the SEC as not containing any material non-public information with respect to the Loan Parties or their securities for purposes of United States federal and state securities laws. All Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated as "Public Investor" (or another similar term). Agent and its Affiliates and the Lenders shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" or that are not at any time filed with the SEC as being suitable only for posting on a portion of the Platform not marked as "Public Investor" (or such other similar term).
- 17.10. <u>Survival</u>. All representations and warranties made by the Loan Parties in the Loan Documents and in the certificates or other instruments delivered in connection with or pursuant to this Agreement or any other Loan Document shall be considered to have been relied upon by the other parties hereto and shall survive the execution and delivery of the Loan Documents and the making of any Loans and issuance of any Letters of Credit, regardless of any investigation made by any such other party or on its behalf and notwithstanding that Agent, Issuing Bank, or any Lender may have had notice or knowledge of any Default or Event of Default or incorrect representation or warranty at the time any credit is extended hereunder, and shall continue in full force and effect as long as the principal of, or any accrued interest on, any

Loan or any fee or any other amount payable under this Agreement is outstanding or unpaid or any Letter of Credit is outstanding and so long as the Commitments have not expired or been terminated.

- 17.11. Patriot Act. Each Lender that is subject to the requirements of the Patriot Act hereby notifies Borrower that pursuant to the requirements of the Act, it is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow such Lender to identify Borrower in accordance with the Patriot Act. In addition, if Agent is required by law or regulation or internal policies to do so, it shall have the right to periodically conduct (a) Patriot Act searches, OFAC/PEP searches, and customary individual background checks for the Loan Parties and (b) OFAC/PEP searches and customary individual background checks for the Loan Parties' senior management and key principals, and Borrower agrees to cooperate in respect of the conduct of such searches and further agrees that the reasonable costs and charges for such searches shall constitute Lender Group Expenses hereunder and be for the account of Borrower.
- 17.12. <u>Integration</u>. This Agreement, together with the other Loan Documents, reflects the entire understanding of the parties with respect to the transactions contemplated hereby and shall not be contradicted or qualified by any other agreement, oral or written, before the Seventh Amendment Effective Date. The foregoing to the contrary notwithstanding, all Bank Product Agreements, if any, are independent agreements governed by the written provisions of such Bank Product Agreements, which will remain in full force and effect, unaffected by any repayment, prepayments, acceleration, reduction, increase, or change in the terms of any credit extended hereunder, except as otherwise expressly provided in such Bank Product Agreement.

[Signature pages to follow.]

PARENT:	LIVEVOX HOLDINGS, INC., a Delaware corporation
	By: Name: Title:
BORROWER:	LIVEVOX, INC., a Delaware corporation
	By: Name: Title:
074658.16087/130783594v.4	

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed and delivered as of the date first

above written.

AGENT AND SOLE INITIAL LENDER:

PNC BANK, NATIONAL ASSOCIATION, a national banking association

By: Name: Christopher Duranto Title: Vice President

Schedule C-1

Commitments

Lender	Revolver Commitment	Outstanding Closing Date Term Loan, First Amendment Term Loan and Third Amendment Term Loan as of SeventhEighth Amendment Effective Date
PNC Bank, National Association	\$4,132,039.00	\$ 46,059,047.22 45,363,729.48
Bank Hapoalim B.M.	\$867,961.00	\$ 9,675,000.78 9,528,908.52
All Lenders	\$5,000,000.00	\$ 55,734,048.00 <u>54,892,638.00</u>

Schedule 1.1

As used in the Agreement, the following terms shall have the following definitions:

- "Accounting Changes" means changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants (or successor thereto or any agency with similar functions).
- "Acquired Indebtedness" means Indebtedness of a Person whose assets or Equity Interests are acquired by Parent or any of its Subsidiaries in a Permitted Acquisition; provided, that such Indebtedness (a) is either purchase money Indebtedness or a Capital Lease with respect to Equipment or mortgage financing with respect to Real Property, (b) was in existence prior to the date of such Permitted Acquisition, and (c) was not incurred in connection with, or in contemplation of, such Permitted Acquisition.
 - "Acquisition" has the meaning specified in the definition of Permitted Acquisition.
 - "Additional Documents" has the meaning specified therefor in Section 5.12 of the Agreement.
 - "Administrative Questionnaire" has the meaning specified therefor in Section 13.1(a) of the Agreement.
- "Advisory Agreement" means the Advisory Agreement, dated as of the Closing Date, by and between Parent, Borrower, LiveVox TopCo, LLC and GGC Administration, LLC, as in effect on the Closing Date.
 - "Affected Lender" has the meaning specified therefor in Section 2.17(b) of the Agreement.
- "Affiliate" means, as applied to any Person, any other Person who controls, is controlled by, or is under common control with, such Person. For purposes of this definition, "control" means the possession, directly or indirectly through one or more intermediaries, of the power to direct the management and policies of a Person, whether through the ownership of Equity Interests, by contract, or otherwise; provided, that, for purposes of Section 6.10 of the Agreement: (a) any Person which owns directly or indirectly 10% or more of the Equity Interests having ordinary voting power for the election of directors or other members of the governing body of a Person or 10% or more of the partnership or other ownership interests of a Person (other than as a limited partner of such Person) shall be deemed an Affiliate of such Person, (b) each director (or comparable manager) of a Person shall be deemed to be an Affiliate of such Person, and (c) each partnership in which a Person is a general partner shall be deemed an Affiliate of such Person.
 - "Agent" has the meaning specified therefor in the preamble to the Agreement.
 - "Agent-Related Persons" means Agent, together with its Affiliates, officers, directors, employees, attorneys, and agents.
- "Agent's Liens" means the Liens granted by Parent, Borrower or their Subsidiaries to Agent under the Loan Documents and securing the Obligations.

Schedule 1.1 – Page 1

"Agreement" means the Credit Agreement to which this Schedule 1.1 is attached.

"Alternate Base Rate" means shall mean, for any day, a rate per annum equal to the highest of (a) the Base Rate in effect on such day, (b) the sum of the Federal Funds Open Overnight Bank Funding Rate in effect on such day plus one half of one percent (0.5%), and (c) the sum of the Daily LIBOR RateSimple SOFR in effect on such day plus one percent (1.0%), so long as a Daily LIBOR RateSimple SOFR is offered, ascertainable and not unlawful; provided, however, if the Alternate Base Rate as determined above would be less than zero, then such rate shall be deemed to be zero. Any change in the Alternate Base Rate (or any component thereof) shall take effect at the opening of business on the day such change occurs.

"Anti-Corruption Laws" means all laws, rules and regulations of any jurisdiction applicable to Parent or its Subsidiaries from time to time concerning or relating to bribery, corruption or money laundering.

"Applicable Affiliate Transaction Documents" means the agreements to which Borrower is a party evidencing the transactions permitted pursuant to Section 6.10(1) of the Agreement.

"Applicable Base Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"Applicable LIBOR Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"Applicable Margin" means (a) in the case of Swing Loans and all other Loans that are, Base Rate Loans, 3.754.00% percentage points (the percentage points in this subclause (a), the "Applicable Base Rate Margin"), and (b) in the case of Loans other than Swing Loans that are, in each case, LIBORTerm SOFR Rate Loans, 4.755.00% percentage points (the percentage points in this subclause (b), the "Applicable LIBORTerm SOFR Rate Margin").

"Applicable Percentage" means, as applied to Excess Cash Flow for any fiscal year, (a) fifty percent (50%) if the Leverage Ratio as of the end of such fiscal year is greater than 3.5 to 1.0, (b) twenty-five percent (25%) if the Leverage Ratio as of the end of such fiscal year is equal to or less than 3.5 to 1.0 but greater than 2.5 to 1.0 and (c) zero percent (0%) if the Leverage Ratio as of the end of such fiscal year is equal to or less than 2.5 to 1.0.

"Applicable Term SOFR Rate Margin" has the meaning set forth in the definition of Applicable Margin.

"Applicable Unused Line Fee Rate" means, as of any date of determination (a) prior to April 1, 2017, 0.50% per annum and (b) on and after April 1, 2017, the per annum rate set forth in the table below that corresponds to the daily amount of Availability:

Availability Amount	Applicable Unused Line Fee Rate
Greater than or equal to 50% of the Maximum Revolver Amount	0.50%
Less than 50% of the Maximum Revolver Amount	0.25%

"Application Event" means the occurrence of (a) a failure by Borrower to repay all of the Obligations in full on the Maturity Date, or (b) an Event of Default and the election by Agent or the Required Lenders to require that Distributions be applied pursuant to Section 2.12(a)(ii) of the Agreement or otherwise accelerate all or any material portion of the Obligations.

"Assignee" has the meaning specified therefor in Section 13.1(a) of the Agreement.

"Assignment and Acceptance" means an Assignment and Acceptance Agreement substantially in the form of Exhibit A-1 to the Agreement.

"<u>Authorized Person</u>" means any one of the individuals identified on <u>Schedule A-1</u> to the Agreement, as such schedule is updated from time to time by written notice from Borrower to Agent accompanied by a certificate as to the incumbency of such individual as an officer of Borrower executed by the corporate secretary or other similar officer of Borrower.

"Availability" means, as of any date of determination, the amount that Borrower is entitled to borrow as Revolving Loans under Section 2.1 of the Agreement (after giving effect to the then outstanding Revolver Usage).

"Bank Product" means any one or more of the following financial products or accommodations extended to Parent or its Subsidiaries by a Bank Product Provider: (a) credit cards (including commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards")), (b) credit card processing services, (c) debit cards, (d) stored value cards, (e) Cash Management Services, or (f) transactions under Hedge Agreements.

"Bank Product Agreements" means those agreements entered into from time to time by Parent or its Subsidiaries with a Bank Product Provider in connection with the obtaining or providing of any of the Bank Products.

"Bank Product Collateralization" means providing cash collateral (pursuant to documentation reasonably satisfactory to Agent) to be held by Agent for the benefit of the Bank Product Providers (other than the Hedge Providers with respect to Hedge Obligations that are allowed by the applicable Hedge Provider to remain outstanding without being required to be repaid or cash collateralized) in an amount determined by Agent as sufficient to satisfy the reasonably estimated credit exposure with respect to the then existing Bank Product Obligations (other than Hedge Obligations as provided above).

"Bank Product Obligations" means (a) all obligations, liabilities, reimbursement obligations, fees, or expenses owing by Parent or its Subsidiaries to any Bank Product Provider pursuant to or evidenced by a Bank Product Agreement and irrespective of whether for the

payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, (b) all Hedge Obligations, and (c) all amounts that Agent or any Lender is obligated to pay to a Bank Product Provider as a result of Agent or such Lender purchasing participations from, or executing guarantees or indemnities or reimbursement obligations to, a Bank Product Provider with respect to the Bank Products provided by such Bank Product Provider to Parent or its Subsidiaries.

"Bank Product Provider" means PNC or any of its Affiliates, including each of the foregoing in its capacity, if applicable, as a Hedge Provider.

"Bank Product Reserves" means, as of any date of determination, those reserves that Agent deems necessary or appropriate to establish (based upon the Bank Product Providers' determination of the liabilities and obligations of Parent and its Subsidiaries in respect of Bank Product Obligations) in respect of Bank Products then provided or outstanding. Notwithstanding the foregoing, the aggregate amount of all Bank Product Reserves outstanding at any one time when taken together with the aggregate amount of all Protective Advances then outstanding, shall not exceed Maximum Applicable Reserve.

"Bankruptcy Code" means title 11 of the United States Code, as in effect from time to time.

"Base Rate" means the base commercial lending rate of PNC as publicly announced to be in effect from time to time, such rate to be adjusted automatically, without notice, on the effective date of any change in such rate. This rate of interest is determined from time to time by PNC as a means of pricing some loans to its customers and is neither tied to any external rate of interest or index nor does it necessarily reflect the lowest rate of interest actually charged by PNC to any particular class or category of customers of PNC.

"Base Rate Loan" means each portion of the Revolving Loans or the Term Loan that bears interest at a rate determined by reference to the Alternate Base Rate.

"Beneficial Owner" shall mean, for the Borrower, each of the following: (a) each individual, if any, who, directly or indirectly, owns 25% or more of such Borrower's Equity Interests; and (b) a single individual with significant responsibility to control, manage, or direct such Borrower.

"Benefit Plan" means a "defined benefit plan" (as defined in Section 3(35) of ERISA) for which Parent or any of its Subsidiaries or ERISA Affiliates has been an "employer" (as defined in Section 3(5) of ERISA) within the past six years.

"Board of Directors" means, as to any Person, the board of directors (or comparable managers) of such Person, or any committee thereof duly authorized to act on behalf of the board of directors (or comparable managers).

"Board of Governors" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" has the meaning specified therefor in the preamble to the Agreement.

"Borrower Materials" has the meaning specified therefor in Section 17.9(c) of the Agreement.

"Borrowing" means a borrowing consisting of Revolving Loans made on the same day by the Lenders (or Agent on behalf thereof), or by Swing Loan Lender in the case of a Swing Loan, or by Agent in the case of a Protective Advance.

"Business Day" means" shall mean any day other than Saturday or Sunday or a legal holiday on which commercial banks are authorized or required by https://linear.com/law-to-be-closed for business in East Brunswick, New Jersey-and, if the applicable; provided that when used in connection with an amount that bears interest at a rate based on SOFR or any direct or indirect calculation or determination of SOFR, the "Business Day-relates to any LIBOR Rate Loans;" means any such day must(hat is also be a day on which dealings are carried on in the London interbank market U.S. Government Securities Business Day.

'Capital Expenditures' means, with respect to any Person for any period, the amount of all expenditures by such Person and its Subsidiaries during such period that are capital expenditures as determined in accordance with GAAP, whether such expenditures are paid in cash or financed, but excluding, without duplication (a) expenditures made during such period in connection with the replacement, substitution, or restoration of assets or properties pursuant to 4-Section 2.12(d)(ii) of the Agreement, (b) with respect to the purchase price of assets that are purchased substantially contemporaneously with the trade-in of existing assets during such period, the amount that the gross amount of such purchase price is reduced by the credit granted by the seller of such assets for the assets being traded in at such time, (c) expenditures made during such period to consummate one or more Permitted Acquisitions, (d) expenditures made during such period to the extent made with the identifiable proceeds of an equity investment in Parent or any of its Subsidiaries by Sponsor which equity investment is made substantially contemporaneously with the making of the expenditure, (e) capitalized software development costs to the extent such costs are deducted from net earnings under the definition of EBITDA for such period, (f) expenditures during such period to the extent made out of the identifiable proceeds of insurance, condemnation awards, casualty awards, or property or asset sales or dispositions, (g) expenditures during such period that, pursuant to a written agreement, are reimbursable by a third Person (excluding Parent or any of its Affiliates) (provided that if such reimbursement has not been received before the end of the fiscal year when the relevant Capital Expenditure was made, such amount shall be included in Capital Expenditure calculation for the immediately succeeding fiscal year to the extent such reimbursement is not received in such immediately succeeding fiscal year), and (h) credits given to Parent or a Subsidiary with respect to used Equipment traded in contemporaneously with the purchase of new Equipment and such credits (or proceeds thereof) are applied to the purchase of such new Equipment.

"Capitalized Lease Obligation" means that portion of the obligations under a Capital Lease that is required to be capitalized in accordance with GAAP.

"Capital Lease" means a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

"Cares Act" means the Coronavirus Aid, Relief, and Economic Security Act, Pub. L. 116-136 and any amendment thereof, or administrative or other guidance or legislation published with respect thereto by any Governmental Authority Body.

"Cash Equivalents" means (a) marketable direct obligations issued by, or unconditionally guaranteed by, the United States or issued by any agency thereof and backed by the full faith and credit of the United States, in each case maturing within 1 year from the date of acquisition thereof, (b) marketable direct obligations issued or fully guaranteed by any state of the United States or any political subdivision of any such state or any public instrumentality thereof maturing within 1 year from the date of acquisition thereof and, at the time of acquisition,

having one of the two highest ratings obtainable from either Standard & Poor's Rating Group ("<u>S&P</u>") or Moody's Investors Service, Inc. ("<u>Moody's</u>"), (c) commercial paper maturing no more than 270 days from the date of creation thereof and, at the time of acquisition, having a rating of at least A-1 from S&P or at least P-1 from Moody's, (d) certificates of deposit, time deposits, overnight bank deposits or bankers' acceptances maturing within 1 year from the date of acquisition thereof issued by any bank organized under the laws of the United States or any state thereof or the District of Columbia or any United States branch of a foreign bank having at the date of acquisition thereof combined capital and surplus of not less than \$1,000,000,000, (e) Deposit Accounts maintained with (i) any bank that satisfies the criteria described in clause (d) above, or (ii) any other bank organized under the laws of the United States or any state thereof so long as the full amount maintained with any such other bank is insured by the Federal Deposit Insurance Corporation, (f) repurchase obligations of any commercial bank satisfying the requirements of clause (d) of this definition or recognized securities dealer having combined capital and surplus of not less than \$1,000,000,000, having a term of not more than seven days, with respect to securities satisfying the criteria in clauses (a) or (d) above, (g) debt securities with maturities of six months or less from the date of acquisition backed by standby letters of credit issued by any commercial bank satisfying the criteria described in clause (d) above, and (h) Investments in money market funds substantially all of whose assets are invested in the types of assets described in clauses (a) through (g) above.

"<u>Cash Management Services</u>" means any cash management or related services including treasury, depository, return items, overdraft, controlled disbursement, merchant store value cards, e-payables services, electronic funds transfer, interstate depository network, automatic clearing house transfer (including the Automated Clearing House processing of electronic funds transfers through the direct Federal Reserve Fedline system) and other cash management arrangements.

"Certificate of Beneficial Ownership" shall mean, for the Borrower, a certificate in form and substance acceptable to Agent (as amended or modified by Agent from time to time in its sole discretion), certifying, among other things, the Beneficial Owner of such Borrower.

"CFC" means a controlled foreign corporation (as that term is defined in the IRC).

"Change of Control" means that:

- (a) Sponsor fails to own and control, directly or indirectly, fifty-one percent (51%) or more of the Equity Interests of Public Holdco entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Public Holdco,
- (b) Public Holdco fails to own and control, directly or indirectly, all of the Equity Interests of Parent entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Parent,
- (c) any "person" or "group" (within the meaning of Sections 13(d) and 14(d) of the Exchange Act but excluding any employee benefit plan and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan), other than Permitted Holders, becomes the beneficial owner (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of 35%, or more, of the Equity Interests of Parent entitled (without regard to the occurrence of any contingency) to vote for the election of members of the Board of Directors of Parent, or

(d) Parent fails to own and control, directly or indirectly, 100% of the Equity Interests of each other Loan Party other than as a result of a Permitted Disposition or otherwise expressly permitted under the Agreement.

"Change in Law" means the occurrence after the date of the Agreement of: (a) the adoption or effectiveness of any law, rule, regulation, judicial ruling, judgment or treaty, (b) any change in any law, rule, regulation, judicial ruling, judgment or treaty or in the administration, interpretation, implementation or application by any Governmental AuthorityBody of any law, rule, regulation, guideline or treaty, or (c) the making or issuance by any Governmental AuthorityBody of any request, rule, guideline or directive, whether or not having the force of law; provided that notwithstanding anything in the Agreement to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives concerning capital adequacy promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities shall, in each case, be deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Closing Date" means the date of the making of the Term Loan under the Agreement.

"Closing Date Disbursement Letter" means the letter executed by the Borrower and delivered to Agent on the Closing Date directing Agent to disburse the proceeds of the Closing Date Term Loan as detailed therein.

"Closing Date Term Loan" has the meaning specified in Section 2.3 hereof.

"Code" means the New York Uniform Commercial Code, as in effect from time to time.

"Collateral" means all assets and interests in assets and proceeds thereof now owned or hereafter acquired by Parent or its Subsidiaries in or upon which a Lien is granted by such Person in favor of Agent or the Lenders under any of the Loan Documents.

"Commitment" means, with respect to each Lender, its Revolver Commitment or its Term Loan Commitment, as the context requires, and, with respect to all Lenders, their Revolver Commitments or their Term Loan Commitments, as the context requires, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on Schedule C to the Agreement or in the Assignment and Acceptance pursuant to which such Lender became a Lender under the Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 13.1 of the Agreement.

"Competitor" means any Person which is a direct competitor of Borrower or its Subsidiaries identified on Schedule C-2 and any private equity fund that is an equity owner of a Competitor identified by * on Schedule C-2.

"Compliance Certificate" means a certificate substantially in the form of Exhibit C-1 to the Agreement delivered by the chief financial officer of Borrower to Agent.

"Confidential Information" has the meaning specified therefor in Section 17.9(a) of the Agreement.

"Conforming Changes" means, with respect to the Term SOFR Rate or any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Alternate Base Rate," the definition of "Business Day," the definition of "Interest Period," timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that the Agent decides may be appropriate to reflect the adoption and implementation of the Term SOFR Rate or such Benchmark Replacement and to permit the administration thereof by the Agent in a manner substantially consistent with market practice (or, if the Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Agent determines that no market practice for the administration of the Term SOFR Rate or the Benchmark Replacement exists, in such other manner of administration as the Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Continuing Director" means (a) any member of the Board of Directors who was a director (or comparable manager) of Parent on the Closing Date, and (b) any individual who becomes a member of the Board of Directors after the Closing Date if such individual was approved, appointed or nominated for election to the Board of Directors by either the Permitted Holders or a majority of the Continuing Directors, but excluding any such individual originally proposed for election in opposition to the Board of Directors in office at the Closing Date in an actual or threatened election contest relating to the election of the directors (or comparable managers) of Parent and whose initial assumption of office resulted from such contest or the settlement thereof.

"Contract Rate" has the meaning specified therefor in Section 2.14(a) of the Agreement.

"Control Agreement" means a control agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by Parent or one of its Subsidiaries, Agent, and the applicable securities intermediary (with respect to a Securities Account) or bank (with respect to a Deposit Account).

"Copyright Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"<u>Current Assets</u>" means, as at any date of determination, the total assets of Parent and its Subsidiaries (other than cash and Cash Equivalents) which may properly be classified as current assets on a consolidated balance sheet of Parent and its Subsidiaries in accordance with GAAP.

"Current Liabilities" means, as at any date of determination, the total liabilities of Parent and its Subsidiaries which may properly be classified as current liabilities (other than the current portion of the Term Loan, the Swing Loans and the Revolving Loans) on a consolidated balance sheet of Parent and its Subsidiaries in accordance with GAAP.

"Daily <u>LIBOR RateSimple SOFR</u>" means, for any day (a "SOFR Rate Day"), the <u>interest</u> rate per annum determined by Agent by dividing (x) the <u>Published Rate</u>the resulting quotient rounded upwards, at Agent's discretion, to the nearest 1/100th of 1%) (A) SOFR for the day (the "SOFR Determination Date") that is two (2) Business Days prior to (i) such SOFR Rate Day if such SOFR Rate Day is a Business Day or (ii) the Business Day immediately preceding such SOFR Rate Day if such SOFR Rate Day is not a Business Day, by (yB) a number equal to 1.00 minus the <u>SOFR</u> Reserve Percentage; provided, however, that if the Daily LIBOR Rate, in

each case, as such SOFR is published by the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate) on the website of the Federal Reserve Bank of New York, currently at http://www.newyorkfed.org, or any successor source identified by the Federal Reserve Bank of New York or its successor administrator for the secured overnight financing rate from time to time, plus the SOFR Adjustment for a 1-month Interest Period. If Daily Simple SOFR as determined as provided above would be less than zero, such rate the SOFR Floor, then Daily Simple SOFR shall be deemed to be zero for purposes of the Agreement. the SOFR Floor. If SOFR for any SOFR Determination Date has not been published or replaced with a Benchmark Replacement by 5:00 p.m. (Pittsburgh, Pennsylvania time) on the second Business Day immediately following such SOFR Determination Date, then SOFR for such SOFR Determination Date will be SOFR for the first Business Day preceding such SOFR Determination Date for which SOFR was published in accordance with the definition of "SOFR"; provided that SOFR determined pursuant to this sentence shall be used for purposes of calculating Daily Simple SOFR for no more than 3 consecutive SOFR Rate Days. If and when Daily Simple SOFR as determined above changes, any applicable rate of interest based on Daily Simple SOFR will change automatically without notice to the Borrower, effective on the date of any such change.

"<u>Debt Reorganization Securities</u>" means Reorganization Securities that consist of debt obligations or Disqualified Equity Interests of the reorganized debtor.

"<u>Default</u>" means an event, condition, or default that, with the giving of notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" has the meaning specified therefor in <u>Section 2.14(a)</u> of the Agreement.

"Defaulting Lender" means any Lender that: (a) has failed, within two (2) Business Days of the date required to be funded or paid, to (i) fund any portion of its Pro Rata Share of Loans, (ii) if applicable, fund any portion of its Participation Commitment in Letters of Credit or Swing Loans or (iii) pay over to Agent, Issuing Bank, Swing Loan Lender or any Lender any other amount required to be paid by it hereunder, unless, in the case of clause (i) above, such Lender notifies Agent in writing that such failure is the result of such Lender's good faith determination that a condition precedent to funding (specifically identified and including a particular Default or Event of Default, if any) has not been satisfied; (b) has notified Borrower or Agent in writing, or has made a public statement to the effect, that it does not intend or expect to comply with any of its funding obligations under this Agreement (unless such writing or public statement indicates that such position is based on such Lender's good faith determination that a condition precedent (specifically identified and including a particular Default or Event of Default, if any) to funding a loan under this Agreement cannot be satisfied) or generally under other agreements in which it commits to extend credit; (c) has failed, within two (2) Business Days after request by Agent, acting in good faith, to provide a certification in writing from an authorized officer of such Lender that it will comply with its obligations (and is financially able to meet such obligations) to fund prospective Loans and, if applicable, participations in then outstanding Letters of Credit and Swing Loans under this Agreement; provided, that, such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon Agent's receipt of such certification in form and substance satisfactory to the Agent; (d) has become the subject of an Insolvency Event; or (e) has failed at any time to comply with the provisions of Section 2.6(e) of the Agreement with respect to purchasing participations from the other Lenders, whereby such Lender's share of any payment received, whether by setoff or otherwise, is in excess of its pro rata share of such payments due and payable to all of the Lenders.

"Deposit Account" means any deposit account (as that term is defined in the Code).

"Disqualified Equity Interests" means any Equity Interest that, by its terms (or by the terms of any security or other Equity Interests into which it is convertible or for which it is exchangeable), or upon the happening of any event or condition (a) matures or is mandatorily redeemable (other than solely for Qualified Equity Interests), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event shall be subject to the prior repayment in full of the Loans and all other Obligations that are accrued and payable and the termination of the Commitments), (b) is redeemable at the option of the holder thereof (other than solely for Qualified Equity Interests), in whole or in part, (c) provides for the scheduled payments of dividends in cash, or (d) is or becomes convertible into or exchangeable for Indebtedness or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is 180 days after the Maturity Date.

"<u>Distributions</u>" means (a) all payments made by any Loan Party on account of the Obligations, (b) all proceeds of insurance policies and awards or other payments with respect to any condemnation or similar proceeding, and (c) all proceeds of Collateral (including proceeds of Collateral received in connection with an exercise of remedies, but excluding Permitted Reorganization Securities).

"Domestic Subsidiary" means a Subsidiary that is not a Foreign Subsidiary.

"Dollars" or "\$" means United States dollars.

"Drawing Date" has the meaning specified therefor in Section 2.11(d)(ii) of the Agreement.

"Earn-Outs" means unsecured liabilities of a Loan Party arising under an agreement to make any deferred payment as a part of the Purchase Price paid for a Permitted Acquisition, including the (i) the SpeechIQ Incentive Payment, (ii) the Teckst Incentive Payment, (iii) the Engage Earn-Out and (iv) the performance bonuses or consulting payments in any related services, employment, or similar agreement, in an amount that is subject to or contingent upon the revenues, income, cash flow, or profits (or the like) of the underlying target.

"EBITDA" means, for any specified period, with respect to Parent and its Subsidiaries on a consolidated basis and determined in accordance with GAAP:

(a) net income (or loss) for such period,

minus

- (b) without duplication and to the extent included in determining net income (or loss) for such period, the sum of:
 - (i) any extraordinary, unusual, or non-recurring gains,
 - (ii) interest income,
 - (iii) any software development costs to the extent capitalized during such period,

(iv) non-cash exchange, translation or performance gains relating to any hedging transactions or foreign currency

fluctuations,

- (v) income arising by reason of the application of FASB Accounting Standards Codification 805,
- (vi) income from the receipt of business interruption insurance proceeds, and
- (vii) non-cash gains increasing net income of such Person for such period, excluding any non-cash gains to the extent they represent the reversal of an accrual or reserve for a potential cash item that reduced EBITDA in any prior period and any non-cash gains with respect to cash actually received in a prior period so long as such cash did not increase EBITDA in such prior period,

plus

- (c) without duplication and to the extent included in determining net earnings (or loss) for such period, the sum of:
 - (i) any extraordinary, unusual, or non-recurring non-cash losses,
 - (ii) Interest Expense (including the Unused Line Fee and Letter of Credit Fees),
- (iii) tax expense based on income, profits or capital, including federal, foreign, state, local, excise, franchise and similar taxes (and for the avoidance of doubt, specifically excluding any sales taxes or any other taxes held in trust for a Governmental AuthorityBody),
 - (iv) depreciation and amortization for such period,
- (v) with respect to any Permitted Acquisition: (A) purchase accounting adjustments, including, without limitation, a dollar for dollar adjustment for that portion of revenue that would have been recorded in the relevant period had the balance of deferred revenue (unearned income) recorded on the closing balance sheet and before application of purchase accounting not been adjusted downward to fair value to be recorded on the pro forma balance sheet delivered to Agent in connection therewith in accordance with GAAP purchase accounting rules; and (B) non-cash adjustments in accordance with GAAP purchase accounting rules under FASB Statement No. 141 and 141R and EITF Issue No. 01-2 (including deferred revenue), in the event that such an adjustment is required by the Loan Parties' independent auditors, in each case, as determined in accordance with GAAP,
- (vi) (A) non-cash compensation expense (including deferred non-cash compensation expense), or other non-cash expenses or charges, arising from the sale or issuance of Equity Interests, stock options, stock appreciation rights, or similar arrangements) minus (B) the amount of any such expenses or charges when paid in cash to the extent not deducted in the computation of net earnings (or loss) for such period,
- (vii) fees, costs and expenses incurred in connection with a Permitted Acquisition which are consistent with the sources and uses for such Permitted Acquisition provided to Agent by the Loan Parties in connection therewith, or if not consummated, not to exceed 10% of EBITDA for such period on a pro forma basis,

- (viii) non-cash exchange, translation, or performance losses relating to any hedging transaction or foreign currency fluctuations.
- (ix) fees and expenses for third party professionals, agents and advisors and other transaction costs and expenses incurred in connection with the closing of the Loan Documents on the Seventh Amendment Effective Date and the initial public offering and any secondary or follow-on public offering (whether or not consummated); provided, that the amount of such fees and expenses are actually incurred within 180 days of the effective date of such transaction,
- (x) up to \$500,000 in the aggregate for any 12-month period of payments in the nature of compensation or expense reimbursements made to members of the Loan Parties' Board of Directors, in each case, to the extent permitted to be paid under the terms of the Agreement,
- (xi) up to \$250,000 in the aggregate for any 12-month period of management or other fees, expenses and reimbursements paid to Sponsor as permitted under <u>Section 6.106.10(e)</u> of the Agreement,
- (xii) up to \$300,000 in the aggregate for fees, costs and expenses incurred during the period from July 1, 2016 through December 31, 2016 in connection with consulting arrangements entered into with Bulger Partners,
- (xiii) up to \$100,000 in the aggregate for retention, recruiting, relocation and signing bonuses and expenses and severance costs,
- (xiv) one-time cost savings associated with synergies or reductions and/or restructurings in force made within twelve (12) months after the closing date for a Permitted Acquisition calculated on a pro-forma, adjusted basis, to the extent such cost savings are factually supportable, calculated in good faith based upon reasonable assumptions and reasonably expected to be realized within 12 months following the applicable Permitted Acquisition; provided that, if the amount of such costs savings during such period exceed the greater of \$1,000,000 or seven and one-half percent (7.5%) of EBITDA for such period on a pro forma basis, then such cost savings shall be supported by a "quality of earnings" or a similar third party report, in each case, reasonably acceptable to Agent,
- (xv) any amounts paid pursuant to the purchase price or net working capital adjustments, Earn-Outs or other deferred purchase payments paid pursuant to a Permitted Acquisition,
 - (xvi) non-cash charges,
 - (xvii) non-cash losses on Permitted Dispositions of fixed assets or write-downs of fixed or intangible assets,
- (xviii) expenses and payments that are covered by indemnification, reimbursement, guaranty or purchase price adjustment provisions in any agreement entered into by Parent or any of its Subsidiaries to the extent such expenses and payments have been reimbursed pursuant to the applicable indemnity, guaranty or acquisition agreement during such period,
- (xix) the aggregate amount of expenses or losses incurred by Parent or one of its Subsidiaries relating to business interruption to the extent covered by insurance

provided by an unaffiliated insurance company and actually reimbursed or otherwise paid to Parent or such Subsidiary during such period, and

(xx) any amounts actually paid (and permitted to be paid under the Agreement) with respect to the repurchase of any Equity Interests of Parent from former employees, officers, or directors (or any spouses, ex-spouses, or estates of any of the foregoing) of Parent which are expensed as compensation for such parties in Parent's consolidated financial statements.

For the purposes of calculating EBITDA for any period of 12 consecutive months (each, a "Reference Period") for any purpose other than determining Excess Cash Flow, (a) if at any time during such Reference Period (and after the Closing Date), Parent or any of its Subsidiaries shall have made a Permitted Acquisition, EBITDA for such Reference Period shall be calculated after giving pro forma effect thereto, including pro forma adjustments arising out of events which are directly attributable to such Permitted Acquisition, are factually supportable, and are expected to have a continuing impact, so long as such adjustments (i) are recommended by any due diligence financial review conducted by financial advisors or accountants retained by Borrower, (ii) are on a basis consistent with Article 11 of Regulation S-X promulgated under the Securities Act and as interpreted by the staff of the SEC, or (iii) are reasonably acceptable to Agent acting in good faith) as if any such Permitted Acquisition or adjustment occurred on the first day of such Reference Period.

"ECF Threshold Amount" has the meaning specified therefor in Section 2.12(d)(vi) of the Agreement.

"Effective Federal Funds Rate" means for any day the rate per annum (based on a year of 360 days and actual days elapsed and rounded upward to the nearest 1/100 of 1% announced by the Federal Reserve Bank of New York (or any successor) on such day as being the weighted average of the rates on overnight federal funds transactions arranged by federal funds brokers on the previous trading day, as computed and announced by such Federal Reserve Bank (or any successor) in substantially the same manner as such Federal Reserve Bank computes and announces the weighted average it refers to as the "Effective Federal Funds Rate" as of the date of this Agreement; provided that if such Federal Reserve Bank (or its successor) does not announce such rate on any day, the "Effective Federal Funds Rate" for such day shall be the Effective Federal Funds Rate for the last day on which such rate was announced. Notwithstanding the foregoing, if the Effective Federal Funds Rate as determined under any method above would be less than zero percent (0.00%), such rate shall be deemed to be zero percent (0.00%) for purposes of this Agreement.

"Eighth Amendment" means that certain Eighth Amendment to Credit Agreement, dated as of the Eighth Amendment Effective

Date.

"Eighth Amendment Effective Date" has the meaning specified in the Eighth Amendment.

"Eighth Amendment Fee Letter" means that certain Third Amended and Restated Fee Letter, dated as of the Eighth Amendment Effective Date, by and between Borrower and Agent.

"Eligible Transferee" means (a) any Lender (other than a Defaulting Lender), any Affiliate of any Lender and any Related Fund of any Lender; and (b) (i) a commercial bank organized under the laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (ii) a savings and loan association or savings bank organized under the

laws of the United States or any state thereof, and having total assets in excess of \$1,000,000,000; (iii) a commercial bank organized under the laws of any other country or a political subdivision thereof; <u>provided</u> that (A) (x) such bank is acting through a branch or agency located in the United States or (y) such bank is organized under the laws of a country that is a member of the Organization for Economic Cooperation and Development or a political subdivision of such country, and (B) such bank has total assets in excess of \$1,000,000,000,000; (d) any other entity (other than a natural person) that is an "accredited investor" (as defined in Regulation D under the Securities Act) that extends credit or buys loans as one of its businesses including insurance companies, investment or mutual funds and lease financing companies, and having total assets in excess of \$1,000,000,000,000; and (f) during the continuation of an Event of Default, any other Person approved by Agent; <u>provided</u>, that, except pursuant to <u>Section 13.1(a)(ii)(B)</u>, no Sponsor Affiliated Entity shall qualify as an Eligible Transferee.

"Engage Acquisition" shall mean the transactions contemplated by the Engage Acquisition Agreement.

"Engage Acquisition Agreement" means that certain Unit Purchase Agreement, dated as of the Sixth Amendment Date, by and among Borrower, the Engage Sellers and Kyle F. Hanson, solely in the capacity of the Seller Rep (as defined therein).

"Engage Acquisition Documentation" shall mean, collectively, the Engage Acquisition Agreement and all schedules, exhibits and annexes thereto, all amendments, waivers and modifications thereto, and all material agreements, instruments, and documents relating to the Engage Acquisition executed by any Loan Party (or any of their Affiliates or Subsidiaries) and affecting the terms thereof or entered into by any Loan Party (or any of their Affiliates or Subsidiaries) in connection therewith.

"Engage Acquisition Agreement Representations" shall mean the representations and warranties made by the Engage Sellers (or any of their Affiliates or Subsidiaries) in the Engage Acquisition Agreement to the extent that Borrower (or any of its Affiliates or Subsidiaries) has the right not to consummate the Engage Acquisition or the right to terminate (or cause the termination of) such Person's obligations under the Engage Acquisition Agreement (giving effect to materiality qualifiers contained in the Engage Acquisition Agreement) as a result of a breach of such representations in the Engage Acquisition Agreement.

"Engage Earn-Out" shall mean that certain Earn-Out, made by the Borrower in favor of the Engage Sellers in accordance with Section 2.5 of the Engage Acquisition Agreement.

"Engage Earn-Out Equity Offset" means, with respect to any payment of the Engage Earn-Out, the extent to which (x) such payment is funded with cash proceeds of equity contributions or any issuance of equity interests that are not prohibited under this Agreement or cash proceeds of Permitted Indebtedness or (y) within three (3) months following the date of such payment, Borrower receives cash proceeds of equity contributions or any issuance of equity interests that are not prohibited under this Agreement.

"Engage PPP Lender" means the "PPP Lender," as defined in the Engage Acquisition Agreement.

"Engage PPP Loan" means the "PPP Loan," as defined in the Engage Acquisition Agreement.

"Engage PPP Loan Documents" shall mean, collectively, the "PPP Loan Note," as defined in the Engage Acquisition Agreement, and all of the other agreements, documents and instruments executed and delivered in connection with or related to the Engage PPP Loan.

"Engage PPP Loan Escrow Agreement" means the PPP Loan Escrow Agreement (as defined in the Engage Acquisition Agreement).

"Engage Sellers" shall mean the Sellers (as defined in the Engage Acquisition Agreement).

"Environmental Action" means any written complaint, summons, citation, notice, directive, order, claim, litigation, investigation, judicial or administrative proceeding, judgment, letter, or other written communication from any Governmental AuthorityBody, or any third party involving violations of Environmental Laws or releases of Hazardous Materials (a) from any assets, properties, or businesses of Borrower, any Subsidiary of Borrower, or any of their predecessors in interest, (b) from adjoining properties or businesses, or (c) from or onto any facilities which received Hazardous Materials generated by Borrower, any Subsidiary of Borrower, or any of their predecessors in interest.

"Environmental Law" means any applicable federal, state, provincial, foreign or local statute, law, rule, regulation, ordinance, code, binding and enforceable guideline, binding and enforceable written policy, or rule of common law now or hereafter in effect and in each case as amended, or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, in each case, to the extent binding on Parent or its Subsidiaries, relating to the environment, the effect of the environment on employee health, or Hazardous Materials, in each case as amended from time to time.

"Environmental Liabilities" means all liabilities, monetary obligations, losses, damages, costs and expenses (including all reasonable fees, disbursements and expenses of counsel, experts, or consultants, and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any claim or demand, or Remedial Action required, by any Governmental Authority Body or any third party, and which relate to any Environmental Action.

"Environmental Lien" means any Lien in favor of any Governmental Authority Body for Environmental Liabilities.

"Equipment" means equipment (as that term is defined in the Code).

"Equity Interest" means, with respect to a Person, all of the shares, options, warrants, interests, participations, or other equivalents (regardless of how designated) of or in such Person, whether voting or nonvoting, including capital stock (or other ownership or profit interests or units), preferred stock, or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act).

"Equity Reorganization Securities" means Reorganization Securities which are not Debt Reorganization Securities.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and any successor statute thereto.

"ERISA Affiliate" means (a) any Person subject to ERISA whose employees are treated as employed by the same employer as the employees of Parent or its Subsidiaries under

IRC Section 414(b), (b) any trade or business subject to ERISA whose employees are treated as employed by the same employer as the employees of Parent or its Subsidiaries under IRC Section 414(c), (c) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any organization subject to ERISA that is a member of an affiliated service group of which Parent or any of its Subsidiaries is a member under IRC Section 414(m), or (d) solely for purposes of Section 302 of ERISA and Section 412 of the IRC, any Person subject to ERISA that is a party to an arrangement with Parent or any of its Subsidiaries and whose employees are aggregated with the employees of Parent or its Subsidiaries under IRC Section 414(o).

"Event of Default" has the meaning specified therefor in Section 8 of the Agreement.

"Excess Cash Flow" means, with respect to any fiscal period and with respect to Parent and its Subsidiaries determined on a consolidated basis in accordance with GAAP the result of:

TTM EBITDA,

plus

- (b) the sum of
 - (i) foreign, United States, state, or local tax refunds paid in cash,
 - (ii) cash interest income, and
- (iii) post-closing Purchase Price adjustments received in cash during such period in connection with a Permitted Acquisition, and
- (iv) the amount of any decrease in Net Working Capital for such period; provided that, notwithstanding the foregoing, in the event that any Loan Party consummates a Permitted Acquisition during such fiscal period, the decrease in Net Working Capital included in such calculation for the target of such Permitted Acquisition shall be the decrease in Net Working Capital for such target for the period from the date of consummation of such Permitted Acquisition to the end of such fiscal period,

minus

- (c) the sum of
 - (i) the cash portion of Interest Expense (including the Unused Line Fee and Letter of Credit Fees),
- (ii) the cash portion of taxes (based on income, profits or capital, including federal, foreign, state, local, excise, franchise and similar taxes (and for the avoidance of doubt, specifically excluding any sales taxes or any other taxes held in trust for a Governmental AuthorityBody)) paid during such period,
- (iii) all scheduled and voluntary principal payments of Indebtedness permitted under the Agreement during such period, excluding voluntary prepayments in respect of the Term Loan,

- (iv) the cash portion of Capital Expenditures (net of (y) any proceeds reinvested in accordance with the proviso to Section 2.12(d)(ii) of the Agreement, and (z) any proceeds of related financings with respect to such expenditures) made during such period,
- (v) up to \$250,000 in the aggregate for any 12-month period of management or other fees, expenses and reimbursements paid to Sponsor as permitted under <u>Section 6.106.10(e)</u> of the Agreement,
- (vi) cash payments made in respect of Permitted Acquisitions (in each case, to the extent such payments are not made with the proceeds of Indebtedness (other than Revolving Loans) or equity contributions made by Sponsor),
- (vii) fees, costs, and expenses paid in cash during such period and incurred in connection with a Permitted Acquisition which were added back to net income (or loss) under clause (c)(vii) of the definition of EBITDA for such period,
- (viii) payments in cash in the nature of compensation or expense reimbursements made to board members, in each case, to the extent added back to net income (or loss) under clause (c)(x) of the definition of EBITDA for such period,
- (ix) any cash amounts actually paid with respect to the repurchase of any Equity Interests of Parent from former employees, officers, or directors (or any spouses, ex-spouses, or estates of any of the foregoing) of Parent which are expensed as compensation for such parties in Parent's consolidated financial statements, not to exceed the amount permitted in Section 6.7(a) of the Agreement
- (x) the cash portion of one-time cost savings associated with synergies or reductions and/or restructurings in force made within twelve (12) months after the closing date for a Permitted Acquisition, in each case to the extent added back to net income (or loss) under clause (c)(xiv) of the definition of EBITDA for such period,
- (xi) the amount of any increase in Net Working Capital for such period; provided that, notwithstanding the foregoing, in the event that any Loan Party consummates a Permitted Acquisition during such fiscal period, the increase in Net Working Capital included in such calculation for the target of such Permitted Acquisition shall be the increase in Net Working Capital for such target for the period from the date of consummation of such Permitted Acquisition to the end of such fiscal period, and
- (xii) any non-cash purchase accounting adjustments with respect to a Permitted Acquisition added to net income (or loss) pursuant to clause (c)(v) of the definition of EBITDA.

"Exchange Act" means the Securities Exchange Act of 1934, as in effect from time to time.

"Excluded Deposit and Securities Accounts" has the meaning specified therefor in the Guaranty and Security Agreement.

"Excluded Domestic Subsidiary" means any Domestic Subsidiary of a Grantor all the assets of which consist of the equity of a Foreign Subsidiary that is a CFC and de minimis assets incidental thereto, and all of the business activities consist of owning such stock and stock equivalents and activities incidental thereto, whose guaranty of the Obligations (or grant of security with respect thereto or pledge of stock of such Domestic Subsidiary) under this Credit

Agreement could reasonably be expected to result in a material adverse tax effect to any Loan Party under Section 956 of the IRC.

"Excluded Taxes" means (i) any tax imposed on the net income or net profits of any Lender or any Participant (including any branch profits taxes), in each case imposed by the jurisdiction (or by any political subdivision or taxing authority thereof) in which such Lender or such Participant is organized or the jurisdiction (or by any political subdivision or taxing authority thereof) in which such Lender's or such Participant's principal office is located in each case as a result of a present or former connection between such Lender or such Participant and the jurisdiction or taxing authority imposing the tax (other than any such connection arising solely from such Lender or such Participant having executed, delivered or performed its obligations or received payment under, or enforced its rights or remedies under the Agreement or any other Loan Document); (ii) taxes resulting from a Lender's or a Participant's failure to comply with (or cure, if applicable) the requirements of Section 16.2 of the Agreement, (iii) any United States federal withholding taxes that would be imposed on amounts payable to a Foreign Lender based upon the applicable withholding rate in effect at the time such Foreign Lender becomes a party to the Agreement (or designates a new lending office), except that Taxes shall include (A) any amount that such Foreign Lender (or its assignor, if any) was previously entitled to receive pursuant to Section 16.1 of the Agreement, if any, with respect to such withholding tax at the time such Foreign Lender becomes a party to the Agreement (or designates a new lending office), as a result of a change in law, rule, regulation, order or other decision with respect to any of the foregoing by any Governmental AuthorityBody, and (iv) any United States federal withholding taxes imposed under FATCA.

"FATCA" means Sections 1471 through 1474 of the IRC, as of the date of the Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof.

"Fee Letter" means, individually and collectively, as the context may require, that certain fee letter, dated as of even date with the Agreement, between Borrower and Agent, in each case in form and substance reasonably satisfactory to Agent, as the same may be amended, amended and restated, modified or supplemented from time to time.

"Federal Funds Effective Rate" means for any day the rate per annum (based on a year of 360 days and actual days clapsed and rounded upward to the nearest 1/100 of 1%) announced by the Federal Reserve Bank of New York (or any successor) on such day as being the weighted average of the rates on overnight federal funds transactions arranged by federal funds brokers on the previous trading day, as computed and announced by such Federal Reserve Bank (or any successor) in substantially the same manner as such Federal Reserve Bank computes and announces the weighted average it refers to as the "Federal Funds Effective Rate" as of the date of this Agreement; provided, if such Federal Reserve Bank (or its successor) does not announce such rate on any day, the "Federal Funds Effective Rate" for such day shall be the Federal Funds Effective Rate for the last day on which such rate was announced.

"Federal Funds Open Rate" means for any day the rate per annum (based on a year of 360 days and actual days elapsed) which is the daily federal funds open rate as quoted by ICAP North America, Inc. (or any successor) as set forth on the Bloomberg Screen BTMM for that day opposite the caption "OPEN" (or on such other substitute Bloomberg Screen that displays such rate), or as set forth on such other recognized electronic source used for the purpose of displaying such rate as selected by Agent (an "Alternate Source") (or if such rate for

such day does not appear on the Bloomberg Screen BTMM (or any substitute screen) or on any Alternate Source, or if there shall at any time, for any reason, no longer exist a Bloomberg Screen BTMM (or any substitute screen) or any Alternate Source, a comparable replacement rate determined by Agent at such time (which determination shall be conclusive absent manifest error); provided however, that if such day is not a Business Day, the Federal Funds Open Rate for such day shall be the "open" rate on the immediately preceding Business Day. If and when the Federal Funds Open Rate changes, the rate of interest with respect to any advance to which the Federal Funds Open Rate applies will change automatically without notice to Borrower, effective on the date of any such change.

"Fifth Amendment" means that certain Consent and Fifth Amendment to Credit Agreement, dated as of the Fifth Amendment Effective Date.

"Fifth Amendment Effective Date" means January 13, 2021.

"First Amendment" means that certain First Amendment to Credit Agreement, dated as of the First Amendment Effective Date.

"First Amendment Effective Date" has the meaning specified in the First Amendment.

"<u>First Amendment Fee Letter</u>" means that certain fee letter, dated as of February 28, 2018, between Parent, Borrower and Agent.

"First Amendment Term Loan" has the meaning specified in Section 2.3 hereof.

"<u>Foreign Lender</u>" means any Lender or Participant that is not a United States person within the meaning of IRC section 7701(a) (30).

"<u>Foreign Subsidiary</u>" means each Subsidiary which is organized under the laws of a jurisdiction other than the United States of America (excluding any territory thereof) or any state thereof or the District of Columbia.

"Fourth Amendment" means that certain Fourth Amendment to Credit Agreement, dated as of the Fourth Amendment Effective Date.

"Fourth Amendment Effective Date" means December 18, 2019.

"Funded Debt" means, with respect to Parent and its Subsidiaries on a consolidated basis determined on a consolidated basis in accordance with GAAP, without duplication, all Indebtedness for borrowed money, Indebtedness evidenced by notes, bonds, debentures, or similar evidences of Indebtedness, Capitalized Lease Obligations, current maturities of long-term debt, revolving credit and short term debt extendible beyond one year at the option of the debtor, and also including, without limitation, the Obligations and, without duplication, Indebtedness consisting of guaranties of Funded Debt of other Persons; provided, however, that for purposes of determining the amount of Funded Debt with respect to (a) the Obligations, the amount of Funded Debt shall be equal to the sum of (i) the outstanding Term Loan as of the date of determination, plus (ii) the quotient of (A) the Revolver Usage for each day of the most recently ended fiscal quarter, divided by (B) the number of such days in such fiscal quarter and (b) Indebtedness consisting of Earn-Outs (solely to the extent not paid within three (3) Business Days of becoming earned, due and payable), such amount shall, for purposes hereof, include secured and unsecured amounts. For the avoidance of doubt, Funded Debt shall include the SpeechIQ Incentive Payment, the SpeechIQ Holdback, the Teckst Incentive Payment

and Teckst Holdback. Notwithstanding the forgoing, Funded Debt shall not include the Engage Earn-Out or the Engage PPP Loan.

"Funding Account" shall mean the deposit account of Borrower established with PNC for purposes of receiving proceeds of Loans.

"GAAP" means generally accepted accounting principles as in effect from time to time in the United States, consistently applied.

"Golden Gate" means Golden Gate Private Equity, Inc., a Delaware corporation, and/or one or more of the investment funds managed by it.

"Governing Documents" means, with respect to any Person, the certificate or articles of incorporation, by-laws, or other organizational documents of such Person.

"Governmental Authority" means the government of Body" shall mean any nation or government, any state or other political subdivision thereof, whether at the national, state, territorial, provincial, municipal or any other level, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity or any entity, authority, agency, division or department exercising the executive, legislative, judicial, taxing, regulatory or administrative powers or functions of, or pertaining to, a government (including any supranational bodies such as the European Union or the European Central Bank), and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

"Guarantor" means (a) each Subsidiary of Parent (other than a Subsidiary that is a CFC or an Excluded Domestic Subsidiary), (b) Parent, (c) LiveVox International Guarantor, (d) SpeechIQ Guarantor, (e) Engage Guarantor and (f) each other Person that becomes a guarantor after the Closing Date pursuant to Section 5.11 of the Agreement.

"Guaranty and Security Agreement" means a guaranty and security agreement, dated as of even date with the Agreement, in form and substance reasonably satisfactory to Agent, executed and delivered by Borrower and each of the Guarantors to Agent.

"Hazardous Materials" means (a) substances that are defined or listed in, or otherwise classified pursuant to, any applicable laws or regulations as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list, or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity, or "EP toxicity", (b) oil, petroleum, or petroleum derived substances, natural gas, natural gas liquids, synthetic gas, drilling fluids, produced waters, and other wastes associated with the exploration, development, or production of crude oil, natural gas, or geothermal resources, (c) any flammable substances or explosives or any radioactive materials, and (d) asbestos in any form or electrical equipment that contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of 50 parts per million.

"Hedge Agreement" means (a) a "swap agreement" as that term is defined in Section 101(53B)(A) of the Bankruptcy Code and (b) any agreement entered into by Parent and/or any of its Subsidiaries with a Hedge Provider with respect to (i) foreign exchange transaction, including spot and forward foreign currency purchases and sales, listed or over-the-counter options on foreign currencies, non-deliverable forwards and options, foreign currency swap

agreements, currency exchange rate price hedging arrangements, and any other similar transaction providing for the purchase of one currency in exchange for the sale of another currency and/or (ii) interest rate exchange, collar, cap, swap, floor, adjustable strike cap, adjustable strike corridor, cross-currency swap or similar transactions entered into in order to provide protection to, or minimize the impact upon, Parent or such Subsidiary of increasing floating rates of interest applicable to Indebtedness.

"Hedge Obligations" means any and all obligations or liabilities, whether absolute or contingent, due or to become due, now existing or hereafter arising, of Parent or its Subsidiaries arising under, owing pursuant to, or existing in respect of Hedge Agreements entered into with one or more of the Hedge Providers.

"Hedge Provider" means PNC or any of its Affiliates.

"Indebtedness" as to any Person means (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments and all reimbursement or other obligations in respect of letters of credit, bankers acceptances, or other financial products, (c) all obligations of such Person as a lessee under Capital Leases, (d) all obligations or liabilities of others secured by a Lien on any asset of such Person, irrespective of whether such obligation or liability is assumed, (e) all obligations of such Person to pay the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business and repayable in accordance with customary trade practices and, for the avoidance of doubt, other than royalty payments payable in the ordinary course of business in respect of non-exclusive licenses), (f) all monetary obligations of such Person owing under Hedge Agreements (which amount shall be calculated based on the amount that would be payable by such Person if the Hedge Agreement were terminated on the date of determination), (g) any Disqualified Equity Interests of such Person, and (h) any obligation of such Person guaranteeing or intended to guarantee (whether directly or indirectly guaranteed, endorsed, co-made, discounted, or sold with recourse) any obligation of any other Person that constitutes Indebtedness under any of clauses (a) through (g) above. For purposes of this definition, (i) the amount of any Indebtedness represented by a guaranty or other similar instrument shall be the lesser of the principal amount of the obligations guaranteed and still outstanding and the maximum amount for which the guaranteeing Person may be liable pursuant to the terms of the instrument embodying such Indebtedness, and (ii) the amount of any Indebtedness which is limited or is non-recourse to a Person or for which recourse is limited to an identified asset shall be valued at the lesser of (A) if applicable, the limited amount of such obligations, and (B) if applicable, the fair market value of such assets securing such obligation. Notwithstanding anything to the contrary set forth in this definition of Indebtedness or otherwise, Earn-Outs shall only be included as Indebtedness for all purposes hereunder (including for the purpose of calculating any financial ratio) to the extent a specific dollar amount has been determined to be earned and the same has not been paid within three (3) Business Days of the same becoming due and payable.

"Indemnified Liabilities" has the meaning specified therefor in Section 10.3 of the Agreement.

"Indemnified Person" has the meaning specified therefor in Section 10.3 of the Agreement.

"Indemnified Taxes" means, any Taxes other than Excluded Taxes.

"Ineligible Institution" means the Persons identified on Schedule I-1.

"Insolvency Proceeding" means any proceeding commenced by or against any Person under any provision of the Bankruptcy Code or under any other state or federal bankruptcy or insolvency law, assignments for the benefit of creditors, formal or informal moratoria, compositions, extensions generally with creditors, or proceedings seeking reorganization, arrangement, or other similar relief.

"Intercompany Subordination Agreement" means an intercompany subordination agreement, dated as of even date with the Agreement, executed and delivered by Parent, each of its Subsidiaries each of the other Loan Parties, and Agent, the form and substance of which is reasonably satisfactory to Agent, as amended, amended and restated, supplemented or otherwise modified from time to time.

"Interest Expense" means, for any period, the aggregate of the interest expense of Parent and its Subsidiaries for such period, determined on a consolidated basis in accordance with GAAP.

"Interest Period" means the period provided for any LIBORTerm SOFR Rate Loan pursuant to Section 2.2(b) hereof.

"Investment" means, with respect to any Person, any investment by such Person in any other Person (including Affiliates) in the form of loans, guarantees, advances, capital contributions (excluding (a) commission, travel, and similar advances to officers and employees of such Person made in the ordinary course of business, and (b) bona fide accounts receivable arising in the ordinary course of business), or acquisitions of Indebtedness, Equity Interests, or all or substantially all of the assets of such other Person (or of any division or business line of such other Person), and any other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP. The amount of any Investment shall be the original cost of such Investment plus the cost of all additions thereto, without any adjustment for increases or decreases in value, or write-ups, write-downs, or write-offs with respect to such Investment.

"IP Reporting Certificate" means the certificate in the form attached to the Agreement as Exhibit I-1.

"IPO" means the first underwritten public offering by Parent (or its direct or indirect parent company) of Equity Interests in Parent (or in its direct or indirect parent company, as the case may be) after the Closing Date pursuant to a registration statement filed with the SEC in accordance with the Securities Act.

"IRC" means the Internal Revenue Code of 1986, as in effect from time to time.

"Issuer Document" means, with respect to any Letter of Credit, a Letter of Credit Application, a letter of credit agreement, or any other document, agreement or instrument entered into (or to be entered into) by Borrower in favor of Issuing Bank and relating to such Letter of Credit.

"Issuing Bank" means PNC or any other Lender that, at the request of Borrower and with the consent of Agent, agrees, in such Lender's sole discretion, to become an Issuing Bank for the purpose of issuing Letters of Credit pursuant to Section 2.11 of the Agreement, and Issuing Bank shall be a Lender.

"Lender" has the meaning set forth in the preamble to the Agreement, shall include Issuing Bank and the Swing Loan Lender, and shall also include any other Person made a

party to the Agreement pursuant to the provisions of <u>Section 13.1</u> of the Agreement and "Lenders" means each of the Lenders or any one or more of them.

"Lender Group" means each of the Lenders (including Issuing Bank and the Swing Loan Lender), Agent and Bank Product Providers, or any one or more of them.

"Lender Group Expenses" means all (a) costs or expenses (including taxes and insurance premiums) required to be paid by Parent or its Subsidiaries under any of the Loan Documents that are paid, advanced, or incurred by the Lender Group, (b) reasonable documented out-of-pocket fees or charges paid or incurred by Agent in connection with the Lender Group's transactions with Parent or its Subsidiaries under any of the Loan Documents, including, photocopying, notarization, couriers and messengers, telecommunication, public record searches, filing fees, recording fees, publication, real estate surveys, real estate title policies and endorsements, and environmental audits, (c) Agent's customary fees and charges imposed or incurred in connection with any background checks or OFAC/PEP searches related to the Loan Parties and any applicable Affiliates thereof, (d) Agent's and Issuing Bank's customary fees and charges (as adjusted from time to time) with respect to the disbursement of funds (or the receipt of funds) to or for the account of Borrower (whether by wire transfer or otherwise) or the issuing of Letters of Credit, together with any out-of-pocket costs and expenses incurred in connection therewith, (e) customary charges imposed or incurred by Agent resulting from the dishonor of checks payable by or to any Loan Party, (f) reasonable documented out-of-pocket costs and expenses paid or incurred by the Agent (limited to one primary outside counsel and, if applicable, local counsel in each applicable jurisdiction), to correct any default or enforce any provision of the Loan Documents, or during the continuance of an Event of Default, in gaining possession of, maintaining, handling, preserving, storing, shipping, selling, preparing for sale, or advertising to sell the Collateral, or any portion thereof, irrespective of whether a sale is consummated, (g) financial examination, appraisal, and valuation fees on the terms set forth herein and reasonable out-of-pocket expenses of Agent related to any financial examinations, appraisals, or valuation to the extent of the fees and charges (and up to the amount of any limitation) provided in Section 2.15(c) of the Agreement, (h) Agent's reasonable out-of-pocket costs and expenses (including reasonable documented attorneys' fees of outside counsel (limited to one primary outside counsel and, if applicable, local counsel in each applicable jurisdiction) and the fees and expenses of in-house attorneys to the extent incurred in connection with matters for which outside counsel is not also engaged) relative to third party claims or any other lawsuit or adverse proceeding paid or incurred, whether in enforcing or defending the Loan Documents or otherwise in connection with the transactions contemplated by the Loan Documents, Agent's Liens in and to the Collateral, or the Lender Group's relationship with Parent or any of its Subsidiaries, (i) Agent's reasonable documented costs and expenses (including reasonable documented attorneys' fees of outside counsel (limited to one primary outside counsel and, if applicable, local counsel in each applicable jurisdiction) and the fees and expenses of in-house attorneys to the extent incurred in connection with matters for which outside counsel is not also engaged) and due diligence expenses) incurred in advising, structuring, drafting, reviewing, administering (including travel, meals, and lodging), syndicating (including reasonable costs and expenses relative to the rating of the Term Loan, CUSIP, DXSyndicateTM, SyndTrak or other communication costs incurred in connection with a syndication of the loan facilities), or amending, waiving, or modifying the Loan Documents, and (j) Agent's and each Lender's reasonable documented costs and expenses (including reasonable documented attorneys', accountants', consultants', and other advisors' fees and expenses, but excluding the fees and expenses of in-house attorneys to the extent incurred in connection with matters for which outside counsel is also engaged) incurred in terminating, enforcing (including attorneys', accountants', consultants', and other advisors' fees and expenses incurred in connection with a "workout," a "restructuring," or an Insolvency Proceeding concerning Parent or any of its Subsidiaries or in exercising rights or remedies under the Loan Documents, but excluding the

fees and expenses of in-house attorneys to the extent incurred in connection with matters for which outside counsel is also engaged and limited to one primary outside counsel and, if applicable, local counsel in each applicable jurisdiction), or defending the Loan Documents, irrespective of whether a lawsuit or other adverse proceeding is brought, or in taking any enforcement action or any Remedial Action with respect to the Collateral.

"Lender Group Representatives" has the meaning specified therefor in Section 17.9 of the Agreement.

"Lender-Related Person" means, with respect to any Lender, such Lender, together with such Lender's Affiliates, and its and their respective officers, directors, employees, attorneys, and agents.

"Letter of Credit Application" shall have the meaning set forth in Section 2.11(b) of the Agreement.

"Letter of Credit Borrowing" shall have the meaning set forth in Section 2.11(d)(iv) of the Agreement.

"Letter of Credit Fees" shall have the meaning set forth in Section 2.14(b) of the Agreement.

"Letter of Credit Sublimit" means \$1,500,000.

"Letter of Credit Collateralization" means either (a) providing cash collateral (pursuant to documentation reasonably satisfactory to Agent, including provisions that specify that the Letter of Credit Fees and all commissions, fees, charges and expenses provided for in Section 2.11 of the Agreement will continue to accrue while the Letters of Credit are outstanding) to be held by Agent for the benefit of the Revolving Lenders in an amount equal to 105% of the Maximum Undrawn Amount of all then outstanding Letters of Credit, (b) delivering to Agent documentation executed by all beneficiaries under the Letters of Credit, in form and substance reasonably satisfactory to Agent and Issuing Bank, terminating all of such beneficiaries' rights under the Letters of Credit, or (c) providing Agent with a standby letter of credit, in form and substance reasonably satisfactory to Agent, from a commercial bank acceptable to Agent (in its sole discretion) in an amount equal to 105% of the Maximum Undrawn Amount of all then outstanding Letters of Credit (it being understood that the Letter of Credit Fees set forth in the Agreement will continue to accrue while the Letters of Credit are outstanding and that any such fees that accrue must be an amount that can be drawn under any such standby letter of credit).

"Letters of Credit" shall have the meaning set forth in Section 2.11(a)(i) of the Agreement.

"Leverage Ratio" means, with respect to Parent and its Subsidiaries on a consolidated basis in accordance with GAAP, for any date of determination, the ratio of (a) Funded Debt as of such date of determination to (b) EBITDA for the most recently ended LTM period.

"LIBOR Alternate Source" has the meaning set forth in the definition of LIBOR Rate.

"LIBOR Rate" for any LIBOR Rate Loan for the then current Interest Period relating thereto, the interest rate per annum determined by Agent by dividing (the resulting

quotient rounded upwards, if necessary, to the nearest 1/100th of 1% per annum) (a) the rate which appears on the Bloomberg Page BBAM1 (or on such other substitute Bloomberg page that displays rates at which Dollar deposits are offered by leading banks in the London interbank deposit market), or, if not available from Bloomberg, the rate which is quoted by another source selected by Agent as an authorized information vendor for the purpose of displaying rates at which Dollar deposits are offered by leading banks in the London interbank deposit market (a "LIBOR Alternate Source"), at approximately 11:00 a.m., London time, two (2) Business Days prior to the commencement of such Interest Period as the London interbank offered rate for Dollars for an amount comparable to such LIBOR Rate Loan and having a borrowing date and a maturity comparable to such Interest Period (or if there shall at any time, for any reason, no longer exist a Bloomberg Page BBAM1 (or any substitute page) or any LIBOR Alternate Source, a comparable replacement rate determined by Agent at such time (which determination shall be conclusive absent manifest error)), by (b) a number equal to 1.00 minus the Reserve Percentage; provided, however, that if the LIBOR Rate determined as provided above would be less than zero, such rate shall be deemed to be zero for purposes of this Agreement. The LIBOR Rate shall be adjusted with respect to any LIBOR Rate Loan that is outstanding on the effective date of any change in the Reserve Percentage as of such effective date. Agent shall give reasonably prompt notice to the Borrower of the LIBOR Rate as determined or adjusted in accordance herewith, which determination shall be presumed conclusive absent manifest error.

"LIBOR Rate Loan" means each portion of a Revolving Loan or the Term Loan that bears interest at a rate determined by reference to the LIBOR Rate.

"<u>Lien</u>" means any mortgage, deed of trust, pledge, hypothecation, assignment, charge, deposit arrangement, encumbrance, easement, lien (statutory or other), security interest, or other security arrangement and any other preference, priority, or preferential arrangement of any kind or nature whatsoever, including any conditional sale contract or other title retention agreement, the interest of a lessor under a Capital Lease and any synthetic or other financing lease having substantially the same economic effect as any of the foregoing.

"<u>Limited Condition Acquisition</u>" means any Permitted Acquisition or Permitted Investment by a Loan Party the consummation of which is not conditioned on the availability of, or on obtaining, third party financing which Limited Condition Acquisition shall have been consummated within 180 days after the definitive agreement for such Limited Condition Acquisition is executed.

"Liquidity" means, as of any date of determination, the sum of Availability and Qualified Cash.

"Loan" means each extension of credit made (or to be made) under the Agreement, including the Revolving Loans, the Term Loan, Swing Loans, Protective Advances and Letters of Credit.

"Loan Account" has the meaning specified therefor in Section 2.10 of the Agreement.

"Loan Documents" means the Agreement, the First Amendment, the Second Amendment, the Third Amendment, the Fourth Amendment, the Fifth Amendment, the Sixth Amendment, the Seventh Amendment, the Control Agreements, the Copyright Security Agreement, the Fee Letter, the First Amendment Fee Letter, the Seventh Amendment Fee Letter, the Guaranty and Security Agreement, the Intercompany Subordination Agreement, any Issuer Documents, the Letters of Credit, the Mortgages, the Patent Security Agreement, the Trademark Security Agreement, the Notes and any other notes executed by Borrower in connection with the

Agreement and payable to any member of the Lender Group, and any other instrument or agreement entered into, now or in the future, by Parent or any of its Subsidiaries and (or in favor of) any member of the Lender Group in connection with the Agreement.

"Loan Party" means Borrower or any Guarantor.

"LTM Period" means, as of any date of determination, the period of four (4) consequence fiscal quarters recently ended for which financial statements required by Section 5.1 have been (or were not required to be) delivered hereunder.

"Management Incentive Documentation" means (i) that certain Amended and Restated Limited Liability Company Agreement, by and among Golden Gate Capital Opportunity Fund, L.P., Golden Gate Capital Opportunity Fund-A, L.P., GGCOF Third-Party Co-Invest, L.P., GGCOF Executive Co-Invest, L.P., GGCOF IRA Co-Invest, L.P. and the other parties deemed signatory thereto (the "LiveVox TopCoLLC Agreement"), and (ii) those certain Letters Regarding Cancellation Of Value Creation Incentive Plan Grant, by and between Borrower and the other parties signatory thereto.

"Management Incentive Unit Agreements" means those certain Management Incentive Unit Agreements, by and between LiveVox TopCo, LLC, a Delaware limited liability company, and the other parties signatories thereto.

"Margin Stock" as defined in Regulation U of the Board of Governors as in effect from time to time.

"Material Adverse Effect" means (a) a material adverse effect in the business, operations, results of operations, assets, liabilities or financial condition of Parent and its Subsidiaries, taken as a whole, (b) a material impairment of Parent's and its Subsidiaries ability to perform their obligations under the Loan Documents to which they are parties or of the Lender Group's ability to enforce the Obligations or realize upon the Collateral (other than as a result of as a result of an action taken or not taken that is solely in the control of Agent), or (c) a material impairment of the enforceability or priority of Agent's Liens with respect to all or a material portion of the Collateral.

"Material Contract" means, with respect to any Person, (i) each contract or agreement to which such Person or any of its Subsidiaries is a party involving aggregate consideration payable to or by such Person or such Subsidiary of an amount equal to 10% or more of projected Recurring Revenue for the 12 months immediately after the date of determination (other than purchase orders in the ordinary course of the business of such Person or such Subsidiary and other than contracts that by their terms may be terminated by such Person or Subsidiary in the ordinary course of its business upon less than 60 days' notice without penalty or premium) (ii) the Advisory Agreement, (iii) the Applicable Affiliate Transaction Documents, and (iv) all other contracts or agreements, the loss of which could reasonably be expected to result in a Material Adverse Effect.

"Maturity Date" means December 31, 2025.

"Maximum Applicable Reserve" has the meaning specified therefor in Section 2.1(b)(i) of the Agreement.

"Maximum Revolver Amount" means \$5,000,000.

"Maximum Swing Loan Advance Amount" means \$500,000.

"Maximum Undrawn Amount" means, with respect to any outstanding Letter of Credit as of any date, the amount of such Letter of Credit that is or may become available to be drawn, including all automatic increases provided for in such Letter of Credit, whether or not any such automatic increase has become effective.

"Merger Agreement" shall have the meaning set forth in the Fifth Amendment.

"Merger Documentation" means the Merger Agreement and any Transaction Agreements (as defined in the Merger Agreement) executed on the date of the Merger Agreement.

"Merger Documentation Representations" shall mean the representations and warranties made by Public Holdco, First Merger Sub (as defined in the Fifth Amendment) and Second Merger Sub (as defined in the Fifth Amendment) (or any of their Affiliates or Subsidiaries) in the Merger Agreement to the extent that Parent (or any of its Affiliates or Subsidiaries) has the right not to consummate the Mergers (as defined in the Fifth Amendment) or the right to terminate (or cause the termination of) such Person's obligations under the Merger Agreement (giving effect to materiality qualifiers contained in the Merger Agreement) as a result of a breach of such representations in the Merger Agreement.

"Moody's" has the meaning specified therefor in the definition of Cash Equivalents.

"Mortgages" means, individually and collectively, one or more mortgages, deeds of trust, or deeds to secure debt, executed and delivered by Parent or its Subsidiaries in favor of Agent, in form and substance reasonably satisfactory to Agent, that encumber the Real Property Collateral.

"Net Cash Proceeds" means:

(a) with respect to any sale or disposition by Parent or any of its Subsidiaries of assets, the amount of cash proceeds received (directly or indirectly) from time to time (whether as initial consideration or through the payment of deferred consideration) by or on behalf of Parent or its Subsidiaries, in connection therewith after deducting therefrom only (i) the amount of any Indebtedness secured by any Permitted Lien on any asset (other than (A) Indebtedness owing to Agent or any Lender under the Agreement or the other Loan Documents and (B) Indebtedness assumed by the purchaser of such asset) which is required to be, and is, repaid in connection with such sale or disposition, (ii) reasonable fees, commissions, and expenses related thereto and required to be paid by Parent or such Subsidiary in connection with such sale or disposition, (iii) taxes paid or payable to any taxing authorities by Parent or such Subsidiary in connection with such sale or disposition, in each case to the extent, but only to the extent, that the amounts so deducted are, at the time of receipt of such cash, actually paid or payable to a Person that is not an Affiliate of Parent or any of its Subsidiaries, and are properly attributable to such transaction, other than fees payable under the Advisory Agreement permitted by Section 6.10(f); and (iv) all amounts that are set aside as a reserve (A) for adjustments in respect of the purchase price of such assets, (B) for any liabilities associated with such sale or casualty, to the extent such reserve is required by GAAP, and (C) for the payment of unassumed liabilities relating to the assets sold or otherwise disposed of at the time of, or within 30 days after, the date of such sale or other disposition, to the extent that in each case the funds described above in this clause (iv) are (x) deposited into escrow with a third party escrow agent or set aside in a separate Deposit Account that is subject to a Control Agreement in favor of Agent and (y) paid to Agent as a prepayment of the applic

Agreement at such time when such amounts are no longer required to be set aside as such a reserve; and

(db) with respect to the issuance or incurrence of any Indebtedness by Parent or any of its Subsidiaries, or the issuance by Parent or any of its Subsidiaries of any Equity Interests, the aggregate amount of cash received (directly or indirectly) from time to time (whether as initial consideration or through the payment or disposition of deferred consideration) by or on behalf of Parent or such Subsidiary in connection with such issuance or incurrence, after deducting therefrom only (i) reasonable fees, commissions, and expenses related thereto and required to be paid by Parent or such Subsidiary in connection with such issuance or incurrence, (ii) taxes paid or payable to any taxing authorities by Parent or such Subsidiary in connection with such issuance or incurrence, in each case to the extent, but only to the extent, that the amounts so deducted are, at the time of receipt of such cash, actually paid or payable to a Person that is not an Affiliate of Parent or any of its Subsidiaries, and are properly attributable to such transaction, other than fees payable under the Advisory Agreement permitted by Section 6.10(f) of the Agreement.

"Net Working Capital" means, as of any date of determination, Current Assets as of such date minus Current Liabilities as of such date.

"Non-Consenting Lender" has the meaning specified therefor in Section 14.2(a) of the Agreement.

"Non-Defaulting Lender" means each Revolver Lender other than a Defaulting Lender.

"Notes" means the Revolving Loan Notes, the Swing Loan Note and the Term Loan Notes, in each case as amended, supplemented, restated, amended and restated, or otherwise modified from time to time, and "Note" means any of them.

"Obligations" means (a) all Loans, debts, principal, interest (including any interest that accrues after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), reimbursement or indemnification obligations with respect to Letters of Credit (irrespective of whether contingent), premiums, liabilities (including all amounts charged to the Loan Account pursuant to the Agreement), obligations (including indemnification obligations), fees (including the fees provided for in the Fee Letter), Lender Group Expenses (including any fees or expenses that accrue after the commencement of an Insolvency Proceeding, regardless of whether allowed or allowable in whole or in part as a claim in any such Insolvency Proceeding), guaranties, and all covenants and duties of any other kind and description owing by any Loan Party arising out of, under, pursuant to, in connection with, or evidenced by the Agreement or any of the other Loan Documents and irrespective of whether for the payment of money, whether direct or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, and including all interest not paid when due and all other expenses or other amounts that Borrower is required to pay or reimburse by the Loan Documents or by law or otherwise in connection with the Loan Documents, and (b) all Bank Product Obligations. Without limiting the generality of the foregoing, the Obligations of Borrower under the Loan Documents include the obligation to pay (i) the principal of the Revolving Loans and the Term Loan, (iii) interest accrued on the Revolving Loans and the Term Loan, (iii) the amount necessary to reimburse Issuing Bank for amounts paid or payable pursuant to Letters of Credit, (iv) Letter of Credit commissions, fees (including fronting fees) and charges, (v) Lender Group Expenses, (vi) fees payable under the Agreement or any of the other Loan Documents, and (vii) indemnities and other amounts pa

Documents to the Obligations shall include all or any portion thereof and any extensions, modifications, renewals, or alterations thereof, both prior and subsequent to any Insolvency Proceeding.

"OFAC" means the Office of Foreign Assets Control of the U.S. Department of the Treasury.

"Originating Lender" has the meaning specified therefor in Section 13.1(e) of the Agreement.

"Other Taxes" has the meaning set forth in Section 16.1.

"Overnight Bank Funding Rate" shall mean, for any day, the rate per annum (based on a year of 360 days and actual days elapsed) comprised of both overnight federal funds and overnight Eurocurrency borrowings by U.S.-managed banking offices of depository institutions, as such composite rate shall be determined by the Federal Reserve Bank of New York, as set forth on its public website from time to time, and as published on the next succeeding Business Day as the overnight bank funding rate by such Federal Reserve Bank (or by such other recognized electronic source (such as Bloomberg) selected by the Agent for the purpose of displaying such rate) (an "Alternate Source"); provided, that if such day is not a Business Day, the Overnight Bank Funding Rate for such day shall be such rate on the immediately preceding Business Day; provided, further, that if such rate shall at any time, for any reason, no longer exist, a comparable replacement rate determined by the Agent at such time (which determination shall be conclusive absent manifest error). If the Overnight Bank Funding Rate determined as above would be less than zero, then such rate shall be deemed to be zero. The rate of interest charged shall be adjusted as of each Business Day based on changes in the Overnight Bank Funding Rate without notice to the Borrower.

"Parent" has the meaning specified therefor in the preamble to the Agreement.

"Participant" has the meaning specified therefor in Section 13.1(e) of the Agreement.

"Participant Register" has the meaning set forth in Section 13.1(i) of the Agreement.

"Participation Advance" shall have the meaning set forth in Section 2.11(d)(iv) of the Agreement.

"Participation Commitment" means the obligation hereunder of each Lender holding a Revolving Commitment to buy a participation equal to its Pro Rata Share (subject to any reallocation pursuant to Section 2.9(b)(iii) of the Agreement) in the Swing Loans made by Swing Loan Lender hereunder as provided for in Section 2.4(c) of the Agreement and in the Letters of Credit issued as provided for in Section 2.11(d)(i) of the Agreement.

"Patent Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Patriot Act" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, P.L. 107-56, as amended.

"Payment Office" means, initially, Two Tower Center Boulevard, East Brunswick, New Jersey 08816 and such bank account of Agent designated to Borrower; thereafter, such other office and/or account of Agent, if any, which it may designate by notice to Borrower and to each Lender to be the Payment Office.

"Perfection Certificate" means a certificate in the form of Exhibit P-1 to the Agreement.

"Permitted Acquisition" means any acquisition of (i) all or substantially all of the Equity Interests of any Person, or (ii) all or substantially all of the assets of any Person (or of a line of business or division of such an entity) (an "Acquisition") made by a Loan Party (or by a Subsidiary thereof) so long as:

- (a) immediately prior to, and after giving effect to, such Acquisition, Liquidity is and would not be less than \$8,000,000,
- (eb) subject to Section 1.7, no Default or Event of Default shall have occurred and be continuing at the time such Acquisition is consummated or would immediately arise after giving pro forma effect to such Acquisition,
 - (c) such Acquisition is consensual (other than in connection with a bankruptcy court approved sale),
- (d) no Indebtedness would be incurred, assumed or would otherwise exist after giving pro forma effect to such Acquisition other than Permitted Indebtedness, and no Liens will be incurred, assumed or would otherwise exist after giving pro forma effect to such Acquisition other than Permitted Liens,
- (e) the applicable Loan Party has provided Agent with written confirmation, supported by reasonably detailed calculations and otherwise in form satisfactory to Agent, that, on a pro forma basis after giving effect to such Acquisition (and to any Indebtedness incurred, continuing to exist or assumed in connection therewith), Parent and its Subsidiaries on a consolidated basis would have been in compliance with the then-applicable covenants set forth in Section 7 of the Agreement as of the end of the fiscal quarter ended immediately prior to the consummation of the proposed Acquisition (calculated on a pro-forma basis as though such Acquisition had been consummated on the first day of the applicable fiscal measurement period for each such applicable financial covenant),
- upon Agent's reasonable request with respect any Acquisition for which the purchase consideration (including the amount of expected Earnout Obligations (determined as of the date of such Acquisition in accordance with GAAP)) but excluding (x) any working capital adjustments and (y) any portion of the purchase consideration paid or consisting of Equity Interests of a Parent or any of its parent entities (other than Disqualified Equity Interests of a Parent) or the cash proceeds from the sale of such Equity Interests) is greater than \$10,000,000, subject to Agent executing customary non-reliance letters, the applicable Loan Party has provided Agent with its due diligence package relative to the proposed Acquisition, including available historical balance sheets, profit and loss statements, and cash flow statements of the Person or with respect to the assets to be acquired and forecasted, for the four fiscal quarter period ending one year after the date such Acquisition is to be consummated, balance sheets, profit and loss statements, and cash flow statements of the Person or with respect to the assets to be acquired, all prepared on a basis consistent with GAAP and the historical financial statements of such Person or relating to such assets, and in the case of such forecasted statements, together with supporting details and a statement of underlying assumptions,

- (g) the applicable Loan Party has provided Agent with (i) written notice of the proposed Acquisition at least two (2) Business Days (or such shorter period as Agent may consent to) prior to the date such Acquisition is to be consummated and (ii) prior to time such Acquisition is to be consummated, copies of the acquisition agreement and other material documents relative to such Acquisition,
- (h) with respect to any Acquisition, (i) the assets being acquired are located within the United States or (ii) the Person whose Equity Interests are being acquired is organized in a jurisdiction located within the United States; provided that the Loan Parties and their Subsidiaries can acquire assets located outside of the United States and Equity Interests of Persons organized in a jurisdiction not located within the United States so long as the aggregate purchase consideration amount, including deferred payment obligations (including the amount of expected Earnout Obligations (determined as of the date of such Acquisition in accordance with GAAP)) but excluding (x) any working capital adjustments and (y) any portion of the purchase consideration paid or consisting of Equity Interests of a Parent or any of its parent entities (other than Disqualified Equity Interests of a Parent) or the cash proceeds from the sale of such Equity Interests, paid by the Loan Parties does not exceed, so long as Liquidity is an amount equal to or greater than \$40,000,000, an amount equal to \$70,000,000 (less the permitted expenses made under clause (k)(y) of "Permitted Investments") in the aggregate during the term of this Agreement,
 - (i) [reserved],
- (i) after giving effect to such Acquisition, Parent and its Subsidiaries would remain in compliance with <u>Section 6.5</u> and <u>Section 6.13</u> of the Agreement, and
- (k) the applicable Loan Party (or any Subsidiary thereof that is required to become a Loan Party) shall comply with Section 5.11 of the Agreement within the time periods set forth therein and with respect to any such Subsidiary of a Loan Party that is required to become a Loan Party, such Subsidiary shall satisfy, in a manner reasonably acceptable to Agent, customary PATRIOT Act requirements within the time periods set forth in Section 5.11; and
- (I) with respect to all Permitted Acquisitions (including, for the avoidance of doubt, any Permitted Acquisition using the limitations set forth in clause (h) above), the purchase consideration payable in respect of all Permitted Acquisitions (including the proposed Acquisition and including deferred payment obligations (including the amount of expected Earnout Obligations (determined as of the date of such Acquisition in accordance with GAAP)) but excluding (x) any working capital adjustments and (y) any portion of the purchase consideration paid or consisting of Equity Interests of a Parent or any of its parent entities (other than Disqualified Equity Interests of a Parent) or the cash proceeds from the sale of such Equity Interests), shall not exceed the greater of (I) \$110,000,000 in the aggregate during the term of this Agreement or (II) so long as Liquidity is an amount equal to or greater than \$8,000,000.

For the avoidance of doubt, each of the SpeechIQ Acquisition and the Engage Acquisition shall be deemed to be a "Permitted Acquisition" for all purposes under this Agreement.

"Permitted Discretion" means a determination made in the exercise of reasonable (from the perspective of a secured commercial lender) business judgment.

"Permitted Dispositions" means:

- (a) sales, abandonment, or other dispositions of Equipment that is substantially worn, damaged, or obsolete or no longer used or useful in the ordinary course of business and leases or subleases of Real Property not useful in the conduct of the business of Parent and its Subsidiaries,
 - (b) sales of inventory to buyers in the ordinary course of business,
- (c) the use or transfer of money or Cash Equivalents in a manner that is not prohibited by the terms of the Agreement or the other Loan Documents,
- (d) the licensing, on a non-exclusive basis, of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,
 - (e) the granting of Permitted Liens,
- (f) the sale or discount, in each case without recourse, of accounts receivable arising in the ordinary course of business, but only in connection with the compromise or collection thereof,
 - (g) any involuntary loss, damage or destruction of property,
- (h) any involuntary condemnation, seizure or taking, by exercise of the power of eminent domain or otherwise, or confiscation or requisition of use of property,
 - (i) the leasing or subleasing of assets of Parent or its Subsidiaries in the ordinary course of business,
 - (i) the sale or issuance of Equity Interests (other than Disqualified Equity Interests) of Parent,
- (k) (i) the lapse of registered patents, trademarks, copyrights and other intellectual property of Parent and its Subsidiaries to the extent not economically desirable in the conduct of their business or (ii) the abandonment of patents, trademarks, copyrights, or other intellectual property rights in the ordinary course of business so long as (in each case under clauses (i) and (ii)), (A) with respect to copyrights, such copyrights are not material revenue generating copyrights, and (B) such lapse is not materially adverse to the interests of the Lender Group,
 - (1) the making of Restricted Payments that are expressly permitted to be made pursuant to the Agreement,
 - (m) the making of Permitted Investments,
- (fin) so long as no Event of Default has occurred and is continuing or would immediately result therefrom, transfers of assets (i) from Parent or any of its Subsidiaries (other than Borrower) to a Loan Party (other than Parent), and (ii) from any Subsidiary of Parent that is not a Loan Party to any other Subsidiary of Parent,
- (go) dispositions of assets acquired by Parent and its Subsidiaries pursuant to a Permitted Acquisition consummated within 12 months of the date of the proposed disposition so long as (i) the consideration received for the assets to be so disposed is at least equal to the fair market value of such assets and at least 75% payable in cash, (ii) the assets to be so disposed are

not necessary or economically desirable in connection with the business of Borrower and its Subsidiaries, and (iii) the assets to be so disposed are readily identifiable as assets acquired pursuant to the subject Permitted Acquisition, and

(hp) sales or dispositions of assets (other than Equity Interests of Subsidiaries of Parent) not otherwise permitted in clauses (a) through (n) above so long as made at fair market value (and at least 75% payable in cash) and the aggregate fair market value of all assets disposed of in fiscal year (including the proposed disposition) would not exceed \$750,000.

"Permitted Holders" means each of (a) Sponsor, (b) Public Holdco, (c) managers and members of management (including board members and senior advisors) of Public Holdco or its Subsidiaries that have ownership interests in Public Holdco, (d) any other beneficial owner in the common equity of Parent as of the Fifth Amendment Effective Date and (e) any group (within the meaning of Rules 13d-3 and 13d-5 under the Exchange Act) of which any of the Persons described in clauses (a), (c) or (d) above are members (provided that, without giving effect to the existence of such group or any other group, any of the Persons described in clauses (a), (c) and (d) above, collectively, (x) beneficially own Equity Interests representing 50% or more of the total voting power of the Equity Interests of Public Holdco then held by such group and (y) have the power, directly or indirectly, to direct or cause the direction of the management and policies, whether by ownership or general partnership (and not by contract), of such group; provided, further, that such group shall not include any of Sponsor's portfolio companies).

"Permitted Indebtedness" means:

- (a) Indebtedness evidenced by the Agreement or the other Loan Documents,
- (b) Indebtedness set forth on Schedule 4.14 to the Agreement and any Refinancing Indebtedness in respect of such Indebtedness,
- (c) Permitted Purchase Money Indebtedness and any Refinancing Indebtedness in respect of such Indebtedness,
- (d) endorsement of instruments or other payment items for deposit,
- (e) Indebtedness consisting of (i) unsecured guarantees incurred in the ordinary course of business with respect to surety and appeal bonds, performance bonds, bid bonds, appeal bonds, completion guarantee and similar obligations; (ii) unsecured guarantees arising with respect to customary indemnification obligations to purchasers in connection with Permitted Dispositions; and (iii) unsecured guarantees with respect to Indebtedness of Parent or one of its Subsidiaries, to the extent that the Person that is obligated under such guaranty could have incurred such underlying Indebtedness,
- (f) unsecured Indebtedness of Borrower that is incurred on the date of the consummation of a Permitted Acquisition solely for the purpose of consummating such Permitted Acquisition so long as (i) no Event of Default has occurred and is continuing or would result therefrom, (ii) such unsecured Indebtedness is not incurred for working capital purposes, (iii) such unsecured Indebtedness does not mature prior to the date that is 6 months after the Maturity Date, (iv) such unsecured Indebtedness does not amortize until 6 months after the Maturity Date, (v) such unsecured Indebtedness does not provide for the payment of interest thereon in cash or Cash Equivalents prior to the date that is 6 months after the Maturity Date, (v) such unsecured Indebtedness does not provide for the payment of cash prior to the date that is 6 months after the Maturity Date, and (vii) such Indebtedness is subordinated in right of payment

to the Obligations on terms and conditions reasonably satisfactory to Agent, and any Refinancing Indebtedness with respect thereto,

- (g) Acquired Indebtedness in an amount not to exceed \$750,000 outstanding at any one time, and any Refinancing Indebtedness with respect thereto,
 - (h) Indebtedness incurred in the ordinary course of business under performance, surety, statutory, or appeal bonds,
- (i) Indebtedness owed to any Person providing property, casualty, liability, or other insurance to Parent or any of its Subsidiaries, so long as the amount of such Indebtedness is not in excess of the amount of the unpaid cost of, and shall be incurred only to defer the cost of, such insurance for the year in which such Indebtedness is incurred and such Indebtedness is outstanding only during such year,
- (j) the incurrence by Parent or its Subsidiaries of Indebtedness under Hedge Agreements that are incurred for the bona fide purpose of hedging the interest rate, commodity, or foreign currency risks associated with Parent's and its Subsidiaries' operations and not for speculative purposes,
- (k) Indebtedness incurred in the ordinary course of business in respect of credit cards, credit card processing services, debit cards, stored value cards, commercial cards (including so-called "purchase cards", "procurement cards" or "p-cards"), or Cash Management Services,
- (1) unsecured Indebtedness of Parent owing to former employees, officers, or directors (or any spouses, ex-spouses, or estates of any of the foregoing) incurred in connection with the repurchase by Parent of the Equity Interests of Parent that has been issued to such Persons, so long as (i) no Default or Event of Default has occurred and is continuing or would result from the incurrence of such Indebtedness, (ii) the aggregate amount of all such Indebtedness outstanding at any one time does not exceed \$7,000,000, and (iii) such Indebtedness is subordinated to the Obligations on terms and conditions reasonably acceptable to Agent,
- (m) unsecured Indebtedness owing to sellers of assets or Equity Interests to a Loan Party that is incurred by the applicable Loan Party in connection with the consummation of one or more Permitted Acquisitions so long as (i) the aggregate principal amount for all such unsecured Indebtedness does not exceed \$750,000 at any one time outstanding, (ii) is not cash payable before 6 months after the Maturity Date and is otherwise subordinated to the Obligations on terms and conditions reasonably acceptable to Agent, and (iii) is otherwise on terms and conditions (including all economic terms and the absence of covenants) reasonably acceptable to Agent,
- (n) contingent liabilities in respect of any indemnification obligation, adjustment of purchase price, non-compete, or similar obligation of Parent or the applicable Loan Party incurred in connection with the consummation of one or more Permitted Acquisitions,
 - (o) Indebtedness composing Permitted Investments,
- (p) unsecured Indebtedness incurred in respect of netting services, overdraft protection, and other like services, in each case, incurred in the ordinary course of business,
- (g) unsecured Indebtedness of Borrower or its Subsidiaries in respect of (i) the SpeechIQ Holdback, (ii) the SpeechIQ Incentive Payment, (iii) the Teckst Holdback, (iv) the

Teckst Incentive Payment, (v) the Engage Earn-Out and (vi) Earn-Outs owing to sellers of assets or Equity Interests to Borrower or its Subsidiaries that is incurred in connection with the consummation of one or more Permitted Acquisitions so long as solely in the case of this sub-clause (vi) (A) such unsecured Indebtedness is subordinated to the Obligations on terms and conditions acceptable to Agent and (B) such unsecured Indebtedness is otherwise on terms and conditions reasonably acceptable to Agent,

- (r) Indebtedness in an aggregate outstanding principal amount not to exceed \$750,000 at any time outstanding for all Subsidiaries of Parent that are CFCs; <u>provided</u>, that such Indebtedness is not directly or indirectly recourse to any of the Loan Parties or of their respective assets,
- (s) accrual of interest, accretion or amortization of original issue discount, or the payment of interest in kind, in each case, on Indebtedness that otherwise constitutes Permitted Indebtedness,
- (t) the Engage PPP Loan, so long as any potential amount owing with respect to the Engage PPP Loan is fully funded pursuant to an escrow arrangement reasonably satisfactory to the Agent (the Agent hereby acknowledges and agrees that the escrow arrangement shall be in place on the Sixth Amendment Effective Date is satisfactory to the Agent);
- (u) Indebtedness incurred by a non-Loan Party in respect of a working capital facility or the benefit of such non-Loan Party in an aggregate amount not to exceed \$3,000,000, and
- (v) unsecured Indebtedness of Parent owing to former employees, officers, or directors (or any spouses, ex-spouses, or estates of any of the foregoing) incurred in connection with the repurchase by Parent of the Equity Interests of Parent that have been issued to such Persons pursuant to the terms of the Management Incentive Documentation, so long as (i) no Default or Event of Default has occurred and is continuing or would result from the incurrence of such Indebtedness, (ii) the aggregate amount of all such Indebtedness outstanding at any one time does not exceed (A) during the fiscal year ending December 31, 2019, \$11,000,000, (B) during the fiscal year ending December 31, 2020, \$5,500,000 and (C) all other times, \$0, and (iii) such Indebtedness is subordinated to the Obligations on terms and conditions reasonably acceptable to Agent.

"Permitted Intercompany Advances" means loans or advances constituting Indebtedness made by (a) a Loan Party to another Loan Party other than Parent, (b) a Subsidiary of Parent that is not a Loan Party to another Subsidiary of Parent that is not a Loan Party, (c) a Subsidiary of Parent that is not a Loan Party to a Loan Party to another Subsidiary of Parent that is not a Loan Party to a Subsidiary of Borrower that is not a Loan Party so long as (i) the aggregate amount of all such loans or advances (by type, not by the borrower) does not exceed 2,000,000 outstanding at any one time, (ii) at the time of the making of such loan or advance, no Event of Default has occurred and is continuing or would result therefrom, and (iii) Borrower has Liquidity of \$2,500,000 or greater immediately after giving effect to each such loan or advance.

"Permitted Investments" means:

- (a) Investments in cash and Cash Equivalents,
- (b) Investments in negotiable instruments deposited or to be deposited for collection in the ordinary course of business,

- (c) advances made in connection with purchases of goods or services in the ordinary course of business,
- (d) Investments received in settlement of amounts due to any Loan Party or any of its Subsidiaries effected in the ordinary course of business or owing to any Loan Party or any of its Subsidiaries as a result of Insolvency Proceedings involving an account debtor or upon the foreclosure or enforcement of any Lien in favor of a Loan Party or its Subsidiaries,
- (e) Investments owned by any Loan Party or any of its Subsidiaries on the Closing Date, and/or the Third Amendment Effective Date, as applicable, and set forth on <u>Schedule P-1</u> to the Agreement,
 - (f) guarantees permitted under the definition of Permitted Indebtedness,
 - (g) Permitted Intercompany Advances,
- (h) Equity Interests or other securities acquired in connection with the satisfaction or enforcement of Indebtedness or claims due or owing to a Loan Party or its Subsidiaries (in bankruptcy of customers or suppliers or otherwise outside the ordinary course of business) or as security for any such Indebtedness or claims,
 - (i) deposits of cash made in the ordinary course of business to secure performance of operating leases,
- (j) non-cash loans and advances to current or former employees, officers, and directors (or successors in interest to any of the foregoing) of Parent or any of its Subsidiaries for the purpose of purchasing Equity Interests in Parent so long as the proceeds of such loans are used in their entirety to purchase such Equity Interests in Parent not to exceed \$750,000 at any one time,
- (k) (x) Permitted Acquisitions, (y) Investments in any Subsidiary that is not a Loan Party existing under the laws of Canada, the Cayman Islands, England, Wales, Scotland, Ireland, France, Germany, India, Jersey, Israel, Luxembourg, Norway, Spain, Sweden or any other additional jurisdiction reasonably agreed to by Agent solely used to consummate Permitted Acquisitions or to fund the administrative, general corporate and operating expenses of such non-Loan Party and its Subsidiaries in an aggregate amount, together with the amount of all non-U.S. Permitted Acquisitions consummated, not to exceed the amount available to consummate Permitted Acquisitions pursuant to clause (h)(ii) of the definition thereof, and (z) Investments in newly formed Subsidiaries solely for the purposes of consummating a Permitted Acquisition, which amount is actually applied by such Subsidiary to consummate such Permitted Acquisition substantially concurrently with the making of such Investment.
- (1) Investments in the form of capital contributions and the acquisition of Equity Interests made by any Loan Party in any other Loan Party (other than capital contributions to or the acquisition of Equity Interests of Parent),
- (m) Investments resulting from entering into (i) Bank Product Agreements, or (ii) agreements relative to Indebtedness that is permitted under clause (j) of the definition of Permitted Indebtedness,
- (n) equity Investments by any Loan Party in any Subsidiary of such Loan Party which is required by law to maintain a minimum net capital requirement or as may be otherwise required by applicable law,

- (o) Investments held by a Person acquired in a Permitted Acquisition to the extent that such Investments were not made in contemplation of or in connection with such Permitted Acquisition and were in existence on the date of such Permitted Acquisition,
- (p) so long as no Event of Default has occurred and is continuing or would result therefrom, any other Investments in an aggregate amount not to exceed \$3,000,000, and
 - (q) Investments made entirety with Qualified Equity Contributions.

"Permitted Liens" means:

- (a) Liens granted to, or for the benefit of, Agent to secure the Obligations,
- (b) Liens for unpaid taxes, assessments, or other governmental charges or levies that either (i) are not yet delinquent, or (ii) do not have priority over Agent's Liens and the underlying taxes, assessments, or charges or levies are the subject of Permitted Protests,
- (c) judgment Liens arising solely as a result of the existence of judgments, orders, or awards that do not constitute an Event of Default under Section 8.3 of the Agreement,
- (d) Liens set forth on <u>Schedule P-2</u> to the Agreement; <u>provided</u>, that to qualify as a Permitted Lien, any such Lien described on <u>Schedule P-2</u> to the Agreement shall only secure the Indebtedness that it secures on the Closing Date and/or the Third Amendment Effective Date, as applicable, and any Refinancing Indebtedness in respect thereof,
 - (e) the interests of lessors under operating leases and non-exclusive licensors under license agreements,
- (f) purchase money Liens or the interests of lessors under Capital Leases to the extent that such Liens or interests secure Permitted Purchase Money Indebtedness and so long as (i) such Lien attaches only to the asset purchased or acquired and the proceeds thereof, and (ii) such Lien only secures the Indebtedness that was incurred to acquire the asset purchased or acquired or any Refinancing Indebtedness in respect thereof,
- (g) Liens arising by operation of law in favor of warehousemen, landlords, carriers, mechanics, materialmen, laborers, or suppliers, incurred in the ordinary course of business and not in connection with the borrowing of money, and which Liens either (i) are for sums not yet delinquent, or (ii) are the subject of Permitted Protests,
- (h) Liens on amounts deposited to secure Parent's and its Subsidiaries' obligations in connection with worker's compensation or other unemployment insurance,
- (i) Liens on amounts deposited to secure Parent's and its Subsidiaries' obligations in connection with the making or entering into of bids, tenders, and leases, the performance under government contracts, and statutory obligations, in each case in the ordinary course of business and not in connection with the borrowing of money,
- (j) Liens on amounts deposited to secure Parent's and its Subsidiaries' reimbursement obligations with respect to surety or appeal bonds obtained in the ordinary course of business,
- (k) with respect to any Real Property, easements, rights of way, and zoning restrictions that do not materially interfere with or impair the use or operation thereof,

- (1) non-exclusive licenses of patents, trademarks, copyrights, and other intellectual property rights in the ordinary course of business,
- (m) Liens that are replacements of Permitted Liens to the extent that the original Indebtedness is the subject of permitted Refinancing Indebtedness and so long as the replacement Liens only encumber those assets that secured the original Indebtedness,
- (n) rights of setoff or bankers' liens upon deposits of cash in favor of banks or other depository institutions, solely to the extent incurred in connection with the maintenance of such deposit accounts in the ordinary course of business,
- (o) Liens granted in the ordinary course of business on the unearned portion of insurance premiums securing the financing of insurance premiums to the extent the financing is permitted under the definition of Permitted Indebtedness,
- (p) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods,
- (q) Liens solely on any cash earnest money deposits made by Parent or any of its Subsidiaries in connection with any letter of intent or purchase agreement with respect to a Permitted Acquisition,
 - (r) Liens assumed by Borrower or its Subsidiaries in connection with a Permitted Acquisition that secure Acquired Indebtedness,
- (s) Liens on cash and Cash Equivalents maintained in Deposit Accounts or Securities Accounts of Parent or any Subsidiary that secure Permitted Indebtedness incurred by such Person and as to which the aggregate amount of the obligations secured thereby does not exceed \$750,000 outstanding at any time, and
 - (t) Liens securing obligations permitted under clause (u) of the definition of "Permitted Indebtedness" not exceeding \$3,000,000.
- "Permitted Protest" means the right of Parent or any of its Subsidiaries to protest any Lien (other than any Lien that secures the Obligations), taxes (other than payroll taxes or taxes that are the subject of a United States federal tax lien), or rental payment, provided that (a) a reserve with respect to such obligation is established on Parent's or its Subsidiaries' books and records in such amount as is required under GAAP, (b) any such protest is instituted promptly and prosecuted diligently by Parent or its Subsidiary, as applicable, in good faith, and (c) Agent is satisfied that, while any such protest is pending, there will be no impairment of the enforceability, validity, or priority of any of Agent's Liens.
- "Permitted Purchase Money Indebtedness" means, as of any date of determination, Indebtedness (other than the Obligations, but including Capitalized Lease Obligations), incurred after the Closing Date and at the time of, or within 20 days after, the acquisition of any fixed assets for the purpose of financing all or any part of the acquisition cost thereof, in an aggregate principal amount outstanding at any one time not in excess of \$7,000,000.
- "Permitted Term Out Fee" means payment, pursuant to the Advisory Agreement, of a fee in lieu of the annual management fee otherwise due thereunder, equal to the amount of such annual management fee that would have been paid over the then remaining term of the Advisory Agreement so long as (a) such fee is paid upon the consummation of an IPO, (b) no

Default or Event of Default has occurred and is continuing or would result therefrom, (c) after giving pro forma effect to such payment the Loan Parties would be in compliance with the applicable covenants set forth in <u>Section 7</u> as of the most recently ended fiscal quarter and (d) except to the extent such payment is made with proceeds of such IPO, after taking into account all such payments to be made on such date, the Loan Parties would have Liquidity of at least \$5,000,000.

"Person" means natural persons, corporations, limited liability companies, limited partnerships, general partnerships, limited liability partnerships, joint ventures, trusts, land trusts, business trusts, or other organizations, irrespective of whether they are legal entities, and governments and agencies and political subdivisions thereof.

"Plan" means a confirmed plan of reorganization or similar dispositive restructuring plan affecting creditors' rights generally pursuant to an Insolvency Proceeding.

"Platform" has the meaning specified therefor in Section 17.9(c) of the Agreement.

"PNC" means PNC Bank, National Association, a national banking association.

"PPP Program" shall mean the "Paycheck Protection Program" added to Section 7(a) of the Small Business Act, as enacted pursuant to the terms of Title 1 (Keeping American Workers Paid and Employed Act) of the CARES Act and administered by the SBA.

"Projections" means Parent's Public Holdco's consolidated forecasted (a) balance sheets, (b) profit and loss statements, and (c) cash flow statements, all prepared on a basis consistent with Parent's Public Holdco's historical financial statements, together with appropriate supporting details and a statement of underlying assumptions.

"Pro Rata Share" means, as of any date of determination:

with respect to a Lender's obligation to make all or a portion of the Revolving Loans, with respect to such Lender's right to receive payments of interest, fees, and principal with respect to the Revolving Loans, and with respect to all other computations and other matters related to the Revolver Commitments or the Revolving Loans, the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender by (ii) the aggregate Revolving Loan Exposure of all Lenders,

with respect to a Lender's obligation to participate in the Letters of Credit, with respect to such Lender's obligation to reimburse Issuing Bank, and with respect to such Lender's right to receive payments of Letter of Credit Fees, and with respect to all other computations and other matters related to the Letters of Credit, the percentage obtained by dividing (i) the Revolving Loan Exposure of such Lender by (ii) the aggregate Revolving Loan Exposure of all Lenders; <u>provided</u>, that if all of the Revolving Loans have been repaid in full and all Revolver Commitments have been terminated, but Letters of Credit remain outstanding, Pro Rata Share under this clause shall be determined as if the Revolver Commitments had not been terminated and based upon the Revolver Commitments as they existed immediately prior to their termination.

with respect to a Lender's obligation to make the Term Loan and right to receive payments of interest, fees, and principal with respect thereto, (i) prior to the making of the Term Loan, the percentage obtained by dividing (y) such Lender's Term Loan Commitment, by (z) the aggregate amount of all Lenders' Term Loan Commitments, and (ii) from and after the making

of the Term Loan, the percentage obtained by dividing (y) the principal amount of such Lender's portion of the Term Loan by (z) the principal amount of the Term Loan, and

with respect to all other matters and for all other matters as to a particular Lender (including the indemnification obligations arising under Section 15.7 of the Agreement), the percentage obtained by dividing (i) the sum of the Term Loan Exposure of such Lender plus the Revolving Loan Exposure of such Lender by (ii) the sum of the aggregate Term Loan Exposure of all Lenders plus the aggregate Revolving Loan Exposure of all Lenders, in any such case as the applicable percentage may be adjusted by assignments permitted pursuant to Section 13.1; provided, that if all of the Loans have been repaid in full, all Letters of Credit have been made the subject of Letter of Credit Collateralization, and all Commitments have been terminated, Pro Rata Share under this clause shall be determined as if the Revolving Loan Exposures and Term Loan Exposures had not been repaid, collateralized, or terminated and shall be based upon the Revolving Loan Exposures and Term Loan Exposures as they existed immediately prior to their repayment, collateralization, or termination.

"Protective Advances" has the meaning specified therefor in Section 2.1(b)(i) of the Agreement.

"Public Holdco" means LiveVox Holdings, Inc., a Delaware corporation (f/k/a Crescent Acquisition Corp.)

"Public Lender" has the meaning specified therefor in Section 17.9(c) of the Agreement.

"Published Rate" means the rate of interest published each Business Day in the Wall Street Journal "Money Rates" listing under the caption "London Interbank Offered Rates" for a one month period (or, if no such rate is published therein for any reason, then the Published Rate shall be the LIBOR Rate for a one month period as published in another publication selected by Agent).

"Purchase Price" means, with respect to any Acquisition, an amount equal to the aggregate consideration, whether cash, property or securities (including the fair market value of any Equity Interests of Parent issued in connection with such Acquisition and including the maximum amount of Earn-Outs), paid or delivered by Parent or one of its Subsidiaries in connection with such Acquisition (whether paid at the closing thereof or payable thereafter and whether fixed or contingent), but excluding therefrom (a) any cash of the seller and its Affiliates used to fund any portion of such consideration and (b) any cash or Cash Equivalents acquired in connection with such Acquisition.

"Qualified Cash" means, as of any date of determination, the amount of unrestricted cash and Cash Equivalents of Borrower and its Subsidiaries that is in Deposit Accounts or in Securities Accounts, or any combination thereof, and which such Deposit Account or Securities Account is the subject of a Control Agreement and is maintained by a branch office of the bank or securities intermediary located within the United States; provided, that so long as Borrower is in compliance with Section 5.14 of the Agreement, until the 90th day after the Closing Date, all unrestricted cash and Cash Equivalents of Borrower and its Subsidiaries that are required to become subject to Control Agreements or closed under Section 5.14 of the Agreement shall constitute Qualified Cash.

"Qualified Equity Contribution" means an investment of Qualified Equity Interests.

"Qualified Equity Interest" means and refers to any Equity Interests issued by Parent (and not by one or more of its Subsidiaries) that is not a Disqualified Equity Interest.

"Real Property" means any estates or interests in real property now owned or hereafter acquired by Parent or its Subsidiaries and the improvements thereto.

"Real Property Collateral" means (a) the Real Property identified on Schedule R-1 to the Agreement and (b) any Real Property hereafter acquired by Parent or its Subsidiaries after the Closing Date.

"Record" means information that is inscribed on a tangible medium or that is stored in an electronic or other medium and is retrievable in perceivable form.

"Recipient" has the meaning set forth in Section 16.2(e) of the Agreement.

"Recurring Revenues" means, with respect to any period, all customer license-based or usage-based revenue attributable to the software owned, licensed or maintained by Borrower or its Subsidiaries which shall include, for the avoidance of doubt, all minute-based charges, text messaging revenue, call-based charges, local caller ID packages, agent-leg minutes, agent recurring charges, call recording charges, telco charges and any similar recurring fees from the Applicable Affiliate Transaction Documents.

"Reference Period" has the meaning set forth in the definition of EBITDA.

"Refinancing Indebtedness" means refinancings, renewals, or extensions of Indebtedness so long as:

- (a) such refinancings, renewals, or extensions do not result in an increase in the principal amount of the Indebtedness so refinanced, renewed, or extended, other than by the amount of premiums paid thereon and the fees and expenses incurred in connection therewith and by the amount of unfunded commitments with respect thereto,
- (b) such refinancings, renewals, or extensions do not result in a shortening of the average weighted maturity (measured as of the refinancing, renewal, or extension) of the Indebtedness so refinanced, renewed, or extended, nor are they on terms or conditions that, taken as a whole, are or could reasonably be expected to be materially adverse to the interests of the Lenders,
- (c) if the Indebtedness that is refinanced, renewed, or extended was subordinated in right of payment to the Obligations, then the terms and conditions of the refinancing, renewal, or extension must include subordination terms and conditions that are at least as favorable to the Lender Group as those that were applicable to the refinanced, renewed, or extended Indebtedness, and
- (d) the Indebtedness that is refinanced, renewed, or extended is not recourse to any Person that is liable on account of the Obligations other than those Persons which were obligated with respect to the Indebtedness that was refinanced, renewed, or extended.

"Register" has the meaning set forth in Section 13.1(h) of the Agreement.

"Registered Loan" has the meaning set forth in Section 13.1(h) of the Agreement.

"Reimbursement Obligation" has the meaning specified therefor in Section 2.11(d)(ii) of the Agreement.

"Related Fund" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course and that is administered, advised or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers, advises or manages a Lender.

"Remedial Action" means all actions taken to (a) clean up, remove, remediate, contain, treat, monitor, assess, evaluate, or in any way address Hazardous Materials in the indoor or outdoor environment, (b) prevent or minimize a release or threatened release of Hazardous Materials so they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, (c) restore or reclaim natural resources or the environment, (d) perform any pre-remedial studies, investigations, or post-remedial operation and maintenance activities, or (e) conduct any other actions with respect to Hazardous Materials required by Environmental Laws.

"Reorganization Securities" means any notes, equity interests, or other securities (whether debt, equity, or otherwise) issued by a reorganized debtor that are distributed pursuant to a Plan on account of the Obligations in any Insolvency Proceeding of a Loan Party.

"Replacement Lender" has the meaning specified therefor in Section 2.17(b) of the Agreement.

"Report" has the meaning specified therefor in Section 15.16 of the Agreement.

"Required Lenders" means, at any time, Lenders having or holding more than 50% of the sum of (a) the aggregate Revolving Loan Exposure of all Lenders, plus (b) the aggregate Term Loan Exposure of all Lenders; provided, that (i) the Revolving Loan Exposure and Term Loan Exposure of any Defaulting Lender shall be disregarded in the determination of the Required Lenders, and (ii) subject to Section 13.1(k) of the Agreement, any portion of the Revolving Loan Exposure or Term Loan Exposure held by Sponsor Affiliated Entities shall not be included in the calculation of "Required Lenders".

"Reserve Percentage" means as of any day the maximum effective percentage in effect on such day as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the reserve requirements (including supplemental, marginal and emergency reserve requirements) with respect to eurocurrency funding (currently referred to as "Eurocurrency Liabilities").

"Restricted Payment" means to (a) declare or pay any dividend or make any other payment or distribution, directly or indirectly, on account of Equity Interests issued by Parent (including any payment in connection with any merger or consolidation involving Parent) or to the direct or indirect holders of Equity Interests issued by Parent in their capacity as such (other than dividends or distributions payable in Qualified Equity Interests issued by Parent, or (b) purchase, redeem, make any sinking fund or similar payment, or otherwise acquire or retire for value (including in connection with any merger or consolidation involving Parent) any Equity Interests issued by Parent, and (c) make any payment to retire, or to obtain the surrender of, any outstanding warrants, options, or other rights to acquire Equity Interests of Parent now or hereafter outstanding.

"Revolver Commitment" means, with respect to each Revolving Lender, its Revolver Commitment, and, with respect to all Revolving Lenders, their Revolver Commitments, in each case as such Dollar amounts are set forth beside such Revolving Lender's name under the applicable heading on Schedule C-1 to Exhibit A to the Seventh Amendment or in the Assignment and Acceptance pursuant to which such Revolving Lender became a Revolving Lender under the Agreement, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 13.1 of the Agreement.

"Revolver Usage" means, as of any date of determination, the sum of (a) the amount of outstanding Loans, other than the Term Loan, plus (b) the Maximum Undrawn Amount of all then outstanding Letters of Credit.

"Revolving Interest Rate" means (a) with respect to Loans, other than the Term Loan, that are, in each case, Base Rate Loans, an interest rate per annum equal to the sum of the Applicable Base Rate Margin <u>plus</u> the Alternate Base Rate and (b) with respect to Revolving Loans that are <u>LIBORTerm SOFR</u> Rate Loans, an interest rate per annum equal to the sum of the Applicable <u>LIBORTerm SOFR</u> Rate Margin <u>plus</u> the <u>greater of (i) the LIBORTerm SOFR</u> Rate <u>and (ii) 1.00%</u>.

"Revolving Lender" means a Lender that has a Revolving Loan Commitment or Participation Commitment or that has an outstanding Revolving Loan.

"Revolving Loan Exposure" means, with respect to any Revolving Lender, as of any date of determination (a) prior to the termination of the Revolver Commitments, the amount of such Lender's Revolver Commitment, and (b) after the termination of the Revolver Commitments, the aggregate outstanding principal amount of the Loans (other than the Term Loan) of such Lender and such Lender's Participation Commitment with respect to then outstanding Letters of Credit.

"Revolving Loan Note" has the meaning specified in Section 2.1(a) of the Agreement.

"Revolving Loans" means the Loans made under Section 2.1(a) of the Agreement.

"Sanctioned Country" means, at any time, a country or territory which is itself the subject or target of any Sanctions (which, as of the date of the Credit Agreement, include Crimea, Cuba, Iran, North Korea, Sudan and Syria).

"Sanctioned Person" means, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by OFAC or the U.S. Department of State or by the United Nations Security Council, the European Union, any European Union member state, Her Majesty's Treasury of the United Kingdom or other relevant sanctions authority, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned or controlled by any such Person or Persons.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by (a) the U.S. government, including those administered by OFAC or the U.S. Department of State or (b) the United Nations Security Council, the European Union, any European Union member state, Her Majesty's Treasury of the United Kingdom or other relevant sanctions authority.

"S&P" has the meaning specified therefor in the definition of Cash Equivalents.

"SEC" means the United States Securities and Exchange Commission and any successor thereto.

"Second Amendment" means that certain Second Amendment to Credit Agreement, dated as of the Second Amendment Effective Date.

"Second Amendment Effective Date" means May 6, 2019.

"Securities Account" means a securities account (as that term is defined in the Code).

"Securities Act" means the Securities Act of 1933, as amended from time to time, and any successor statute.

"Settlement" has the meaning specified therefor in Section 2.6(d) of the Agreement.

"Settlement Date" has the meaning specified therefor in Section 2.6(d) of the Agreement.

"Seventh Amendment" means that certain Seventh Amendment to Credit Agreement, dated as of the Seventh Amendment Effective Date.

"Seventh Amendment Effective Date" has the meaning specified in the Seventh Amendment.

"Seventh Amendment Fee Letter" means that certain Second Amended and Restated Fee Letter, dated as of the Seventh Amendment Effective Date, by and between Borrower and Agent.

"SOFR" shall mean, for any day, a rate equal to the secured overnight financing rate as administered by the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Adjustment" shall mean, for Term SOFR, the following:

SOFR Adjustment	<u>Interest Period</u>
10 basis points (0.10%)	For a 1-month Interest Period
15 basis points (0.15%)	For a 3-month Interest Period

"SOFR Floor" means a rate of interest per annum equal to 100 basis points (1.00%).

"SOFR Reserve Percentage" shall mean, for any day, the maximum effective percentage in effect on such day, if any, as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the reserve requirements (including, without limitation, supplemental, marginal and emergency reserve requirements) with respect to SOFR funding.

"Solvent" means, with respect to any Person as of any date of determination, that (a) at fair valuations, the sum of such Person's debts (including contingent liabilities) is less than all of such Person's assets, (b) such Person is not engaged or about to engage in a business or transaction for which the remaining assets of such Person are unreasonably small in relation to the business or transaction or for which the property remaining with such Person is an unreasonably small capital, and (c) such Person has not incurred and does not intend to incur, or reasonably believe that it will incur, debts beyond its ability to pay such debts as they become due (whether at maturity or otherwise), and (d) such Person is "solvent" or not "insolvent", as applicable within the meaning given those terms and similar terms under applicable laws relating to fraudulent transfers and conveyances. For purposes of this definition, the amount of any contingent liability at any time shall be computed as the amount that, in light of all of the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability (irrespective of whether such contingent liabilities meet the criteria for accrual under Statement of Financial Accounting Standard No. 5).

"Specified Event of Default" means an Event of Default under Section 8.1 (including, without limitation, due to a failure to pay any of the Obligations upon the acceleration thereof), Section 8.4 or Section 8.5 of the Agreement.

"SpeechIQ Acquisition" shall mean the transactions contemplated by the SpeechIQ Acquisition Agreement.

"SpeechIQ Acquisition Agreement" shall mean the Unit Purchase Agreement, dated on or about the Third Amendment Effective Date, by and among Borrower, SpeechIQ Sellers and Kyle F. Hanson, solely in the capacity of the Seller Rep (as defined therein).

"SpeechIQ Acquisition Documentation" shall mean, collectively, the SpeechIQ Acquisition Agreement and all schedules, exhibits and annexes thereto, all amendments, waivers and modifications thereto, and all material agreements, instruments, and documents relating to the SpeechIQ Acquisition executed by any Loan Party (or any of their Affiliates or Subsidiaries) and affecting the terms thereof or entered into by any Loan Party (or any of their Affiliates or Subsidiaries) in connection therewith.

"SpeechIQ Acquisition Agreement Representations" shall mean the representations and warranties made by the SpeechIQ Sellers (or any of their Affiliates or Subsidiaries) in the SpeechIQ Acquisition Agreement to the extent that Borrower (or any of its Affiliates or Subsidiaries) has the right not to consummate the SpeechIQ Acquisition or the right to terminate (or cause the termination of) such Person's obligations under the SpeechIQ Acquisition Agreement (giving effect to materiality qualifiers contained in the SpeechIQ Acquisition Agreement) as a result of a breach of such representations in the SpeechIQ Acquisition Agreement.

"SpeechIQ Holdback" shall mean that unsecured obligation owing by Borrower to the SpeechIQ Sellers to pay the amount pursuant to Section 7.9 of the SpeechIQ Acquisition Agreement, in an aggregate amount not to exceed \$1,200,000.

"SpeechIQ Incentive Payment" shall mean that unsecured obligation owing by Borrower to management or any other employees, as applicable, to make an incentive payment pursuant to the SpeechIQ Acquisition Agreement, in an aggregate amount not to exceed \$1,000,000.

"SpeechIQ Sellers" shall mean the Sellers (as defined in the SpeechIQ Acquisition Agreement).

"Sponsor" means Golden Gate Capital.

"Sponsor Affiliated Entity" means Sponsor or any of its Affiliates (other than Loan Parties or their Subsidiaries and other than operating portfolio companies of Sponsor and its Affiliates.

"Subordinated Indebtedness" means any unsecured Indebtedness of Parent or its Subsidiaries incurred from time to time that is subordinated in right of payment to the Obligations and (a) that is only guaranteed by the Guarantors, (b) that is not subject to any cash payments, scheduled amortization, redemption, sinking fund or similar payment and does not have a final maturity, in each case, on or before the date that is six months after the Maturity Date, (c) that does not include any financial covenants or any covenant or agreement that is more restrictive or onerous on any Loan Party in any material respect than any comparable covenant in the Agreement, (d) shall be limited to cross-payment default and cross-acceleration to designated "senior debt" (including the Obligations"), and (e) the terms and conditions of the subordination are reasonably acceptable to Agent.

"Subsidiary" of a Person means a corporation, partnership, limited liability company, or other entity in which that Person directly or indirectly owns or controls the Equity Interests having ordinary voting power to elect a majority of the Board of Directors of such corporation, partnership, limited liability company, or other entity.

"Swing Loan Lender" means PNC or any other Lender that, at the request of Borrower and with the consent of Agent agrees, in such Lender's sole discretion, to become the Swing Loan Lender under Section 2.4) of the Agreement.

"Swing Loans" has the meaning specified therefor in Section 2.4(a) of the Agreement.

"Swing Loan Note" has the meaning specified in Section 2.4(a) of the Agreement.

"Taxes" means any taxes, levies, imposts, duties, fees, assessments or other charges of whatever nature now or hereafter imposed by any jurisdiction or by any political subdivision or taxing authority thereof or therein, and all interest, penalties or similar liabilities with respect thereto.

"Tax Group" has the meaning specified therefor in Section 6.7(c) of the Agreement.

"Tax Lender" has the meaning specified therefor in Section 14.2(a) of the Agreement.

"<u>Teckst Acquisition Agreement</u>" shall mean the Asset Purchase Agreement, dated as of October 16, 2019, by and among Borrower and Teckst, Inc., a Delaware corporation.

"<u>Teckst Holdback</u>" shall mean that unsecured obligation owing by Borrower to Teckst Seller to make a holdback adjustment pursuant to the Teckst Acquisition Agreement, in an aggregate amount not to exceed \$450,000.

"<u>Teckst Incentive Payment</u>" shall mean that unsecured obligation owing by Borrower to management or any other employees, as applicable, to make a management incentive payment pursuant to the Teckst Acquisition Agreement, in an aggregate amount not to exceed \$1,000,000.

"Teckst Seller" means Teckst, Inc., a Delaware corporation.

"Term Loan" has the meaning specified therefor in Section 2.3 of the Agreement.

"Term Loan Commitment" means, with respect to each Lender, its Term Loan Commitment, and, with respect to all Lenders, their Term Loan Commitments, in each case as such Dollar amounts are set forth beside such Lender's name under the applicable heading on Schedule C-1 to Exhibit A to the Seventh Amendment and in case as may have been adjusted by an Assignment and Acceptance pursuant to which such Lender became a Lender, as such amounts may be reduced or increased from time to time pursuant to assignments made in accordance with the provisions of Section 13.1 of the Agreement.

"Term Loan Exposure" means, with respect to any Term Loan Lender, as of any date of determination (a) prior to the funding of the Third Amendment Term Loan, the amount of such Lender's Term Loan Commitment, and (b) after the funding of the Third Amendment Term Loan, the outstanding principal amount of the Term Loan held by such Lender.

"Term Loan Lender" means a Lender that has a Term Loan Commitment or that has a portion of the Term Loan.

"Term Loan Note" has the meaning specified in Section 2.3 of the Agreement.

"Term Loan Rate" means (a) with respect to portions of the Term Loan that are Base Rate Loans, an interest rate per annum equal to the sum of the Applicable Base Rate Margin plus the Alternate Base Rate and (b) with respect to portions of the Term Loan that are LIBORTerm SOFR Rate Loans, an interest rate per annum equal to the sum of the Applicable LIBORTerm SOFR Rate Margin plus the greater of (i) the LIBORTerm SOFR Rate and (ii) 1.00%.

"Term SOFR Administrator" means CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by Agent in its reasonable discretion).

"Term SOFR Rate" shall mean, with respect to any Term SOFR Rate Loan for any Interest Period, the interest rate per annum determined by Agent by dividing (the resulting quotient rounded upwards, at Agent's discretion, to the nearest 1/100th of 1%) (A) the Term SOFR Reference Rate for a tenor comparable to such Interest Period on the day (the "Term SOFR Determination Date") that is two (2) Business Days prior to the first day of such Interest Period, as such rate is published by the Term SOFR Administrator, by (B) a number equal to 1.00 minus the SOFR Reserve Percentage, plus the SOFR Adjustment for the applicable Interest Period. If the Term SOFR Reference Rate for the applicable tenor has not been published or replaced with a Benchmark Replacement by 5:00 p.m. (Pittsburgh, Pennsylvania time) on the Term SOFR Determination Date, then the Term SOFR Reference Rate, for purposes of clause (A) in the preceding sentence, shall be the Term SOFR Reference Rate for such tenor on the first Business Day preceding such Term SOFR Determination Date for which such Term SOFR Reference Rate for such tenor was published in accordance herewith, so long as such first preceding Business Day is not more than three (3) Business Days prior to such Term SOFR Determination Date. If the Term SOFR Rate, determined as provided above, would be less than the SOFR Floor, then the Term SOFR Rate shall be deemed to be the SOFR Floor. The Term SOFR Rate shall be adjusted automatically without notice to the Borrower on and as of (i) the first day of each Interest Period, and (ii) the effective date of any change in the SOFR Reserve Percentage.

"Term SOFR Rate Loan" means a Loan that bears interest based on Term SOFR Rate.

"Term SOFR Reference Rate" shall mean the forward-looking term rate based on SOFR.

"Third Amendment" means that certain Third Amendment and Waiver to Credit Agreement, dated as of the Third Amendment Effective Date.

"Third Amendment Disbursement Letter" means the letter executed by Borrower and delivered to Agent on the Third Amendment Effective Date directing Agent to disburse the proceeds of the Third Amendment Term Loan as detailed therein.

"Third Amendment Effective Date" has the meaning specified in the Third Amendment.

"Third Amendment Term Loan" has the meaning specified in Section 2.3 hereof.

"Trademark Security Agreement" has the meaning specified therefor in the Guaranty and Security Agreement.

"Triggering Event" means the occurrence of (a) an exercise of remedies by Agent against all Collateral or a material portion of the Collateral, (b) and during the continuation of an Event of Default under Section 8.1 of the Agreement with respect to any payment of (i) principal, (ii) interest, or (iii) other Obligations in excess of \$250,000 in the case of this subclause (b) until the payment underlying such Event of Default is paid, or (c) an Event of Default under Sections 8.4 or 8.5 of the Agreement.

"TTM EBITDA" means, as of any date of determination, EBITDA of Parent determined on a consolidated basis in accordance with GAAP, for the 12 month period most recently ended.

"<u>Unfunded Capital Expenditures</u>" means, as to Parent and its Subsidiaries on a consolidated basis in accordance with GAAP, without duplication, Capital Expenditures funded (a) from the Parent's and/or its Subsidiaries' internally generated cash flow or (b) with the proceeds of a Revolving Loan, Letter of Credit or Swing Loan.

"United States" means the United States of America.

"Unused Line Fee" has the meaning specified therefor in Section 2.15(b) of the Agreement.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States governmental securities.

"Voidable Transfer" has the meaning specified therefor in Section 17.8 of the Agreement.

Schedule 3.1

The obligation of each Lender to make its initial extension of credit provided for in the Agreement is subject to the fulfillment, to the satisfaction of each Lender (the making of such initial extension of credit by any Lender being conclusively deemed to be its satisfaction or waiver of the following), of each of the following conditions precedent:

- (a) Agent shall have received each of the following documents, in form and substance reasonably satisfactory to Agent and Lenders, duly executed and delivered, and each such document shall be in full force and effect:
- (i) a letter duly executed by Borrower and Parent authorizing Agent and its agents (including legal counsel) to file appropriate financing statements in such office or offices as may be necessary or, in the opinion of Agent, desirable to perfect the security interests to be created by the Loan Documents,
 - (ii) this Agreement,
 - (iii) the Fee Letter,
 - (iv) the Closing Date Disbursement Letter,
 - (v) the Guaranty and Security Agreement,
 - (vi) the Intercompany Subordination Agreement,
 - (vii) a Perfection Certificate,

(viii) a letter, in form and substance reasonably satisfactory to Agent, from Wells Fargo Bank, National Association (in its capacity as agent "Existing Agent") and the lenders for whom it is acting as agent (collectively, the "Existing Lenders") respecting the amount necessary to repay in full all of the obligations (other than unasserted contingent indemnification obligations) of the Loan Parties owing under, or in connection with, the credit facility provided by Existing Lenders to Borrower and releasing all of the liens existing in favor of Existing Agent and/or the Existing Lenders in and to the assets of the Loan Parties, together with termination statements and other documentation evidencing the termination by Existing Agent and/or Existing Lenders of their Liens in and to the properties and assets of the Loan Parties,

- (ib) Agent shall have received a certificate from the Secretary of each Loan Party (i) attesting to the resolutions of such Loan Party's board of directors authorizing its execution, delivery, and performance of the Loan Documents to which it is a party, (ii) authorizing specific officers of such Loan Party to execute the same, and (iii) attesting to the incumbency and signatures of such specific officers of such Loan Party;
- (jc) Agent shall have received copies of each Loan Party's Governing Documents, as amended, modified, or supplemented to the Closing Date, certified by the Secretary of such Loan Party;
- (kd) Agent shall have received a certificate of status or compliance or the applicable equivalent thereof with respect to each Loan Party, dated within 30 days of the Closing Date, such certificate to be issued by the appropriate officer of the jurisdiction of organization of such Loan Party, which certificate shall indicate that such Loan Party is in good standing in such jurisdiction (to the extent the applicable jurisdiction provides such a certificate);

- (Le) Agent shall have received certificates of status or compliance or the applicable equivalent thereof with respect to each Loan Party, each dated within 30 days of the Closing Date, such certificates to be issued by the appropriate officer of the jurisdictions (other than the jurisdiction of organization of such Loan Party) in which its failure to be duly qualified or licensed would constitute a Material Adverse Effect, which certificates shall indicate that such Loan Party is in good standing in such jurisdictions (to the extent the applicable jurisdiction provides such a certificate);
- (mf) Agent shall have received legal opinions from counsel to each Loan Party in form and substance reasonably satisfactory to Agent;
- (ng) After giving effect to the initial extensions of credit under the Agreement, the repayment of amounts owing to the Existing Lenders and the payment of all fees and expenses required to be paid by Borrower on the Closing Date under the Agreement or the other Loan Documents: (i) no Revolving Loans shall be outstanding and (ii) Borrower shall have Qualified Cash of not less than \$1,500,000;
- (oh) Borrower shall have paid all Lender Group Expenses incurred in connection with the transactions evidenced by the Agreement and the other Loan Documents;
- (pi) Agent shall have received true, correct, and complete copies of the Advisory Agreement, the other Material Contracts and such other licenses, developer agreements, re-seller agreements and vendor supply agreements as Agent shall have reasonably requested at least five (5) Business Days prior to the Closing Date;
- (qi) The final legal and capital structure of Parent and its Subsidiaries shall be acceptable to Agent, including, but not limited to, Sponsor's existing cash equity contribution in the Borrower of not less than \$54,000,000;
- (†k) (i) No litigation, investigation or proceeding before or by any arbitrator or Governmental AuthorityBody shall be continuing or threatened against any Loan Party or against the officers or directors of any Loan Party (A) in connection with the Agreement, the other Loan Documents, or any of the transactions contemplated hereby or referenced herein which, in the reasonable opinion of Agent, is deemed material or (B) which could, in the reasonable opinion of Agent, have a Material Adverse Effect; and (ii) no injunction, writ, restraining order or other order of any nature materially adverse to any Loan Party or the conduct of its business or inconsistent with the due consummation of the s shall have been issued by any Governmental AuthorityBody;
- (si) Each (i) Uniform Commercial Code financing statement and filing with the United States Patent and Trademark Office and the United States Copyright Office required by this Agreement, any other Loan Document, or under applicable law requested by the Agent to be filed, registered or recorded in order to create, in favor of Agent, a perfected security interest in or lien upon the Collateral subject thereto shall have been delivered to Agent in proper form for filing, registration or recordation in each jurisdiction in which the filing, registration or recordation thereof is so required or requested by Agent together with payment of any necessary fee, tax or expense relating thereto and (ii) copies of stock certificates evidencing Collateral, together with copies of transfer powers executed in blank, and copies of each promissory note constituting Collateral, together with copies of execute allonges, shall have been received by Agent or its counsel;
- (tm) Agent shall have received, and been satisfied with its review of, a harvest scenario/recurring revenue valuation report and historical churn analysis with respect to the Borrower and its Subsidiaries, performed by CTS Capital Advisors;

- (<u>HI</u>) Agent's receipt and satisfactory review of evidence that for the trailing 12 month period ending June 30, 2016 the Borrower has earnings before interest, taxes, depreciation and amortization of not less than \$6,400,000;
- (vo) Satisfactory receipt and review by Agent of a financial condition certificate in the form attached to the Agreement as Exhibit 3.1(o) attaching thereto (i) annual and interim historical financial statements for Parent and its Subsidiaries; (ii) quarterly consolidated projections for Parent and its Subsidiaries for the first 12 months following the Closing Date; (iii) quarterly consolidated projections for Parent and its Subsidiaries for the 2nd through 5th years following the Closing Date; and (iv) a pro forma opening balance sheet, which shall, as applicable, reflect the final pro forma capital structure of Parent and its Subsidiaries and demonstrate the Borrower's ability to service the Obligations;
- (wp) The Loan Parties shall be in compliance (i) in all respects with anti-terrorism laws, anti-money laundering and international trade laws and (ii) in all material respects with all other pertinent Federal, state, and local laws including, but not limited to, Environmental Laws, the Federal Occupational Safety and Health Act and ERISA;
- (*q) Agent shall have received in form and substance satisfactory to Agent in its Permitted Discretion, (i) evidence that adequate insurance, including without limitation, casualty and liability insurance, required to be maintained under the Agreement is in full force and effect, (ii) insurance certificates issued by the Loan Parties' insurance broker containing such information regarding the Loan Parties' casualty and liability insurance policies as Agent shall request and naming Agent as an additional insured, lenders loss payee and/or mortgagee, as applicable, and (iii) loss payable endorsements issued by the Loan Parties' insurer naming Agent as lenders loss payee and mortgagee, as applicable;
- (yr) Agent shall have received all information requested by Agent to complete all necessary "know your customer" regulatory compliance checks and background checks, and the results thereof shall be satisfactory to Agent in its sole discretion;
 - (25) Borrower shall have established the Funding Account;
 - (aat) Since December 31, 2015, there has not occurred any Material Adverse Effect with respect to Parent and its Subsidiaries;
- (bbu) No Default or Event of Default under the Loan Documents shall have occurred or shall result from the making of the Loans and other extension of credit by Lenders; and
- (eey) All other documents and legal matters in connection with the transactions contemplated by the Agreement shall have been delivered, executed, or recorded and shall be in form and substance reasonably satisfactory to Agent.

Schedule 3.6

- (a) Within 3 Business Days after the Closing, Agent or its counsel shall have received original stock certificates evidencing Collateral, together with original transfer powers executed in blank, and each original promissory note constituting Collateral, together with original execute allonges.
- (b) Within 90 days after the Closing Date, Agent shall have received duly executed Control Agreements with respect to each Deposit Account of the Loan Parties maintained with any bank other than PNC (other than Excluded Deposit and Securities Accounts); provided that Loan Parties shall use commercially reasonable efforts to obtain such Control Agreements, or provide Agent with reasonably satisfactory evidence of the closure of such accounts, prior to such 90th day.
- (ddc) Borrower shall use commercially reasonable efforts to deliver to Agent, within 60 days after the Closing Date, a landlord waiver in form and substance satisfactory to Agent duly executed by the landlord of the Borrower's headquarters location.
- (eed) Within 120 days after the Closing Date, Agent shall have received evidence reasonably satisfactory to Agent that the Loan Parties have established policies and procedures that are sufficient to ensure compliance by the Loan Parties, their Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions.

The time periods for all of the deliverables required in clauses (a) through (d) above may be extended as determined by Agent from time to time in its sole discretion.

Schedule 3.6 - Page 1

Schedule 5.1

Deliver to Agent, with copies to each Lender, each of the financial statements, reports, or other items set forth below at the following times in form reasonably satisfactory to Agent:

within 30 days after the end of each month (other than months constituting the third month in any of Parent's fiscal quarters) during each of Parent's fiscal years,	(a) an unaudited consolidated and consolidating balance sheet (including information thereon sufficient to identify capitalized software expenses), income statement, statement of cash flow, and statement of Shareholder's equity covering Parent's and it Subsidiaries' operations during such period, and (b) a Compliance Certificate along with a separate worksheet containing detailed financial covenant calculations with respect to the applicable financial covenants set forth in Section 7 of the Agreement.
within 45 days after the end of each of Parent's fiscal quarters (other than the fourth quarter in any of Parent's fiscal years) during each of Parent's fiscal years,	(c) solely to the extent Public Holdco has not filed its (i) Form 10-Q quarterly report for such fiscal quarter within 45 days after the end of such fiscal quarter, an unaudited consolidated and consolidating balance sheet, income statement, statement of cash flow, and statement of shareholder's equity covering Parent'sPublic Holdco's and its Subsidiaries' operations during such period and compared to the prior period and plan, together with a corresponding discussion and analysis of results from management (such management discussion & analysis shall include a discussion of the number of new eustomer contracts executed in such period, and, to the extent available, shall include management's estimate of the potential aggregate annual revenue to be contributed by such new customer contracts), and and (d) a Compliance Certificate along with a separate worksheet containing detailed financial covenant calculations with respect to the applicable financial covenants set forth in Section 7 of the Agreement.

	T
within 120 days after the end of each of Parent's fiscal years,	(e) solely to the extent Public Holdco has not filed its Form 10-K annual report for such fiscal year within ninety (90) days after the end of such fiscal year, consolidated and consolidating financial statements of ParentPublic Holdco and its Subsidiaries for each such fiscal year, audited by independent certified public accountants reasonably acceptable to Agent (it being understood and agreed that Deloitte, Ernst & Young, KPMG, GrantThorton and PWC shall be acceptable to Agent) and certified, without any qualifications (including any qualification as to (A) "going concern" or like qualification or exception as to the scope of such audit, or (C) qualification which relates to the treatment or classification of any item and which, as a condition to the removal of such qualification, would require an adjustment to such item, the effect of which would be to cause any noncompliance with the provisions of Section 7 of the Agreement (provided, however, that it shall not be a violation of the foregoing clause (A) if the report and opinion accompanying the financial statements for the fiscal year ending immediately prior to the Maturity Date is subject to a "going concern" or other qualification solely as a result of such impending Maturity Date)), by such accountants to have been prepared in accordance with IFRSGAAP and without any explanation, supplemental comment, or other comment concerning the ability of the applicable Person to continue as a going concern or concerning the scope of the audit (such audited financial statements to include a balance sheet, income statement, statement of cash flow, and statement of shareholder's equity, and, if prepared, such accountants' final letter to management), (f) Compliance Certificate along with a separate worksheet containing detailed financial covenant calculations, and
within 45 days after the start of each of Parent's fiscal years,	(h) copies of Parent's Public Holdco's Projections, in form and substance (including as to scope and underlying assumptions) satisfactory to Agent, in its Permitted Discretion, for the forthcoming fiscal year, quarter by quarter, certified by the chief financial officer of Parent as being such officer's good faith estimate of the financial performance of ParentPublic Holdco and its Subsidiaries during the period covered thereby (it being recognized by Agent and the Lenders that such Projections and estimates as to future events are subject to significant uncertainties and contingencies, and although reflecting Parent's good faith projections or forecasts based on methods and assumptions which Parent believes to be reasonable and data which Parent believes to be accurate at the time such Projections were prepared, are not to be viewed as facts and that the actual results during the period or periods covered by any such Projections and estimates may differ materially from projected or estimated results).
if and when filed by ParentPublic Holdco,	(i) Form 10-Q quarterly reports, Form 10-K annual reports, and Form 8-K current reports, and (j) any other filings made by ParentPublic Holdco with the SEC.

promptly, but in any event within 5 Business Days after any Loan Party has knowledge of any event or condition that constitutes an Event of Default,	(k) notice of such event or condition and a statement of the curative action that Borrower (or any other Loan Party) proposes to take with respect thereto.
promptly after the commencement thereof, but in any event within 5 Business Days after the service of process with respect thereto on Parent or any of its Subsidiaries,	(l) notice of all actions, suits, or proceedings brought by or against Parent or any of its Subsidiaries before any Governmental <u>AuthorityBody</u> which reasonably could be expected to result in a Material Adverse Effect.
promptly after the execution thereof, but in any event within 5 Business Days thereafter,	(m) copies of each Applicable Affiliate <u>Transaction</u> Document.
promptly after the request of Agent,	(n) any other information reasonably requested relating to the financial condition of Parent or its Subsidiaries, except for (i) any information subject to a confidentiality agreement in favor of a third party (unless Agent has executed a confidentiality agreement with respect thereto or such disclosure is otherwise permitted under such confidentiality agreement), and (ii) any information protected by attorney-client privilege or similar privilege, or attorney work product.

Schedule 5.2

Provide Agent, with copies to each Lender, with each of the documents set forth below at the following times in form reasonably satisfactory to Agent:

Monthly (within 30 days (45 days in the case of a month that is the end of a fiscal quarter) after the end of each month)	 (a) a report summarizing the following (i) Recurring Revenues for the prior month, and (ii) Recurring Revenues for the trailing twelve months, and (b) a detailed report regarding Parent's and its Subsidiaries' cash and Cash Equivalents, including an indication of which accounts constitute Qualified Cash.
Quarterly (no later than 45 days following the end of each of the first three fiscal quarterquarters of each fiscal year)	 (c) a schedule of the minimum monthly contract value by customer (which shall be for the last month of such quarter), and (d) attrition data for the prior fiscal quarter consistent with what was previously provided.
Annually (no later than 45 days following the end of each fiscal year)	 (f) a Perfection Certificate or a supplement to the Perfection Certificate setting forth any updates to the most recent Perfection Certificate delivered to Agent not included in the IP Reporting Certificate, and (g) an IP Reporting Certificate.
Promptly upon request by Agent	(h) such other financial reports, as Agent may reasonably request (it being acknowledged by Agent and Lenders that, so long as no Default or Event of Default has occurred or is continuing, Agent's and Lenders' request for other financial reports shall be limited to those reports that are regularly prepared by Parent or its Subsidiaries) except for (i) any information subject to a confidentiality agreement in favor of a third party (unless Agent has executed a confidentiality agreement with respect thereto or such disclosure is otherwise permitted under such confidentiality agreement), and (ii) any information protected by attorney-client privilege or similar privilege, or attorney work product, and (i) all documentation and other information that Agent or any Lender reasonably requests in order to comply with its ongoing obligations under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act.

Certification of the Chief Executive Officer, pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, John DiLullo, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of LiveVox Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during
 the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2023 **By:**/s/ John DiLullo
John DiLullo

Chief Executive Officer and Director (Principal Executive Officer)

Certification of the Chief Financial Officer, pursuant to Rule 13a-14(a)/15d-14(a), as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Gregg Clevenger, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of LiveVox Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during
 the period in which this report is being prepared;
 - Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2023 By:

/s/ Gregg Clevenger
Gregg Clevenger
Executive Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)

Certification of the Chief Executive Officer, pursuant to Rule 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of LiveVox Holdings, Inc. (the "Company") on Form 10-Q for the quarter ended March 31, 2023 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, John DiLullo, Chief Executive Officer of the Company, hereby certify, to the best of my knowledge and belief, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to section 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2023 By: /s/ John DiLullo

John DiLullo

Chief Executive Officer and Director (Principal Executive Officer)

Certification of the Chief Financial Officer, pursuant to Rule 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report of LiveVox Holdings, Inc. (the "Company") on Form 10-Q for the quarter ended March 31, 2023 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Gregg Clevenger, Executive Vice President and Chief Financial Officer of the Company, hereby certify, to the best of my knowledge and belief, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to section 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2023 By: /s/ Gregg Clevenger

Gregg Clevenger

Executive Vice President and Chief Financial Officer (Principal Financial and Accounting Officer)